TRANSCRIPTIONAL REGULATION OF HUMAN UDP-GLUCURONOSYLTRANSFERASES

Dione Anne Gardner-Stephen

B. Biotech. (Hons)

Department of Clinical Pharmacology

Faculty of Health Sciences, School of Medicine

Flinders University

A thesis submitted in fulfilment of the requirements for the

degree of Doctor of Philosophy

2008

TABLE OF CONTENTS

List o	of Figures	viii
List o	of Tables	xi
Sum	mary	xii
Decla	aration	xiv
Ackn	nowledgements	xv
Publi	ications arising directly from this thesis	vvii
Addi	tional publications related to this thesis	
Awa	utonal publications related to this thesis	
Awa	rds in support of this thesis	.xvm
Abbr	reviations	XIX
Chap	oter One: General Introduction	1
1.1	Towards disease prevention and designer therapies	1
1.2	Genetic variation in metabolic pathways; implications for disease susceptibility	
	and treatment	3
1.3	Biotransformation of small lipophilic molecules	5
1.4	Glucuronidation and the UDP-Glucuronosyltransferases	10
1.5	Genomic organisation of the human UGT gene family	14
	1.5.1. The human UGT1 locus	14
	1.5.2. The human UGT2 locus	16
1.6	UGT substrates	17
1.7	Location and distribution of human UGT gene products	17
1.8	Causes and consequences of variability in UGT expression and activity	23
	1.8.1. UGT expression during human development	23
	1.8.2. Gender effects on human glucuronidation	26
	1.8.3. Xenobiotic exposure and glucuronidation	26
	1.8.4. Pregnancy and UGT expression	30
	1.8.5. Health and glucuronidation	31
	1.8.6. Genetic variation and glucuronidation	33
1.9	Regulation of human UGT genes	43
	1.9.1. Transcriptional regulation: transcription factors and co-regulators	43
	1.9.2. Gene regulatory elements: the core promoter, the proximal promoter	
	and enhancer elements	45

	1.9.3. Regulation of human UGT genes	47
1.10	Experimental aims	53

Chap	ter Tw	o: Materials and Methods	55
2.1	Materi	als	55
	2.1.1.	Chemicals and molecular biology reagents	55
	2.1.2.	General buffers	59
	2.1.3.	Eukaryotic and prokaryotic cell lines	60
	2.1.4.	Mammalian reporter and expression vectors	60
	2.1.5.	Cloning vectors	61
2.2	Genera	al methods	61
	2.2.1.	Maintenance of mammalian cell lines	61
	2.2.2.	Bacterial culture and preparation of competent cells	62
	2.2.3.	Extraction of total RNA and generation of cDNA	63
	2.2.4.	Extraction of genomic DNA from HEK293T cells	65
	2.2.5.	Plasmid and BAC preparation	66
	2.2.6.	Polymerase chain reaction amplification	68
	2.2.7.	Restriction digests, calf intestinal alkaline phosphatase treatment,	
		ligation and transformation	72
	2.2.8.	Electrophoresis and quantification of DNA and RNA	74
	2.2.9.	Sequencing	75
	2.2.10.	Transient transfection of HepG2, Caco-2 and HEK293T cells	75
	2.2.11.	Luciferase assay	76
	2.2.12.	Preparation of nuclear extracts	77
	2.2.13.	Labelling oligonucleotide probes	78
	2.2.14.	Electrophoretic mobility-shift assays	79
	2.2.15.	Super-shift EMSA	80
	2.2.16.	Western blot	80
	2.2.17.	Statistical analysis	81
2.3	Clonin	g of liver-enriched transcription factors	82
	2.3.1.	Hepatocyte nuclear factor 1β (HNF1 β)	82
	2.3.2.	Hepatocyte nuclear factor 4α (HNF4α)	83
	2.3.3.	Hepatocyte nuclear factor 6 (HNF6, OneCut 1)	85
	2.3.4.	Forkhead box A factors (FoxA, hepatocyte nuclear factor 3 factors)	85
	2.3.5.	CAATT enhancer binding protein factors (C/EBP)	86
	2.3.6.	Pregnane X receptor (PXR)	86

Cha UC7	pter 1 n	ree: In Vitro Characterisation of the UGIIA3, UGIIA4 and	00
3 1	Introd	luction	00
5.1	3 1 1	The UGT1A3 UGT1A4 and UGT1A5 game cluster	80 88
	3.1.1.	Francession of UGT143, UGT144 and UGT145	80 80
	313	Substrates of UGT143, UGT144 and UGT145	90
	311	Regulatory controls of UGT1A3, UGT1A4 and UGT1A5	90 92
	315	The hepatocyte nuclear factor 1 transcription factor family	92
	316	Aims	9 99
32	Metho	nds	
5.2	<i>3.2.1</i> .	Isolation of the UGT1A3-3.3k and UGT1A4-3.4k promoters	99
	3.2.2.	Generation of UGT1A3 and UGT1A4 promoter deletion constructs an	d
		mutants	102
	3.2.3.	Isolation of the UGT1A5 promoter	103
	3.2.4.	HNF1α and HNF1β expression vectors	104
	3.2.5.	Transient transfection and luciferase reporter assay	104
	3.2.6.	Electrophoretic mobility-shift assay	104
	3.2.7.	Statistical analysis	105
3.3	Resul	ts and discussion	105
	3.3.1.	Basal activities of the UGT1A3, UGT1A4 and UGT1A5 proximal	
		promoters	105
	3.3.2.	HNF1 is required for basal activity of the UGT1A3 and UGT1A4	
		proximal promoters	108
	3.3.3.	<i>HNF1β can transactivate the</i> UGT1A3 <i>promoter</i> in vitro	114
	3.3.4.	The UGT1A3 and UGT1A4 proximal promoters differ in their HNF1	
		responses	115
	3.3.5.	Factors bound to at least two elements in the UGT1A3 and UGT1A4	
		promoters co-operate with HNF1a	123
3.4	Gener	al discussion and summary	126
	3.4.1.	Achievement of aims	126
	3.4.2.	Future directions	126
	3.4.3.	Relevance to pharmacogenetics	128
	3.4.4.	Summary	129

Pror	noter Re	esponse to HNF4a	130
4.1	Introd	uction	
	4.1.1.	The UGT1A7, UGT1A8, UGT1A9 and UGT1A10 gene cluster	130
	4.1.2.	Expression and substrates of UGT1A9	132
	4.1.3.	Regulatory controls of UGT1A9	
	4.1.4.	The hepatocyte nuclear factor 4 transcription factor family	136
	4.1.5.	Aims	145
4.2	Metho	ds	145
	4.2.1.	Construction of the pGL3+ reporter plasmid	145
	4.2.2.	Generation of the UGT1A9-2k reporter construct	146
	4.2.3.	Generation of UGT1A9 deletion constructs and mutants	150
	4.2.4.	Isolation of the UGT1A7-5k promoter and generation of pGL3-1A7	
		-2.5k	151
	4.2.5.	UGT1A8 and UGT1A10 reporter constructs	152
	4.2.6.	Isolation of the apoCIII-810/+23 promoter	152
	4.2.7.	Transcription factor expression vectors	153
	4.2.8.	Transient transfection and luciferase reporter assay	153
	4.2.9.	Electrophoretic mobility-shift and super-shift assays	154
	4.2.10	. In vitro synthesis of transcription factor proteins	154
	4.2.11	. Western blot	155
	4.2.12	. Statistical analysis	155
4.3	Result	s and discussion	156
	4.3.1.	The UGT1A9 proximal promoter responds strongly to over-expression	1
		of human HNF4 α 2 in HepG2 cells	156
	4.3.2.	The UGT1A9 promoter -372 to -360 HNF4 α -binding site is not	
		necessary for the UGT1A9 promoter response to human HNF4 α 2	156
	4.3.3.	The UGT1A9 response to $HNF4\alpha$ in vitro is dependent on a second	
		$HNF4\alpha$ -binding element	158
	4.3.4.	The UGT1A9 proximal promoter response to HNF4 requires HNF1a.	160
	4.3.5.	A second HNF1 site is essential for synergistic regulation of the	
		UGT1A9 promoter by HNF1α and HNF4α	164
	4.3.6.	Additional evidence for a direct interaction of HNF4 α with UGT1A9.	167
	4.3.7.	Nucleotide substitutions in the second HNF4 and HNF1 sites may part	tly
		explain the uniqueness of the UGT1A9 promoter response to $HNF4\alpha$.	175
	4.3.8.	The UGT1A9 initiator-like region contributes to its $HNF4\alpha$ -response.	176

Chanter Form UNET Transprintion Factors are Essential for th - - -.....

	4.3.9. HNF1 β supports activation of the UGT1A9 promoter by HNF4 α	179
	4.3.10. Differences between published work and this study are not explained	d
	by usage of HNF4α1 or the UGT1A9 T-275A polymorphism	
4.4	General discussion and summary	
	4.4.1. Achievement of aims	
	4.4.2. Future directions	
	4.4.3. Relevance to glucuronidation in humans	
	4.4.4. Summary	

Chapter Five: Regulation of Endogenous UGT Expression in HEPG2 Cells

by L	liver-En	riched Transcription Factors	. 197
5.1	Introd	uction	197
	5.1.1.	Promoter-reporter assays versus endogenous gene expression	197
	5.1.2.	Aims	198
	5.1.3.	Rationale	200
5.2	Metho	ods	219
	5.2.1.	Generation of liver-enriched transcription factor expression plasmids	219
	5.2.2.	Transfection of HepG2 cells and extraction of total RNA	220
	5.2.3.	Trichostatin A treatment of HepG2 cells	220
	5.2.4.	Rifampicin treatment of HepG2 cells	220
	5.2.5.	Reverse transcription and quantitative real-time PCR	221
	5.2.6.	Cloning of the UGT1A3-9.4k and UGT1A4-5k promoters	225
	5.2.7.	Transfection of HepG2 cells and dual-luciferase assay	228
	5.2.8.	Hepatocyte nuclear factor 1a Western blot	228
5.3	Result	s and discussion	228
	5.3.1.	Validation of 18S rRNA as a suitable internal control for $HepG2$ cells	
		over-expressing liver-enriched transcription factors and/or treated with	
		trichostatin A or rifampicin	228
	5.3.2.	Basal levels of UGT1A and UGT2B mRNA transcripts in HepG2 cells	231
	5.3.3.	Effect of rifampicin-liganded pregnane X receptor on endogenous UGT	
		expression in HepG2 cells	234
	5.3.4.	UGT1A3 promoter-reporter constructs lack ligand-dependent PXR	
		responsiveness in HepG2 cells	237
	5.3.5.	Effect of trichostatin A treatment on endogenous UGT expression in	
		HepG2 cells	241

	5.3.6.	Effect of trichostatin A treatment on endogenous expression of HNF1 α	
		in HepG2 cells	247
	5.3.7.	Effect of over-expressed liver-enriched transcription factors on	
		endogenous UGT expression in HepG2 cells	250
	5.3.8.	Validation of the FoxA2 and C/EBPβ expression vectors	283
5.4	Genera	al discussion and summary	285
	5.4.1.	Achievement of aims	285
	5.4.2.	Broader directions for future investigations	286
	5.4.3.	Relevance to pharmacogenetics and disease	293
	5.4.4.	Summary	295

Cha _]	pter Six:	Genetic Determinants of Human UGT1A3 Expression	. 296
6.1	Introdu	action	296
	6.1.1.	Genetic variation in human UGT genes	296
	6.1.2.	HNF1α gene polymorphisms in humans	297
	6.1.3.	Aims	299
6.2	Metho	ds	300
	6.2.1.	Amplification of UGT1A3 sequences from genomic DNA	300
	6.2.2.	HNF1a expression vectors	302
	6.2.3.	Transient transfection and luciferase reporter assay	302
	6.2.4.	Statistical analysis	302
6.3	Results	s and discussion	302
	6.3.1.	The UGT1A3 promoter contains multiple SNPs, resulting in five alleles.	302
	6.3.2.	Activities of the newly defined UGT1A3 promoter alleles in vitro	305
	6.3.3.	Regulation of the UGT1A3 promoter by HNF1a variants	309
	6.3.4.	The UGT1A3 Promoter 2 allele is under-represented in a colon cancer	
		cohort and is associated with the W11R/V47A protein variant	312
6.4	Genera	Il discussion and summary	316
	6.4.1.	Achievement of aims	316
	6.4.2.	Future directions	316
	6.4.3.	Relevance to pharmacogenetics	318
	6.4.4.	Summary	320

Chapter Seven: General Summary and Conclusions			
7.1	Towards disease prevention and designer therapies	.322	
7.2	Summary of the research findings presented in this thesis	.323	

Appe	Appendix Two: pGL3+ reporter vector map		
Appe	ndix O	ne: pGL3-Basic reporter vector map	. 332
7.5	Final r	emarks	330
7.4	Model	ling human UGT gene regulation	329
7.3	Releva	ant work published over the duration of this PhD candidature	327
	7.2.4.	Chapter 6: Genetic determinants of human UGT1A3 expression	326
		by liver-enriched transcription factors	324
	7.2.3.	Chapter 5: Regulation of endogenous UGT expression in HepG2 cells	
		promoter response to HNF4α	324
	7.2.2.	Chapter 4: HNF1 transcription factors are essential for the UGT1A9	
		UGT1A5 proximal promoters	323
	7.2.1.	Chapter 3: In vitro characterisation of the UGT1A3, UGT1A4 and	

LIST OF FIGURES

Figure 1.1:	Alternative metabolic pathways for 2-amino-1-methyl-6- phenylimidazo[4,5- <i>b</i>]pyridine (PhIP) after CYP activation, resulting in DNA adducts or detoxification
Figure 1.2:	Schematic representation of the glucuronidation reaction11
Figure 1.3:	Dendogram depicting relationships between the primary amino acid sequences of human UGTs
Figure 1.4:	Schematic representation of the human UGT1A locus15
Figure 1.5:	Schematic representation of the human UGT2 locus16
Figure 1.6:	Excerpt from the current label for Camptosar (Irinotecan HCl), as obtained from the Food and Drug Administration website
Figure 1.7:	Idealised structure of an eukaryotic gene46
Figure 1.8:	The <i>UGT1A</i> and <i>UGT2B</i> core promoters showing putative TATA box and initiator-like elements
Figure 3.1:	Successive deletion constructs of the <i>UGT1A3</i> and <i>UGT1A4</i> promoters reveal positive regulatory elements necessary for basal activity
Figure 3.2:	The human <i>UGT1A3</i> , <i>UGT1A4</i> and <i>UGT1A5</i> promoters contain putative HNF1-binding sites
Figure 3.3:	The HNF1-binding site is required for maximal basal activity of the <i>UGT1A3</i> and <i>UGT1A4</i> proximal promoters
Figure 3.4:	HNF1α from HepG2 nuclear extracts binds to nucleotides -156 to -128 of the <i>UGT1A3</i> and <i>UGT1A4/UGT1A5</i> promoters
Figure 3.5:	HNF1α from Caco-2 nuclear extracts binds to nucleotides -156 to -128 of the <i>UGT1A3</i> and <i>UGT1A4/UGT1A5</i> promoters
Figure 3.6:	Activation of the <i>UGT1A3</i> -500bp promoter by HNF1β in HEK293T cells
Figure 3.7:	Mutation of the <i>UGT1A3</i> HNF1-binding site to the equivalent <i>UGT1A4</i> sequence does not confer HNF1-responsiveness in HepG2 cells
Figure 3.8:	The <i>UGT1A3</i> -200 to -150bp region contains at least two regulatory elements that can positively influence transcription in HepG2 cells120
Figure 3.9:	Characterisation of the UGT1A3 promoter site 1 regulatory element121
Figure 3.10:	The <i>UGT1A3</i> and <i>UGT1A4</i> -200 to -150bp regions contain at least two regulatory elements that can positively influence their HNF1 α -response in HEK293T cells
Figure 4.1:	Alignment of the -391 bp <i>UGT1A9</i> proximal promoter with the corresponding regions of <i>UGT1A7</i> , <i>UGT1A8</i> and <i>UGT1A10</i> 131
Figure 4.2:	Over-expression of human HNF4α in HepG2 cells strongly up- regulates the human <i>UGT1A9</i> proximal promoter157

Figure 4.3:	Assessment of HNF4α binding to <i>UGT1A9</i> promoter sequences <i>in vitro</i> .	159
Figure 4.4:	HNF1α synergistically regulates the <i>UGT1A9</i> promoter with HNF4α through a second HNF1-binding site	161
Figure 4.5:	HNF1 α only moderately increases activation of the <i>UGT1A9</i> proximal promoter by HNF4 α in HepG2 cells.	163
Figure 4.6:	Mutation of a second putative HNF1-binding site decreases the co- operative activation of the $UGT1A9$ proximal promoter by HNF1 α and HNF4 α in HEK293T cells.	165
Figure 4.7:	The second UGT1A9 HNF1-response element binds HNF1α in vitro	166
Figure 4.8:	Alignment of the oestrogen receptor β and HNF4 α DNA-binding domains	169
Figure 4.9:	The HNF4α R127W, DG69/70AA and RR76/77AE mutants have decreased ability to up-regulate the <i>UGT1A9</i> promoter <i>in vitro</i>	170
Figure 4.10:	Expression and binding capacity of wild-type and mutant HNF4α proteins <i>in vitro</i>	172
Figure 4.11:	S-nitroso-N-acetyl penicillamine (SNAP) treatment decreases the ability of HNF4 α to up-regulate the UGT1A9 promoter in HepG2 cells.	173
Figure 4.12:	Introduction of the <i>UGT1A9</i> HNF1s2 and HNF4s2 elements into the <i>UGT1A8</i> promoter does not confer HNF4α-responsiveness to pGL3-1A8-2k.	177
Figure 4.13:	The Inr-like region of $UGT1A9$ is more supportive of HNF4 α -responsiveness than the equivalent region of $UGT1A8$	178
Figure 4.14:	HNF1 β facilitates regulation of the <i>UGT1A9</i> proximal promoter by HNF4 α	180
Figure 4.15:	Comparison of human HNF4 α 2, human HNF4 α 1 and rat HNF4 α 1 pCMX expression vectors for effectiveness against the <i>UGT1A9</i> promoter.	183
Figure 4.16:	The <i>UGT1A9</i> -275 SNP decreases stimulation of the <i>UGT1A9</i> promoter by HNF4α2 <i>in vitro</i>	186
Figure 5.1:	All known human <i>UGT1A</i> and <i>UGT2B</i> promoters contain putative HNF1-binding sites.	204
Figure 5.2:	18S rRNA levels in treated HepG2 cells as a proportion of total extracted RNA.	230
Figure 5.3:	Expression of UGT mRNA in HepG2 cells under basal culture conditions.	233
Figure 5.4:	Endogenous UGT mRNA levels in HepG2 cells treated with 10 μ M rifampicin and/or over-expressed pregnane X receptor	235
Figure 5.5:	PXR up-regulates the <i>UGT1A3</i> and <i>UGT1A4</i> promoters <i>in vitro</i> in a ligand-independent manner.	238
Figure 5.6:	Effects of trichostatin A on UGT mRNA levels in HepG2 cells	243
Figure 5.7:	Effect of ethanol, trichostatin A and pCMX-HNF1 α on the levels of HNF1 α in HepG2 cells.	249

Figure 5.8A-C:	Effect of over-expressed liver-enriched transcription factors on endogenous UGT1A1, UGT1A3 and UGT1A4 expression in HepG2 cells	251
Figure 5.8D-F:	Effect of over-expressed liver-enriched transcription factors on endogenous UGT1A6, UGT1A7 and UGT1A9 expression in HepG2 cells	252
Figure 5.8G-I:	Effect of over-expressed liver-enriched transcription factors on endogenous UGT2B4, UGT2B7 and UGT2B10 expression in HepG2 cells	253
Figure 5.8J-L:	Effect of over-expressed liver-enriched transcription factors on endogenous UGT2B11, UGT2B15 and UGT2B17 expression in HepG2 cells.	254
Figure 5.8M:	Effect of over-expressed liver-enriched transcription factors on endogenous UGT2B28 expression in HepG2 cells.	255
Figure 5.9:	The UGT1A3 and UGT1A4 promoter responses to TSA and HNF1a	262
Figure 5.10:	Response of the $UGT1A6$ proximal promoter to HNF1 α and HNF4 α in HepG2 cells.	264
Figure 5.11:	HNF6 regulates the transiently transfected <i>UGT1A4</i> promoter in HepG2 cells.	274
Figure 5.12:	The pCMX-FoxA2 and pCMX-C/EBPβ expression plasmids are active in HepG2 cells	284
Figure 6.1:	Alternative <i>UGT1A3</i> promoter alleles vary less than 1.4-fold in basal activity.	306
Figure 6.2:	Alternative <i>UGT1A3</i> promoter alleles are not decreased in their ability to respond to over-expressed HNF1α	307
Figure 6.3:	Known HNF1α protein variants are insufficient to explain the variability of UGT1A3 mRNA levels in humans.	310
Figure 6.4:	Two $UGT1A3$ promoter variants interact with the tested HNF1 α protein variants similarly to the $UGT1A3$ reference promoter sequence.	311
	•	

LIST OF TABLES

Table 1.1:	Human UDP-Glucuronosyltransferases and selected substrates
Table 1.2:	Distribution of UDP-Glucuronosyltransferase mRNA in humans22
Table 2.1:	Reagents used in experimental procedures
Table 2.2:	Primers used for cloning transcription factor cDNAs
Table 3.1:	Primers used for cloning the <i>UGT1A3-1A5</i> proximal promoters, and in generating deleted or mutated promoter-reporter constructs100
Table 3.2:	Oligonucleotides used for EMSA and super-shift experiments105
Table 4.1:	Primers used in generating pGL3+, in cloning the <i>UGT1A7</i> , <i>UGT1A9</i> and <i>apoCIII</i> proximal promoters, and in generating deleted and mutated promoter-reporter and transcription factor expression constructs
Table 4.2:	Oligonucleotides used for EMSA and super-shift experiments155
Table 5.1:	Primers and annealing conditions used for quantification of human UGT and 18S RNA transcripts222
Table 6.1:	Primers used in amplifying and cloning the human <i>UGT1A3</i> proximal promoter from genomic DNA samples
Table 6.2:	Single nucleotide polymorphisms detected in a German population304
Table 6.3:	Association of UGT1A3 promoter genotype with colon cancer

SUMMARY

The UDP-glucuronosyltransferases (UGTs) are a superfamily of enzymes that glucuronidate small, lipophilic molecules, thereby altering their biological activity and excretion. In humans, important examples of UGT substrates include molecules of both endogenous and xenobiotic origin; thus, UGTs are considered essential contributors to homeostatic regulation and an important defence mechanism against chemical insult. In keeping with both roles, UGTs are most strongly expressed in the liver, a predominant organ involved in detoxification.

Rates of glucuronidation in humans are neither uniform among individuals, nor constant in an individual over time. Genetic determinants and non-endogenous signals are both known to influence the expression of UGTs, which in turn may affect the efficacy of certain pharmaceutical treatments or alter long-term risk of developing disease. Thus, this thesis focuses on the transcriptional regulation of *UGT* genes in humans, particularly on mechanisms that are likely to be relevant to their expression and variation in the liver. Two major approaches were used: firstly, extensive studies of several *UGT* promoters were performed to identify and characterise transcriptional elements that are important for UGT expression; and secondly, important hepatic transcription factors were investigated as potential regulators of *UGT* genes.

UGT1A3, *UGT1A4* and *UGT1A5* are a subset of highly related, but independently regulated, genes of the human *UGT1* subfamily. UGT1A3 and UGT1A4 are expressed in the liver, whereas UGT1A5 is not. The presented analysis of the *UGT1A3*, *UGT1A4* and *UGT1A5* proximal promoters demonstrates that a hepatocyte nuclear factor (HNF)1-binding site common to all three promoters is important for

UGT1A3 and UGT1A4 promoter activity *in vitro*, but is insufficient to drive UGT1A5 expression. Two additional elements required for the maximal activity of the *UGT1A3* promoter were also identified that may distinguish this gene from *UGT1A4*. *UGT1A3* was investigated further, focusing on mechanisms that may contribute to interindividual variation in UGT1A3 expression. Polymorphisms in the *UGT1A3* proximal promoter were identified and their functional consequences tested. Known variants of HNF1 α were also tested for altered activity towards the *UGT1A3* gene.

UGT1A9 is the only hepatic member of the UGT1A7-1A10 subgroup of UGT1 enzymes. Previous work had identified HNF1-binding sites in all four genes, and HNF4 α as an *UGT1A9*-specific regulator. The work presented herein extends these findings to show that HNF1 factors and HNF4 α synergistically regulate *UGT1A9*, and that HNF4 α is not the only transcription factor responsible for the unique presence of UGT1A9 in the liver.

Liver-enriched transcription factors screened as potential *UGT* regulators were chosen from the HNF1, HNF4, HNF6, FoxA and C/EBP protein families. Functional interactions newly identified by this work were HNF4 α with *UGT1A1* and *UGT1A6*, HNF6 with *UGT1A4* and *UGT2B11*, FoxA1 and FoxA3 with *UGT2B11*, *UGT2B15* and *UGT2B28* and C/EBP α with *UGT2B17*. Observations were also made regarding different patterns of interaction between each *UGT* and the transcription factors tested, particularly HNF1 α .

These studies significantly advance the understanding of the transcriptional control of human *UGT* genes. In time, it is hoped that a detailed knowledge of UGTs will be useful in developing better therapeutic and prophylactic medical treatments.

DECLARATION

I certify that this thesis does not incorporate without acknowledgement any material previously submitted for a degree or diploma in any university; and that to the best of my knowledge and belief it does not contain any material previously published or written by another person except where due reference is made in the text.

Dione Gardner-Stephen

ACKNOWLEDGEMENTS

I wish to convey sincere, heartfelt gratitude to all those who have made this thesis possible. Whether specifically listed below or otherwise, your encouragement, support, feedback and sacrifices have been invaluable; I am and continue to be most grateful.

Firstly, I offer profuse thanks to my supervisor, Professor Peter Mackenzie. Thank you for not only encouraging me to take on this project, but also for following through with friendly, effective guidance and constant availability. I have long admired your ability to balance excellent science with other important commitments, especially your family.

Also to my other colleagues, both past and present, I owe a great debt of thanks for their work and friendship. In particular, I thank Dr. Rikke Lewinsky and Dr. Philip Gregory, for providing much of the groundwork on which this thesis is built. Extra thanks are also due to Phil for teaching me many of the techniques used in this project. I am also grateful to Anne Rogers and Joanna Treloar for their excellent technical assistance, to Heather Aubert and Karli Goodwin for their unsurpassed administrative support, and to the numerous other friends and colleagues in the Department of Clinical Pharmacology who have made my PhD candidature a pleasant and productive one.

To my husband, best friend and co-PhD-student, Paul, I am forever grateful. Your love, support, encouragement and (worn out) listening ear mean more to me than words could ever express. I love you dearly and I hope you find some use for all the molecular biology you have accidentally learnt in the last few years! To our unborn child, I owe the great debt of motivation: your impending arrival was a great incentive to finish this thesis and a very welcome distraction from the more mundane aspects of "writing-up". I also want to recognise the contributions that my immediate and extended family, my friends and my church family have all made to my wellbeing over the course of my PhD. Thank you to you all. Finally, I willingly acknowledge the claim that Christ has on my life and praise Him for His unfailing provision for me; past, present and future.

PUBLICATIONS ARISING DIRECTLY FROM THIS THESIS

Gardner-Stephen, D.A. and Mackenzie, P.I. (2007) Isolation of the UDPglucuronosyltransferase 1A3 and 1A4 proximal promoters and characterization of their dependence on the transcription factor hepatocyte nuclear factor 1alpha. *Drug Metab. Dispos.* **35**: 116-120.

Gardner-Stephen, D.A. and Mackenzie, P.I. (2007) Hepatocyte nuclear factor 1 transcription factors are essential for the UDP-glucuronosyltransferase 1A9 promoter response to hepatocyte nuclear factor 4alpha. *Pharmacogenet. Genomics* **17**: 25-36.

Gardner-Stephen, D.A. and Mackenzie, P.I. (2005) Identification and characterisation of functional hepatocyte nuclear factor 1 binding sites in UDP-glucuronosyltransferase genes. *Methods Enzymol.* **400**: 22-46.

ADDITIONAL PUBLICATIONS RELATED TO THIS THESIS

Gregory, P.A. **Gardner-Stephen**, **D.A.**, Rogers, A., Michael, M.Z., and Mackenzie, P.I. (2006) The caudal-related homeodomain protein Cdx2 and hepatocyte nuclear factor 1alpha cooperatively regulate the UDP-glucuronosyltransferase 2B7 gene promoter. *Pharmacogenet. Genomics* **16**: 527-536.

Finel M., Li X., **Gardner-Stephen D.A.**, Bratton S., Mackenzie P.I. and Radominska-Pandya A. (2005) Human UDP-glucuronosyltransferase 1A5: identification, expression, and activity. *J. Pharmacol. Exp. Ther.* **315**: 1143-1149.

Mackenzie P.I., Gregory P.A., Lewinsky R.H., Yasmin S.N., Height T., McKinnon R.A. and **Gardner-Stephen D.A.** (2005) Polymorphic variations in the expression of the chemical detoxifying UDP glucuronosyltransferases. *Toxicol. Appl. Pharmacol.* **207**: S77-83.

Gregory P.A., Lewinsky R.H., Gardner-Stephen D.A. and Mackenzie P.I. (2004) Regulation of UDP glucuronosyltransferases in the gastrointestinal tract. *Toxicol. Appl. Pharmacol.* **199**: 354-363.

Gregory P.A., Lewinsky R.H., **Gardner-Stephen D.A.** and Mackenzie P.I. (2004) Coordinate regulation of the human UDP-glucuronosyltransferase 1A8, 1A9, and 1A10 genes by hepatocyte nuclear factor 1alpha and the caudal-related homeodomain protein 2. *Mol. Pharmacol.* **65**: 953-963.

Uchaipichat V., Mackenzie P.I., Guo X.H., **Gardner-Stephen D.A.**, Galetin A., Houston J.B. and Miners J.O. (2004) Human UDP-glucuronosyltransferases: isoform selectivity and kinetics of 4-methylumbelliferone and 1-naphthol glucuronidation,

effects of organic solvents, and inhibition by diclofenac and probenecid. *Drug Metab. Dispos.* **32**: 413-423.

Gardner-Stephen D.A., Heydel J.M., Goyal A., Lu Y., Xie W., Lindblom T., Mackenzie P.I. and Radominska-Pandya A. (2004) Human PXR variants and their differential effects on the regulation of human UDP-glucuronosyltransferase gene expression. *Drug Metab. Dispos.* **32**: 340-347.

Gregory P.A., Gardner-Stephen D.A., Lewinsky R.H., Duncliffe K.N. and Mackenzie P.I. (2003) Cloning and characterization of the human UDP-glucuronosyltransferase 1A8, 1A9, and 1A10 gene promoters: differential regulation through an initiator-like region. *J. Biol. Chem.* **278**: 36107-36114.

Mackenzie P.I., Gregory P.A., **Gardner-Stephen D.A.**, Lewinsky R.H., Jorgensen B.R., Nishiyama T., Xie W. and Radominska-Pandya A. (2003) Regulation of UDP glucuronosyltransferase genes. *Curr. Drug Metab.* **4**: 249-257.

AWARDS IN SUPPORT OF THIS THESIS

Australasian Society of Clinical and Experimental Pharmacologists and Toxicologists Percy Student Poster Prize: Australian Health and Medical Research Congress 2004.

Australian Society for Biochemistry and Molecular Biology Student Poster Prize: ComBio2005 Combined National Conference 2005.

Australasian Society of Clinical and Experimental Pharmacologists and Toxicologists Peter Meffin Student Poster Prize (honourable mention): National Joint Meeting of the Australasian Society of Clinical and Experimental Pharmacologists and Toxicologists and the Australasian Pharmaceutical Science Association 2005.

Outstanding Laboratory Research Presentation Oral Prize: Flinders Cancer Control Alliance Research Day 2007.

ABBREVIATIONS

ADH	alcohol dehydrogenase
AF	activation function
AhR	aryl hydrocarbon receptor
apo	apolipoprotein
ATCC	American Type Culture Collection
ATP	adenosine triphosphate
BAC	bacterial artificial chromosome
β-ΜΕ	β-mercaptoethanol
BSA	bovine serum albumin
bZIP	basic region leucine zipper
C/EBP	CCAAT/enhancer binding protein
CAR	constitutive androstane receptor
CBP	CREB-binding protein
CDCA	chenodeoxycholic acid
Cdx	caudal-related homeodomain protein
CIP	calf intestinal alkaline phosphatase
CMV	cytomegalovirus
СоА	coenzyme A
com	counts per minute
CREB	cAMP-response-element-binding protein
CYP	cvtochrome P450
DBP	D-site binding protein
DCoH	dimerisation co-factor of HNF1
DD	dihydrodiol dehydrogenase
DMEM	Dulbecco's modified Eagle's medium
DMSO	dimethyl sulphoxide
dNTP	deoxynucleotide-triphosphate
DTT	dithiothreitol
E. coli	Escherichia coli
EDTA	ethylenediaminetetra-acetic acid
EMSA	electrophoretic mobility-shift assay
ERß	oestrogen recentor ß
fabn	fatty acid-binding protein
FoxA	forkhead box A
FXR	farmesoid X recentor
olut?	glucose transporter 2
GR	glucocorticoid recentor
GRIP	glucocorticoid receptor interacting protein
НАТ	histone acetyltransferase
HCV	hepatitis C virus
HDAC	histone deacetylase
HNF	hepatocyte nuclear factor
II	interleukin
IL	initiator
	liver anriched transcriptional activator protain
	Luria broth
LD	

LETF	liver-enriched transcription factor
LIP	liver-enriched transcriptional inhibitory protein
LXR	liver X receptor
MODY	mature onset diabetes of the young
MRP	multidrug resistance protein
NCoR	nuclear receptor co-repressor
NNAL	4-(methylnitrosamino)-1-(3-pyridyl)-1-butanol
NRRE	nuclear receptor response element
NSAID	non-steroidal anti-inflammatory drug
OAT	organic anion transporter
Oct	octamer transcription factor
P/CAF	p300/CBP-associated factor
PBREM	phenobarbital response enhancer module
PBS	phosphate buffered saline
pBSII	pBlueScript II
PCR	polymerase chain reaction
PGC	PPAR-gamma co-activator
PhIP	2-amino-1-methyl-6-phenylimidazo[4,5-b]pyridine
POU	Pit-1, Oct-1 and Oct-2, and Unc-86
PPAR	peroxisome-proliferator-activated receptor
PXR	pregnane X receptor
QPCR	quantitative real-time PCR
rRNA	ribosomal RNA
rs	reference SNP
RXR	retinoid X receptor
SDS	sodium dodecyl sulphate
SHP	small heterodimer partner
siRNA	small interfering RNA
SMP	skim milk powder
SMRT	silencing mediator of retinoid and thyroid hormone receptor
SN-38	7-ethyl-10-hydroxycamptothecine
SNAP	S-nitroso-N-acetyl penicillamine
SNP	single nucleotide polymorphism
SRC	steroid receptor co-activator
TBE	tris-borate EDTA
TBST	tris-buffered saline/tween-20
TFII	transcription factor II
Tris	tris[hydroxymethyl]aminomethane
TSA	trichostatin A
TSS	transcription start site
UDP	uridine diphosphate
UGT	UDP-glucuronosyltransferase
XRE	xenobiotic response element
	1

CHAPTER ONE GENERAL INTRODUCTION

1.1 Towards disease prevention and designer therapies

Human genetic variation, particularly in relation to disease, is becoming an increasingly important focus of medical research. The recent accomplishments of the Human Genome Project (International Human Genome Sequencing Consortium, 2001), the ongoing collection of single nucleotide polymorphism (SNP) data by the International HapMap Project and other groups (International HapMap Consortium, 2003; Johnson et al., 2005) and continuing advancements in bioinformatics, research methodologies and high-throughput technologies are providing an unprecedented wealth of information regarding human molecular biology. The ability to analyse our own genome on such a comprehensive scale is generating a collective belief that science can eventually tailor preventative and curative health-care right down to the level of an individual's genetic shortcomings. The ultimate aspiration is to unravel the complex aetiology of the common diseases that are a major burden to Western society. These include cancer, cardiovascular disease and type 2 diabetes mellitus; ailments that were collectively responsible for over 18% of Australian medical expenditure in 2000-01 and 70% of deaths in 2001 (Australian Institute of Health and Welfare, 2005). Once the genetic susceptibility and causative environmental factors of such diseases are understood, it will then, at least theoretically, be possible to customise recommendations for a person's health-care and life-style according to the risk factors that he or she has inherited (Haga et al., 2003; Ordovas and Mooser, 2004). For example, it is thought that many common clinical diagnoses, such as type 2 diabetes mellitus or hypertension, will be redefined into multiple subgroups based

on distinct molecular mechanistic causes: each subtype requiring its own treatment strategy for the best response (Heijmans *et al.*, 2002; Lindpaintner, 2002). In addition, if relationships between genotype and cancer risk can be established, it may also be possible to develop measures to mitigate an individual's probability of developing malignancies. Other multifactorial, polygenic disorders that are attracting such attention include psychiatric illness, stroke, obesity, neurodegenerative conditions including Parkinson and Alzheimer diseases, and inappropriate inflammatory/autoimmune responses including asthma and arthritis (Van Eerdewegh *et al.*, 2002; El-Omar *et al.*, 2003; de la Chapelle, 2004; Ordovas and Mooser, 2004; Zee *et al.*, 2004; Bernardini *et al.*, 2005).

However, despite current progress, it will be many years or perhaps many decades, before personalised genomic medicine will have widespread usefulness for predicting an individual's disease predispositions and metabolic limitations. This is due to a number of economic, ethical and technological hurdles that will not be considered in the scope of this thesis, but are discussed in detail by other authors (Issa, 2000; Lindpaintner, 2002; Haga *et al.*, 2003; Paul and Roses, 2003). Nonetheless, in the meantime, information gleaned from studying basic gene regulation and genetic variation can better our comprehension of general biological concepts and lead to medical advances applicable to all humans, or to large genetic subgroups. For example, research focused on human drug metabolising enzymes is increasingly being taken into account at early stages of drug design and development, as well as in rational drug use, resulting in more targeted efforts in medical progress, and accordingly, improvements in safety and treatment outcomes. Genomic research is identifying new drugable targets, as well as revealing the mechanisms of previously observed drug-drug interactions (Lindpaintner, 2002;

Ross and Ginsburg, 2002; Evans and McLeod, 2003). Advances in molecular epidemiology are contributing to the identification of subpopulations (based on traits such as age, gender or genetic predisposition) that may have an elevated risk of developing cancer or other diseases following exposure to certain causative environmental factors (Perera, 2000). A better understanding of molecular targets and pharmacological parameters is also identifying specific patient genetic subgroups for whom certain drug regimes are most appropriate, or conversely, people for whom the use of specific therapies carry significant risk of harm. In time this will hopefully translate into improved prescribing, more accurate dosing and a reduction in adverse drug reactions (Weinshilboum, 2003; Ross *et al.*, 2004), and is indeed already a reality for a select group of examples in some clinics, such as 6-mercaptopurine, irinotecan and several antidepressants (Kootstra-Ros *et al.*, 2006).

1.2 Genetic variation in metabolic pathways; implications for disease susceptibility and treatment

While the multifactorial causes of complex diseases are still poorly defined, it has long been evident that not all individuals exposed to a given set of external triggers will develop ill-health. Thus the chance that any particular individual will develop one or more such ailments is presumably determined not only by the environmental risk factors to which they are exposed, but also by intrinsic determinants such as age, gender and genetic disposition (Sanchez *et al.*, 2001). Genetic risk for complex, multifactorial diseases is proving particularly difficult to characterise, as it is likely to be a product of seemingly trivial variations in multiple genes spread over the entire genome (Paul and Roses, 2003; Pharoah *et al.*, 2004); however, variation in genes that affect an individual's ability to detoxify and remove harmful chemicals from the body, whether of endogenous or external origin, is one example where genetic variance may affect disease susceptibility. Many dietary constituents and environmental substances are potentially toxic or carcinogenic; therefore, enzymes and transporters that inactivate such compounds or facilitate their removal from the body minimise the damage incurred in the event of exposure (Perera, 2000; Perera and Weinstein, 2000; Hoffmann and Kroemer, 2004). In addition, metabolic enzymes and transport proteins affect the local and circulating levels of endogenous substances, many of which, such as steroids, have important homeostatic roles and must be maintained at appropriate levels (Nebert, 1994). Thus, sequence variation in the coding or regulatory regions of these genes, or in the factors that control their expression, may partially explain the observed variance in disease susceptibility between individuals (Perera and Weinstein, 2000; Desai *et al.*, 2003; Leslie *et al.*, 2005).

It is not only the likelihood of developing ill health or the severity of disease that can be affected by interindividual genetic variation and environmental stimuli. The degree to which standard medical intervention for a disease is successful also varies extensively between patients. The molecular mechanisms that determine the pharmacokinetic and pharmacodynamic properties of therapeutic drugs in humans can also be subject to genetic diversity and variation brought about by external cues. Thus, individuals can differ, or the same patient may vary temporally, in how effectively a drug will reach its target, the extent and type of modifications a drug will undergo, a drug's clearance, the extent to which undesirable adverse effects are triggered and the clinical response obtained (Evans and Relling, 1999; Lindpaintner, 2002; Evans and McLeod, 2003; Weinshilboum, 2003). These issues extend not only to treatment of non-communicable disorders such as those mentioned above, but are equally applicable to the treatment of infectious diseases and pharmaceutical management of pain.

It is evident that the more complete our understanding of human metabolism and its regulation becomes, the more effectively we will be able to devise and evaluate possible strategies for treating, or even preventing, disease and discomfit. Therefore, this thesis is designed to contribute further insights into the mechanisms regulating the metabolism of xenobiotic substances, endocrine signalling molecules and endogenous waste products in humans. In particular, this research investigates the regulation of the uridine diphosphate (UDP)-glucuronosyltransferase (*UGT*) genes, a subset of the many genes that participate in the biotransformation of lipophilic chemicals in humans and other higher organisms. General mechanisms, as well as those that may contribute to the differences that occur between individuals, are considered.

1.3 Biotransformation of small lipophilic molecules

The human body is constantly exposed to many potentially dangerous compounds, some arising from internal metabolic and catabolic processes, others being encountered directly through dietary intake, therapeutic and illicit drug use, or environmental contact (Wogan *et al.*, 2004). Lipid-soluble chemicals of low molecular weight typically enter cells by passive diffusion across the membrane, those from external sources being readily absorbed via the gastrointestinal tract and lungs, and to a lesser extent, the skin (Artursson and Karlsson, 1991; Walle and Walle, 1999; Gunaratna, 2000). To avoid harm from carcinogens and toxins, or undesirable alterations in gene expression patterns by ligands, the body must regulate the intracellular concentration and biological activity of these substances through

chemical modification and/or active efflux through the cellular membrane. Members of the adenosine triphosphate (ATP)-binding cassette transporter protein superfamily, such as P-glycoprotein and the multidrug resistance proteins (MRP) can transport a large variety of unaltered lipophilic molecules out of the cell, returning them to the intestinal lumen or removing them from the systemic circulation for excretion via bile or urine. However, efficient efflux of small lipophilic chemicals, particularly by MRP transporters, normally requires or is significantly enhanced by prior biotransformation (Hoffmann and Kroemer, 2004).

Biotransformation is the act by which a chemical substrate is structurally modified by one or more enzyme-catalysed reactions. These reactions include such diverse processes as oxidation, reduction, conjugation and nucleophilic trapping, and are classically divided into two stages. However, although categorised as "Phase I/functionalisation" and "Phase II/conjugative" reactions, biotransformation events do not necessarily progress through the two stages sequentially, nor do they adhere strictly to the accepted generalisations for each class of reaction (Josephy et al., 2005). Therefore, for clarity, this thesis avoids the use of this nomenclature, referring specifically to the enzymes or reactions in question. Furthermore, while biotransformation enzymes also include oxygenases, dehydrogenases, reductases, deaminases, aminotransferases, methyltransferases and hydrolases, the description that follows concentrates primarily on the relevance of the cytochrome P450 (CYP) oxidoreductases and the glucuronosyltransferases, sulphotransferases, Nacetyltransferases and glutathione-S-transferases to human health.

Overall, biotransformation is considered to be protective in nature, as the terminal metabolic products of this process tend to be less biologically active than their parent compounds and more readily excreted. Oxidation, a common biotransformation

event predominately catalysed by the CYP superfamily, often results in the direct inactivation and elimination of a substrate. Moreover, with the exception of a small number of known examples, metabolites resulting from conjugative reactions are nearly always lacking in significant pharmacological or toxicological activity; glucuronides especially being renowned for their inert nature. In addition, many glucuronides, sulphates and glutathione-conjugates are excellent substrates for MRP transporters; thus they are more readily excreted from the cell than their parent compound. Elimination from the body is further enhanced by the inability of conjugated organic anions to passively re-enter cells by diffusion, due to their decreased lipid-solubility (Sanchez *et al.*, 2001; Hoffmann and Kroemer, 2004).

It is important to note, however, that not all oxidations or conjugations generate inactive metabolites. Instead, metabolites from either type of biotransformation may have similar, increased or completely new actions compared with the original compound. One therapeutically relevant example is the requirement for codeine to be metabolised to morphine by CYP2D6 before an analgesic effect is experienced. Furthermore, a minor glucuronidation product of morphine, morphine-6-glucuronide, is believed to exhibit greater analgesic potency than morphine itself (Sawe *et al.*, 1985; Osborne *et al.*, 1992; Caraco *et al.*, 1997). The glucuronides of certain steroids, retinoids and bile acids are also highly bioactive, with retinoyl betaglucuronide having similar therapeutic benefit to retinoic acid but without the associated adverse effects (Ritter, 2000; Barua and Sidell, 2004). In contrast to these beneficial examples, oxidation and reduction reactions can also generate highly reactive electrophilic intermediates or nucleophilic radicals, capable of interacting deleteriously with cellular macromolecules such as proteins and DNA. Many environmental and dietary carcinogens, such as those found in tobacco smoke and burnt meat have little or no mutagenic activity per se, but become highly reactive in vivo when metabolised by CYP or other enzymes. Examples of CYP-activated carcinogens include 4-(methylnitrosamino)-1-(3-pyridyl)-1-butanone and other nitrosamines. polycyclic aromatic hydrocarbons such as benzo[*a*]pyrene, heterocyclic aromatic amines including 2-amino-1-methyl-6-phenylimidazo[4,5b]pyridine (PhIP), as well as aldehydes, phenols and aflatoxin (Nowell *et al.*, 1999; Guengerich, 2000; Fang et al., 2002; Wiener et al., 2004; Wogan et al., 2004). A limited number of conjugates have also been shown to form DNA or protein adducts, enhance tumourigenesis, trigger toxic/immunological reactions or possess decreased water-solubility. Certain glutathione-S-transferase, *N*-acetlytransferase and sulphotransferase catalysed reactions have been found to contribute to the activity of environmental carcinogens, while there is some evidence that acyl glucuronides of carboxylic acid moieties such as non-steroidal anti-inflammatory drugs (NSAIDs) may trigger idiosyncratic adverse drug reactions. Furthermore, D-ring glucuronides of oestrogens such as oestradiol and ethinyl-oestradiol can cause cholestasis in rats, although ethinyl-oestradiol is used safely in human females as an oral contraceptive. Finally, while a given glucuronide may be inert, it may be subject to hydrolysis by β -glucuronidase to the parent aglycone, which may then be re-absorbed, or to a toxic intermediate as in the case of cleavage of 3-benzo[a]pyrene glucuronide. For some substances, this may actually lead to a prolonged exposure compared with alternative metabolic pathways. Acyl glucuronides are especially susceptible to hydrolysis and futile cycling, a factor that can be clinically important in patients with renal dysfunction (Kari et al., 1984; Minchin et al., 1992; Chou et al., 1995; Sperker et al., 1997; Grubb et al., 1999; Bailey and Dickinson, 2003; Sanchez Pozzi et al., 2003; Anders, 2004; Josephy et al., 2005).

The actual impact of deleterious biotransformation events rests in the body's ability to either circumvent them by directing parent compounds into alternative pathways that generate less harmful products, or by further metabolising reactive intermediates to inert substances. Glucuronidation is one such means by which cells are often protected from the effects of electrophilic metabolites. For example, PhIP is oxidised in the liver to *N*-OH–PhIP, a metabolite capable of forming DNA adducts that can also be further metabolised to even more highly reactive electrophilic species by acetyltransferases and sulphotransferases. Alternatively *N*-OH–PhIP can be inactivated by *N*-glucuronidation and safely excreted, avoiding the formation of the PhIP *N*-acetoxy and *N*-sulphonyoxy esters and their consequent DNA adducts, and hence the potential for carcinogenesis (see Figure 1.1) (Nowell *et al.*, 1999; Hecht, 2003).



Figure 1.1: Alternative metabolic pathways for 2-amino-1-methyl-6phenylimidazo[4,5-*b*]pyridine (PhIP) after CYP activation, resulting in DNA adducts or detoxification. Figure generated from information detailed by Nowell *et al.* (1999). Thus, the balance of biotransformation enzymes present in an individual controls the metabolic fate of many small lipophilic pharmaceutical drugs, xenobiotics and endogenous ligands, determining both their efficacy and toxicity. Genetic variations and environmental or inherent triggers that shift this balance will alter the individual's response to drug therapies and change their likelihood of developing disease, whether for better or for worse. As glucuronidation is a key mechanism in the metabolism of many therapeutic compounds, dietary constituents, carcinogens, toxins, steroids, fatty acids, neurotransmitters and endogenous waste products such as bilirubin, an understanding of the factors that regulate this pathway is paramount to improved drug design and disease prevention strategies.

1.4 Glucuronidation and the UDP-glucuronosyltransferases

Glucuronidation is the process by which glucuronic acid, from the nucleotide sugar donor uridine 5'-diphosphoglucuronic acid, is transferred to a suitable functional group on an acceptor molecule. As a result, a β -D-glucopyranosiduronic acid derivative (glucuronide) of the original substrate is generated and UDP is released (see Figure 1.2). Multiple functional groups are amenable to glucuronidation, including hydroxyl (phenol and alcohol), carboxyl and thiol residues, as well as primary, secondary and tertiary amino groups. This diversity allows a large array of molecules from unrelated chemical classes to form glucuronides (Radominska-Pandya *et al.*, 1999; King *et al.*, 2000; Tukey and Strassburg, 2000). The glucuronidation reaction is thought to proceed through an acid/base bimolecular nucleophilic substitution (S_N2) reaction mechanism, as reviewed by Radominska-Pandya *et al.* (1999), and is catalysed by the members of a large enzyme family known as the UDP-glucuronosyltransferases (UGT; EC 2.4.1.17). The UGTs are in

turn members of the broader superfamily of UDP-glycosyltransferases, enzymes that transfer glycosyl groups to lipophilic substances from a variety of UDP-sugars. All UDP-glycosyltransferases, including UGTs, share a 44-amino acid characteristic "signature sequence" in their carboxyl-terminal half that is thought to be involved in the binding of the UDP moiety of the nucleotide sugar (Mackenzie *et al.*, 1997; Mackenzie *et al.*, 2005b).



Figure 1.2: Schematic representation of the glucuronidation reaction. Adapted from Timmers *et al.* (1997).

UGTs, or the glucuronides that they produce, have been identified in a diverse range of vertebrate species, including humans, other primates, other mammalian species (such as cow, dog, cat, sheep, pig and various rodents), birds and fish (Ohyama et al., 2004; Mackenzie et al., 2005b). For those species that have been investigated in detail, multiple UGT forms have been found, with each enzyme having its own distinct set of target substrates. Thus, each organism can metabolise an extensive range of lipophilic chemicals through the collective activity of its UGTs. Furthermore, many small compounds are recognised by more than one UGT enzyme within a species due to a considerable degree of overlap in UGT substrate selectivity. This allows for a valuable measure of redundancy in the glucuronidation system; as a result, glucuronidation is not only a versatile metabolic pathway, but also a relatively robust one. Accordingly, isolated mutations in single human UGT genes generally have no overt link to disease, with the notable exception of mutations in UGT1A1, as discussed in section 1.8.6.1. However, this cannot be taken to imply that UGTs other than UGT1A1 are unimportant in the prevention of disease. While over 1000 mutated genes have been linked to disorders caused by single gene aberrations, their incidence is low and they account for only a small proportion of the total disease burden in humans (van Ommen, 2002). Indeed, mutations in UGT1A1 that would be fatal without treatment or liver transplant only affect one in 1×10^6 newborns (Bosma, 2003). In contrast, conditions that have genetic predisposition as one component of a more complex aetiology are common, and it is certainly feasible that UGTs can influence the development of some such disorders through more subtle changes in metabolism.

The mammalian UGTs have been divided into two major families; UGT1 and UGT2, on the basis of amino acid sequence similarities (see Figure 1.3). In general,

members within each family share more than 45% amino acid identity but are $\leq 45\%$ identical to UGTs of the alternative family. The UGT2 family has been further divided into subfamilies; UGT2A and UGT2B; the members of each sharing $\geq 70\%$ amino acid identity. To date, humans are known to possess 19 potentially functional UGTs, as well as 9 pseudogenes (possessing at least one incomplete exon) and many gene remnants. The intact human genes include 9 members of the *UGT1* family (known as *UGT1A* genes), three *UGT2A* genes and 7 members of the *UGT2B* subfamily (Mackenzie *et al.*, 1997; Mackenzie *et al.*, 2005b).



Figure 1.3: Dendogram depicting relationships between the primary amino acid sequences of human UGTs. Alignment of the amino acid sequences was performed with Clustal X v1.81 (Thompson *et al.*, 1997) and visualisation of the dendogram was achieved using TreeView v1.6.6 (Page, 1996).

1.5 Genomic organisation of the human *UGT* gene family

1.5.1. The human UGT1 locus

The human UGT1A coding sequences are found on a single locus located on chromosome 2q37, spanning approximately 200 kb. This locus contains 13 unique exons (exon 1), which encode the N-terminal portion of 13 potential UGT1A forms, and 4 exons (exons 2-5) that are shared by all full-length UGT1A transcripts as illustrated in Figure 1.4 (Ritter et al., 1992b; Gong et al., 2001). As a result, the UGT1A enzymes possess unique amino-terminal ends that provide functional diversity, while the 245 carboxyl-terminal amino acids of each are identical. Each UGT1 unique first exon is preceded by a core promoter region that facilitates transcription of the corresponding UGT1A message, a process that is thought to occur independently of other UGT1A members. mRNA transcripts for specific UGT1A forms have been found to be initiated at transcription start sites (TSSs) located 16 to 112 nucleotides upstream of the initiation codon of their appropriate exon 1 (Ritter et al., 1992b; Gong et al., 2001; Gregory et al., 2003). Each exon 1 sequence is followed by a donor splice site, allowing it to be joined to the first 5' receptor splice site in the UGT1A locus, which precedes exon 2. As such, only the first exon 1 sequence on each mRNA transcript can be spliced to the shared exons, and alternative UGT1A transcripts are generated from alternative transcription initiation events followed by conventional splicing (Ritter et al., 1992b). Although not strictly separate genes, the accepted convention for the UGT1A family is to consider the spliced transcripts as arising from such; thus they are named accordingly (Mackenzie et al., 2005b).



Figure 1.4: Schematic representation of the human *UGT1A* **locus.** Each exon 1 is represented by a coloured rectangle (not drawn to scale), and its position relative to the common exons 2 to 5 (shown in grey) is indicated. The grey common exons are numbered 2-5, left to right: the black common exon represents a newly discovered exon, named 5b (Levesque *et al.*, 2007b). Pseudogenes are labelled with a "P". The entire *UGT1A* locus extends over approximately 200 kb. Alternative splicing events resulting in UGT1A1, UGT1A6 and UGT1A1_v2 mRNA transcripts are illustrated as examples. Figure adapted from the latest UGT nomenclature update as published by Mackenzie *et al.* (2005b).

It was only very recently that a fifth common exon was discovered in the human *UGT1A* locus (Levesque *et al.*, 2007b). Usage of this new exon, named common exon 5b, results in truncated UGT mRNA transcripts such as UGT1A1_v2 (Figure 1.4). In this case, the resulting protein (UGT1A1_i1) behaves as an inhibitor of full-length UGT1A1 function. Truncated UGT1A mRNAs have been demonstrated in the liver, kidney, colon, oesophagus and small intestine, and it is anticipated that all UGT1A mRNAs can be spliced to form truncated variants (Levesque *et al.*, 2007b).

Of the UGT1A first exons, four contain mutations that render them non-functional.

These UGT1 pseudogenes are UGT1A2P, UGT1A11P, UGT1A12P and UGT1A13P.

The remaining genes, UGT1A1, and UGT1A3 through to UGT1A10, are separated

into clusters based on sequence relatedness (see Figure 1.3). The UGT1A3, UGT1A4
and *UGT1A5* genes produce proteins of greater than 93% homology, while the enzymes of the *UGT1A7-1A10* cluster share between 89 and 95% identity.

1.5.2. The human UGT2 locus

In contrast to the *UGT1* gene, the *UGT2* genes are almost all generated from discreet genes. This includes all of the *UGT2B* subfamily and *UGT2A3*. The only known exceptions are *UGT2A1* and *UGT2A2*, which are generated through exon sharing in a similar manner to the *UGT1A* locus. All of the *UGT2* genes are found on chromosome 4 at position 4q13, with the *UGT2A* and *UGT2B* members interdispersed as depicted in Figure 1.5. The *UGT2B* genes all consist of six exons and share similar intron/exon boundaries, although intron lengths vary between genes. Despite originating from separate genes, the carboxyl halves of the UGT2B enzymes are still highly conserved within the UGT2B family and, to a lesser extent, with the UGT1A subfamily (Turgeon *et al.*, 2000; Tukey and Strassburg, 2001; Mackenzie *et al.*, 2005b).



Figure 1.5: Schematic representation of the human UGT2 locus. Each UGT2B gene, consisting of six exons is represented by a coloured rectangle (not drawn to scale), except 2A1/2, which represents seven exons. The UGT2A1 and UGT2A2 genes contain unique first exons (2A1 and 2A2) and a shared set of five downstream exons (exons 2 to 6 in grey); their exon arrangement is depicted at the bottom of the figure. Pseudogenes are labelled with a "P". The entire UGT2 locus extends over approximately 1.45 Mb. Figure adapted from the latest UGT nomenclature update as published by Mackenzie *et al.* (2005b).

Of the UGT2 family, disparately little is known about the role of the UGT2A forms in human health and disease. Human UGT2A1 has been found to be expressed mainly in olfactory tissue, and is known to conjugate phenolic, aliphatic and monoterpenoid odorants, as well as certain coumarins, flavonoids, therapeutic drugs and steroid hormones (Jedlitschky *et al.*, 1999). UGT2A2 transcripts have been detected in liver and small intestine, but the substrate specificity of the protein is currently unknown (Tukey and Strassburg, 2001). UGT2A3 is the most recent member of the human UGT2A family to be recognised, being first reported by a project specifically designed to identify novel human secreted and transmembrane proteins in 2003 (Clark *et al.*, 2003). Neither expression nor substrate data is currently available for this UGT form. Therefore, whilst presumably important in its own right, this subfamily will not be addressed further by this thesis.

1.6 UGT substrates

The known substrates of human UGTs are numerous, varied and continually increasing. Accordingly, the following summary of human UGTs and their substrates (Table 1.1) is not an exhaustive list, but rather aims to highlight some important target substances, or substrate classes, of each UGT form. More comprehensive lists of substrates are given in the appropriate chapters for those forms that have been investigated in detail in the presented work.

1.7 Location and distribution of human UGT gene products

Within the cell, UGT proteins are anchored in the membrane of the endoplasmic reticulum. This is achieved by virtue of a hydrophobic 17-amino acid domain located in the carboxyl-terminal portion of each enzyme. Only approximately 20 amino acids of the UGT proteins are located on the cytosolic side of the endoplasmic reticulum

UGT	Substrate	Relevance/Action	References
UGT1A1	Bilirubin	Toxic waste product of haem metabolism	Bosma et al. (1994)
	Oestrogens	Steroid hormones	Cheng et al. (1998b)
	Paracetamol	Analgesic; hepatotoxic and nephrotoxic	Court <i>et al.</i> (2001)
	SN-38	Chemotherapeutic agent	Hanioka et al. (2001)
	Benzo[<i>a</i>]pyrene metabolites	Carcinogens	Dellinger et al. (2006)
	Thyroid hormones	Development and homeostasis	Findlay et al. (2000)
	N-hydroxy-PhIP	Carcinogen	Malfatti and Felton (2001)
UGT1A3	Oestrogens	Steroid hormones	Cheng et al. (1998b)
	Amines	Pharmaceuticals	Breyer-Pfaff et al. (2000)
	Benzo[<i>a</i>]pyrene metabolites	Carcinogens	Dellinger et al. (2006)
	NSAIDs	Anti-inflammatory agents	Green et al. (1998a)
	Bile acids	Endogenous detergents	Gall et al. (1999)
UGT1A4	Primary amines	Carcinogens	Green and Tephly (1996)
	Tertiary amines	Pharmaceuticals such as antipsychotics	Breyer-Pfaff et al. (2000)
	Progestins	Steroid hormones	Green and Tephly (1996)
UGT1A5	1-hydroxypyrene	Polyaromatic hydrocarbon metabolite	Finel <i>et al.</i> (2005)
UGT1A6	Serotonin	Neurotransmitter	Krishnaswamy <i>et al.</i> (2003)
	Planar phenols	Semiochemicals, toxins, carcinogens	Ebner and Burchell (1993)
	Paracetamol	Analgesic; hepatotoxic and nephrotoxic	Court <i>et al.</i> (2001)
UGT1A7	Benzo[<i>a</i>]pyrene metabolites	Carcinogens	Dellinger et al. (2006)
	SN-38	Chemotherapeutic agent	Gagne et al. (2002)

Table 1.1:	Human	UDP-Glu	curonosyl	transferases	and selected	substrates.

Table 1.1 continued.

UGT	Substrate	Relevance/Action	References
UGT1A8	Oestrogens	Steroid hormones	Cheng et al. (1998a)
	Flavonoids and coumarins	Antioxidants	Cheng et al. (1998a)
	N-hydroxy-PhIP	Carcinogen	Nowell et al. (1999)
	Mycophenolic acid	Immunosuppressant	Bernard and Guillemette (2004)
UGT1A9	Thyroid hormones	Development and homeostasis	Findlay et al. (2000)
	Benzo[<i>a</i>]pyrene metabolites	Carcinogens	Dellinger et al. (2006)
	Paracetamol	Analgesic; hepatotoxic and nephrotoxic	Court <i>et al.</i> (2001)
	Mycophenolic acid	Immunosuppressant	Picard et al. (2005)
	SN-38	Chemotherapeutic agent	Gagne et al. (2002)
	N-hydroxy-PhIP	Carcinogen	Nowell et al. (1999)
	NNAL	Carcinogen	Ren et al. (2000)
	Propofol	Anaesthetic	Soars <i>et al.</i> (2004)
UGT1A10	Flavonoids	Antioxidants	Lewinsky et al. (2005)
	Oestrogens	Steroid hormones	Strassburg et al. (1998a)
	Benzo[<i>a</i>]pyrene metabolites	Carcinogens	Dellinger et al. (2006)
UGT2B4	Fatty acids	Cellular membrane component	Turgeon et al. (2003b)
	Bile acids	Endogenous detergents	Ritter et al. (1992a)
	Oestrogens	Steroid hormones	Turgeon et al. (2001)
	Eugenol	Antiseptic, anaesthetic	Turgeon et al. (2001)
UGT2B7	Fatty acids	Cellular membrane component	Turgeon et al. (2003b)
	Bile acids	Endogenous detergents	Ritter et al. (1992a)
	Oestrogens	Steroid hormones	Turgeon et al. (2001)
	Retinoids	Development and differentiation	Samokyszyn et al. (2000)
	Opioids	Analgesics	Coffman et al. (1998)

UGT	Substrate	Relevance /Action	References		
UGT2B7	NSAIDs	Anti-inflammatory agents	Jin et al. (1993)		
continued	Zidovudine	Anti-retroviral agent	Barbier et al. (2000)		
	Mycophenolic acid	Immunosuppressant	Picard et al. (2005)		
	NNAL	Carcinogen	Ren et al. (2000)		
	Eugenol	Antiseptic, anaesthetic	Turgeon et al. (2001)		
UGT2B10	Fatty acids	Cellular membrane component	Turgeon et al. (2003b)		
	Nicotine	Addictive stimulant	Kaivosaari et al. (2007)		
UGT2B11	Fatty acids	Cellular membrane component	Turgeon <i>et al.</i> (2003b)		
UGT2B15	Androgens and oestrogens	Steroid hormones	Turgeon <i>et al.</i> (2001)		
	Flavonoids and coumarins	Antioxidants	Green et al. (1994)		
	Eugenol	Antiseptic, anaesthetic	Green et al. (1994)		
UGT2B17	Androgens	Steroid hormones	Turgeon <i>et al.</i> (2001)		
	Coumarins	Antioxidants	Turgeon et al. (2003a)		
	Anthraquinones	Pharmaceuticals	Turgeon et al. (2003a)		
	Eugenol	Antiseptic, anaesthetic	Turgeon et al. (2003a)		
UGT2B28	Bile acids	Endogenous detergents	Levesque et al. (2001)		
	Eugenol	Antiseptic, anaesthetic	Levesque et al. (2001)		
	Androgens	Steroid hormones	Levesque et al. (2001)		

Table 1.1 continued.

PhIP: 2-amino-1-methyl-6-phenylimidazo[4,5-*b*]pyridine; NSAID: non-steroidal anti-inflammatory drug; SN-38: 7-ethyl-10-hydroxycamptothecine;. NNAL: 4-(methylnitrosamino)-1-(3-pyridyl)-1-butanol;

membrane, with the majority of each protein residing within the lumen. Nascent UGT protein molecules include a signal peptide that is cleaved after integration into the endoplasmic reticulum membrane, generating a mature protein of approximately 505 amino acids (Iyanagi *et al.*, 1986; Mackenzie, 1986; Meech and Mackenzie, 1997a).

In their native environment, UGTs are thought to form dimers, including heterodimers, and may also form higher order complexes with other cellular proteins such as CYP enzymes. The formation of UGT heterodimers is potentially important *in vivo*, as it may increase the rate of glucuronidation and/or the breadth of substrates metabolised by co-expressed UGT enzymes. Conversely, UGT mutants that behave in a dominant negative manner may exacerbate disease (*e.g.* Crigler-Najjar syndrome type II) by reducing the effectiveness of remaining functional UGT protein through formation of inactive dimers (Koiwai *et al.*, 1996; Ikushiro *et al.*, 1997; Meech and Mackenzie, 1997b; Ghosh *et al.*, 2001; Ishii *et al.*, 2001; Kurkela *et al.*, 2003; Ishii *et al.*, 2004). Association with CYPs or other proteins may also modulate UGT activity (Taura *et al.*, 2000; Taura *et al.*, 2004).

Each UGT enzyme has been found to have its own unique tissue expression profile, which is subject to both developmental and cell-type determinants. Table 1.2 summarises the current understanding of adult human UGT mRNA distribution by tissue type. It should be noted that there is some disagreement within the literature about the presence of certain UGT forms in several organs (as indicated). Most of these discrepancies have likely arisen from either differing sensitivities of detection or polymorphic expression between individuals. For example, UGT1A1, UGT1A3 and UGT1A6 have all been demonstrated to only be expressed in the stomach of approximately one third of individuals (Strassburg *et al.*, 1998b), while UGT1A10 transcripts were only detected in lung when amplified separately rather than in multiplex polymerase chain reaction (PCR) (Dellinger *et al.*, 2006). The liver is considered to be the singly most important organ for glucuronidation in humans.

	Liver	Kidney	Lung	Brain	Oesophagus	Stomach	Small intestine	Colon	Prostate	Breast	Testis	Skin
UGT1A1	+	+	-	-	-	+/-	+/-	+	+	+/-	+	-
UGT1A3	+	ND	-	ND	-	+/-	+/-	+	ND	+	ND	ND
UGT1A4	+	+	-	-	-	-	+/-	+	ND	+	ND	ND
UGT1A5	§	ND	ND	ND	-	-	§	-	ND	ND	ND	ND
UGT1A6	+	+	+/-	+	+/-	+/-	+/-	+	ND	ND	+	ND
UGT1A7	-	ND	-	ND	+	+	-	-	ND	ND	ND	ND
UGT1A8	-	-	-	ND	+/-	-	+/-	+	ND	+	ND	ND
UGT1A9	+	+	-	-	+	-	-	+	+	+	+	+
UGT1A10	-	-	+/-	ND	+	+	+	+	ND	ND	ND	ND
UGT2B4	+	+	+	ND	-	ND	+/-	-	+	+	+	+
UGT2B7	+	+	+/-	+/-	+/-	-	+/-	+	-	+	-	ND
UGT2B10	+	+/-	+/-	-	+/-	+/-	-	+/-	+	+	+	ND
UGT2B11	+	+	+	ND	ND	ND	ND	ND	+	+	-	+
UGT2B15	+	+	+	ND	+/-	ND	+/-	+	+	+	+	+
UGT2B17	+	+	+	ND	-	ND	ND	ND	+	+	+	+
UGT2B28	+	-	-	ND	ND	-	-	ND	-	+	-	-

 Table 1.2: Distribution of UDP-Glucuronosyltransferase mRNA in humans.

Data compiled from Munzel *et al.* (1996), Strassburg *et al.* (1998a), Albert *et al.* (1999), King *et al.* (2000), Tukey and Strassburg (2000), Levesque *et al.* (2001), Tukey and Strassburg (2001), Turgeon *et al.* (2001), Vallee *et al.* (2001), Zheng *et al.* (2002), Belanger *et al.* (2003), Finel *et al.* (2005), Chouinard *et al.* (2006), Dellinger *et al.* (2006) and Kaivosaari *et al.* (2007). The majority of information presented was originally compiled by Dr. Philip Gregory. "+" UGT mRNA is present in tissue; "-" UGT mRNA is absent in tissue; "+/-" UGT mRNA presence is either known to be polymorphic or remains a subject of controversy. §Small amounts of UGT1A5 mRNA have been demonstrated in human liver (Finel *et al.*, 2005) and duodenum (Tukey and Strassburg, 2001), but the functional relevance of these discoveries is still to be determined. ND = not determined.

1.8 Causes and consequences of variability in UGT expression and activity

Despite the robust nature of glucuronidation in humans, this metabolic pathway is still susceptible to factors that either change the activity of UGT enzymes or affect their expression profiles. Glucuronidation is a major route of elimination for many xenobiotics and endogenous molecules; therefore, deviation from the normal rate of glucuronidation can modulate the concentration and/or effect of compounds that are UGT substrates. In the case of pharmaceutical drugs, altered metabolism can have acute clinical consequences in the form of adverse drug reactions or lack of drug efficacy, particularly for drugs that have narrow therapeutic indices. On the other hand, changes in UGT activity may either improve or impair the body's ability to protect itself against chemical insult and, over the longer term, protect against or predispose to cancer or other diseases caused by chronic chemical toxicity. It has been suggested that even small changes in glucuronidation may lead to disproportionately large increases in bioactivation for substances where the former is a quantitatively major pathway of elimination, especially if there are no alternative eliminating pathways, or those that exist are readily saturable (Wells *et al.*, 2004). Known influences on glucuronidation rates in humans include developmental stage, gender, genetic variation, pregnancy, xenobiotic exposure and health status.

1.8.1. UGT expression during human development

Glucuronidation rates in the human foetus and neonate are significantly lower than the adult for most substrates, and increase during the first few months or years of life. The time frame required for glucuronidation to reach maturity depends on the substrate involved, presumably due to differential regulation of *UGT* genes. While the significance of low UGT activity in the foetus is uncertain, poor understanding of neonatal glucuronidation has had fatal consequences (Robertson, 2003).

Several features of foetal development appear to contribute to inferior glucuronidation at this stage of life, including the structural immaturity of important metabolic organs, and the temporal control of UGT expression. Although UGT protein is expressed in an adult-like distribution in the metanephric kidney around 12 weeks gestation (Hume et al., 1995), the foetal kidney has low blood flow and excretes metabolites into the amniotic fluid, from which they can be re-absorbed (Morgan, 1997). On the other hand, the foetal liver lacks much of the UGT protein expressed in the adult organ. In one study, no UGT transcripts were detected in human foetal liver at 20 weeks gestation (Strassburg et al., 2002a). Studies using microsomes or homogenates from foetal liver of 15-27 weeks gestation have revealed low rates of glucuronidation for substrates such as bilirubin, 2-aminophenol, testosterone, morphine and 1-naphthol, but higher rates for oestrone and serotonin. The latter were found to be glucuronidated at 30% and > 100% of adult rates respectively, with the next highest rate of conjugation being for morphine at 10-16% (Kawade and Onishi, 1981; Pacifici et al., 1982; Leakey et al., 1987). Although outdated due to improvements in available human UGT anti-sera, an immunoblot analysis of foetal liver supports the hypothesis that most UGT forms are underexpressed before birth (Coughtrie et al., 1988).

During pregnancy, the foetus may be partially metabolically protected by the placenta. The human placenta expresses multiple UGT enzymes, and glucuronides formed *in situ* are preferentially transported into the maternal circulation (Collier *et al.*, 2002b; Collier *et al.*, 2004). Placental UGT expression appears to vary with gestational age; UGT1A and UGT2B forms are present in placenta during the first

trimester, however, only UGT2B forms are evident at term (Collier *et al.*, 2002a; Collier *et al.*, 2002b). Nonetheless, maternal metabolism is probably the major determinant of foetal exposure to potentially harmful UGT substrates (Morgan, 1997).

In human neonates, the rates of hepatic glucuronidation for many substances initially remain low (Leakey et al., 1987). As a result, newborns are particularly sensitive to the adverse effects of drugs that are glucuronidated or that inhibit glucuronidation, such as chloramphenicol and novobiocin, respectively (Robertson, 2003). Adult-like glucuronidation of most substrates appears to then develop steadily over the first few months of life. For example, at birth the elimination half-lives of extensively glucuronidated drugs such as morphine, naloxone, lorazepam and zidovudine exceed three times those observed when administered to adults, but these differences are lost between two and six months of age (de Wildt et al., 1999; Ginsberg et al., 2002; Bouwmeester et al., 2004). Correspondingly, at six months of age, UGT1A1, UGT1A3, UGT1A4, UGT1A6, UGT2B7, UGT2B10 and UGT2B15 mRNA transcript levels are comparable to those in adult liver, and UGT1A1, UGT1A6 and UGT2B7 protein levels in paediatric and adult liver samples are similar (Strassburg et al., 2002a). Furthermore, in liver microsomes, adult-level activities towards bilirubin and 2-aminophenol can be observed within three months of birth (Onishi et al., 1979). However, as mentioned earlier, the temporal postpartum acquisition of UGT activity does not occur equally swiftly for all UGTs. Trifluoperazine glucuronidation by liver microsomes does not plateau at adult levels until approximately 17 months of age (Miyagi and Collier, 2007), and at the molecular level, hepatic UGT1A9 and UGT2B4 mRNA content is still low in infants at 24 months relative to adults (Strassburg et al., 2002a). In addition, young children (6-24

months) may have much lower capacities than adults for glucuronidation of substances such as oestrone and buprenorphine (Strassburg *et al.*, 2002a), although it should be noted that the functional integrity of the paediatric microsomal preparations used to generate these data were not adequately demonstrated.

1.8.2. Gender effects on human glucuronidation

In adulthood, gender has some influence on glucuronidation, but the effect is limited to a subset of UGT forms. UGT1A6 is the only UGT form that has been directly shown to be differentially expressed due to gender, being found at a higher level in male liver (Court *et al.*, 2001). However, substrates known to be subject to sexrelated differences in glucuronidation include paracetamol, propranolol, oxazepam and mycophenolic acid; all having higher clearances in men (Greenblatt *et al.*, 1980; Mucklow *et al.*, 1980; Abernethy *et al.*, 1982; Miners *et al.*, 1983; Walle *et al.*, 1989; Bock *et al.*, 1994; Court *et al.*, 2001; Morissette *et al.*, 2001; Court *et al.*, 2004). The clinical relevance of these differences, at least in terms of immediate therapeutic outcome after drug administration, is currently thought to be minimal (Miners and Mackenzie, 1991).

1.8.3. Xenobiotic exposure and glucuronidation

A number of xenobiotics absorbed from the diet or the environment, or taken deliberately as pharmaceuticals, have been observed to affect UGT expression or activity in humans and other mammals. The UGT activity of freshly isolated or immortalised hepatocytes, liver microsomes, or microsomes from cells expressing heterologous UGT can also be altered by numerous compounds when treated *in vitro*. Severe acute adverse events due to altered glucuronidation in humans have been reported (Hirata-Koizumi *et al.*, 2007), and lesser alterations in glucuronidation

that cause chronic under- or over-dosing of therapeutic substances may also be clinically relevant (Kiang *et al.*, 2005; Weintraub *et al.*, 2005).

Well documented examples of therapeutic agents that increase glucuronidation in vivo include the antibiotic rifampicin, the anticonvulsants phenytoin, carbamazepine and phenobarbital, and oral contraceptives (Miners and Mackenzie, 1991). Rifampicin treatment has been observed to increase the glucuronidation of bilirubin (Ellis et al., 2006), the anti-epileptic lamotrigine (Ebert et al., 2000), lorazepam (Chung et al., 2005) and the anti-retroviral zidovudine (Burger et al., 1993; Gallicano et al., 1999). Likewise, phenytoin, phenobarbital and carbamazepine can increase the clearance of drugs such as lamotrigine (Weintraub et al., 2005), paracetamol (Miners et al., 1984a) and oxazepam (Seideman et al., 1981), while phenobarbital can be used therapeutically to increase bilirubin conjugation in some patients with genetic deficiencies in UGT1A1 (Jansen, 1999). Oral contraceptive steroids have been shown to affect the metabolism of numerous drugs with which they are often co-administered, including lamotrigine, paracetamol and clofibric acid. All three examples exhibit decreased plasma levels or increased clearance in oral contraceptive users of approximately 50% that can be attributed to changes in glucuronidation (Miners et al., 1983; Miners et al., 1984b; Sabers et al., 2003). Finally, it has recently been suggested that induction of UGTs by antiepileptic drugs may be the cause of altered thyroid hormone homoeostasis seen in patients treated with phenobarbital, phenytoin, and carbamazepine (Benedetti et al., 2005).

Conversely, glucuronidation in humans can also be inhibited by administration of therapeutics. A clinically relevant example is the previously mentioned administration of novobiocin to neonates, where novobiocin competes for the limited UGT1A1 expressed in newborns and causes hyperbilirubinemia (Robertson, 2003).

Probenecid has also been implicated in the inhibition of numerous glucuronidated drugs *in vivo* (Miners and Mackenzie, 1991). Further examples where glucuronidation of one substance may be inhibited by the pre- or co-administration of another include: decreased glucuronidation of morphine in the presence of nortriptyline, amitriptyline, clomipramine or diazepam (Yue *et al.*, 1990; Wahlstrom *et al.*, 1994); decreased glucuronidation of testosterone in the presence of amitriptyline, imipramine or chlorpromazine (Sharp *et al.*, 1992); decreased glucuronidation of paracetamol in the presence of propranolol (Baraka *et al.*, 1990); and decreased glucuronidation or clearance of zidovudine, lamotrigine and lorazepam in the presence of valproic acid (Yuen *et al.*, 1992; Lertora *et al.*, 1994; Chung *et al.*, 2005). It must be noted however, that the *in vivo* relevance of many interactions is still to be demonstrated, as much of the available data is from experiments utilising human liver microsomes *in vitro*.

Although in the majority, not all substances known to affect glucuronidation in humans are pharmaceuticals. Glucuronidation of paracetamol is increased in people on a diet of cruciferous vegetables (Pantuck *et al.*, 1984), and the relative ratio of carbohydrate to protein in the human diet can also affect glucuronidation; with high carbohydrate intake being associated with increased glucuronidation of paracetamol and oxazepam at the expense of other metabolic pathways (Pantuck *et al.*, 1991). A further dietary study reported a decrease in bilirubin levels in subjects who had a high intake of cruciferous vegetables, but that this effect was also restricted to individuals with a specific *UGT1A1* genotype (Peterson *et al.*, 2005). Acute ethanol consumption has been reported to slightly impair the clearance of lorazepam in humans (Hoyumpa *et al.*, 1981), while UGT1A6 mRNA and protein levels have been found to be 2-fold higher in liver samples from patients with a history of excessive

alcohol usage than those without (Krishnaswamy et al., 2005a). Smoking of tobacco induces the glucuronidation of mexiletine and propranolol (Grech-Belanger et al., 1985; Walle et al., 1987), and has also been associated with increased glucuronidation or clearance of paracetamol (Mucklow et al., 1980; Bock et al., 1987; Bock et al., 1994), although several other studies have not supported this latter observation (Miners et al., 1984a; Scavone et al., 1990; Krishnaswamy et al., 2005a). Whether smoking habits (frequency, cigarette strength or co-consumption of substances such as alcohol), can explain the discrepancies between studies remains unclear. Interestingly, co-abuse of ethanol and tobacco by pregnant women synergistically increases UGT expression in placenta compared to either substance alone (Collier *et al.*, 2002b), illustrating the complexities involved in delineating the relationships between UGT expression/activity and lifestyle. Other dietary components known to affect UGT expression or activity in human cell culture or rats include chrysin, guercetin, tumeric, curcumin, retinol, tannic acid, flavone and coumarin, although any relevance of these observations to human health is yet to be demonstrated (Galijatovic et al., 2000; Grancharov et al., 2001; Haberkorn et al., 2002; van der Logt et al., 2003; Naganuma et al., 2006).

There are a number of ways in which xenobiotics can alter glucuronidation in humans. The most extensively studied are increased glucuronidation through increased expression of one or more UGTs, and decreased glucuronidation via enzyme inhibition. Generally, where mechanistic studies have been performed to investigate the former, it has been found that increases in UGT expression caused by xenobiotic exposure are mediated at the transcriptional level by nuclear receptor transcription factors. These findings will be discussed at further length in section 1.9.3.2.

1.8.4. Pregnancy and UGT expression

During pregnancy there are many physiological changes that alter drug absorption, distribution and metabolism, including alterations in the expression of certain CYP and UGT enzymes (Anderson, 2005). Notably, a clinically relevant increase in clearance of the antiepileptic drug lamotrigine has been observed in pregnant women during all three trimesters (Pennell *et al.*, 2004). While this effect has historically been postulated to be the consequence of an increase in UGT1A4 expression, a recent study shows that UGT2B7 is also a significant contributor to lamotrigine glucuronidation (Rowland *et al.*, 2006). Additional evidence that expectant mothers may express more UGT2B7 than their non-pregnant counterparts comes from studies of zidovudine, morphine and oxazepam clearances. However, metabolism of these drugs by multiple UGT forms, and/or high basal interindividual variability in their observed clearances, have made the available data difficult to interpret (Anderson, 2005).

Paracetamol is another good example of a drug with increased clearance during human pregnancy, with both glucuronidation and oxidative pathways increased in the third trimester (Miners *et al.*, 1986). Paracetamol is metabolised by multiple UGT forms including UGT1A1 and UGT1A6 (see Table 1.1). Interestingly, transgenic mice bearing the human *UGT1A* locus have been shown to express higher levels of UGT1A1, UGT1A4 and UGT1A6 mRNA during pregnancy, and UGT1A4 and UGT1A6 during postpartum lactation (Chen *et al.*, 2005a). One suggested mechanism for the increase in lamotrigine and paracetamol clearances during human pregnancy is a transcriptional response of one or more *UGT* genes to increased hormone levels, as similar effects are seen in oral contraceptive users (Miners *et al.*, 1983; Sabers *et al.*, 2003; Anderson, 2005).

1.8.5. Health and glucuronidation

A number of common diseases and suboptimal body states are known to influence biotransformation pathways in humans. Whilst most research has focused on the effects of health on CYP-mediated drug metabolism, due to the magnitude of the changes observed and their immediate clinical significance, there is some evidence that UGT expression is also affected, albeit more subtly, by conditions such as inflammation, cancer and obesity.

Inflammation and sepsis have been shown to decrease hepatic glucuronidation in rodents (Strasser et al., 1998; Richardson et al., 2006), and may also have some effect in humans. The severity of inflammation during viral infection of the liver is associated with a reduction of UGT mRNA in human hepatocytes. This phenomenon differentially affects the UGT genes, with UGT1A4, UGT1A9 and UGT2B7 being particularly affected, and is specific to hepatitis, as expression of these UGTs in chronic liver fibrosis is maintained (Congiu et al., 2002). The one exception to the unaltered status of UGT expression in fibrotic liver may be UGT2B17 (Congiu et al., 2002). Proinflammatory cytokines such as interleukin (IL)-1B, IL-6 and tumour necrosis factor α , may be indirectly responsible for the loss of UGT expression in inflamed tissue, by altering the expression of key transcription factors (Assenat et al., 2004; Aitken et al., 2006; Richardson et al., 2006). Furthermore, while the basal expression levels of some UGTs do not appear to be greatly affected by inflammation, inducible expression may still be affected, as is the case for constitutive androstane receptor (CAR)-mediated induction of human UGT1A1 in the presence of IL-1 β (Assenat *et al.*, 2004). Furthermore, glucuronide hydrolysis may also be increased at inflamed sites, due to immune-mediated release of endogenous β -glucuronidase and decreased pH. This may further diminish the net effectiveness of local UGT activity in inflammation (Shimoi *et al.*, 2001).

Many malignant tissues have been reported to have decreased UGT expression, relative to healthy tissue procured from the same organ and donor. Examples include cancers of the urinary bladder, colon, liver, stomach and biliary tissue (Strassburg *et al.*, 1997a; Strassburg *et al.*, 1998b; Giuliani *et al.*, 2005). However, the loss of UGT expression in tumours is not universal, even within cancer subgroups. Whether decreases in UGT expression can contribute to the progression of a cell to malignancy, or is simply a result of the dysregulation evident in cancerous cells, is currently unknown. However, when it occurs, the loss of UGT expression appears to be an early event in neoplastic transformation (Giuliani *et al.*, 2005).

Another disease that may result in altered glucuronidation is hypothyroidism. The clearances of oxazepam and paracetamol are lower in patients with severe hypothyroidism than in those who have had their thyroid hormone levels corrected by pharmaceutical intervention (Sonne *et al.*, 1990). However, the exact relationship between UGTs and thyroid hormones in humans is still unclear, and regulation of human UGT expression by these substrates is yet to be demonstrated, although is has been shown in rats (Haberkorn *et al.*, 2002).

Obesity has been associated with an increased glucuronidation capacity in both males and females. Total metabolic clearances for lorazepam, oxazepam and paracetamol were all increased in obese subjects after adjustment for body weight (Abernethy *et al.*, 1983). Obese men also have altered steroid hormone profiles compared with their lean counterparts, with lower androgen and higher oestrogen levels (Tchernof *et al.*, 1999). These observations may be the consequence of altered

glucuronidation, as obese subjects have both an increased liver size (relative to body weight) and an expanded adipose mass, and both organs are sites of glucuronidation. In particular, UGT2B15, a steroid metabolising UGT form, is expressed in adipose tissue (Tchernof *et al.*, 1999). In obese rats, a similar increase in glucuronidation capacity has been observed, with no concomitant increase in rates of sulphation or glutathione conjugation (Chaudhary *et al.*, 1993).

1.8.6. Genetic variation and glucuronidation

Functional genetic variations have been found in many *UGT* coding regions and/or promoters, including those of *UGT1A1*, *UGT1A3*, *UGT1A4*, *UGT1A6*, *UGT1A7*, *UGT1A8*, *UGT1A9*, *UGT1A10*, *UGT2B7* and *UGT2B17* (Bosma *et al.*, 1995; Jinno *et al.*, 2003; Villeneuve *et al.*, 2003; Bernard and Guillemette, 2004; Duguay *et al.*, 2004a; Ehmer *et al.*, 2004; Iwai *et al.*, 2004; Wilson *et al.*, 2004; Krishnaswamy *et al.*, 2005b). As highlighted in sections 1.8.6.1 and 1.8.6.2, a number of polymorphisms in these genes have been associated with health outcomes in humans. Overall, it seems that UGTs represent good candidates for low-penetrance susceptibility genes that may contribute to disease risk by influencing homeostasis and altering the effects of carcinogen exposure.

So far, studies into the impact of genetic variation on glucuronidation in humans have largely been restricted to testing the effects of polymorphisms on UGT protein activity or mRNA levels. The latter is generally assumed to be a product of altered transcriptional rates, resulting from altered promoter function. However, it should be noted that genetic diversity can also potentially affect UGT expression through mechanisms not yet investigated for this gene set. Polymorphisms within genes may also cause differences in mRNA processing, pre-mRNA splicing, exon skipping, mRNA stability, mRNA trafficking and production of regulatory RNA transcripts (reviewed in Johnson *et al.* (2005)), and may be relevant to *UGT* and other biotransformation enzyme genes. Furthermore, variation in *trans*-acting regulators (*e.g.* transcription factors) appears to be a major contributor to interindividual differences in mRNA profiles (Morley *et al.*, 2004), and therefore present a potential source of variation for UGT expression. For example, the transcription factor hepatocyte nuclear factor (HNF)1 α is a positive regulator of a number of *UGT* genes; thus, polymorphisms that alter the expression or activity of HNF1 α may affect UGT expression (Ryffel, 2001; Toide *et al.*, 2002; Mackenzie *et al.*, 2005a). As regulatory factors for *UGT* genes are identified, the likely impact of their polymorphic activity or expression level on UGT expression can be assessed.

1.8.6.1. Genetic variability in UGTs: association with disease

The only UGT currently known to harbour genetic mutations that lead directly and unequivocally to disease in the absence of any xenobiotic exposure is UGT1A1. Severe UGT1A1 deficiency leads to the accumulation of the endobiotic bilirubin to toxic levels, as this is the only human UGT with any appreciable activity towards bilirubin (Bosma *et al.*, 1994). The clinical outcome of an UGT1A1 deficiency relies on the degree to which UGT1A1 activity is compromised. Unconjugated hyperbilirubinemias range in severity from no detectable symptoms (most Gilbert syndrome patients), through to severe toxicity in Crigler-Najjar syndrome type II patients and fatal accumulation of bilirubin in Crigler-Najjar syndrome type I patients. Over 110 mutations have been described in the UGT1A1 gene, the majority of which are associated with Crigler-Najjar syndrome type I or type II phenotypes (UGT Nomenclature Committee, 2005). Several UGT1A1 mutants (UGT1A1 F83L, P229Q and R367G), and two UGT1A1 promoter polymorphisms (-3279(T>G) and $A(TA)_6TAA$ to $A(TA)_7TAA$) have also been associated with Gilbert syndrome. The former promoter polymorphism is located in a nucleotide element important for the *UGT1A1* transcriptional response to xenobiotics (see section 1.9.3.2), while the latter is located in the TATA box of the *UGT1A1* promoter and results in lower expression of UGT1A1 (Aono *et al.*, 1995; Bosma *et al.*, 1995; Sugatani *et al.*, 2002; Sutomo *et al.*, 2002).

Another significant group of diseases with aetiologies that may be influenced by variability in glucuronidation is cancer. The ability of UGTs to inactivate known carcinogens, as well as substances that support cell growth and survival (such as steroid hormones) suggests that UGTs may be protective against chemically-induced mutagenesis. A study that thoroughly illustrates this point measured the cytotoxicity of benzo[*a*]pyrene metabolites on lymphocytes from normal subjects. A 200-fold variation in UGT activities against benzo[*a*]pyrene metabolites was found between samples, and decreased UGT activity correlated with decreased protection against covalent binding of benzo[*a*]pyrene to cellular proteins and increased cytotoxicity of several benzo[*a*]pyrene metabolites (Hu and Wells, 2004).

Specific associations reported between genetic variation in *UGTs* and cancer include: genetic variation in *UGT1A1* with risk of breast, endometrial and colorectal cancer; genetic variation in *UGT1A7* with risk of oral, gastrointestinal, colorectal, lung and liver cancer; genetic variation in *UGT1A10* with risk of orolaryngeal cancer; genetic variation in *UGT2B15* with risk of prostate cancer; and genetic variation in *UGT2B17* with risk of lung and prostate cancer. It should be noted however, that most of these associations are weak, with odds ratios of less than three, meaning that this information is likely to be of most use when it can be considered in conjunction with other risk factors. There are also significant discrepancies between some studies investigating the relationships of particular *UGT* polymorphisms with cancer risk, suggesting that a subset of the reported associations are likely to eventually be declared false positives. The following paragraphs detail the relationships between *UGT* genes and cancer reported so far.

For *UGT1A1*, the lower activity A(TA)₇TAA TATA box allele (*UGT1A1*28*) was found to be associated with development of breast cancer in premenopausal African-American and Chinese women under 40 years old (Guillemette *et al.*, 2000; Adegoke *et al.*, 2004), but not in three other studies involving Greek Caucasian, African and postmenopausal American Caucasian women (Guillemette *et al.*, 2001; Huo *et al.*, 2007; Tsezou *et al.*, 2007). In fact, the Nigerian study found that low activity *UGT1A1* promoter alleles were protective against breast cancer, but only in premenopausal women (Huo *et al.*, 2007). Yet, variation in the *UGT1A1* gene at the TATA box has also been associated with breast cancer characteristics such as age at diagnosis and tumour grade in Caucasian women (Shatalova *et al.*, 2005). Therefore, it remains unclear whether *UGT1A1* genotype is a relevant risk factor for breast cancer in certain populations, and if so, how this risk is modified by age, environment and/or ethnicity.

Other studies investigating the links between UGT1A1 genotype and cancer risk have reported that the UGT1A1 G71R amino acid change may predispose to development of colorectal cancer in Taiwanese men (Tang *et al.*, 2005), and conversely, that the A(TA)₇TAA *UGT1A1* allele may be protective against the development of endometrial cancer (Duguay *et al.*, 2004b).

For *UGT1A7*, several alleles, *UGT1A7*2*, *UGT1A7*3* and *UGT1A7*4*, have been shown to have low or very low activity towards PhIP and several benzo[*a*]pyrene

metabolites, all known carcinogens normally glucuronidated by UGT1A7 (Strassburg et al., 2002b). Allelic variants of UGT1A7 have been associated with higher risk of developing colorectal cancer in four studies (Strassburg *et al.*, 2002b; van der Logt et al., 2004; Tang et al., 2005; Chen et al., 2006a), with some evidence suggesting that cigarette smokers who carry UGT1A7 polymorphisms are at a greater risk for colorectal cancer than smokers with wild-type UGT1A7 (Chen et al., 2006a). However, a fifth study found no association between low activity UGT1A7 genotypes and risk of developing colon cancer, except in individuals with a high exposure to heterocyclic amines (Butler et al., 2005), again suggesting that any associations between UGTs and cancer susceptibility are likely to be affected, even confounded, by other genetic and lifestyle factors that vary between populations. UGT1A7*3 and other allelic variants have also been associated with higher risk of developing lung cancer (Araki et al., 2005), hepatocellular carcinoma (Vogel et al., 2001; Wang et al., 2004b; Tseng et al., 2005) and proximal digestive tract cancers (specifically orolaryngeal, but also possibly oesophageal) (Zheng et al., 2001; Vogel et al., 2002). Interestingly, the study by Zheng and co-workers also showed that UGT1A7 genotype was only predictive of cancer risk in patients who smoked (Zheng et al., 2001). Some questions remain over the likely mechanism of the association between UGT1A7 genotype and liver and lung cancers, as UGT1A7 is not expressed in these tissues, and the effect of UGT1A7 polymorphisms on circulating levels of relevant carcinogens is yet to be investigated. It may transpire that in such cases UGT1A7 is behaving as a biomarker rather than a causative risk factor. Finally, one study also reported an association between UGT1A7*3 and increased risk of chronic pancreatitis and pancreatic cancer (Ockenga et al., 2003), but two further studies have failed to replicate this result (Verlaan et al., 2005; Piepoli et al., 2006).

For UGT1A10, the only association between genotype and cancer risk to date is for orolaryngeal cancer and the UGT1A10 protein variant E139K (from UGT1A10*2). Individuals with one or more UGT1A10*2 alleles were found to have decreased risk of developing orolaryngeal cancer (Elahi *et al.*, 2003); however, this variant has subsequently been shown to have less activity towards benzo[a]pyrene metabolites than wild-type UGT1A10 (Dellinger *et al.*, 2006). Thus, the reason for this observation remains unknown.

For UGT2B15, two highly prevalent alleles (UGT2B15*1 and UGT2B15*2) exist, each occurring at frequencies of approximately 50% in Caucasians (Levesque et al., 1997; Gsur et al., 2002; Park et al., 2004c). The UGT2B15*2 allele encodes a protein variant (UGT2B15 D85Y) that has approximately 2-fold increased activity towards dihydrotestosterone than the variant encoded by UGT2B15*1 (Levesque et al., 1997). Since higher androgen exposure may predispose to prostate cancer, this UGT form has been extensively investigated for associations between genotype and risk of prostate cancer. Three studies have reported a positive association between the lower activity UGT2B15.1 variant and risk of developing prostate cancer in Asian and Caucasian ethnic groups (MacLeod et al., 2000; Park et al., 2004c; Okugi et al., 2006); however, three more studies disagree. Firstly, Gsur and colleagues found no association between UGT2B15 genotype and prostate cancer in Austrians, while Hajdinjak and co-workers found no association between prostate cancer incidence and UGT2B15 genotype in Slovenians, but that UGT2B15 genotype was associated with pathological grade (Gsur et al., 2002; Hajdinjak et al., 2004). This second study is completely at odds with the similar study in Japanese men that showed an association of UGT2B15 genotype with prostate cancer risk, but not pathological grade (Okugi et al., 2006). Thirdly, a study that examined 46 polymorphisms in the

androgen and oestrogen metabolic pathways found that there was no evidence of an association between *UGT2B15* genotype and prostate cancer risk in sporadic or familial prostate cancer patients (Cunningham *et al.*, 2007). Thus, the usefulness of this genotype as a risk marker for prostate cancer continues to be debated.

For *UGT2B17*, a major polymorphic variation exists in the form of a gene deletion, resulting in the complete absence of this UGT form in up to 85% of individuals, depending on ethnicity and study population (Wilson *et al.*, 2004; Terakura *et al.*, 2005; Jakobsson *et al.*, 2006; Park *et al.*, 2006). On the basis that similarly to UGT2B15, UGT2B17 also metabolises androgens (Jakobsson *et al.*, 2006), one research group has recently studied the association of this gene deletion with prostate cancer risk. A positive association was reported (Park *et al.*, 2006); however, since this deletion is known to be in high linkage disequilibrium with *UGT2B15* genotype (Wilson *et al.*, 2004), it is difficult to draw conclusions regarding the possible individual contribution of either gene to this association. A positive association between the complete absence of the *UGT2B17* gene, 4-(methylnitrosamino)-1-(3-pyridyl)-1-butanol (NNAL) glucuronidation and lung adenocarcinoma in Caucasian women, but not men, has also been recently reported (Gallagher *et al.*, 2007).

Finally, genetic variation in *UGT1A1* has also been associated with altered risk of cardiovascular and coronary heart disease. In a study of 1780 unrelated individuals, it was found that individuals homozygous for the *UGT1A1* A(TA)₇TAA promoter allele had approximately one third the risks for cardiovascular and coronary heart disease than carriers of the A(TA)₆TAA allele (Lin *et al.*, 2006). The protective effect may be due to higher circulating levels of bilirubin in homozygous carriers of the *UGT1A1* A(TA)₇TAA promoter allele, as bilirubin has antioxidant properties.

1.8.6.2. Genetic variability in UGTs: association with adverse drug reactions, altered drug efficacy and outcomes of organ transplantation

Several important relationships between the outcome of medical treatment and patient genotype have been reported for *UGTs*. The best known example is the association between *UGT1A1* and irinotecan disposition, which has now resulted in clinically relevant recommendations on irinotecan usage.

Irinotecan is an anticancer prodrug, primarily used to treat colorectal cancer. The active therapeutic, 7-ethyl-10-hydroxycamptothecin (SN-38), is generated by carboxylesterase metabolism and is eliminated mostly by glucuronidation (reviewed in Garcia-Carbonero et al. (2002)). SN-38 has a narrow therapeutic window, and over-dosing can cause life-threatening toxicities including diarrhoea and neutropenia. UGT1A1, UGT1A7, UGT1A9 and UGT1A10 have been proposed to be the major catalysts of SN-38 glucuronide formation (Gagne et al., 2002; Oguri et al., 2004), and accordingly, UGT1A1, UGT1A7 and UGT1A9 genetic variations that decrease SN-38 glucuronidation have been associated with altered treatment outcomes. In particular, the UGT1A1 A(TA)₇TAA allele (UGT1A1*28), -3156G>A promoter polymorphism (UGT1A1*93) and UGT1A1*6 (UGT1A1 G71R) have been associated with increased risk of irinotecan-induced toxicity, particularly neutropenia (Ando et al., 2000; Innocenti et al., 2004; Kitagawa et al., 2005; Minami et al., 2007), while UGT1A7*2 (UGT1A7 N129K/R131K), UGT1A7*3 (N129K/R131K/W208R) and UGT1A9*1b (UGT1A9 -118(dT)_{9>10} promoter SNP) have been associated with decreased diarrhoea and increased efficacy (Carlini et al., 2005). The association of UGT1A1 A(TA)7TAA with irinotecan-mediated toxicity is sufficiently well established that the Food and Drug Administration has recently added recommendations for the testing of patient UGTIA1 genotype prior to

irinotecan treatment to the drug label (Figure 1.6) (Maitland *et al.*, 2006). The relative importance of the other implicated genotypes in irinotecan-treated patients remains unresolved; however, there is some evidence that the *UGT1A1*28* allele is insufficient to predict severe toxicity in some populations, such as the Japanese and other Asian people groups (Minami *et al.*, 2007; Sandanaraj *et al.*, 2007).

physicians may wish to consider Corruse in individual patients experiencing significant neutropenia.

Patients with Reduced UGT1A1 Activity

Individuals who are homozygous for the UGT1A1*28 allele are at increased risk for neutropenia following initiation of CAMPTOSAR treatment. A reduced initial dose should be considered for patients known to be homozygous for the UGT1A1*28 allele (see DOSAGE AND ADMINISTRATION). Heterozygous patients (carriers of one variant allele and one wild-type allele which results in intermediate UGT1A1 activity) may be at increased risk for neutropenia; however, clinical results have been variable and such patients have been shown to tolerate normal starting doses. **Hypersensitivity**

Intersensitivity reactions including severe anaphylactic or anaphylactoid reaction

Figure 1.6: Excerpt from the current label for Camptosar (Irinotecan HCl), as obtained from the Food and Drug Administration website. (Food and Drug Administration, 2005).

Other examples where *UGT* genotype may affect drug disposition in humans exist for *UGT1A6*, *UGT1A9*, *UGT2B7* and *UGT2B15*. Two studies have shown that *UGT1A6* genotypes modulate the protective effect of aspirin on the risk of developing colorectal adenoma, suggesting that aspirin use by individuals with wildtype UGT1A6 does not confer any chemopreventative benefits, but individuals with low activity UGT1A6 variants can reduce their risk of developing colorectal adenoma by regular aspirin consumption (Bigler *et al.*, 2001; Chan *et al.*, 2005). Yet, the results of two further studies conflict with these and each other; one showing that low activity *UGT1A6* genotypes are protective against colorectal adenoma recurrence irrespective of aspirin intake (Hubner *et al.*, 2006); the other that NSAIDs (including or excluding aspirin) are protective against colorectal adenoma regardless of *UGT1A6* genotype. Furthermore, the importance of the UGT1A6 enzyme in aspirin metabolism it is still under considerable debate (Miners and Day, 2007; van Oijen *et al.*, 2007). Further research into the interaction between *UGT1A6* genotype, aspirin use and colorectal adenoma is still clearly needed.

On the other hand, UGT1A9 and UGT2B7 are two enzymes important in the glucuronidation of mycophenolic acid, an immunosuppressant with a low therapeutic index and considerable interindividual variation in pharmacokinetics (Picard *et al.*, 2005). In healthy volunteers, the *UGT1A9* promoter -275T>A/-2152C>T, *UGT1A9*3* and *UGT2B7*2* (UGT2B7 H268Y) alleles have been associated with alterations in mycophenolic acid exposure, enterohepatic recycling and production of the toxic acyl-glucuronide metabolite (Levesque *et al.*, 2007a). The clinical importance of these findings is still to be determined, but is certainly of interest. Similarly, UGT2B15 is an important enzyme for the metabolism of the anxiolytic drugs oxazepam and lorazepam, and the UGT2B15 D85Y (*UGT2B15*2*) variant has been associated with lower glucuronidation of oxazepam in human liver and lower clearance of lorazepam in healthy volunteers (Court *et al.*, 2004; Chung *et al.*, 2005). Whether this polymorphism has clinically relevant affects on lorazepam safety and/or efficacy in humans remains to be seen, but it seems probable, as this drug also has a relatively low therapeutic index (Chung *et al.*, 2005).

Finally, the *UGT2B17* gene deletion may be a risk factor for transplant-related mortality in recipients of haematopoietic stem cells. The UGT2B17 protein is immunogenic in individuals that are genetically devoid of the *UGT2B17* gene, and may be responsible for a heightened risk of complications in recipients given transplants from donors mismatched for *UGT2B17* (Terakura *et al.*, 2005).

1.9 Regulation of human UGT genes

Even though differential UGT expression and activity between human individuals may be clinically important in multiple medical disciplines, our knowledge of the mechanisms that control *UGT* expression and interindividual variation in UGT levels remains limited. Therefore, this thesis was designed to expand the understanding of *UGT* gene regulation, based on the conviction that, in the future, such knowledge will be useful for identifying and understanding key points at which interindividual variation can occur, and ultimately, for identifying pathways that can be taken into account or manipulated for therapeutic benefit. The following sections discuss the extent of the knowledge base regarding the regulation of human *UGT* genes at the time this thesis was commenced. Additional advancements made during the period of this candidature are discussed in the following chapters as relevant to the work presented therein, and/or summarised in Chapter 7.

1.9.1. Transcriptional regulation: transcription factors and co-regulators

Proteins that bind to specific DNA sequences to control gene expression are collectively known as transcription factors. DNA sequences targeted by such proteins are referred to as regulatory elements and are typically found upstream of a gene's coding region; however, they can also reside within coding regions, within introns, downstream of a gene's coding region, and even within distant or interchromosomal DNA sequences that are not obviously part of the target gene (Brooks *et al.*, 1994; Harrow *et al.*, 2004; Patrinos *et al.*, 2004; Kleinjan and van Heyningen, 2005). The general purpose of transcription factors is to facilitate appropriate interaction of the transcriptional machinery with each gene target, and as such, they may have a positive or negative role in this process. Furthermore, multiple

transcription factors bind and influence the expression of each gene, occurring in seemingly endless combinations that allow a high degree of control to be exerted over individual genes using a large but limited number of regulatory proteins. Differential expression of transcription factors (with some expressed ubiquitously, and others being tissue-restricted) allows further differentiation of gene expression profiles of different cell types. Additional complexity is provided by a second class of regulatory proteins known as co-regulators, which do not directly bind DNA, but are recruited by bound transcription factors and interact with the transcriptional machinery and/or the chromatin environment in which the target gene resides. Some proteins can also behave either as transcription factors or co-regulators, depending on the gene context. A comprehensive review of general gene regulation can be found in Schrem *et al.* (2002).

Gene expression requires a permissive DNA environment with a relatively open chromatin structure to proceed. Thus, the purpose of some transcription factors, coregulators and other chromatin remodelling complexes is to modify chromatin to allow other regulatory proteins and the transcriptional machinery access to appropriate genes. The histones, around which DNA is wrapped to form chromatin, can be modified by processes such as acetylation, phosphorylation and ubiquitination, while the DNA itself can be methylated. Methylation and deacetylation of chromatin are two linked processes that contribute to the compaction of chromatin and the silencing of genes *in vivo* (Schrem *et al.*, 2002). Of particular pertinence to work in this thesis, actively expressed genes are found in highly acetylated chromatin, and the acetylation status of chromatin is regulated by two groups of enzymes with opposing activities, the histone acetyltransferases (HATs) and histone deacetylases (HDACs). Many co-activators, including p300/cAMP-response-element-binding protein (CREB)-binding protein (p300/CBP), p300/CBP-associated factor (P/CAF), steroid receptor co-activator (SRC)-1 and peroxisome-proliferator-activated receptor (PPAR)-gamma co-activator (PGC)-1, are known to either possess or recruit HAT activity to target genes (Ogryzko *et al.*, 1996; Yang *et al.*, 1996; Spencer *et al.*, 1997; Puigserver and Spiegelman, 2003). Some transcription factors also have intrinsic chromatin-opening activity such as the forkhead box A (FoxA) proteins (Cirillo *et al.*, 2002). Conversely, many corepressors, such as silencing mediator of retinoid and thyroid hormone receptor (SMRT) and nuclear receptor co-repressor (NCoR), recruit HDAC activity (Ng and Bird, 2000). In turn, many of these co-activators and co-repressors are known to physically interact with transcription factors investigated in this thesis. The properties of relevant transcription factors and their co-regulators are discussed in detail in the chapters to which they are pertinent.

1.9.2. Gene regulatory elements: the core promoter, the proximal promoter and enhancer elements

Immediately upstream of the initiation codon of a gene, is a region known as the core promoter. This region typically contains the minimal DNA elements required for RNA polymerase II and the other components of the transcriptional machinery to bind a gene and initiate transcription. Common elements in the core promoter include the TATA box, initiator (Inr) region, transcription factor II (TFII)B recognition region and downstream core promoter element, located within a DNA stretch of approximately 70 nucleotides and centred roughly around the TSS (see Figure 1.7). Any particular core promoter may contain some, all or none of these elements (Smale and Kadonaga, 2003; Buckland, 2006), and the human *UGT* gene

family contains examples of both TATA box-reliant and TATA-less promoters (see section 1.9.3.1).



Figure 1.7: Idealised structure of an eukaryotic gene. Common regulatory elements and regions found in eukaryotic genes (figure adapted from Smale and Kadonaga (2003)). TRE: TFIIB recognition element; Inr: Initiator region; DPE: Downstream core promoter element. Numbering is relative to the transcriptional start site. Consensus sequences for the four core promoter elements depicted are given in parentheses (S = G + C, R = A + G, W = A + T, Y = C + T, V = G + C + A).

The nucleotides immediately upstream of a gene's initiation codon, encompassing and extending several hundred base pairs beyond the core promoter, are generally referred to as the proximal, or regulatory, promoter. Often this region will be sufficient to drive transcription in *in vitro* assays, and typically contains multiple transcription factor binding sites that are involved in recruiting and positioning the transcription machinery (Cooper *et al.*, 2006). Mutation of these sites *in vitro* can often demonstrate their importance for transcription, and even highly related sequences can vary significantly in function through relatively few nucleotide substitutions (Buckland, 2006). However, when these sequences are integrated into the genome; *i.e.* in stable transfections or knock-in animal models, they are often insufficient to drive transcription, or cannot confer appropriate tissue-type specific and insertion-point independent expression patterns. Such behaviour typically requires longer sequences that contain additional transcription factor binding sites known as enhancers and locus control regions (Brooks et al., 1994; Harrow et al., 2004; Kleinjan and van Heyningen, 2005). There is no real distinction between proximal promoter and enhancer elements, except that the latter tends to be used to refer to elements that reside long distances from the TSS, up to many kilobases upor downstream, and includes sites necessary for regulating the structure and nuclear localisation of chromatin (Blackwood and Kadonaga, 1998). Like the proximal promoter, enhancer regions often possess several transcription factor binding sites within close proximity of each other, facilitating the formation of protein complexes with specific functions. Combinatorial binding of transcription factors (in both enhancers and proximal promoters) is of sufficient importance that regulatory regions containing transcription factor site combinations conserved between orthologues, or genes of similar expression pattern within an organism, are considered highly likely to be functionally significant (Liu et al., 2003; Johnson et al., 2005).

1.9.3. Regulation of human UGT genes

At the commencement of this thesis, human *UGT* proximal promoters that had been cloned and subjected to some degree of functional analysis included *UGT1A1*, *UGT1A6*, *UGT1A8*, *UGT1A9*, *UGT1A10*, *UGT2B4*, *UGT2B7*, *UGT2B15* and *UGT2B17*.

1.9.3.1. Regulation of human UGT genes by transcription factors other than nuclear receptors

The first human *UGT* promoter clone to be published was *UGT1A1*, in work by Bosma and colleagues that showed that the length of the *UGT1A1* TATA box affected promoter activity in transient transfections of HuH7 cells (Bosma *et al.*, 1995). The *UGT1A1* promoter containing the $A(TA)_7TAA$ sequence associated with Gilbert syndrome was found to have only 18-33% of the activity achieved by the same promoter with the $A(TA)_6TAA$ sequence (Bosma *et al.*, 1995). Recently, it has been shown that increasing the number of thymine-adenine repeats in the *UGT1A1* promoter causes a decrease in TATA-binding protein *in vitro*, providing a plausible mechanism for the observed concomitant decrease in promoter activity (Hsieh *et al.*, 2007).

The only other *UGT* gene subsequently found to possess functional variants of the TATA box to date is *UGT1A7*, where a T to G transversion at nucleotide position -57 results in a 70% reduction in promoter activity *in vitro* (Lankisch *et al.*, 2005). Indeed, some human *UGT* genes actually appear to be TATA-less (Figure 1.8). While canonical TATA boxes have been predicted for *UGT1A1*, *UGT1A3*, *UGT1A4*, *UGT1A5*, *UGT1A6* and *UGT1A7* (Ritter *et al.*, 1992b; Bosma *et al.*, 1995; Lankisch *et al.*, 2005), *UGT1A8*, *UGT1A9* and *UGT1A10* have been shown to rely on an Inr-like region for initiation of transcription (Gregory *et al.*, 2003) and are seemingly TATA-less. Likewise, the *UGT2B7* gene does not have a canonical TATA box (Ishii *et al.*, 2000), a feature shared by *UGT2B4*. In contrast, the remaining known human *UGT2B* genes contain the nucleotide sequence "TATAA", predicted to be a TATA box, positioned at nucleotides -63 to -59 relative to the *UGT2B17* initiation codon.

Accordingly, the TSS locations mapped for *UGT2B7* and *UGT2B17* are completely different (Gregory *et al.*, 2000; Ishii *et al.*, 2000) (see Figure 1.8).

UGT1A1	5' TATA box 3' GGTGTATCGATTGGTTTTTGCCATATATATATATATATAT
UGT1A3 UGT1A4 UGT1A5	5' * 3' GTAATTAAGATGAAGAAAGCAAATGTAGCAGGGCACAGCGTGGGGTGGACAGTCAGCTGTCCGTGTCTTCTGCTGAGATG GTAATTAAGGCGAAGGAAACAAATGTAGCAGGCACAGCGTGGGGTGGACAGTCAGCTGTCGGTGGCTTCTGCTGAGATG GTAATTAAGACGAAGGAAACAATTCTAGGAGGCACAACGTGGGGTGGACAGTCAGCTGTCGGTGGCTTCTGCTGAGATG TATA box
UGT1A6 UGT1A6	5' TATA box * * TGGGTTCTTACA <mark>TATCAAA</mark> BGGTAAAATTCAGAGCAAGGGAGAGGGAGAGGACAGGACCTGTGAA A AGCAGTGGTTAGTTTA 3' GGGAAAATACCTAGGAGCCCTGTGATTTGGAGAGTGAAAACTCTTTATTACCGTTGTTACTTTTAACTCTTTCCAGG ATG
UGT1A7 UGT1A8 UGT1A9 UGT1A10 UGT1A7 UGT1A8 UGT1A9 UGT1A10	5' T-region GTAAATCATTGGCAGTGAATGTGAATTTTTTTTTATAATGAATGAATAAGTACACGCCTTCTTTTGAGGGCAGGGTTT GTAAATCATTGGCAGTGAGTGTGATTTTTTTTTTTTTTT
UGT2B4 UGT2B7 UGT2B10 UGT2B11 UGT2B15 UGT2B17 UGT2B28	5' * * 3' ATTTATA-CTGGATGTCACCATGAGAAATGACAGAAAGGAAGGAGGAGCAGCAACTGGAAAACAAGCATTGCATTGCACCAGG ATG ATTTATCTTTGGACATAACCATGAGAAATGACAGAAAGGAAGG

Figure 1.8: The UGT1A and UGT2B core promoters showing putative TATA box and initiator-like elements. The putative elements are boxed, and experimentally demonstrated transcription start sites (TSSs) are indicated in bold and are marked by asterisks or dots. For the UGT1A7-1A10 cluster, conflicting explanations of transcriptional initiation have been published. Gong and colleagues postulated that the T-region shown behaves as the anchoring site for the transcriptional machinery and reported the TSSs marked by dots (Gong et al., 2001). In contrast, Gregory and co-workers reported an initiator (Inr)-like region further downstream and the TSSs marked by asterisks (Gregory et al., 2003). Two different cap sites have also been found for UGT1A6 by different research groups (Ritter et al., 1992b; Munzel et al., 1998). Such differences may indicate that UGT TSSs are at least partly determined by cell type (Munzel et al., 1998). The remaining data were obtained from Ritter et al. (1992b), Ishii et al. (2000), Lankisch et al. (2005) and Gregory et al. (2000).

A large proportion of the initial studies addressing basic human UGT promoter

function concentrated on the role of HNF1 transcription factors. After a report that

HNF1α bound and activated the rat UGT2B1 promoter was published (Hansen et al., 1997), work by Bernard et al. (1999) showed that the UGT1A1 promoter could also be regulated by HNF1 α and HNF1 β in HEK293 cells. This report was closely followed by three others that also implicated HNF1 α in the regulation of two additional human UGT genes. Studies of the isolated UGT2B7 promoter in HepG2 cells showed that HNF1 α , but not HNF1 β , could drive transcription from a proximal HNF1-binding site, and that octamer transcription factor-1 (Oct-1) could interact with HNF1α as a co-activator on this promoter to further increase transcription (Ishii et al., 2000). Likewise, a proximal HNF1-binding element was found to confer in *vitro* responsiveness to HNF1 α , but not HNF1 β , to the UGT2B17 gene promoter in HepG2 cells (Gregory et al., 2000). Although the HNF1-binding element of the UGT2B17 promoter is at the same position relative to the initiation codon as in UGT2B7, Oct-1 was found to have a negative effect on transcription from the UGT2B17 promoter in HepG2 cells (Gregory et al., 2000). Furthermore, it was found that binding of pre-B cell homeobox and related factors to a site immediately adjacent to the HNF1-binding site of UGT2B17 modulated transcription in vitro by restricting access of HNF1a to the HNF1-binding site (Gregory and Mackenzie, 2002).

One further study published prior to 2003 regarding *UGT* regulation in humans identified two promoter elements in the *UGT2B15* promoter that were important for *in vitro* function in prostate-derived LNCaP cells, and that were not shared by the closely related *UGT2B17* promoter. The most proximal of these two elements was suggested to be a FoxA2 (HNF3 β) binding site (Turgeon *et al.*, 2000); however, no experimental data was provided (then nor since) to confirm the identity of the binding sites or proteins important for these observations.

1.9.3.2. Regulation of human UGT genes by ligand-dependent nuclear receptors

Changes in the environment or health state of humans can alter the activity or level of many transcription factors, thus causing indirect alterations in the expression of their target genes. However, there is also a specific subset of transcription factors that overtly rely on xenobiotic and endogenous compounds as ligands to regulate their activity and consequent expression of target genes. These transcription factors are known as ligand-dependent nuclear receptors, and regulation of genes by these factors is generally considered "inducible" because the regulatory pathways that rely on these transcription factors are typically inactive or repressed in the absence of ligand. Such pathways are thought to be important for the appropriate expression of UGTs in response to chemical exposure, and in allowing co-ordinate expression of biotransformation enzymes from different stages of chemical metabolism and elimination (Xu *et al.*, 2005; Trottier *et al.*, 2006a).

The earliest evidence that ligand-dependent nuclear receptors were involved in the regulation of UGTs in humans was the discovery that the aryl hydrocarbon receptor (AhR) binds to a xenobiotic response element (XRE) in the human *UGT1A6* promoter, and that the presence of an AhR agonist, 2,3,7,8-tetrachlorodibenzo-*p*-dioxin, increases transcriptional activity from this gene *in vitro* (Munzel *et al.*, 1998). Further work demonstrated that UGT1A1, UGT1A9 and UGT2B7 expression could also be increased by exposure to 2,3,7,8-tetrachlorodibenzo-*p*-dioxin and polyaromatic hydrocarbons (AhR agonists) and/or antioxidant-type inducers that normally work through antioxidant response elements (Bock *et al.*, 1998; Munzel *et al.*, 1999; Ritter *et al.*, 1999; Walle and Walle, 2002). The presence of a XRE in the *UGT1A1* gene promoter was subsequently demonstrated by Yueh *et al.* (2003) and further studies of the *UGT1A6* promoter indicated that the response of this particular
gene to antioxidants is through an atypical mechanism that probably involves AhR (Munzel *et al.*, 2003).

Other nuclear receptors that were recognised as UGT regulators by 2003 include CAR, pregnane X receptor (PXR), PPARa, PPARy and farnesoid X receptor (FXR). CAR is a constitutively active nuclear receptor that regulates UGT1A1 gene expression in response to agonists such as phenobarbital and antagonists such as androstenol, through a nuclear receptor response element (NRRE) that resides within a short DNA sequence known as the UGT1A1 phenobarbital response enhancer module (PBREM: UGT1A1 -3499 to -3210). The PBREM also consists of at least six other NRRE motifs that are functionally active (Sugatani et al., 2001; Sugatani et al., 2005b), and a polymorphism in one NRRE of the UGT1A1 PBREM (UGT1A1 -3279T>G) has been associated with Gilbert syndrome in Japanese patients (Sugatani et al., 2002). Other NRREs in the UGTIAI PBREM include the XRE mentioned earlier (Yueh et al., 2003) and a PXR binding site (Xie et al., 2003). PXR activates UGT1A1 gene expression when liganded with xenobiotics such as rifampicin and can bind to three of the UGT1A1 PBREM NRREs. However, the majority of UGT1A1 gene activation appears to rely on one particular PBREM NRRE at nucleotide position -3430 to -3386, which can also be bound by CAR (Xie et al., 2003). PXR has also been shown to regulate the human UGT1A3, UGT1A4 and UGT1A6 genes, although the DNA sequences required for these effects have not been identified (Rae et al., 2001; Gardner-Stephen et al., 2004).

The closely related nuclear receptors PPAR α and PPAR γ were first recognised as human *UGT* regulators when Barbier and colleagues demonstrated that PPAR ligands could increase UGT1A9 expression in human hepatocytes, and that the *UGT1A9* promoter contains a functional PPAR response element (Barbier *et al.*, 2003c). Similarly, the same research group demonstrated a role for PPAR α in the regulation of *UGT2B4* (Barbier *et al.*, 2003a), and concurrently identified *UGT2B4* as a gene target of FXR (Barbier *et al.*, 2003b).

1.10 Experimental aims

Despite much evidence that UGTs are important in at least three major aspects of human health (homeostasis, xenobiotic defence and drug efficacy), the understanding of these enzymes and the factors that determine inter- and intra-individual variation in their expression remains limited. If the goal of personalised medicine is to become realised with widespread impact, it will be necessary to understand the biology of human UGTs (and other biotransformation enzymes such as the CYPs and sulphotransferases, N-acetyltransferases and glutathione-S-transferases) to a much greater extent than the current knowledge allows. Of particular concern, research into the basal regulation of human UGTs, and the interplay of this with genetic variation was almost nonexistent at the commencement of this PhD candidature, with the exception of limited studies into the role of HNF1 factors. Moreover, while a considerable amount of progress had been made on inducible expression of UGTs by nuclear receptors, there clearly remained much to be learnt. Therefore, the overall aim of this thesis was to substantially expand our knowledge of UGT regulation in humans, with an emphasis on how this may relate to the variability observed in UGT expression. To achieve this end, the specific aims of this thesis were to:

- 1. Identify DNA elements important for the promoter function of several lesser studied *UGT1A* genes; in particular, *UGT1A3*, *UGT1A4* and *UGT1A9*;
- 2. Identify transcription factors involved in the regulation of human *UGTs*, with a specific focus on those that may be important for hepatic UGT expression;

- 3. Provide further information on the mechanisms by which selected transcription factors control *UGT* promoter activity and how this may differ between *UGTs* that share similar transcription factor sets;
- 4. Test the UGT1A3 promoter and HNF1 α protein variants for effects that may be responsible for the observed variation in UGT1A3 expression levels between individuals.

CHAPTER TWO MATERIALS AND METHODS

2.1 Materials

2.1.1. Chemicals and molecular biology reagents

The suppliers of all reagents and kits used throughout this thesis are listed in Table

2.1. All chemicals were of analytical reagent grade.

Reagent	Supplier
Buffer Chemicals	
Acetic acid	BDH AnalaR (Merck), Kilsyth, VIC, Australia
Boric acid	BDH AnalaR (Merck)
Bovine serum albumin (BSA) solution (100 mg/ml)	New England Biolabs, Beverly, MA
Bromophenol blue	Sigma Chemical Co, St Louis, MO
CaCl ₂ .2H ₂ O	Ajax Finechem, Seven Hills, NSW, Australia
Dimethyl sulfoxide (DMSO)	Ajax Finechem
Dithiothreitol (DTT)	Fischer Biotech, West Perth, WA, Australia
Ethylenediaminetetra-acetic acid, di-sodium salt (EDTA)	BDH AnalaR (Merck)
Glycerol	BDH AnalaR (Merck)
Glycine	Amresco, Solon, OH
HCl	BDH AnalaR (Merck)
Isopropanol	Mallinckrodt Australia, South Oakleigh, VIC, Australia
KCl	BDH AnalaR (Merck)
KH ₂ PO ₄	BDH AnalaR (Merck)

Table 2.1: Reagents used in experimental procedures.

Reagent	Supplier	
Methanol	BDH AnalaR (Merck)	
MgCl ₂ .6H ₂ O	Amresco	
Na ₂ HPO ₄	Ajax Finechem	
NaCl	Ajax Finechem	
Nonidet P-40	Sigma Chemical Co	
Proteinase K	Merck, Darmstadt, Germany	
Sucrose	BDH AnalaR (Merck)	
Sodium dodecyl sulphate (SDS)	BDH AnalaR (Merck)	
Tris[hydroxymethyl]aminomethane (Tris)	Amresco	
Xylene cyanol FF	Pharmacia LKB Biotechnology, Bromma, Sweden	
<u>Mammalian Tissue Culture</u>		
Dulbecco's modified Eagle's medium (DMEM)	Invitrogen, Carlsbad, CA	
Foetal calf serum	Trace Scientific, Melbourne, VIC, Australia	
MEM non-essential amino acids	Invitrogen	
MEM sodium pyruvate	Invitrogen	
Rifampicin	Sigma Chemical Co	
Tissue culture flasks and plates	Nunc, Roskilde, Denmark	
Trichostatin A (TSA)	Sigma Chemical Co	
Trypsin-EDTA	Invitrogen	
Trypan blue	Sigma Chemical Co	
Transfection and Reporter Gene Assays		
Dual-luciferase TM Reporter Assay System	Promega, Madison, WI	
Lipofectamine [™] 2000	Invitrogen	
Passive lysis buffer	Promega	

Reagent	Supplier
Bacterial Culture	
Agar	Difco Laboratories, Livonia, MI
Ampicillin	CSL, Parkville, VIC, Australia
5-Bromo-4-chloro-3-indolyl-α-D-galactopyranoside	Astral Scientific, Gymea, NSW, Australia
Chloramphenicol	Sigma Chemical Co
Isopropyl-β -D-thiogalactopyranoside	Sigma Chemical Co
Kanamycin	Sigma Chemical Co
Luria Broth (LB) EZMix TM	Sigma Chemical Co
DNA Detection, Purification and Modification	
Agarose	Amresco
Buffer QG	Qiagen, Clifton Hill, VIC, Australia
Calf intestinal phosphatase (CIP)	New England Biolabs
Ethidium bromide	Amresco
Restriction enzymes	New England Biolabs
QIAEXII Gel Extraction kit	Qiagen
QIAGEN Large-Construct kit	Qiagen
QIAGEN Plasmid Midiprep kit	Qiagen
QIAprep Spin Miniprep kit	Qiagen
QIAquick Gel Extraction kit	Qiagen
QIAquick PCR Purification kit	Qiagen
Quick Ligation kit	New England Biolabs
RNA Purification and cDNA Synthesis	
Amplification grade DNase I	Invitrogen
β-Mercaptoethanol (β-ME)	BDH AnalaR (Merck)
RNeasy Midi kit	Qiagen

Reagent	Supplier	
SuperScript [™] FirstStrand Synthesis System for RT-PCR	Invitrogen	
Polymerase Chain Reaction (PCR)		
Deoxynucleotide-triphosphate mix (dNTP)	Astral Scientific	
Expand [™] Long Template PCR system	Roche Diagnostics, Mannheim, Germany	
Oligonucleotides	Sigma-Genosys, Castle Hill, NSW, Australia	
<i>PfuTurbo</i> [®] polymerase	Stratagene, La Jolla, CA	
QuantiTect SYBR Green PCR Kit	Qiagen	
Taq Extender™ PCR additive	Stratagene	
Taq polymerase	New England Biolabs	
Electrophoretic Mobility-Shift Assays and Wes	tern Blot	
40% (w/v) Acrylamide/Bis solution (19:1)	Bio-Rad, Hercules, CA	
40% (w/v) Acrylamide/Bis solution (29:1)	Bio-Rad	
Ammonium persulphate	Amresco	
Bio-Rad Protein Assay Reagent	Bio-Rad	
Complete [™] protease inhibitor tablets	Roche Diagnostics	
Deoxyribonucleic acid, sodium salt, type III: from salmon testes	Sigma Chemical Co	
ECL TM Western blotting detection reagent	Amersham Biosciences, Piscataway, NJ	
γ^{32} P-ATP	PerkinElmer, Wellesley, MA	
G25 columns	Amersham Biosciences	
Goat anti-HNF1α antibody (sc-6547X)	Santa Cruz Biotechnology, Santa Cruz, CA	
Goat anti-HNF1β antibody (sc-7411X)	Santa Cruz Biotechnology	
Goat anti-HNF4α antibody (sc-6556X)	Santa Cruz Biotechnology	
Photographic reagents: Developer G153 and Fixer G354	AGFA, Mortsel, Belgium	

Reagent	Supplier	
Poly(dI-dC)	Sigma-Aldrich	
N,N,N',N'-Tetramethyl-1-,2-diaminomethane	Amresco	
Rabbit anti-Goat IgG-Horseradish Peroxidase conjugate	Pierce Biotechnology, Rockford, IL	
Skim milk powder (SMP)	Bonland Dairies, Rowville, Vic, Australia	
Slide-A-Lyzer® dialysis cassettes	Pierce Biotechnology	
T4 polynucleotide kinase New England Biolabs		
TNT® Quick Coupled Transcription/Translation kit Promega		
rans-Blot Transfer Medium (nitrocellulose, 0.45 μm) Bio-Rad		
X-Omat Blue XB-1 autoradiographic film	Eastman-Kodak, Rochester, NY	

2.1.2. General buffers

The following buffer formulae were obtained from Sambrook and Russell (2001).

<u>1 × DNA gel loading buffer:</u> 0.04% (w/v) bromophenol blue, 0.04% (w/v) xylene cyanol FF, 5% (v/v) glycerol.

<u>1 × Phosphate buffered saline (PBS)</u>: 137 mM NaCl, 2.7 mM KCl, 10 mM Na₂HPO₄ and 2 mM KH₂PO₄, pH 7.4

<u> $1 \times$ Tris-acetate EDTA electrophoresis buffer:</u> 40 mM Tris-acetate, 1 mM EDTA

1 × Tris-EDTA buffer: 10 mM Tris-HCL, pH 7.5, 1 mM EDTA

 $0.5 \times \text{Tris-borate EDTA (TBE) electrophoresis buffer:} 45\text{mM Tris-borate, 1mM}$ EDTA

2.1.3. Eukaryotic and prokaryotic cell lines

Human embryonic kidney (HEK293T), human hepatocarcinoma (HepG2) and human colorectal adenocarcinoma (Caco-2) cells were all obtained from the American Type Culture Collection (ATCC, Manassas, VA). The DH5α *Escherichia coli* (*E. coli*) strain was also bought from the ATCC. DH10B *E. coli* stocks containing clones from the RPCI-11 bacterial artificial chromosome (BAC) library (Osoegawa *et al.*, 2001) were obtained from BACPAC Resources (Children's Hospital Oakland Research Institute, Oakland, CA). Top10 One-Shot chemically super-competent *E. coli* were purchased from Invitrogen.

2.1.4. Mammalian reporter and expression vectors

The reporter vectors pGL3-basic and pRL-Null were purchased from Promega. A schematic map of pGL3-basic showing the relevant restriction sites can be found in Appendix 1. The construction of pGL3 daughter plasmids carrying promoter inserts is described for each vector in the "Methods" sections of the appropriate chapters. The empty mammalian expression vector pCMX-PL2 was the kind gift of Dr. Ronald Evans (The Salk Institute for Biological Sciences, San Diego, CA) and is described in Umesono *et al.* (1991). pCMX vectors express high levels of recombinant protein through a cytomegalovirus (CMV) promoter and contain a T7 promoter sequence to facilitate *in vitro* transcription and translation.

pCMX-HNF1 α (containing the human HNF1 α -A variant cDNA) and subsequently derived pCMX-HNF1 α WT+21, I27L, A98V, S487N and P291finsC mutant plasmids were constructed by Tamara Height (University of South Australia, Australia) (Mackenzie *et al.*, 2005a). The pBJ5-HNF1 α and pBJ5-HNF1 β expression vectors were kindly provided by Dr. Gerald Crabtree (Stanford University, Stanford, CA) and express the murine orthologues of HNF1 α and HNF1 β . The pRB-HNF3 α and pGEM-HNF3 β plasmids, containing rat FoxA1 and FoxA2 cDNAs respectively, were the kind gift of Dr. Guntram Suske (Klinikum Der Philipps-Universitat Marburg, Germany). Expression vectors for the rat CCAAT/enhancer binding protein (C/EBP) α and C/EBP β transcription factors were generously provided by Dr. Peter Johnson (National Cancer Institute, Frederick, MD). All vectors listed in this section confer ampicillin resistance.

2.1.5. Cloning vectors

The shuttle vectors pCR-blunt and pCR-2.1, used for cloning PCR products without prior restriction, were purchased from Invitrogen. pCR-blunt requires 50 μ g/ml kanamycin for selection, whilst pCR-2.1-derived plasmids can be selected with 50 μ g/ml kanamycin or 100 μ g/ml ampicillin.

2.2 General methods

2.2.1. Maintenance of mammalian cell lines

All mammalian cell lines were cultured under sterile conditions in DMEM supplemented with 10% (v/v) foetal calf serum, 0.1 mM non-essential amino acids and 1 mM sodium pyruvate. Cells were grown at 37°C in a humidified atmosphere of 5% CO₂. All cells were PCR screened for mycoplasma infection by Anne Rogers before use and periodically thereafter. Cell stocks were preserved in foetal calf serum containing 10% (v/v) DMSO and stored in liquid nitrogen.

All cells were routinely passaged at approximately 80% confluence. HepG2 and Caco-2 cells were released from the surfaces of tissue culture flasks by incubation in 0.05% (w/v) trypsin at 37°C after washing in PBS. It was necessary to pass resuspended HepG2 cells through a sterile stepper syringe (Nichiryo, Tokyo, Japan)

before re-plating, to disperse clumps and maintain correct morphology. Moderately vigorous pipetting without prior trypsin digestion was adequate to release and disperse HEK293T cells for re-plating. Cell cultures were replaced from frozen stocks after 15-25 passages.

2.2.2. Bacterial culture and preparation of competent cells

All bacterial strains were grown at 37°C in LB liquid culture medium with vigorous shaking or on LB plates solidified with 15 g/L agar. Each culture was maintained under antibiotic selection if and as appropriate: bacteria carrying reporter or expression vectors were grown in 100 μ g/ml ampicillin, with the exceptions of those harbouring BAC constructs (20 μ g/ml chloramphenicol) and cells transformed with pCR-Blunt-derived plasmids (50 μ g/ml kanamycin). Long term bacterial stocks were stored at -80°C in LB containing 17.5% (v/v) glycerol.

To generate competent cells, DH5 α *E. coli* were grown overnight to stationary phase in antibiotic-free LB and subsequently diluted 1:60 in fresh medium. The diluted cells were then grown to log-phase (an optical density of approximately 0.4 at 600 nm) and harvested by centrifugation at 2,000 × g, 4°C for 15 minutes in a Sigma 4K15 centrifuge. The cell pellet was then re-suspended in ice-cold 50 mM CaCl₂ to 50% of their original volume and incubated on ice for 30 minutes. After a second centrifugation, the bacteria were re-suspended in one tenth of their original volume of ice-cold 50 mM CaCl₂ and glycerol was added to a final concentration of 15% (v/v). Competent DH5 α were used immediately, or stored at -80°C and thawed for 10 minutes on ice before use.

2.2.3. Extraction of total RNA and generation of cDNA

2.2.3.1. Extraction of RNA from HepG2 cells

HepG2 cells (untreated or transiently transfected) were harvested from single wells of 6-well tissue culture plates before they reached 80% confluence. To extract total RNA, the supernatant was aspirated, and the attached cells washed once with PBS and lysed by the addition of 350 µl Buffer RLT (Qiagen) containing 0.01% (v/v) β -ME. The lysate was then passed five times through a 21-gauge needle, and total RNA was isolated from each sample using the RNeasy Mini kit. To do this, one volume of 70% (v/v) ethanol was mixed with the homogenate and the mixture placed on an RNeasy spin column. The column was then centrifuged in a Beckman Microfuge 18 centrifuge at 8,000 × g for 15 seconds, and then washed by addition of 500 µl Buffer RW1 and centrifugation at 8,000 × g for 15 seconds, followed by two washes with 750 µl Buffer RPE under the same conditions. The column was dried by an additional centrifugation at 18,000 × g for 1 minute. Purified RNA was eluted from each column with 30 µl RNAse-free water and centrifugation at 8,000 × g for 1 minute. Finally, the eluant was passed through the column again to elute any RNA remaining on the column. All RNA samples were stored at -20°C.

2.2.3.2. Extraction of RNA from murine liver

Two hundred and fifty milligrams of murine liver (originating from a (CBAxC57BI/6J)xC57BI/6J h*UGT1A8*-7k transgenic mouse sacrificed under Flinders Medical Centre Animal Welfare Committee approval #574/04) was disrupted and homogenised immediately after collection in 16 µl/mg Buffer RLT (Qiagen) and 10 µl/ml β-ME using a hand micropestle followed by 10 passages through a 21-gauge needle. The homogenate was then frozen on dry ice and stored at -80°C until required for RNA extraction. Before extraction of RNA, the liver

homogenate was thawed at room temperature and then heated to 37°C for 20 minutes.

To extract total RNA from the murine liver, the homogenate was processed using an RNeasy Midi kit according to the manufacturer's instructions. Briefly, the thawed homogenate was centrifuged at $5000 \times g$ for 10 minutes in a Sigma 4K15 centrifuge and the supernatant transferred to a fresh tube. One volume of 70% (v/v) ethanol was then added to the homogenate, which was mixed vigorously and passed through an RNeasy midi column by centrifugation at $5000 \times g$ for 5 minutes. The column was then washed once by addition of 4 ml Buffer RW1 (Qiagen) and centrifugation at $5000 \times g$ for 5 minutes, and then twice by addition of 2.5 ml Buffer RPE (Qiagen) followed by centrifugation at $5000 \times g$ for 2 minutes. After the third wash, the column was dried by a final centrifugation at $5000 \times g$ for 10 minutes. Finally, the RNA was eluted with 250 µl RNase-free water, and the eluate passed through the column a second time to collect any remaining RNA. The yield obtained was approximately 0.5 µg RNA/mg of liver tissue.

2.2.3.3. DNase I treatment of RNA

Column-purified total RNA (2.5 μ g) was treated with one unit amplification grade DNase I in 20 mM Tris-HCl (pH 8.4), 50 mM KCl and 2 mM MgCl₂ for 15 minutes at room temperature. EDTA was added to 2.3 mM and the sample was incubated at 65°C for 15 minutes to inactivate the DNAse I.

2.2.3.4. Generation of cDNA

cDNA was generated from total HepG2 or mouse RNA in a random hexamer-primed SuperScript II reverse transcriptase reaction using the SuperScript First-Strand Synthesis System (Invitrogen). Briefly, one microgram of RNA was added to a 10 µl reaction containing 1 mM dNTPs and 5 ng/µg random hexamer oligonucleotides, heated to 65°C for 5 minutes and cooled on ice for one minute. The reaction mix was then made up to a volume of 19 µl, such that when 1 µl of reverse transcriptase was also added later, the final composition was: 50 ng/µl RNA, 0.5 mM dNTPs, 2.5 ng/µl random hexamers, 20 mM Tris-HCl (pH 8.4), 50 mM KCl, 5 mM MgCl₂, 0.01 mM DTT and 0.1 units/µl RNaseOUTTM Recombinant Ribonuclease Inhibitor. After preheating the reaction mix to 25°C for 2 minutes, 50 Units (1 µl) of Superscript II reverse transcriptase were added and the reaction incubated at 25°C for 10 minutes, then at 42°C for a further 50 minutes. cDNA synthesis was terminated at 70°C for 15 minutes, after which 2 units of *E. coli* RNase H were added and the RNA digested at 37°C for 20 minutes. Finally, the cDNA was diluted 1:5 in DNAse-free water before use. All DNA samples, whether genomic, cDNA, plasmid or PCR product, were stored at -20°C.

2.2.4. Extraction of genomic DNA from HEK293T cells

HEK293T cells were grown to near-confluence in a 75 cm² tissue culture flask and washed twice in PBS. Cells were then lysed *in situ* by addition of 5 ml lysis buffer (10 mM Tris pH 8.0, 5 mM EDTA, 0.2% (w/v) SDS, 0.2 M NaCl and 0.1 mg/ml proteinase K) and incubation at 37°C for 2 hours with occasional shaking. After lysis, an equal volume of isopropanol was added to the lysate and the DNA collected by physically scooping the precipitate out of the supernatant. The DNA was then blotted on tissue paper and dissolved in 10 mM Tris pH 7.5, aided by incubation at 37°C for 2 hours.

To prepare the genomic DNA for use in PCR as template, 2.5 μ g was digested with 20 Units *Not*I restriction endonuclease in a 50 μ I reaction buffered with NEB buffer

3 (50 mM Tris-HCl, 100 mM NaCl, 10 mM MgCl₂, 1 mM DTT, pH 7.9) plus 0.1 mg/ml BSA. Restriction was performed for 3 hrs at 37°C.

2.2.5. Plasmid and BAC preparation

Small scale plasmid DNA preparations (minipreps), suitable for use as PCR template, sequencing template or in restriction analysis, were generated from 3 ml overnight bacterial cultures using the QIAprep Spin Miniprep kit as per the manufacturer's instructions. The bacterial pellet was collected by centrifugation of the overnight culture at $6,000 \times g$ for 15 minutes in a Sigma 4K15 centrifuge and the supernatant discarded. The pellet was then resuspended in 250 µl Buffer P1 containing 100 µg/ml RNase A (Qiagen), lysed by addition of 250 µl Buffer P2, and the bacterial proteins and genomic DNA precipitated by the addition of 350 µl Buffer N3. This precipitate was then pelleted by centrifugation at $18,000 \times g$ in a Beckman Microfuge 18 centrifuge for 10 minutes and the supernatant applied to a Qiaprep spin column attached to a Qiavac manifold. The DNA solution was drawn through the column by application of a vacuum. The column was washed with 500 µl Buffer PB, followed by 750 µl Buffer PE, both applied via the vacuum system. The column was then dried by centrifugation at $18,000 \times g$ for 1 minute and the DNA eluted by application of 50 μ l 10 mM Tris, pH 8.5 and a further centrifugation at 18,000 \times g for 1 minute.

Large scale plasmid DNA preparations (midipreps), used for mammalian cell transfection, were generated from 50 ml overnight bacterial cultures using the QIAGEN Plasmid Midiprep kit as per the manufacturer's instructions. The bacterial pellet was collected by centrifugation of the overnight culture at $6,000 \times g$ for 15 minutes in a Sigma 4K15 centrifuge and the supernatant discarded. The bacteria

were then resuspended in 4 ml Buffer P1 containing 100 µg/ml RNase A (Qiagen), and 4 ml Buffer P2 (Qiagen) was added to facilitate lysis. After incubation at room temperature for 5 minutes, 4 ml ice-cold Buffer P3 (Qiagen) was added to the bacterial lysate, which was mixed vigorously and incubated on ice for 15 minutes. All precipitate was pelleted by two sequential centrifugations in a JM20 rotor/Beckman J2-21M/E ultracentrifuge at 20,400 × g, 4°C, transferring the supernatant to a fresh tube in between. The initial centrifugation was for 30 minutes, and the second, 15 minutes. The cleared supernatant was then added to a Qiagen-tip 100 that had been pre-equilibrated with 10 ml Buffer QBT (Qiagen), and allowed to pass through the resin under gravity. The Qiagen-tip 100 was then washed twice with 10 ml Buffer QC (Qiagen) and the DNA eluted with 5 ml Buffer QF (Qiagen). 3.5 ml isopropanol was then added to precipitate the eluted DNA, and the DNA pelleted by centrifugation at 20,400 \times g, 4°C for 30 minutes. Finally the DNA was washed in 2 ml 70% (v/v) ethanol, recovered by centrifugation at $20,400 \times g$, 4°C for 10 minutes and air-dried. Midiprep DNA pellets were re-suspended in 500 µl 10 mM Tris, pH 8.5.

BAC DNA was extracted from 3 ml overnight bacterial cultures by the following method. Overnight cultures were harvested by centrifugation in a Sigma 4K15 centrifuge at $6000 \times g$ for 15 minutes and the pellets resuspended in 300 µl Buffer P1 and 100 µg/ml RNase A (Qiagen). The bacteria were then lysed by mixing the suspension with 300 µl Buffer P2 (Qiagen), followed by incubation at room temperature for 5 minutes. 300 µl Buffer P3 (Qiagen) was then added slowly to the cell lysate with gentle mixing, and the mixture placed on ice for 5 minutes. Following this second incubation, the mixture was centrifuge at 15,600 × g in an IEC Centra-M centrifuge at 4°C for 10 minutes and the supernatant added to a fresh

tube containing 800 μ l ice-cold isopropanol. After mixing, the DNA was allowed to precipitate by incubation on ice for 10 minutes and was collected by centrifugation at 15,600 × g, 4°C for 15 minutes. The DNA pellet was then washed with 500 μ l 70% (v/v) ethanol, re-collected by centrifugation at 15,600 × g, 4°C for 5 minutes, air-dried until translucent, and resuspended in 40 μ l 10 mM Tris, pH 8.5.

2.2.6. Polymerase chain reaction amplification

2.2.6.1. Equipment

Non-quantitative PCR reactions were performed on Cetus (PerkinElmer), RoboCycler Gradient 96 (Stratagene) or iCycler (Bio-Rad) thermal cyclers. The particular instrument used is not specified for each reaction, except in instances where this altered the PCR outcome or an exact annealing temperature was required. For quantitative real-time PCR (QPCR) reactions, a Rotor-GeneTM 3000 (Corbett Life Science, Mortlake, NSW, Australia) thermal cycler was used. Analysis of realtime PCR results was performed using Rotor-Gene 6 software (Corbett Life Science).

2.2.6.2. Primers

The nucleotide sequences of all oligonucleotides used for cloning and screening PCRs can be found in the "Methods" sections of each appropriate chapter, unless listed in Table 2.2. In addition, the sense sequences of the primers pairs used in site-directed mutagenesis reactions and electrophoretic mobility-shift assays (EMSAs) are also recorded in the relevant chapters. Oligonucleotides used for site-directed mutagenesis were polyacrylamide gel electrophoresis purified by Sigma-Genosys. Otherwise, primers were purchased desalted.

2.2.6.3. PCR for cloning

PCRs intended to generate DNA fragments for cloning were performed with the proof-reading *PfuTurbo*[®] DNA polymerase unless otherwise stated, to minimise the occurrence of introduced sequence errors. PfuTurbo is a mixture of Pfu DNA polymerase and the ArchaemaxxTM polymerase-enhancing factor, and has an error rate of 1.3×10^{-6} mutations per base pair per duplication. *PfuTurbo* PCR reactions were performed in the supplied PfuTurbo buffer (20 mM Tris-HCl pH8.8, 2 mM MgSO₄, 10 mM KCl, 10 mM (NH₄)₂SO₄, 0.1% (v/v) Triton-X 100 and 0.1 mg/ml BSA) with 0.05 Units/µl PfuTurbo, 200 nM dNTPs and 5 ng/µl each primer. The templates, primer sequences and specific cycling conditions for each reaction are detailed in the appropriate sections. In cases where *PfuTurbo* proved unsuitable for amplification, either the more robust *Taq* DNA polymerase was used as described in section 2.2.6.4 or the Expand[™] Long Template PCR system was utilised. For templates exceeding 2.5 kb where Taq DNA polymerase was used, equal units of Tag ExtenderTM PCR additive and *Tag* were added to the PCR to improve product yield. The Expand[™] Long Template PCR System, which utilises a mixture of *Taq* and Tgo DNA polymerases to optimise amplification range and fidelity (4.8×10^{-6}) mutations per base pair per duplication), was only used to clone the UGT1A3-9.4kb promoter (Chapter 5, section 5.2.6)

After PCR, the remaining dNTPs, primers and salts were removed from the PCR products using the QIAquick PCR Purification kit. Five volumes of Buffer PB (Qiagen) were mixed with the PCR product and the entire mixture placed on a QIAquick spin column (Qiagen). After centrifugation in a Beckman Microfuge 18 centrifuge at $18,000 \times g$ for 1 minute, the column was washed by addition of 750 µl Buffer PE and centrifugation at $18,000 \times g$ for 1 minute. The column was then dried

by centrifugation at 18,000 \times g for 1 minute and the DNA eluted in 50 μ l 10 mM Tris, pH 8.5.

If non-specific products persisted after optimisation of the PCR reaction, the PCR reaction was subjected to electrophoresis through an agarose gel (described in section 2.2.8) and the desired product excised. The DNA was then retrieved from the agarose slice using the QIAquick PCR Purification kit and Buffer QG (Qiagen) as per the manufacturer's instructions (described in section 2.2.8).

2.2.6.4. PCR for genotyping or screening

In cases where fidelity was not critical, or if a target could not be amplified with *PfuTurbo, Taq* DNA polymerase was used. *Taq* DNA polymerase has an error rate of 1×10^{-4} to 2×10^{-5} mutations per base pair per duplication. *Taq* mediated-PCR reactions were performed in the supplied ThermoPol buffer (20 mM Tris-HCl pH8.8, 10 mM (NH₄)₂SO₄, 10 mM KCl, 2 mM MgSO₄, and 0.1% (v/v) Triton-X 100) with 200 nM dNTPs and 5 ng/µl each primer. Again, the specific conditions for each reaction are detailed in the relevant sections of this thesis. If the resulting PCR product required purification before use downstream, this was performed as described in section 2.2.6.3. Generally, it was found that it was unnecessary to purify PCR products before sequencing, provided that the PCR product concentration was sufficiently high that the reaction mix could be diluted by at least 1:5 before submission to the DNA Sequencing Core Facility.

2.2.6.5. Site-directed mutagenesis

To insert mutations into plasmid constructs in a controlled manner, two complementary oligonucleotides were designed over the target site according to the rules established by Stratagene's QuikChange[®] site-directed mutagenesis kit. Thus,

where practicable, the primers pairs were completely complimentary and contained the desired mutation(s) flanked by at least 10 bases of correct sequence on both sides. They also possessed melting temperatures of \geq 78°C, a minimum GC content of 40% and one or more C or G terminating residues. Mutagenesis reactions were performed in a total volume of 50 µl containing PfuTurbo buffer, 200 nM dNTPs, 2.5 Units PfuTurbo DNA polymerase, 250 ng each oligonucleotide and 50 ng plasmid template. The default reaction conditions were 30 seconds of denaturation at 95°C, followed by 16 rounds of 30 seconds at 95°C, 1 minute annealing at 55°C and an extension time of 2 minutes per kb of target plasmid. Deviations from these conditions are recorded in the appropriate thesis sections. 10 Units of *DpnI* were added to the amplified vector and incubated at 37°C for 1 hour. This step degrades the wild-type parent vector, but not the mutated transcripts, as *Dpn*I recognition of restriction sites is methylation sensitive and PCR-generated plasmids are not methylated. One microlitre of digested PCR product was then used to transform 100 μl competent DH5α E. coli as described in section 2.2.7. The presence of the desired mutation(s) was confirmed by sequencing of miniprep DNA prepared from representative colonies.

2.2.6.6. Quantitative real-time PCR

To quantify the levels of UGT mRNA transcripts and 18S ribosomal RNA (rRNA) present in RNA extracted from HepG2 cells, real-time PCR was used. The primer sets and annealing temperatures used are detailed in Chapter 5, section 5.2.5. However, the generic set-up used for all reactions were: 20 μ l aliquots containing 1 \times QuantiTect SYBR Green PCR master mix (comprised of HotStarTaq DNA polymerase, QuantiTect SYBR Green I, ROX passive reference dye and 2.5 mM MgCl₂), pH 8.7), dNTP mix, SYBR Green I, ROX passive reference dye and 2.5 mM MgCl₂),

0.5 µM each primer, and template cDNA equivalent to 40 ng input RNA. For 18S PCRs, the amount of template cDNA used per reaction was decreased to the equivalent of 1.25 ng input RNA. The ROX dye is not required for analysis using the Rotor-Gene 3000 and does not interfere with the system. The generic PCR cycling conditions used for quantitative analysis were: an activation period of 15 minutes at 95°C; 40 cycles of 95°C for 15 seconds, specific annealing temperature (see Table 5.1) for 20 seconds, and 72°C for 20 seconds; and a ramped melt analysis between 60 and 95°C with 5 second, 1°C steps. Data was acquired during the 72°C extension phase of each cycle.

2.2.7. Restriction digests, calf intestinal alkaline phosphatase treatment, ligation and transformation

Preparative restriction digests (for cloning) were done with 5 μ g of starting material in 50 μ l reactions. Each reaction also contained 1 × appropriate reaction buffer (see below), 100 μ g/ml BSA and 0.2-0.4 Units/ μ l restriction enzyme. Digestion was performed at 37°C (unless otherwise stated) for 3 hours. Analytical restriction digests were performed with approximately 100 ng plasmid DNA in a reaction volume of 20 μ l containing 100 μ g/ml BSA and 0.1-0.2 Units/ μ l each restriction enzyme. DNA was digested at 37°C (unless otherwise stated) for 1 hour.

For *Kpn*I digests, NEB Buffer 1 (10 mM Bis-Tris-Propane-HCl, 10 mM MgCl₂ and 1 mM DTT, pH 7.0) was used. For *Bam*HI, *Bsr*GI, *Eco*RI, *Hind*III, *Nhe*I, *Pvu*II, *Spe*I, *Xba*I and *Xho*I digests, or double digests, NEB Buffer 2 (10 mM Tris-HCl, 50 mM NaCl, 10 mM MgCl₂ and 1 mM DTT, pH 7.9) was used. For *Mlu*I and *Pst*I digests, NEB Buffer 3 (50 mM Tris-HCl, 100 mM NaCl, 10 mM MgCl₂ and 1 mM DTT, pH 7.9) was used. *Kpn*I/*Mlu*I double digests were performed sequentially in their

respective buffers due to buffer incompatibility. *MluI/XhoI* double digestions were performed in NEB Buffer 3.

For vectors that were to be used in cloning after restriction with only a single enzyme, treatment with calf intestinal alkaline phosphatase (CIP) was employed to prevent self-ligation. CIP treatment was performed by addition of 10 Units CIP to the 50 μ l preparative digestion after the initial 3 hour incubation (no buffer change necessary), followed by incubation at 37°C for an additional hour.

Ligations were performed using the NEB Quick Ligation kit. Twenty microlitre reactions were prepared with 50 ng plasmid DNA, a 3 to 10-fold molar excess of insert DNA and $1 \times$ NEB Quick Ligase buffer (66 mM Tris-HCl, 10 mM MgCl₂, 1 mM DTT, 1 mM ATP and 7.5% (w/v) polyethylene glycol 6000, pH 7.6). One microlitre of Quick T4 DNA Ligase was then added to the mix and incubated for 5 minutes at room temperature. The reaction was chilled on ice before use in transformation.

All transformations were performed using chemically competent DH5 α *E. coli* cells prepared as described in section 2.2.2, unless otherwise stated in the specific methods of each chapter. Competent cells were thawed on ice for 10 minutes before addition of up to 5 µl of ligation product. The cells were then mixed gently and incubated on ice for a further 30 minutes before exposure to heat shock. To facilitate uptake of the ligated DNA, the cells were incubated at 42°C for 45 seconds and then immediately placed on ice for 2 minutes. 400 µl of LB was then added to the shocked cells and they were allowed to recover at 37°C for one hour before being plated on LB/agar plates containing an appropriate antibiotic for selection. Colonies were allowed to form at 37°C overnight before being screened for desired clones.

2.2.8. Electrophoresis and quantification of DNA and RNA

DNA fragments and plasmids were analysed for relative size (based on mobility) and purity (from unwanted DNA fragments and contaminating RNA) by agarose gel electrophoresis. DNA samples were mixed with $6 \times$ DNA gel loading buffer to give the final buffer concentration specified in section 2.1.2 and loaded onto 0.8-2% (w/v) agarose gels in Tris-acetate EDTA electrophoresis buffer (section 2.1.2) and 0.4 µg/ml ethidium bromide. Electrophoresis was performed in a Bio-Rad Mini-Sub Cell GT electrophoresis system by applying 5 volts/cm (between the electrodes) to the gel. The gel was then examined with either a Bio-Rad Gel Doc 1000 system with Molecular Analyst Software (Bio-Rad), or a Gene Genius Bio imaging system (Syngene, Cambridge, England) and GeneSnap version 6.04 software (Syngene).

To recover DNA samples, agarose slices containing the required DNA were swiftly excised under low-intensity UV irradiation and added to 3 volumes of buffer QG (Qiagen). After heating to 50°C for 10 minutes with occasional mixing, one gel volume of isopropanol was added to the sample, which was then mixed and added to a QIAquick column. After centrifugation at $18,000 \times g$ in a Beckman Microfuge 18 centrifuge for 1 minute, the column was washed once with 500 µl Buffer QG and twice with 750 µl Buffer PE, with a 1 minute centrifugation step between each wash. The column was dried by centrifugation at $18,000 \times g$ for 1 minute and the DNA eluted into a clean tube by addition of 50 µl 10 mM TrisCl, pH 8.5 to the column and centrifugation at $18,000 \times g$ for 1 minute.

The concentration and purity of DNA samples were determined by spectrophotometry (optical absorbance at 260 nm versus 280 nm) using a GeneQuant Π (Pharmacia Biotech (GE Healthcare), Buckinghamshire, England) spectrophotometer and/or estimation against a series of known standards after electrophoresis into an agarose gel and staining with ethidium bromide as described earlier. Likewise, the concentration of RNA was determined by spectrophotometry.

2.2.9. Sequencing

All sequencing was performed by the DNA Sequencing Core Facility (Department of Haematology and Genetic Pathology, Flinders University) using Big Dye Terminator Cycle Sequencing Version 3.1 chemistry (Applied Biosytems, Foster City, CA) and an ABI 3100 Genetic Analyser sequencer (Applied Biosystems). Sequencing of pGL3-derived plasmids performed with GL2 (5' was (5' CTTTATGTTTTTGGCGTCTTCC 3') RV3 and GGGACAGCCTATTTTGCTAG 3') primers. Sequencing of pCMX-derived plasmids was performed with a T7 primer (5' TAATACGACTCACTATAGGGAGA 3').

2.2.10. Transient transfection of HepG2, Caco-2 and HEK293T cells

HepG2, Caco-2 or HEK293T cells were re-suspended in growth medium before they reached 80% confluence and counted in the presence of 0.2% (w/v) trypan blue. Only cells that excluded the trypan blue dye were considered viable. For 24-well plates (Nunc, Kamstrupvej, Denmark), HepG2 cells were seeded at a density of 2×10^5 cells per well, Caco-2 at 7.5×10^4 cells per well and HEK293 at 1×10^5 cells per well in at total volume of 500µl. Where 6-well plates (Nunc) were used, HepG2 cells were plated a density of 1×10^6 cells per well in a total volume of 2 ml.

Twenty-four hours after plating, transfections were performed using Lipofectamine 2000. The amount and identity of the DNA used in each transfection was experiment-specific and is detailed in each appropriate chapter. For transfections in 24-well plates, the DNA for each well was added to 50 µl serum-free DMEM, for

experiments in 6-well plates, 250 µl of DMEM was used. Two or 10 µl of the Lipofectamine 2000 cationic lipid transfection agent was then mixed with 50 or 250 µl serum-free DMEM for 24 or 6-well plate experiments respectively, and allowed to stand at room temperature for 5 minutes. The DNA and Lipofectamine 2000 mixtures for each well were then combined and incubated at room temperature for a minimum of 20 minutes. One hundred microlitres (24-well plates) or 500 µl (6-well plates) of Lipofectamine 2000:DNA complexes were then added directly to each well without removing the plating medium. The cells were then incubated overnight before changing the medium for fresh culture medium with all additives, unless chemical treatment of cells was required to commence prior. In this case, cells were incubated with the Lipofectamine 2000:DNA complexes for a minimum of 6 hours before the transfection medium was replaced with culture medium containing the desired treatment. pRL-null, which constitutively expresses the renilla (Renilla reniformis) luciferase gene was added to transfections as an internal control for transfection efficiency, with the exception of transfections that were to be harvested for RNA.

2.2.11. Luciferase assay

Forty-eight hours post-transfection, cells were lysed by addition of passive lysis buffer (100 μ l per well for 24-well plates; 500 μ l per well for 6-well plates) followed by incubation at room temperature with continual rocking for 20 minutes. A 20 μ l sample of lysate was then analysed sequentially for firefly (*Photinus pyralis*) and *renilla* luciferase activity using the Dual-Luciferase Reporter Assay system (Promega) and a TopCount luminescence and scintillation counter (Parkard, Mt Waverly, Victoria, Australia). Lysate was added to alternate wells of 96-well plates (to avoid crossover luminescence from neighbouring wells), and mixed with 50 μ l of firefly luciferase reagent (Luciferase Assay Reagent II, Promega) to measure the activity of firefly luciferase expressed from transiently transfected pGL3-derived vectors. The luminescence of each sample was quantified 100 seconds after addition of Luciferase Assay Reagent II. The samples were then quenched by addition of 50 μ l of the Stop and Glo Reagent (Promega), which also contains the substrate for *renilla* luciferase. The luminescence from the *renilla* luciferase protein was also measured 100 seconds after addition of the relevant substrate. If the firefly luminescence reading exceeded 2 × 10⁶ counts per second, the samples were diluted in passive lysis buffer to ensure that the results were within the linear range of the luminescence counter used.

Each transfection was performed in triplicate, and mean relative luciferase activities were calculated from the three resulting ratios of firefly to *renilla* luciferase activity. Results presented are the mean activity (plus one standard deviation) of each promoter-reporter vector relative to the promoterless pGL3-basic vector (with the activity of pGL3-basic being set arbitrarily to a value of one). Unless otherwise stated, all triplicate transfections presented were performed at least twice in independent experiments.

2.2.12. Preparation of nuclear extracts

Nuclear extracts were prepared by the following method adapted from Schreiber *et al.* (1989) as published in Gardner-Stephen and Mackenzie (2005). All wash and lysis buffers were pre-chilled on ice. HepG2 cells were grown to confluence in 175 cm² flasks, rinsed once in PBS and harvested in 10 ml PBS by scraping. The pellet from each flask was collected by centrifugation at $1500 \times g$ for 5 minutes in a Sigma 4K15 centrifuge, washed in 1 ml PBS, pelleted in an IEC Centra-M centrifuge at

15,600 × g, 4°C for 1 minute and re-suspended in 800 µl buffer A (10 mM Tris-HCl pH 7.9, 10 mM KCl, 1 mM DTT, 1.5 mM MgCl₂ and 1 × Complete protease inhibitor cocktail) with 0.5% (v/v) Nonidet P-40. After 15 minutes on ice the nuclear fraction was pelleted at $15,600 \times \text{g}$ for 1 minute (4°C) and washed in 800 µl buffer A. The washed nuclei from four flasks were pooled and re-suspended in 400 µl ice-cold buffer B (50 mM Tris-HCl pH 7.9, 500 mM KCl, 2mM DTT, 5 mM MgCl₂, 0.1 mM EDTA, 10% (w/v) sucrose, 20% (v/v) glycerol and $1 \times$ Complete protease inhibitor cocktail). The tube containing the pooled sample was then buried in ice in a small beaker and shaken vigorously for an hour to facilitate lysis of the nuclei. The nuclear extract supernatant was separated from the remaining debris by centrifugation for 15 minutes at 15,600 \times g at 4°C and transferred to a Slide-A-Lyzer[®] dialysis cassette. Dialysis was performed for at least 2 hours against 200 ml of buffer TM-1 (25 mM Tris-HCl pH 7.6, 100 mM KCl, 0.5 mM DTT, 5 mM MgCl₂, 0.5 mM EDTA and 10% (v/v) glycerol) at 4°C. The protein concentration of the dialysed nuclear extract was determined by comparison to BSA standards using the Bio-Rad Protein Assay reagent, before storage at -80 °C.

2.2.13. Labelling oligonucleotide probes

To generate double stranded DNA probes from two complementary oligonucleotides, 5 µg of each primer was denatured at 95°C for 2 minutes in 100 µl of annealing buffer (40 mM Tris-HCl pH 7.5, 20 mM MgCl₂ and 50 mM NaCl) in a dry block heater. The probes were then allowed to cool unassisted to room temperature in the heating block over the course of several hours. One hundred nanograms of annealed probe was then end-labelled by incubation with approximately 4 MBq γ^{32} P-ATP and 10 units of T4 polynucleotide kinase in the supplied reaction buffer (70 mM Tris-HCl pH 7.6, 10 mM MgCl₂ and 5 mM DTT) for one hour at 37°C. The labelled probe was then diluted 1:10 in Tris-EDTA buffer (section 2.1.2) and separated from free γ^{32} P-ATP by purification through G25 columns according to the manufacturer's instructions.

2.2.14. Electrophoretic mobility-shift assays

To determine whether nuclear proteins could bind to *UGT* promoter regions of interest, EMSAs were used. The mobility of a DNA oligonucleotide passing through a non-denaturing polyacrylamide gel is retarded if it forms a higher molecular weight complex with supplied proteins. To demonstrate the specificity of any protein-probe interactions formed, up to 500-fold excess unlabelled wild-type or unrelated probes were used to compete for protein binding. Mutations were also introduced into selected EMSA probes to demonstrate the importance of particular nucleotides in the binding of protein complexes. To confirm the identity of proteins bound to a probe, antibodies with reactivity towards the protein of interest were added to the reaction, thereby further increasing the molecular weight of the complex formed if the suspected protein was present (see section 2.2.15).

For HNF1 EMSAs, either 5 µg of HepG2 or Caco-2 nuclear extract, or 1 µl *in vitro* synthesised HNF1 α or HNF1 β protein (generated using the TNT Quick Coupled Transcription/Translation; see Chapter 4, section 4.2.10) were incubated with 1 µg poly(dIdC) for 10 minutes on ice, in a reaction mix made up to a total volume of 15 µl with buffer TM-1. If unlabelled competitor probes were required, they were also included in this 15 µl reaction mix. Fifty thousand counts per minute (cpm) of γ^{32} P-ATP end-labelled double-stranded DNA probe (as defined in each appropriate chapter) were then added to each reaction as an extra 1 µl aliquot. After the addition of radioactive probe, reactions were kept at room temperature for 30 minutes to

allow DNA-protein complexes to form. These were then resolved on a 4% (w/v) non-denaturing polyacrylamide gel made with 29:1 Acrylamide/Bis in $0.5 \times$ TBE. After pre-running the gel at 4°C for 2 hours at 170 V, electrophoresis of samples was achieved by applying 250 V for 2 hours at 4°C using Dual Slab Gel kit electrophoresis equipment (CBS Scientific Company, Del Mar, CA). The gel was then dried under vacuum and exposed to X-Omat Blue XB-1 film at -80°C with the aid of intensifying screens.

For HNF4 α EMSAs, 5 µg of HepG2 nuclear extract was incubated with 1 µg poly(dIdC) and 0.5 µg sonicated salmon sperm DNA for 10 minutes on ice in a reaction mix made up to a total volume of 15 µl with buffer TM-2 (10 mM Hepes-NaOH pH 7.8, 100 mM NaCl, 0.1 mM EDTA, 10% (v/v) glycerol, 1 mg/ml BSA and 0.5 mM DTT) (Pineda-Torra *et al.*, 2002). Addition of unlabelled or labelled probe, incubation, electrophoresis and exposure to film were all performed as described for the HNF1 EMSA experiments.

2.2.15. Super-shift EMSA

For super-shift EMSAs, 2 μ g of anti-HNF1 α , anti-HNF1 β or anti-HNF4 α antibody was added to EMSA reactions immediately after the addition of labelled probe. Samples were then treated in the same way as described for standard EMSA experiments.

2.2.16. Western blot

Twenty micrograms of HepG2 or HEK293T total cell lysate or 1 μ l *in vitro* synthesised HNF1 α or HNF4 α protein (see Chapter 4, section 4.2.10) were subjected to electrophoresis on 10% (w/v) SDS-polyacrylamide gel electrophoresis gels (made with 19:1 Acrylamide/Bis in 0.5 × TBE with a 4% (w/v) stacking gel).

Electrophoresis was performed at room temperature; at 70V until the samples cleared the stacking gel and then at 150V until the tracking dye reached the bottom of the gel using Mini-Protean II Cell equipment (Bio-Rad). The separated proteins were then transferred to Trans-Blot Transfer Medium (nitrocellulose membrane, 0.45 μ m) for 1 hour at 100V, using an ice-cooled Mini Trans-Blot Cell apparatus (Bio-Rad).

To detect HNF1 α or HNF4 α proteins present on membranes after transfer, blots were blocked overnight at 4°C in TBST [Tris-buffered saline (10 mM Tris-HCl, pH 7.4 and 150 mM NaCl) plus 0.05% (v/v) Tween-20] containing 5% (w/v) skim milk powder (SMP). The membrane was then incubated for 2 hours with 1 µg/ml anti-HNF1 α or anti-HNF4 α antibody in TBST/SMP, and washed three times with TBST for 5 minutes each. The membrane was then incubated for a further hour in TBST/SMP containing 1:2,500 diluted rabbit anti-goat IgG-horseradish peroxidaseconjugated secondary antibody and washed three times with TBST for 5 minutes each. Finally, the blot was given a final wash for 5 minutes in Tris-buffered saline, was treated with ECL Western blotting detection reagent according to the manufacturer's instructions and exposed to X-Omat Blue XB-1 autoradiographic film. All washes and antibody incubations were performed at room temperature with gentle rocking.

2.2.17. Statistical analysis

All statistical analyses reported in this thesis were performed using the SPSS software package version 12.0.1 (SPSS Inc., Chicago, IL) or Microsoft Office Excel 2003 software (Microsoft Corporation, WA).

The statistical significance of altered promoter activity in luciferase assays, or of altered levels of endogenous *UGT* mRNA transcripts in treated HepG2 cells, was

determined by independent-samples *t*-tests. Where results were subject to heteroscedasticity (nonconsistant variance), data were log-transformed before analysis. Results were considered statistically significant if P < 0.05.

Statistical analysis of *UGT1A3* promoter variant frequencies and compliance with Hardy-Weinberg equilibrium (tested by Pearson χ^2) were performed using Microsoft Office Excel 2003 software (Microsoft Corporation, WA).

2.3 Cloning of liver-enriched transcription factors

The primers used to amplify the coding regions of each transcription factor are listed in Table 2.2. Details for the amplification of each PCR product are given below in each appropriate section.

2.3.1. Hepatocyte nuclear factor 1β (HNF1β)

pCMX-HNF1β was constructed as described in Gardner-Stephen and Mackenzie (2007a), using HepG2 cDNA as template for amplification of the HNF1β coding region, and primers HNF1βfor and HNF1βrev (Table 2.2). The following *PfuTurbo* PCR (see section 2.2.6.3) was performed on the cDNA equivalent of 0.8 ng/µl of input RNA. The amplification parameters used were: 4 minutes initial denaturation at 95°C; 30 cycles of 95°C for 30 seconds, 55°C for 30 seconds and 72°C for 3 minutes; completed by a 5 minute final extension at 72°C. The resulting PCR product was cleaned using the QIAquick PCR purification kit, digested with *Hind*III and *Bam*HI (section 2.2.7), cloned into pCMX-PL2 and sequenced in full (section 2.2.9).

Oligonucleotide	Nucleotide Sequence (5'→3')	RE
HNF1βfor	AGCCAT <u>AAGCTT</u> ATGGTGTCCAAGCTCACGTCG	HindIII
HNF1βrev	AGCCAT <u>GGATCC</u> TCACCAGGCTTGTAGAGGACAC	BamHI
hHNF4a2for	AGCCAT <u>AAGCTT</u> ATGCGACTCTCCAAAACCCTCGT	HindIII
hHNF4arev	AGCCAT <u>GGATCC</u> TAGATAACTTCCTGCTTGGTGAT	BamHI
QC hHNF4a1for	CACCTCAGCAACGGACAGATGT^CCACCCCTGAGA	NA
	CCCCACAG	
rHNF4a1for	AGCCAT <u>AAGCTT</u> ATGCGACTCTCTAAAACCCTC	HindIII
rHNF4a1rev	AGCCAT <u>GGATCC</u> TAGATGGCTTCCTGCTTGGTGAT	BamHI
HNF6for	AGCCAT <u>AAGCTT</u> ATGAACGCGCAGCTGACCATG	HindIII
HNF6rev	AGCCAT <u>GGATCC</u> TCATGCTTTGGTACAAGTGCTTG	BamHI
hFoxA3for	AGCCAT <u>AAGCTT</u> ATGCTGGGCTCAGTGAAGATG	HindIII
hFoxA3rev	AGCCAT <u>GGATCC</u> TAGGATGCATTAAGCAAAGAG	BamHI
hPXRT1for	AGCCAT <u>AAGCTT</u> ATGGAGGTGAGACCCAAAGA	HindIII
hPXRrev	AGCCAT <u>TCTAGA</u> TCAGCTACCTGTGATGCCGAA	XbaI

 Table 2.2: Primers used for cloning transcription factor cDNAs.

RE: restriction endonuclease site, as underlined. NA: not applicable. The position of the 30 nucleotides deleted from HNF4 α 2 to create HNF4 α 1 is indicated by the circumflex accent (^). The deliberate mutation in the hPXRT1 for primer is highlighted in bold.

2.3.2. Hepatocyte nuclear factor 4α (HNF4α)

2.3.2.1. Human HNF4a splice variant 2

 $HNF4\alpha 2$ transcripts were amplified by PCR from cDNA generated from HepG2 total

RNA as described in Gardner-Stephen and Mackenzie (2007a). The 50µl PCR

contained cDNA equivalent to 40 ng of input RNA and was performed using primers

hHNF4a2for and hHNF4arev (Table 2.2) and PfuTurbo DNA polymerase. The PCR

cycling parameters were: 95°C for 4 minutes; 30 cycles of 95°C for 30 seconds, 58°C for 30 seconds and 72°C for 2 minutes; and a final extension step of 72°C for 5 minutes. The resulting PCR product was cloned into the *Hind*III and *Bam*HI sites of pCMX-PL2 utilising restriction sites incorporated into each primer. The HNF4 α coding region was sequenced in full to verify that the wild-type sequence had been obtained, as HepG2 cells are known to be heterozygous for a single base-pair change at nucleotide 206 of HNF4 α which results in a D69A mutation (Lausen *et al.*, 2000). HNF4 α 2 is the most abundant HNF4 α transcript in the adult liver (Hata *et al.*, 1995).

2.3.2.2. Human HNF4a splice variant 1

The HNF4 α 1 splice variant differs from HNF4 α 2 by a 10 amino acid insertion present in the F domain of HNF4 α 2 relative to HNF4 α 1 (Hata *et al.*, 1992; Chartier *et al.*, 1994; Hata *et al.*, 1995; Drewes *et al.*, 1996; Sladek *et al.*, 1999). To obtain an expression vector for human HNF4 α 1, pCMX-HNF4 α 2 was altered to contain an HNF4 α 1-identical cDNA region. This was achieved using the QuikChange sitedirected mutagenesis protocol (section 2.2.6.5) and the primer pair QC hHNF4 α 1 for and QC hHNF4 α 1rev, where QC hHNF4 α 1 for is listed in Table 2.2 and QC hHNF4 α 1rev is perfectly complementary to QC hHNF4 α 1for. The specific QuikChange PCR parameters for this reaction were: 95°C for 30 seconds; 16 cycles of 95°C for 30 seconds, 55°C for 1 minute and 68°C for 14 minutes.

2.3.2.3. Rat HNF4α splice variant 1

The coding region of the rat HNF4 α 1 splice variant was amplified by PCR from the plasmid pSG5-HNF4 α . The primers used were rHNF4 α 1 for and rHNF4 α 1 rev (Table 2.2). The following PCR was performed as a *PfuTurbo* reaction (section 2.2.6.3) using 0.6 ng/µl pSG5-HNF4 α plasmid template. Cycling conditions were: 95°C for 4 minutes; followed by 30 cycles of 95°C for 30 seconds, 55°C for 30 seconds and

72°C for 4 minutes; finished with an extra extension step of 72°C for 5 minutes. The resulting PCR product was cleaned with the Qiagen PCR purification kit, digested with *Hind*III and *Bam*HI, and ligated into the corresponding sites of pCMX-PL2. The entire insert was then sequenced for nucleotide errors.

2.3.3. Hepatocyte nuclear factor 6 (HNF6, OneCut 1)

Attempts to clone HNF6 from HepG2 cDNA resulted in only truncated or mutated transcripts being obtained. Therefore, HNF6 was cloned from murine liver RNA.

HNF6 was amplified from the murine liver cDNA equivalent of 80 ng of input RNA using *PfuTurbo*. The primers used were mHNF6for and mHNF6rev (see Table 2.2), and the amplification conditions were: 95°C for 4 minutes; followed by 40 cycles of 95°C for 30 seconds, 55°C for 1 minute and 72°C for 5 minutes; finished with an extra extension step of 72°C for 5 minutes. The resulting PCR product was inserted into the *Hind*III and *Bam*HI sites of pCMX-PL2, utilising the restriction sites incorporated into the PCR primers, and sequenced in full.

2.3.4. Forkhead box A factors (FoxA, hepatocyte nuclear factor 3 factors)

2.3.4.1. Rat FoxA1 (HNF3 α) and FoxA2 (HNF3 β)

Plasmids pRB-HNF3 α and pGEM-HNF3 β , containing rat FoxA1 and FoxA2 cDNAs respectively, were restricted with *Eco*RI to release their FoxA encoding inserts. The inserts were then separated from plasmid DNA by gel electrophoresis and purified using the QIAquick Gel Extraction kit. Each FoxA fragment was then nondirectionally cloned into pCMX-PL2 pre-treated with *Eco*RI and CIP. The orientation of the FoxA1 insert, relative to the pCMX-PL2 promoter, was determined by digestion with *Nhe*I, followed by analysis of the restriction pattern by agarose gel electrophoresis. Likewise, the orientation of the inserted FoxA2 cDNA was determined by double digestion with *Pvu*II and *Xho*I. Chosen clones were sequenced in full.

2.3.4.2. *Human FoxA3 (HNF3γ)*

FoxA3 was amplified from HepG2 cDNA using primers hFoxA3 for and hFoxA3 rev (Table 2.2) in a *PfuTurbo* DNA polymerase PCR reaction. The concentration of HepG2 cDNA present in the PCR reaction was the equivalent of 0.8 ng/µl of total RNA used to generate the cDNA. Amplification conditions for FoxA3 were as described for HNF1 β (section 2.3.1), and the resulting PCR product was digested with *Bam*HI and *Hind*III and ligated into pCMX-PL2. Sequencing confirmed that a full-length FoxA3 clone without mutations had been obtained.

2.3.5. CAATT enhancer binding protein factors (C/EBP)

Plasmids containing rat C/EBP α and C/EBP β cDNAs were used as starting material to construct the pCMX-CEBP α and pCMX-CEBP β expression vectors. Similarly to the assembly of pCMX-FoxA1 and pCMX-FoxA2, the C/EBP α and C/EBP β coding fragments were retrieved from their original vector by *Eco*RI digest followed by gel electrophoresis and excision/purification of the desired fragment. The rat C/EBP α and C/EBP β cDNAs were then cloned non-directionally into the *Eco*RI and CIP treated pCMX-PL2 preparation as for FoxA1 and FoxA2, checked for insert orientation by *Pst*I digestion and sequenced.

2.3.6. Pregnane X receptor (PXR)

The T1 splice variant of human PXR was cloned as described in Gardner-Stephen *et al.* (2004). In detail, the PXR T1 transcript was amplified from HepG2 cDNA (at a concentration equivalent to 20 ng/ μ l reverse transcribed RNA) using the primers hPXRT1for and hPXRrev (Table 2.2) and *PfuTurbo* DNA polymerase. After an

initial 4 minutes at 95°C, 40 cycles of 95°C for 45 seconds, 60°C for 45 seconds and 72°C for 4 minutes were performed and the PCR finished with 5 minutes at 72°C. The PCR-amplified cDNA was then inserted into the *Xba*I and *Hind*III sites of pCMV5 (GenBank record AF239249, Andersson *et al.*, 1989) and sequenced. To maintain consistency amongst the transcription factors investigated in this thesis, the PXR cDNA insert was then shuttled from the original pCMV5 vector into the *Hind*III and *Bam*HI sites of pCMX-PL2 to create pCMX-PXR. It should also be noted that the native CTG initiation codon of the human PXR T1 mRNA transcript was replaced by the more conventional ATG initiation codon through primer mismatch in hPXRT1for.
CHAPTER THREE IN VITRO CHARACTERISATION OF THE UGT1A3, UGT1A4 AND UGT1A5 PROXIMAL PROMOTERS

Published in part as: Gardner-Stephen, D.A. and Mackenzie, P.I. (2007) Isolation of the UDP-glucuronosyltransferase 1A3 and 1A4 proximal promoters and characterization of their dependence on the transcription factor hepatocyte nuclear factor 1α. *Drug Metab. Dispos.* **35**:116-120.

3.1 Introduction

3.1.1. The UGT1A3, UGT1A4 and UGT1A5 gene cluster

UGT1A3, UGT1A4 and UGT1A5 are a triad of highly related UGT proteins encoded by the human *UGT1A* locus (see Chapter 1, Figure 1.3). They share greater than 90% identity in their primary amino acid sequences and more than 85% nucleotide sequence identity in their 1 kb proximal promoters (Green and Tephly, 1998; Gong *et al.*, 2001). However, despite these similarities, they vary considerably in their substrate selectivities and in their expression patterns. Furthermore, all three genes are subject to considerable interindividual variation in expression (Strassburg *et al.*, 1998b; Strassburg *et al.*, 2000; Finel *et al.*, 2005). This chapter describes a series of experiments designed to investigate the function of the *UGT1A3*, *UGT1A4* and *UGT1A5* proximal promoters *in vitro*; a first step in understanding the mechanisms that: a) allow these highly related genes to be independently regulated; and b) lead to differential expression of these UGTs between individuals.

3.1.2. Expression of UGT1A3, UGT1A4 and UGT1A5

UGT1A3 mRNA transcripts have been found in liver, biliary tissue, kidney, stomach, small intestine, colon, prostate, testis and breast. Interestingly, the stomach, duodenum, jejunum and ileum only express detectable levels of UGT1A3 transcripts in a subset of individuals (Mojarrabi et al., 1996; Strassburg et al., 1997b; Mojarrabi and Mackenzie, 1998; Strassburg et al., 1998b; Strassburg et al., 2000; Sabolovic et al., 2004; Chouinard et al., 2006). UGT1A4 mRNA has a similar expression pattern to UGT1A3, being found in liver, biliary tissue, breast and colon, and also having polymorphic expression along the small intestine (Strassburg et al., 1997b; Strassburg et al., 1998a; Strassburg et al., 2000; Chouinard et al., 2006). However, this enzyme is notably absent in the stomach (Strassburg et al., 1998b). Although there is weak evidence that UGT1A4 is more highly expressed in the liver and breast than UGT1A3 (Ritter et al., 1992b; Congiu et al., 2002; Chouinard et al., 2006), both proteins are readily detectable in liver extracts by enzyme-specific antibodies (Ikushiro et al., 2006). In contrast, UGT1A5 has not been found to be expressed to any significant extent in any tissues, although highly variable (but very low) expression in liver and gastrointestinal tract has been recently reported (Chen *et al.*, 2005a; Finel et al., 2005).

As well as differing at the level of basal expression, *UGT1A3* and *UGT1A4* are differentially induced by various stimuli. Human UGT1A4, but not UGT1A3, has been found to have increased expression during pregnancy in Tg-*UGT1* transgenic mice (Chen *et al.*, 2005a). On the other hand, UGT1A3 expression is known to be increased by the nuclear receptor PXR in the presence of ligand to a much greater extent than UGT1A4 in HepG2 cells (Gardner-Stephen *et al.*, 2004). *UGT1A5* may also be a PXR-target gene, as rifampicin treated hepatocytes possess increased levels

of UGT1A5 mRNA (Finel *et al.*, 2005). Most recently, *UGT1A3* has been found to be responsive to the liganded liver X receptor (LXR) α in HepG2 cells and the liver of transgenic Tg-*UGT1* mice, although it was not reported whether UGT1A4 is also regulated by LXR α (Verreault *et al.*, 2006). Remarkably, despite their common sequences, the only regulators identified in common to *UGT1A3* and *UGT1A4* so far are HNF1 (this chapter, published by Gardner-Stephen and Mackenzie (2007b)), PXR (Rae *et al.*, 2001; Gardner-Stephen *et al.*, 2004) and PPAR α (Senekeo-Effenberger *et al.*, 2007).

3.1.3. Substrates of UGT1A3, UGT1A4 and UGT1A5

UGT1A3 and UGT1A4 are two UGTs involved in the catalysis of quaternary ammonium-linked glucuronides from tertiary amines, a substance class that includes many important pharmaceuticals and other bioactive molecules. Examples of substrates metabolised by both enzymes include ketotifen (an anti-allergic) (Breyer-Pfaff *et al.*, 2000) and amitriptyline (an anti-depressant) (Green *et al.*, 1998a; Breyer-Pfaff *et al.*, 2000). Further substrates known to be common to UGT1A3 and UGT1A4 are predominantly primary and secondary amines, such as the muscle relaxant afloqualone (Kaji and Kume, 2005) and the carcinogen benzidine (Green and Tephly, 1996; Green *et al.*, 1998a). Generally, amines are better substrates for UGT1A4 than UGT1A3 (Green and Tephly, 1996; Green *et al.*, 1998a).

Substrates for UGT1A3, but not UGT1A4, include the cholesterol absorption inhibitor ezetimibe (Ghosal *et al.*, 2004a), the pesticide methoxychlor (Hazai *et al.*, 2004), oripavine opioids, coumarins such as 4-methylumbelliferone and scopoletin, the flavonoid quercetin, oestrogens, bile acids and carboxylic acid-containing pharmaceuticals such as the anti-inflammatories ketoprofen and ibuprofen (Green *et*

al., 1998a; Gall et al., 1999). In general, other UGTs also metabolise these chemicals and often, more efficiently (Tukey and Strassburg, 2000; Ghosal et al., 2004a; Hazai et al., 2004). However, UGT1A3 activity towards these shared substrates may be important under certain conditions. For example, UGT1A3-mediated glucuronidation of β -oestradiol may be important in individuals that have low UGT1A1 expression (Smith *et al.*, 2005a). Furthermore, it is becoming increasing clear that UGT1A3 is an important biotransformation enzyme in its own right. UGT1A3 was recently reported to be the only enzyme catalyst of 26,26,26,27,27,27-F₆-1 α ,23S,25trihydroxyvitamin D₃ glucuronidation in the treatment of hyperparathyroidism (Kasai et al., 2005); a major contributor to the metabolism of the anti-oestrogenic drug fulvestrant (Chouinard et al., 2006); and a major UGT form responsible for the glucuronidation of two bile acids, chenodeoxycholic acid (CDCA) (Trottier et al., 2006b) and lithocholic acid (Gall et al., 1999; Verreault et al., 2006; Senekeo-Effenberger et al., 2007). In particular, it has been suggested that UGT1A3 is an important regulator of bile acids in humans, and that drugs that increase the expression of this enzyme, such as fibrates, may be effective treatments for cholestasis (Trottier et al., 2006b; Senekeo-Effenberger et al., 2007).

One clinically relevant substrate for UGT1A4 is the anti-fungal agent posaconazole (Ghosal *et al.*, 2004b). In addition, UGT1A4 is a catalyst for the N⁺-glucuronidation of trifluoperazine (an anti-psychotic) (Green and Tephly, 1996), imipramine (an anti-depressant) (Nakajima *et al.*, 2002), tamoxifen (an oestrogen receptor antagonist) (Kaku *et al.*, 2004), lamotrigine (an anti-epileptic) (Green *et al.*, 1995), nicotine and cotinine (both stimulants) (Kuehl and Murphy, 2003) and NNAL (a nicotine-derived carcinogen) (Wiener *et al.*, 2004). Some of these substrates, such as lamotrigine, tamoxifen, cotinine and nicotine, also undergo glucuronidation at functional groups

other than tertiary amines (with or without prior oxidation) or have an ammoniumlinked glucuronidation component catalysed by other UGTs such as UGT1A9, UGT2B7 or UGT2B10 (Nishiyama *et al.*, 2002; Nakajima and Yokoi, 2005; Rowland *et al.*, 2006; Kaivosaari *et al.*, 2007). Nonetheless, UGT1A4 has been postulated to be important in the elimination of each of the listed substrates.

In contrast to UGT1A3, but similarly to UGT1A4, UGT1A5 exhibits very low rates of 4-methylumbelliferone and scopoletin glucuronidation (Finel *et al.*, 2005). In addition, UGT1A5 does not glucuronidate 4-aminobiphenyl, a good substrate for the highly homologous UGT1A4 and also a substrate of UGT1A3 (Green *et al.*, 1998a; Finel *et al.*, 2005). However, 1-hydroxypyrene, another substrate common to UGT1A3 and UGT1A4 (Luukkanen *et al.*, 2005) is glucuronidated by UGT1A5 (Finel *et al.*, 2005). Thus, if individuals or circumstances exist where UGT1A5 is expressed to any significant extent in any human tissues, it is expected that this enzyme will affect the glucuronidation of its own specific subset of substrates. Whether these individuals or circumstances actually exist is still unknown.

3.1.4. Regulatory controls of UGT1A3, UGT1A4 and UGT1A5

Although they may have important ramifications for drug, carcinogen and bile acid metabolism in humans, the factors that determine the expression levels and/or tissue specificity of the *UGT1A3-1A5* cluster are currently not well understood. At the beginning of this PhD candidature, only PXR had been identified as a regulator of *UGT1A3* and *UGT1A4* (Rae *et al.*, 2001; Gardner-Stephen *et al.*, 2004), and little information regarding the molecular function of any of the *UGT1A3*, *UGT1A4* or *UGT1A5* promoters was available. The *UGT1A4* TSS was identified by Ritter and colleagues as nucleotide -44C, relative to the initiation codon, and the TAATTAA

sequence present at nucleotides -75 to -69 predicted to be a TATA box (Ritter *et al.*, 1992b). The only other published observation regarding the regulation of any of these genes was that an element with high homology to the consensus HNF1-binding site, as defined by Tronche and Yaniv (1992), was present in the *UGT1A4* proximal promoter (Tronche *et al.*, 1997). This putative HNF1-binding element had not been tested for functional significance. However, the expression pattern of the UGT1A and UGT2B enzymes overlaps considerably with that of the HNF1 homeoproteins; the liver, and intestine are both important sites of glucuronidation, and both contain significant levels of HNF1. The kidneys and other regions of the gastrointestinal tract, such as the stomach, are further examples of tissues that contain at least one of the HNF1 isoforms and also express a subset of UGTs. It therefore seemed feasible that HNF1 proteins could play an important role in regulating the expression of UGT1A3 and UGT1A4.

3.1.5. The hepatocyte nuclear factor 1 transcription factor family

3.1.5.1. Physical attributes of the HNF1 proteins

The hepatocyte nuclear factor 1 transcription factor family is comprised of two closely related proteins, HNF1 α and HNF1 β . These transcription factors are considered to be divergent members of the POU (Pit-1, Oct-1 and Oct-2, and Unc-86) subgroup of the homeodomain protein superfamily, as both factors contain regions of homology with the POU-specific A-box but not B-box, and a large homeodomain region that contains 21 extra amino acids upstream of helix III when compared with other homeodomain proteins (Herr *et al.*, 1988; Baumhueter *et al.*, 1990; Rey-Campos *et al.*, 1991). Although encoded by separate genes on separate chromosomes (Bach *et al.*, 1991), HNF1 α and HNF1 β share substantial homology. Their amino-terminal dimerisation and internal DNA-binding domains have about

75% and 93% identity, respectively, in both rat and mouse (Mendel *et al.*, 1991a; Rey-Campos *et al.*, 1991; Tronche and Yaniv, 1992). Likewise, the same domains in the human HNF1 α (Entrez Protein accession number AAI04911) and HNF1 β (AAH17714) proteins have 69% and 90% identity, respectively, as determined by alignment of their amino acid sequences with Clustal X (Thompson *et al.*, 1997). These homologous regions allow HNF1 α and HNF1 β , which both bind DNA as dimers, to heterodimerise readily. In addition, homodimers and heterodimers all recognise the same inverted dyad DNA element with the consensus sequence GTTAATNATTAAC (Tronche and Yaniv, 1992; Tronche *et al.*, 1997; Locker *et al.*, 2002). HNF1 dimers are stabilised by the formation of tetramers containing two copies of the dimerisation co-factor of HNF1 (DCoH), also known as pterin-4acarbinolamine dehydratase. HNF1 activity is especially dependent on DCoH when the homeoprotein concentration is low (Mendel *et al.*, 1991b; Rhee *et al.*, 1997).

In contrast to the highly homologous dimerisation and DNA-binding domains of HNF1 α and HNF1 β , the carboxyl-terminal activation domains are more divergent, with only approximately 47% identity between the two proteins. The activation domain of HNF1 α is also considerably larger than that of HNF1 β (Mendel *et al.*, 1991a; Rey-Campos *et al.*, 1991; Tronche and Yaniv, 1992), and because of this, these two proteins are not equal in their ability to transactivate genes containing HNF1-binding sites. However, there is significant overlap in their activity. Generally, of the two transcription factors, HNF1 β is considered to be the weaker transactivator (De Simone *et al.*, 1991; Mendel *et al.*, 1991a; Rey-Campos *et al.*, 1995; Pontoglio *et al.*, 1996; Song *et al.*, 1998; Bernard *et al.*, 1999; Kikuchi *et al.*, 2006).

Due to alternative splicing, both HNF1 α and HNF1 β are expressed as multiple isoforms with varying activities. HNF1 α has three known variants, HNF1 α -A, HNF1 α -B and HNF1 α -C, with the latter two being 5-fold more active than the HNF1 α -A isoform. However, in nearly all human tissues, and in HepG2 and Caco-2 cells, HNF1 α -A is by far the most highly expressed (Bach and Yaniv, 1993). Likewise, HNF1 β has three known variants, HNF1 β -A, HNF1 β -B and HNF1 β -C, with HNF1 β -A being the major mRNA species (Bach and Yaniv, 1993; Ringeisen *et al.*, 1993). HNF1 β -A has a higher transactivational potential than HNF1 β -B (Ringeisen *et al.*, 1993), while HNF1 β -C is a dominant negative inhibitor of HNF1 α (Bach and Yaniv, 1993).

3.1.5.2. Expression profiles of the HNF1 proteins

HNF1 α and HNF1 β have similar but distinct temporal and spatial expression profiles, with HNF1 β expression preceding HNF1 α gene activation in embryonic liver development (De Simone *et al.*, 1991; Rey-Campos *et al.*, 1991; Ryffel, 2001). Whereas both isoforms are found at comparable levels in the adult kidney, HNF1 α is the predominant form in the liver and HNF1 β is exclusively expressed in the lung. Tissues other than the kidney where both HNF1 variants are found include the intestine, stomach and pancreas (De Simone *et al.*, 1991; Mendel *et al.*, 1991a; Rey-Campos *et al.*, 1991; Pontoglio *et al.*, 1996). There is also some evidence that HNF1 α expression increases in enterocytes as they differentiate from crypt to villous tip (Hu and Perlmutter, 1999). Like HNF1, expression of DCoH is tissue restricted, with considerable overlap in the localisation of the two tetramer constituents (Mendel *et al.*, 1991b; Strandmann *et al.*, 1998).

3.1.5.3. Gene targets of the HNF1 proteins

HNF1 α has been implicated in the regulation of numerous genes in human liver, intestine, kidney and pancreas, such as glucose-6-phosphatase, albumin, cystic fibrosis transmembrane conductance regulator, α 1-antitrypsin and the insulin receptor (Boj et al., 2001; Mouchel et al., 2004; Odom et al., 2004; Senkel et al., 2005; Gautier-Stein et al., 2006; Kikuchi et al., 2006). In addition, a number of human biotransformation enzymes and transporter proteins such as CYP1A2 (Chung and Bresnick, 1997), CYP7A1 (Chen et al., 1999), the class I alcohol dehydrogenase (ADH) gene locus (Su et al., 2006), UGT1A1 (Bernard et al., 1999), UGT1A8, UGT1A9 and UGT1A10 (Gregory et al., 2004), UGT2B7 (Ishii et al., 2000), UGT2B17 (Gregory et al., 2000), the organic anion transporter 3 (OAT3) (Kikuchi et al., 2006) and MRP2 (Qadri et al., 2006) have been identified as HNF1 α -target genes. Likewise, HNF1 β has also been shown to regulate the promoters of numerous kidney, pancreatic and foetal liver genes, including glucose-6-phosphatase, OAT3, dipeptidyl peptidase 4, angiotensin converting enzyme 2 and glucose transporter 2 (glut2) (Senkel et al., 2005; Gautier-Stein et al., 2006; Haumaitre et al., 2006; Kikuchi et al., 2006). However, of the UGT promoters shown to be responsive to HNF1a, only UGT1A1 has been shown to functionally interact with HNF1B. UGT1A8, UGT2B7 and UGT2B17 have all been specifically shown not to respond to HNF1β over-expression.

In the chromosomal setting, a single HNF1 site is insufficient to drive targeted expression of a gene. Rather, each gene is regulated by a transcription factor network that is specific both to that gene and to the cell types in which it is active. Promoters that are under the control of HNF1 generally have additional binding sites nearby for other transcription factors that participate in the overall activation of transcription, and/or multiple HNF1 sites. Multiple HNF1 sites are particularly common among genes expressed in the liver (Frain et al., 1990; Song et al., 1998; Schrem et al., 2002; Costa *et al.*, 2003). Other transcription factors known to interact with HNF1 α in hepatocyte or enterocyte-derived cells include HNF4 α (Miura and Tanaka, 1993), FoxA family members (Rouet et al., 1995; Cha et al., 2000), C/EBPa (Wu et al., 1994), Oct-1 (Ishii *et al.*, 2000) and caudal-related homeodomain protein (Cdx)2 (Gregory *et al.*, 2004). Furthermore, HNF1 factors are part of a complex network of liver-enriched transcription factors (LETFs) that are interdependent on each other for expression. HNF1 α and HNF4 α reciprocally bind the promoter of the other's gene (Boj et al., 2001; Odom et al., 2004), targeted deletion of HNF1β in murine pancreas causes increased expression of HNF1a and decreased expression of HNF4a (Wang et al., 2004a), FoxA proteins have a weak, but positive effect on the rat and mouse $hnfl\alpha$ promoters (Kuo *et al.*, 1992), foxA3 is a target gene of HNF1 (Hiemisch *et al.*, 1997) and expression of HNF6 in the pancreas requires HNF1 β (Poll *et al.*, 2006). The regulatory sequences upstream of the *PXR* gene also contain an HNF1-binding site (Uno et al., 2003).

3.1.5.4. HNF1a serves a dual purpose in gene transcription

HNF1 α can influence transcription of its target genes at several levels; firstly by causing chromatin remodelling and secondly, by recruiting general transcription machinery. In the *in vivo* setting, HNF1 α increases the accessibility of promoter elements to other transcription factors and nuclear receptors through modification of the chromatin environment. One study has shown that developmental demethylation of certain genes appears to be under the influence of HNF1 α (Pontoglio *et al.*, 1997). In addition, HNF1 α is thought to induce repositioning or modification of nucleosomes through recruitment and activation of HAT proteins such as p300/CBP

and P/CAF (Pontoglio *et al.*, 1997; Rollini *et al.*, 1999; Soutoglou *et al.*, 2000b; Parrizas *et al.*, 2001; Soutoglou *et al.*, 2001). This HNF1 α -mediated hyperacetylation of histones in target genes is cell-type specific. For example, while HNF1 α interacts with the mouse *glut2* promoter chromatin template in both liver and pancreatic islet cells, only the latter are dependent on HNF1 α for hyperacetylation and transcriptional activity of the *glut2* gene (Parrizas *et al.*, 2001). Most recently, HNF1 β has also been shown be able to interact with both the p300/CBP and P/CAF HAT factors (Barbacci *et al.*, 2004; Hiesberger *et al.*, 2005).

Apart from altering chromatin higher order structure, HNF1 α and HNF1 β can, in concert with appropriate combinations of other transcription factors and coactivators, increase the rate of transcription from promoters containing HNF1 sites. This is thought to be mediated by interaction, either directly or indirectly, with components of the general transcription machinery, providing recruitment and positioning services for the pre-initiation complex (Vorachek et al., 2000; Schrem et al., 2002). This may be particularly important in promoters that lack a TATA box but still have a well defined TSS, such as the mouse *Ugt1a1* gene (Bernard *et al.*, 1999). Because this function can be observed in episomal DNA, in which nucleosomal organisation is considered relatively random compared to the highly organised nature of chromatin (Archer et al., 1992; Liu and Gonzalez, 1995; Smith and Hager, 1997; Soutoglou et al., 2000b; Akiyama and Gonzalez, 2003), it is often considered in isolation from the ability to direct histone acetylation. However, the two functions involve many of the same proteins, and in many cases are likely to be profoundly linked in vivo (Soutoglou et al., 2000b). HNF1a bound to sites in proximal promoters may direct the assembly of the pre-initiation complex by either interacting directly with components of the general transcription machinery such as TFIIB

(Ktistaki and Talianidis, 1997), or through co-activator proteins that provide a bridge between the two. In addition to their intrinsic HAT activity mentioned earlier, CBP and P/CAF have well researched roles as co-activators, linking HNF1 dimers to the transcription apparatus (Dallas *et al.*, 1997; Soutoglou *et al.*, 2000b; Schrem *et al.*, 2002; Dohda *et al.*, 2004).

3.1.6. Aims

The work presented in this chapter arose from three central aims. These were to:

- 1. Clone the proximal promoters of the human *UGT1A3-1A5* genes and investigate their relative abilities to drive reporter gene expression under basal conditions in cells of human hepatocyte and enterocyte origin;
- 2. Identify regions of the *UGT1A3* and *UGT1A4* proximal promoters important for transcriptional activity;
- 3. Test the functionality of the predicted *UGT1A4* HNF1-binding element and of the homologous regions of *UGT1A3* and *UGT1A5*.

3.2 Methods

3.2.1. Isolation of the UGT1A3-3.3k and UGT1A4-3.4k promoters

The proximal 3.3 and 3.4 kb of the *UGT1A3* and *UGT1A4* promoters, respectively, were amplified by nested PCR from *Not*I digested human genomic DNA. Briefly, 0.05 Units/µl *PfuTurbo* (see Chapter 2, section 2.2.6.3) were used to simultaneously amplify both promoters from 30 ng/µl genomic DNA using the primers 1A3/4prom-3.5k and 1A3/4rev-common (Table 3.1). Cycling conditions were: initial denaturation for 4 minutes at 95°C, 35 cycles of 95°C, 30 seconds; 50°C, 30 seconds; 72°C, 10 minutes, followed by a final extension at 72°C for 5 minutes. One tenth of

Oligonucleotide	Nucleotide Sequence $(5, \rightarrow 3)$	Nucleotide Position on Target Gene	RE
1A3/4prom-3.5k	GTTATCATTAAATAATAATCCT	UGTIA3: -3350 to -3329; UGTIA4: -3438 to -3417	NA
1A3/4rev-common	GGCAGGGGAACCTGGAGTCCT	UGT1A3: +29 to +9; UGT1A4: +29 to +9	NA
1A3prom-3.3kHindIII	AGCCAT <u>AAGCTT</u> ATTGGATACCAGTATTGCT	UGTIA3: -3315 to -3297	HindIII
1A3UTRHindIII	AGCCAT <u>AAGCTT</u> CTCAGCAGAAGACACGGACA	<i>UGT1A3</i> : -1 to -20	HindIII
1A4prom-3.4kNheI	AGCCAT <u>GCTAGC</u> ATCTGATATCAGTAATGTG	UGT1A4: -3403 to -3385	NheI
1A4UTRXhoI	AGCCAT <u>CTCGAG</u> CTCAGCAGGAAGCCACCGACA	<i>UGT1A4</i> : -1 to -20; <i>UGT1A5</i> : -1 to -20	XhoI
1A3UTRXho1	AGCCAT <u>CTCGAG</u> CTCAGCAGAAGACACGGACA	<i>UGT1A3</i> : -1 to -20	XhoI
1A3/4prom-0.5k	AGCCAT <u>GCTAGC</u> AAGCTGGCTTAGCAATGTTGT	<i>UGT1A3</i> : -507 to -487; <i>UGT1A4</i> : -506 to -486; <i>UGT1A5</i> : -508 to -488	Nhel
1A3/4prom-1.5k	AGCCAT <u>GCTAGC</u> TAAGGGGGGTTGGAGGAATAGT	<i>UGT1A3</i> : -1539 to -1519; <i>UGT1A4</i> : -1574 to -1554; <i>UGT1A5</i> : -1550 to -1530	Nhel
1A3/4prom-2.5k	AGCCAT <u>GCTAGC</u> AAGAGCCCTTGTGATTACA	UGTIA3: -2541 to -2523; UGTIA4: -2610 to -2590	Nhel
1A3prom-130bp	AGCCAT <u>GCTAGC</u> AGGGCACTCTGTCTTCCA	<i>UGTIA3</i> : -130 to -113	Nhel
1A3prom-150bp	AGCCAT <u>GCTAGC</u> GGTTAATAATTAACTAGAGGA	<i>UGT1A3</i> : -150 to -130	Nhel
1A3prom-165bp	AGCCAT <u>GCTAGC</u> CAGTGATAGATTAATGGT	<i>UGT1A3</i> : -165 to -148	Nhel

l adle 3.1 continued.			
Oligonucleotide	Nucleotide Sequence $(5, \rightarrow 3)$	Nucleotide Position on Target Gene	RE
1A3prom-180bp	AGCCATGCTAGCCCACAGTGAAAAACAGTG	UGT1A3: -180 to -161	Nhel
1A3prom-200bp	AGCCATGCTAGCTCGGTCTTTTCCAGGGT	UGT1A3: -200 to -184	NheI
1A3prom-150 HNF1 mut	AGCCATGCCGGCCAACGCTTCACTAGAGGA	UGT1A3: -150 to -130	Nhel
1A3prom-150 HNF1Δ4	AGCCAT <u>GCTAGC</u> GGGTAATAAGTAACTAGAGGA	UGT1A3: -150 to -130	NheI
1A3prom-165s1 mut 4	AGCCAT <u>GCTAGC</u> CAGTTATAGATTAATGGT	<i>UGTIA3</i> : -165 to -148	NheI
1A3prom-165s1 mut CGC	AGCCAT <u>GCTAGC</u> CAGTGATAGACGCATGGT	<i>UGTIA3</i> : -165 to -148	NheI
1A4prom-130bp	AGCCAT <u>GCTAGC</u> AGGGCACTTTGTCTTCCA	UGT1A4: -130 to -113	Nhel
1A4prom-150bp	AGCCAT <u>GCTAGC</u> GGGTAATAAGTAACTGGAGGA	UGT1A4: -150 to -130; UGT1A5: -150 to -129	Nhel
1A4prom-165bp	AGCCAT <u>GCTAGC</u> CAGTTATAGATTAATGGGT	<i>UGT1A4</i> : -165 to -147	NheI
1A4prom-200bp	AGCCATGCTAGCTTGGTCTTTCCCAGGGT	UGT1A4: -200 to -184	Nhel
1A5prom-2087bp	GAGGTCTTTAGACCACTTAGTC	UGT1A5: -2087 to -2066	NA
1 A 5 rev	GCTCCACAAGACCTATGTATGAT	<i>UGT1A5</i> : +391 to +367	NA
D.F. motini ation and anno loose di	to a undadiand NIA. not andiachle Mutations and highlichte	and the second	a man a strike a strike

RE: restriction endonuclease site, as underlined. NA: not applicable. Mutations are highlighted in bold. Numbering is relative to the initiation codon of each respective gene.

101

the resulting PCR product was then used as template to specifically amplify each promoter in separate reactions. The primers used in the second round of amplification were 1A3prom-3.3kHindIII and 1A3UTRHindIII for the *UGT1A3* promoter, or 1A4prom-3.4kNheI and 1A4UTRXhoI for *UGT1A4* (Table 3.1). The amplification conditions were as for the first round PCR, but with 12 minutes extension during the cycling stage. Both promoters were cloned into pGL3-basic utilising the *Hind*III (*UGT1A3*) or *NheI/XhoI* (*UGT1A4*) restriction sites engineered into the second primer sets. The ends of each clone were sequenced to confirm their identity.

3.2.2. Generation of *UGT1A3* and *UGT1A4* promoter deletion constructs and mutants

The pGL3-1A3-3.3k and pGL3-1A4-3.4k vectors were used as templates to clone the required deletion fragments of each promoter. All PCRs were performed with *PfuTurbo* as described in Chapter 2, section 2.2.6.3. The amplification reaction parameters for all fragments of 500 nucleotides or greater were: 95°C for 4 minutes, followed by 30 cycles of 95°C for 30 seconds, 50°C for 30 seconds and 72°C for 5 minutes, and completed with a single step of 72°C for 5 minutes. These conditions were also used for generating shorter promoter fragments (\leq 200 bp), but with a briefer extension step of one minute. All resulting PCR products were cloned into the *Nhe*I and *Xho*I sites of pGL3-basic and sequenced in full.

The antisense primer sequences for all *UGT1A3* and *UGT1A4* fragments were 1A3UTRXho1 and 1A4UTRXhoI, respectively. Both contain an *Xho*I site, as marked by the underscored text (Table 3.1), while each of the sense primers contains a *Nhe*I site. The sense primers 1A3/4prom-2.5k, 1A3/4prom-1.5k and 1A3/4prom-0.5k annealed to nucleotides -2541 to -2523, -1539 to -1519, and -507 to -487 of the

UGT1A3 promoter, respectively, relative to the translation start site (see Table 3.1). The three *UGT1A4* promoter sub-fragments amplified with the same sense primers had lengths of 2610, 1574 and 506 nucleotides, respectively.

For the shorter *UGT1A3* and *UGT1A4* promoter fragments, the sense primers used annealed to nucleotides -200 to -184, -150 to -130 or -130 to -113 of their respective templates, -165 to -148 of the *UGT1A3* promoter or -165 to -147 of the *UGT1A4* promoter. A construct containing the proximal 180 nucleotides of the *UGT1A3* promoter was also generated using the primer 1A3prom-180bp. To produce the mutated *UGT1A3*-150bp and *UGT1A3*-165bp constructs, the *UGT1A3* promoter was re-amplified with primers containing the desired mutations (Table 3.1).

3.2.3. Isolation of the UGT1A5 promoter

The proximal 1.5 kb of the *UGT1A5* promoter was also amplified using two sequential *PfuTurbo* reactions. The first round of PCR was performed on BAC 1308M2 from the human library RPCI-11 (BACPAC Resources) DNA using primers 1A5prom-2087bp and 1A5rev with the following cycling conditions: 95°C for 4 minutes, followed by 30 cycles of 95°C, 30 seconds; 65°C, 30 seconds; 72°C, 4 minutes and a final extension step of 72°C for 5 minutes. One fiftieth of the resulting products were used as template to amplify the *UGT1A5*-1550bp promoter using the same primers and conditions as for *UGT1A4*-1574bp. The 508 bp and 150 bp fragments of the *UGT1A5* promoter were also amplified using the corresponding *UGT1A4* primers defined above. All size fragments were ligated into the *Xho*I and *Nhe*I sites of pGL3-basic and sequenced in full.

3.2.4. HNF1α and HNF1β expression vectors

The cloning or acquisition of each HNF1 expression vector used in this chapter is detailed in Chapter 2, sections 2.1.4 and 2.3.1. The cDNAs encoded by the pCMX-HNF1 α and pCMX-HNF1 β expression vectors are human in origin and express the A-variants of each transcription factor. The proteins encoded by the pBJ5-HNF1 α and pBJ5-HNF1 β expression vectors are of murine origin and have 94% and 96% identity with their human counterparts at the amino acid level respectively.

3.2.5. Transient transfection and luciferase reporter assay

HepG2, Caco-2 and HEK293T cells were seeded into 24-well plates and transfected as described in Chapter 2, section 2.2.10. Each well was transfected with either 0.5 μ g of empty pGL3-basic or a reporter vector carrying the indicated *UGT1A3*, *UGT1A4* or *UGT1A5* promoter sequences. For induction studies, 0.25 μ g HNF1 α or HNF1 β expression vectors or empty pCMX vector were co-transfected with the pGL3 reporter constructs. pRL-null (0.025 μ g) was added to all transfections as an internal control for transfection efficiency. After 48 hours, cells were lysed in passive lysis buffer and analysed for firefly and *renilla* luciferase activity using the Dual-Luciferase Reporter Assay System as detailed in Chapter 2, section 2.2.11.

3.2.6. Electrophoretic mobility-shift assay

HepG2 and Caco-2 nuclear extracts, prepared as described in Chapter 2, section 2.2.12 was used to perform EMSA and super-shift assays as detailed in Chapter 2, sections 2.2.13, 2.2.14 and 2.2.15. The sense sequences of the double-stranded DNA probes used are listed in Table 3.2. The anti-HNF1 α antibody used in the super-shift assays was sourced from Santa Cruz Biotechnologies (sc-6547).

Oligonucleotide	Nucleotide Sequence (5'→3')
UGT1A3-HNF1	ATTAATG <u>GTTAATAATTAAC</u> TAGAGG
UGT1A4/UGT1A5-HNF1	ATTAATG <u>GGTAATAAGTAAC</u> TGGTGG
UGT1A3-HNF1mut	ATTAATG <u>GCCAACGCTTCAC</u> TAGAGG
HNF1-consensus	TCAG <u>GTTAATCATTAAC</u> GATCT
FXR-consensus	GATCTCAAGAGGTCATTGACCTTTTTG

 Table 3.2: Oligonucleotides used for EMSA and super-shift experiments.

Underlined text indicates the extent of the putative HNF1-binding sites in each probe and deliberate mutations are highlighted in bold. NB: Only the sense strand of each oligonucleotide pair is shown.

3.2.7. Statistical analysis

Statistical treatment of all reporter-promoter assay data was performed as described in Chapter 2, section 2.2.17 using independent-samples *t*-tests.

3.3 Results and discussion

3.3.1. Basal activities of the UGT1A3, UGT1A4 and UGT1A5 proximal promoters

In both liver (HepG2) and colon (Caco-2) derived cell lines, the *UGT1A3* and *UGT1A4*-130bp promoters had minimal activity, exhibiting less than 3-fold increases over basal reporter gene expression by the promoter-less pGL3-basic vector. However, inclusion of a further 20 nucleotides of either promoter substantially increased luciferase expression in both cell types (P < 0.001) (Figure 3.1A and B). Further increases in promoter activity could be obtained in either cell line by inclusion of up to 500 base pairs of the *UGT1A3* or *UGT1A4* promoters ($P \le 0.017$), with the exception of the *UGT1A3* promoter in Caco-2 cells, which showed greatest activity at a length of 200 bp (P = 0.001) (Figure 3.1B). The largest increases obtained in promoter activity for *UGT1A3* were 53-fold in HepG2 and 40-fold in Caco-2 cells; while the *UGT1A4* promoter had maximal activities of 21 and 15 times



Figure 3.1: Successive deletion constructs of the *UGT1A3* and *UGT1A4* promoters reveal positive regulatory elements necessary for basal activity. A. HepG2 or B. Caco-2 cells were transfected in triplicate with 0.5 µg of pGL3 reporter vectors carrying the indicated lengths of the *UGT1A3*, *UGT1A4* or *UGT1A5* promoters and 25 ng of the promoter-less control vector pRL-Null. Forty eight hours post-transfection the cells were lysed and assayed for firefly and *renilla* luciferase reporter gene activities as described in "Methods". Representative results of at least two independent experiments are presented as the mean firefly:*renilla* luciferase ratio relative to pGL3-basic (set arbitrarily to 1) plus one standard deviation. *P* values for the indicated comparisons are **P* < 0.001 and †*P* = 0.001.

that of pGL3-basic in the same cell lines (Figure 3.1A and B). Increasing promoter

length from 500 to 1500 nucleotides resulted in reduced promoter activity for both

UGT genes ($P \le 0.01$), although this phenomenon was more marked in HepG2 cells than Caco-2. It was also found that, for all promoter lengths ≥ 200 bp, the *UGT1A3* gene had greater activity *in vitro* than *UGT1A4* ($P \le 0.005$) regardless of host cell type. In Caco-2 cells, the *UGT1A4*-150bp promoter also had less activity than the *UGT1A3*-150bp fragment (P < 0.001) (Figure 3.1B); but in HepG2 cells, the difference in the activity between these two constructs was statistically insignificant (P = 0.069) (Figure 3.1A).

Pair-wise comparison of the UGT1A5 promoter with the regulatory regions of UGT1A3 and UGT1A4 revealed that the former had the least activity for all three promoter lengths tested in HepG2 ($P \le 0.005$) (Figure 3.1A), and for all but the UGT1A4 and UGT1A5-150 nucleotide promoter pair in Caco-2 cells, where equivalent activities were observed (P = 0.144) (Figure 3.1B). The UGT1A5 promoter also differed from UGT1A3 and UGT1A4 in that the shortest fragment tested, 150 bp, was the most active. Increasing the promoter length to 500 nucleotides decreased promoter function in both HepG2 and Caco-2 cells (P = 0.001HepG2, and P = 0.026 Caco-2), a result in direct opposition to that obtained for UGT1A3 and UGT1A4 (Figure 3.1A and B). UGT1A5 expression has not been detected at a substantial level in any human tissue to date, and it has been suggested that this may be due to lack of a functional promoter (Tukey and Strassburg, 2001). These results support this hypothesis, and suggest that although the UGT1A5 core promoter is sufficient for assembly of a pre-initiation complex, one or more crucial regulatory elements between -150 and -500 bp are missing, and/or that the UGT1A5 promoter contains negative regulatory sequences not present in UGT1A3 and UGT1A4.

3.3.2. HNF1 is required for basal activity of the UGT1A3 and UGT1A4 proximal promoters

For both *UGT1A3* and *UGT1A4*, it was found that nucleotides -130 to -150 were important for basal activity in HepG2 and Caco-2 cells. This region of the *UGT1A4* promoter has previously been predicted to contain an HNF1-binding site (Tronche *et al.*, 1997), based on the high identity (10 of 12 nucleotides) of this region with the HNF1-binding site consensus sequence. Furthermore, the equivalent region of the *UGT1A3* promoter contains a 100% match to the HNF1-binding site consensus, while the *UGT1A5* promoter is identical to *UGT1A4* over these nucleotides (Figure 3.2).

Before this present study, however, no experimental evidence had been presented to ascertain whether these putative HNF1-binding sites are functional in the context of their promoters. To investigate the influence of HNF1 α on the activity of the *UGT1A3* and *UGT1A4* promoters, constructs containing 500 bp or less of each regulatory region were co-transfected with HNF1 α into cells known to express HNF1 factors (HepG2 and Caco-2 cells) or a cell line devoid of HNF1 factors; namely HEK293T (Bernard *et al.*, 1999; Gardner-Stephen and Mackenzie, 2005). In HEK293T cells, it was found that *UGT1A3* and *UGT1A4* promoters of sufficient length to include the putative HNF1-binding site were highly responsive to heterologous expression of HNF1 α (P < 0.001). Reporter gene expression under the control of the *UGT1A3* or *UGT1A4* promoters could be increased up to 22-fold over the basal expression from pGL3 (Figure 3.3A). In contrast, vectors containing only the most proximal 130 bp of the *UGT1A3* or *UGT1A4* promoters were completely unresponsive to the presence of HNF1 α .

	-190	-179	-165	-149 -137
	Site 2	_	Site 1	HNF1
1A3	TTCCAGGGTGGG	GCCCACAGTGAAAA	A <mark>CAGTGATAGATTAA</mark> T	GGTTAATAATTAACTAGA
1A5	TCCCAGGGTGGG	GCCCATAATGCAAG	ACAATTATAGATTAAC	GGTAATAAGTAAC <mark>T</mark> GGA
1A4	TCCCAGGGTTGG	GCCCATAACGAAAG	GCAGTTATAGATTAAT	'GGGTAATAAGTAACTGGA
	* ****** **	**** * * * *	** * *******	* * * * * * * * * * * * * * * * * * * *
				-75
1A3	GGAGGGCACTCI	GTCTTCCAATTACA	CGTTGATTTGCTAAGT	GGCTCAGTGACAAGG TAA
1A5	GGAGGGCACTCI	GTCTTC-AATTACA	TGTTGATTTGCTAGGT	GTCTCAGTGACAAGGTAA
1A4	GGAGGGCACTTI	GTCTTCCAATTACA	TGCTGATTTGCTAGGT	GGCTCAATGACAAGGTAA
	********	*****	* ********	* **** ******
	-69		-44	
	TATA		TSS	
1A3	TTAA GATGAAGA	AAGCAAATGTAGCA	GGCACAGCGTGGGGTG	GACAGTCAGCTGTCCGTG
1A5	TTAAGACGAAGG	GAAACAATTCTAGGA	GGCACAACGTGGGGTG	GACAGTCAGCTGTCGGTG
1A4	TTAAGGCGAAGG	AAACAAATGTAGCA	GG <mark>C</mark> ACAGCGTGGGGTG	GACAGTCAGCTGTCGGTG
	**** ****	** *** * *** *	***** *****	****
		+1		
1A3	TCTTCTGCTGAG	ATG		
1A5	GCTTCTGCTGAG	ATG		
1A4	GCTTCTGCTGAG	ATG		
	********	***		

Figure 3.2: The human UGT1A3, UGT1A4 and UGT1A5 promoters contain putative HNF1-binding sites. The UGT1A3, UGT1A4 and UGT1A5 200-nucleotide proximal promoter regions are aligned, with the conserved putative HNF1-binding sites boxed in red. The initiation codons of each gene are labeled with bold italic text, the UGT1A4 transcription start site (TSS), as defined by Ritter and colleagues is marked in magenta (Ritter *et al.*, 1992b). It has been predicted that the corresponding nucleotides of UGT1A3 and UGT1A5 are also their respective TSSs, based on sequence conservation. The sequence generally accepted to be the TATA box for these genes is bolded and underlined in green. Two UGT1A3 promoter regions found to contain positive transcriptional elements in work described in this chapter are boxed in blue. Asterisks indicate identity between all three promoters and the numbering is relative to the initiation codon of UGT1A3.

In HepG2 or Caco-2 cells, co-transfections of the UGT1A3 or UGT1A4 promoters

with HNF1a resulted in no additional response, or only minor increases in luciferase

expression respectively (Figure 3.3B and C). Since HepG2 and Caco-2 cells express

HNF1 factors (Kuo et al., 1990; Rey-Campos et al., 1991), it seemed likely that the

endogenous levels of HNF1a and/or HNF1B in these cells were sufficient to support

expression of the reporter gene from the UGT1A3 and UGT1A4 promoters in vitro.

Therefore, the putative HNF1-binding site in the UGT1A3-150bp promoter was

mutated to abolish any binding of HNF1α. The functional result of this mutation was

Figure 3.3: The HNF1-binding site is required for maximal basal activity of the *UGT1A3* and *UGT1A4* proximal promoters. The HNF1-binding site of the *UGT1A3*-150bp promoter construct was mutated as described in section 3.2.2. A. HEK293T, B. HepG2 or C. Caco-2 cells were co-transfected with 0.5 μ g pGL3-based vectors containing 500, 200, 150 or 130 nucleotides of the *UGT1A3* or *UGT1A4* promoters or 500 or 150 nucleotides of the *UGT1A5* promoter, 25 ng pRL-Null and 0.25 μ g of pCMX-HNF1 α expression vector. The DNA concentration in control transfections was kept constant by addition of empty pCMX vector as necessary. The results of all experiments are the means of triplicate samples, expressed as a relative value of firefly luciferase activity to the internal *renilla* control, compared to the pGL3-basic control (set to 1). The error bars indicate one standard deviation. ND: Not done. *P* values for the indicated comparisons are **P* < 0.001 and #*P* > 0.05 (not significant).



a loss of basal activity of the *UGT1A3*-150 bp promoter in both HepG2 and Caco-2 cells (P < 0.001), and prevention of HNF1 α -responsiveness in HEK293T cells (P = 0.123). In all cases, the mutated *UGT1A3*-150bp promoter construct behaved in the same manner as the *UGT1A3*-130bp promoter that contains no recognised HNF1 α -binding site (Figure 3.3). In support of the above evidence that the *UGT1A3* and *UGT1A4* promoter HNF1 sites are functional, binding of HNF1 factors from HepG2 and Caco-2 nuclear extracts to these sequences could be demonstrated by EMSA (Figures 3.4 and 3.5). Furthermore, the mutation used to abolish HNF1 α -responsiveness of the *UGT1A3*-150bp promoter also prevented binding of HNF1 factors to this region *in vitro* (Figures 3.4 and 3.5), while neither the mutated nor unrelated (FXR) probes could interfere with HNF1 binding when added in 500-fold excess (Figure 3.4). Subsequently, Caillier *et al.* (2007) also confirmed that the putative *UGT1A3* HNF1-binding site interacts with HNF1 α *in vitro* and is important for *UGT1A3* promoter function.

The *UGT1A5* promoter contains the same putative HNF1-binding site as *UGT1A4*. Therefore, this promoter was also tested for responsiveness to over-expressed HNF1 α . The activity of the *UGT1A5*-500bp promoter was increased by HNF1 α over-expression in all three cell lines tested (P < 0.001), with the greatest response being observed in HEK293T cells (15.8-fold increase relative to pGL3-basic). Interestingly, even though the basal reporter gene expression from the pGL3-1A5-500 construct was only one quarter that of the equivalent *UGT1A4* vector in HepG2 cells, the overall activity of the *UGT1A5*-500bp promoter in the presence of excess HNF1 α was 81% of that achieved for *UGT1A4* (Figure 3.3). The same phenomenon was observed in Caco-2 cells, although the initial contrast between the basal expression levels was not as severe.



Electrophoretic mobility-shift assays were performed using 50,000 cpm of ³²P end-labelled oligonucleotide probes encompassing the putative Figure 3.4: HNF1a from HepG2 nuclear extracts binds to nucleotides -156 to -128 of the UGT1A3 and UGT1A4/UGT1A5 promoters. UGTIA3 or UGTIA4/UGTIA5 HNF1 sites. A. HNF1 complexes were super-shifted with 2 µg of HNF1α-specific antibody, or B. competed with between 10 and 500-fold excess cold probe. The positions of free probe and complexes containing HNF1 α are indicated by parentheses. HNF1: consensus HNF1-binding sequence probe; WT: UGTIA3 or UGTIA4/IA5 wild-type probe; Mut: UGTIA3 mutant probe; FXR: consensus FXR-binding sequence probe.



Figure 3.5: HNF1a from Caco-2 nuclear extracts binds to nucleotides -156 to -128 of the UGT1A3 and UGT1A4/UGT1A5 promoters. Electrophoretic mobility-shift assays were performed using 50,000 cpm of ³²P end-labelled oligonucleotide probes encompassing the putative UGT1A3 or UGT1A4/UGT1A5 HNF1 sites. HNF1 complexes were super-shifted with 2 μ g of HNF1a-specific antibody. The positions of free probe and complexes containing HNF1a are indicated by parentheses. WT: UGT1A3 or UGT1A4/IA5 wild-type probe; Mut: UGT1A3 mutant probe.

The pGL3-1A5-150 reporter vector was also found to be responsive to HNF1a in HepG2 and HEK293T cells (P < 0.001), suggesting that it is the same HNF1-binding element driving the UGT1A5 response as for UGT1A3 and UGT1A4 (Figure 3.3). The presence of a homologous, functional HNF1-binding site in the UGT1A5 promoter indicates that, although the identified HNF1-binding elements are necessary for much of the UGT1A3 and UGT1A4 promoter activities in vitro, they are also insufficient to sustain maximal promoter activity. Since the UGT1A4 and UGT1A5 promoters possess identical HNF1-binding elements and TATA boxes, yet differ in their basal activity, it appears that one or more transcription factors other than HNF1 are also required to promote initiation of transcription. Given that UGT1A5 is not substantially expressed in liver while UGT1A4 is, this observation is consistent with the currently available information regarding transcriptional control of hepatic genes (see section 3.1.5.3). Interestingly, excess HNF1 α overcomes much of the difference between the activity of the UGT1A4 and UGT1A5 promoters in all cell types tested, suggesting that this second factor may normally play a role in recruiting or activating HNF1 factors on the UGT1A4 promoter and/or be under the expressional control of HNF1a.

3.3.3. HNF1β can transactivate the UGT1A3 promoter in vitro

Since HNF1 β can recognise the same nucleotide elements as HNF1 α , but regulates an overlapping, yet separate set of genes, it was investigated whether this factor could also transactivate the *UGT1A3* promoter. HNF1 β activates the *UGT1A1* promoter to the same extent as HNF1 α (Bernard *et al.*, 1999), but does not activate *UGT1A8*, *UGT2B7* or *UGT2B17*, all known to be HNF1 α target genes (Gregory *et al.*, 2000; Ishii *et al.*, 2000; Gregory *et al.*, 2004). Since the published *UGT1A8*, *UGT2B7* and *UGT2B17* work, which was performed in our laboratory, was done with pBJ5 vectors expressing the mouse HNF1 homologues, these vectors were also used in this experiment so that the results would be directly comparable.

Murine HNF1 α increased the expression of luciferase from the *UGT1A3-500bp* construct in HEK293T cells to 14-fold over the basal activity of the pGL3-basic control (P < 0.001) (Figure 3.6), a similar result as that obtained with human pCMX-HNF1 α expression vector (Figure 3.3). Likewise, HNF1 β activated the *UGT1A3-500bp* promoter in HEK293T cells (P < 0.001); however, the 9-fold increase in reporter gene expression produced by HNF1 β over-expression was significantly less than that for HNF1 α (P < 0.001) (Figure 3.6). These results are consistent with the observation that HNF1 β is frequently a weaker transactivator than HNF1 α when these proteins activate a common target (see section 3.1.5.1). Interestingly, the regulation of the *UGT1A3* promoter by HNF1 factors is more similar to the *UGT1A1* gene than *UGT1A8*, *UGT2B7* or *UGT2B17* in that HNF1 α and HNF1 β can both function as positive transcriptional regulators for the *UGT1A3* promoter and their respective HNF1 sites are essential for transcriptional activity *in vitro*.

3.3.4. The UGT1A3 and UGT1A4 proximal promoters differ in their HNF1 responses

During the course of this study, two notable differences between the *UGT1A3* and *UGT1A4* HNF1 responses were observed. Firstly, whereas none of the *UGT1A3* promoter constructs exhibited any activity in HEK293T cells in the absence of HNF1 α , *UGT1A4* promoters of 150 bp or longer could support a small degree of basal activity (*P* < 0.001). This activity, which was 2 to 3-fold greater than the empty vector control, is presumably HNF1-independent and was also observed for the *UGT1A5* promoter (*P* < 0.001) (Figure 3.3A). The second observation was that,



Figure 3.6: Activation of the *UGT1A3*-500bp promoter by HNF1 β in HEK293T cells. HEK293T cells were transfected with 0.5 µg pGL3-basic or pGL3-1A3-500, and 0.25 µg pCMX, pBJ5-HNF1 α or pBJ5-HNF1 β as described in section 3.2.5. In addition, each transfection contained 25 ng of pRL-Null to serve as an internal control. The means of triplicate samples are expressed as a relative value of firefly luciferase activity to the internal *renilla* control, compared to the pGL3-basic control (set to 1). The error bars indicate one standard deviation. *P* value for the indicated comparison is **P* < 0.001.

whereas UGT1A3 promoter activity could not be increased in HepG2 cells by overexpression of HNF1a ($P \ge 0.157$), UGT1A4 promoter activity was increased up to 2.3-fold by excess HNF1a for promoter fragments ≥ 150 bp ($P \le 0.005$) (Figure 3.3B). One possible explanation is that the perfect UGT1A3 HNF1-binding element is fully occupied at physiological HNF1 concentrations, whereas the slightly flawed site of the UGT1A4 promoter is less efficient at competing with the multitude of genomic sites for limited HNF1. Therefore, addition of excess HNF1a into the system can only increase the occupancy rate of the UGT1A4 HNF1-binding site. There is also likely a cell type-specific component to this second difference between the promoters, as it was only observed in cells of hepatic origin (Figure 3.3B).

To test whether the sequence differences between the UGT1A3 and UGT1A4 HNF1binding elements were responsible for their different HNF1 α -responses in HepG2 cells, the HNF1-binding site of the UGT1A3 promoter was mutated to the corresponding UGT1A4 sequence. When the mutated UGT1A3-150bp HNF1 Δ 4 construct was tested, it was found that it had 18% less basal activity than the UGT1A3 wild-type promoter (P = 0.032). However, in the presence of HNF1 α overexpression it had the same activity as the wild-type UGT1A3-150bp promoter (P =0.861), which is significantly less than that of the UGT1A4-150bp promoter under the same conditions (P = 0.029) (Figure 3.7). Thus, reducing the UGT1A3 HNF1binding site's similarity to the consensus sequence adversely affected the basal activity of the promoter, yet this loss of activity could be compensated for by a high HNF1 α concentration. These results support the hypothesis that the UGT1A4 HNF1binding site is less efficient than the corresponding UGT1A3 sequence at competing for limited HNF1 factors, and that a high HNF1 α concentration increases promoter activity by increasing occupancy of the HNF1-binding element. However, the introduction of the UGT1A4 HNF1-binding site into the UGT1A3-150bp promoter did not cause its behaviour to become completely UGT1A4-like: the reporter activity emanating from the HNF1 α -induced UGT1A3-150bp HNF1 Δ 4 construct was still significantly less than the similarly treated UGT1A4-150bp promoter. One feasible explanation for this is that HNF1 factors contribute less to the basal activity of the UGT1A4-150bp promoter than the UGT1A3-150bp promoter in HepG2 cells, but that a second stimulatory transcription factor bound downstream of the HNF1-binding site, on the UGT1A4 promoter only, causes the two promoters to have the same activity in these cells (P = 0.069) (Figure 3.1A). When the HNF1 concentration is increased, HNF1 binding to the UGT1A4 promoter increases, which in combination with the putative second factor, allows the UGT1A4 promoter activity to exceed that of UGT1A3. If this second putative factor is expressed in HepG2 cells but not Caco-2

cells, this hypothesis could also help explain why the difference between the basal activities of the *UGT1A3*-150bp and *UGT1A4*-150bp promoter constructs is much greater in the latter (Figures 3.1 and 3.3), and the cell-type specificity of the *UGT1A4* HNF1 α -response noted earlier (Figure 3.3B).



Figure 3.7: Mutation of the *UGT1A3* HNF1-binding site to the equivalent *UGT1A4* sequence does not confer HNF1-responsiveness in HepG2 cells. The HNF1-binding site of the *UGT1A3* promoter was mutated to the corresponding *UGT1A4* sequence and tested for HNF1 α -responsiveness in HepG2 cells. 0.5 µg of each reporter vector was co-transfected with 0.25 µg pCMX or pCMX-HNF1 α and 25 ng of pRL-Null. Results are expressed as the mean (n = 3) relative value of firefly luciferase activity to the internal *renilla* control, compared to pGL3-basic (set arbitrarily to 1). The error bars indicate one standard deviation. *P* values for the indicated comparisons are $\ddagger P = 0.029, \ddagger P = 0.032$ and #P > 0.05 (not significant).

Therefore, it was concluded that the nucleotide differences of the *UGT1A3* and *UGT1A4* HNF1-binding site sequences are functionally significant, at least under the conditions tested *in vitro*. However, the differing behaviour of these promoters in the presence of excess HNF1 α is also likely to be a consequence of further nucleotide discrepancies downstream of their HNF1-binding sites, and is more complex than

originally hypothesised. Further experiments will be required to elucidate the differences between the *UGT1A3* and *UGT1A4* promoters that cause their divergent behaviour in liver-derived cells.

When pGL3-1A3-165 and pGL3-1A3-180 reporter constructs were constructed and tested in HepG2 cells, two separate elements were found to be important for maximal basal activity. The proximal 165 bases of the *UGT1A3* promoter were more active than the *UGT1A3*-150bp promoter (P = 0.002), revealing the first positive element (site 1); and the *UGT1A3*-180bp promoter had the same activity as the pGL3-1A3-165 construct (P = 0.802), which is less than that for *UGT1A3*-200bp (P = 0.001), indicating that a second positive element (site 2) either resides between nucleotides -180 to -200 or overlaps the -180bp junction point (Figure 3.8).

To search the newly defined, functionally important regions of the *UGT1A3* promoter for known transcription factor binding sites, publicly available MatInspector v6.0 software (Cartharius *et al.*, 2005; http://www.genomatrix.de) was used. The results indicated that, according to current knowledge, the guanine-rich region between nucleotides -187 and -180 constitutes the only likely transcription factor binding site that could account for the difference in activity between the *UGT1A3*-180bp and *UGT1A3*-200bp promoters. Candidate binding proteins were identified as Sp1/Sp3, Kruppel-like factors and MYC-associated zinc finger protein related transcription factor. No attempts have yet been made to confirm the importance of this guanine tract for *UGT1A3* activity, or to identify the transcription factors that bind to site 2.



Figure 3.8: The UGT1A3-200 to -150bp region contains at least two regulatory elements that can positively influence transcription in HepG2 cells. Firefly luciferase reporter constructs carrying the proximal 165 or 180 nucleotides of the UGT1A3 promoter were constructed and transfected into HepG2 cells as described in "Methods" to assess their basal activity. The means of triplicate samples are expressed as a relative value of firefly luciferase activity to the internal *renilla* control, compared to the pGL3-basic control (set to 1). The error bars represent one standard deviation. *P* values for the indicated comparisons are $\dagger P = 0.001$, $\dagger \dagger P = 0.002$ and #P > 0.05 (not significant).

Similar analysis of the UGT1A3 promoter nucleotides -165 to -150 indicated a number of transcription factors that may bind site 1, the majority of which rely on the AT-rich region between nucleotides -156 and -151 for their high "core similarity" scores. To determine whether any of these putative transcription factors may be important contributors to UGT1A3 promoter activity, the AT-rich region was interrupted by mutating nucleotides -155 to -153 to the sequence 'CGC' (pGL3-1A3-165 s1mut CGC). However, when tested, the mutated promoter construct had greater activity than the wild-type sequence (P < 0.001) (Figure 3.9A). Since a decrease in activity was anticipated if these nucleotides formed the core of a biologically



Figure 3.9: Characterisation of the UGT1A3 promoter site 1 regulatory element. Reporter constructs containing A. mutated UGT1A3-165bp promoters or B. the wild-type UGT1A4-165bp promoter were constructed as described in section 3.2.2 and compared to existing constructs. HepG2 cells were transfected in triplicate with 0.5 µg of the indicated pGL3-derived reporter vectors and 25 ng of the promoter-less control vector pRL-Null. Forty eight hours post-transfection the cells were lysed and assayed for firefly and *renilla* luciferase reporter gene activities as described in "Methods". Results are presented as the mean firefly:*renilla* luciferase ratio, relative to pGL3-basic (set to 1) plus one standard deviation. The experiment presented in panel A was only performed once. *P < 0.001 and #P > 0.05 (not significant).

relevant transcription factor binding site, it was deemed unlikely that any of the factors predicted to bind the AT-rich region were contributing to UGT1A3 activity. The only remaining transcription factor element predicted in this region with a core similarity of greater than 0.870 was a binding site for Pbx/Hox heterodimers. This site had a core similarity of 1.000 for nucleotide positions -162 to -159 against the Pbx/Hox matrix. This putative element is not conserved in the UGT1A4 promoter, as it encompasses the only nucleotide between positions -165 and -150 that is not identical between the UGT1A3 and UGT1A4 promoters (nucleotide -161). Since the UGT1A4 promoter does not change significantly in activity when its length is extended from 150 to 165 base pairs (P = 0.063), in contrast to UGT1A3 (P = 0.001) (Figure 3.9B), it was hypothesised that the identity of this nucleotide may be fundamental for site 1 function. Therefore, this base was mutated in the UGT1A3-165bp construct to match the UGT1A4 sequence (pGL3-1A3-165 s1mut Δ 4) and was tested in HepG2 cells. Interestingly, instead of abolishing the fraction of UGT1A3 promoter activity attributable to the 15 nucleotides immediately upstream of the characterised HNF1-binding site, this nucleotide exchange substantially enhanced the activity of the UGT1A3-165bp promoter (P < 0.001) (Figure 3.9B). Therefore, it was concluded that Pbx-Hox factors are unlikely to be the transcription factors causing the 165-nucleotide UGT1A3 promoter to be more active than the 150 bp fragment.

The ability of the *UGT1A4* -165 to -150 nucleotide region to drive the *UGT1A3* promoter but not the *UGT1A4* promoter is interesting. Given that there is only one nucleotide difference in this region, and that both sequences can activate the *UGT1A3* proximal promoter, it seems highly likely that the same factors can bind the *UGT1A3* and *UGT1A4* sequences at this position. If this is true, this would mean that
the apparent functional difference that exists between the *UGT1A3*-165bp and *UGT1A4*-165bp promoters is actually due to elements downstream of the characterised HNF1-binding site. This result could be explained either by: a) factors bound downstream co-operating with the protein(s) bound to the -165 to -150 bp region in the case of *UGT1A3*; or b) factors bound downstream blocking access to the transcriptional machinery or otherwise inhibiting the factor(s) bound to the -165 to -150 bp region in the case of *UGT1A4*. These two mechanisms are not mutually exclusive. Data presented earlier (Figures 3.1 and 3.3) supports the notion that there are important functional differences in the proximal 150 nucleotides of the *UGT1A3* and *UGT1A4* promoters. The pGL3-1A4-150 reporter vector only has approximately half the activity of the corresponding *UGT1A3* construct in Caco-2 cells; the *UGT1A4*-150bp vector is HNF1 α -responsive in HepG2 cells, although the *UGT1A3*-150bp construct is not; while the *UGT1A4*-150bp promoter is the shortest vector to display the *UGT1A4*-specific, HNF1 α -independent promoter activity noted in HEK293T.

3.3.5. Factors bound to at least two elements in the UGT1A3 and UGT1A4 promoters co-operate with HNF1α

In HEK293T cells, both the *UGT1A3*-200bp and *UGT1A4*-200bp promoters had greater HNF1 α -responses than their 150 bp length counterparts (*P* < 0.001) (Figure 3.3). Therefore, it was hypothesised that there was a common element in the *UGT1A3* and *UGT1A4* promoters, within the -200 to -150 nucleotide regions that either binds a second HNF1 dimer, or a transcription factor that co-operates with the HNF1 α bound to the previously characterised site. To narrow down the location of this putative element, reporter vectors carrying the *UGT1A3*-165bp and *UGT1A4*-

165bp promoters were tested for their activity in HNF1 α -co-transfected HEK293T cells.

Interestingly, the results of this experiment showed that, for both genes, two separable elements were required to support the difference in activity between the HNF1 α -induced 200 bp and 150 bp promoters (Figure 3.10). In both cases, the 165 bp proximal promoters had greater activities than the 150 nucleotide fragments in the presence of over-expressed HNF1 α ($P \leq 0.001$), but less than the similarly treated 200 bp promoters ($P \leq 0.001$). Fold-activities over basal also increased as the promoters were extended. This behaviour closely resembles that of the *UGT1A3* promoter in HepG2 cells, suggesting that the same elements may be involved in both systems and that the factors that bind and activate the *UGT1A3* promoter through the two functional regions defined in HepG2 cells may be widely expressed rather than specifically liver-enriched. Whether the factors bound to the -200 to -180 and -165 to -150 bp regions of the *UGT1A3* promoter also co-operate with HNF1 α to drive *UGT1A3* transcription in HepG2 cells remains to be tested.

The involvement of the site 1 region of both promoters in the HNF1 α -response in HEK293T cells is noteworthy, since in HepG2 cells only the *UGT1A3* site has observable function. Given that only one nucleotide substitution separates the two sequences at site 1, and the *UGT1A3*-165bp and *UGT1A4*-165bp promoters behave similarly, it would be reasonable to assume that, at least in HEK293T cells, the factors that bind the *UGT1A3* and *UGT1A4* site 1 regions are the same. If these factors are widely expressed and also bind this region in HepG2 cells, then the hypothesis that the activation of the *UGT1A4*-165bp promoter in HepG2 cells is specifically repressed by a tissue-restricted factor recruited by downstream



Figure 3.10: The *UGT1A3* and *UGT1A4* -200 to -150bp regions contain at least two regulatory elements that can positively influence their HNF1a-response in HEK293T cells. The HNF1a-responses of the *UGT1A3*-165bp and *UGT1A4*-165bp promoters were compared to the respective 200 and 165 nucleotide length promoters in HEK293T cells by co-transfecting triplicate cultures with 0.5 µg of each promoter reporter vector, 25 ng pRL-Null and 0.25 µg of either pCMX or pCMX-HNF1a. The results are expressed as a relative value of firefly luciferase activity to the internal *renilla* control, compared to the pGL3-basic control (set to 1). The error bars indicate one standard deviation. Transfection of HEK293T cells with the pGL3-1A3-165 and pGL3-1A4-165 constructs was only performed once. *P* values for the comparisons indicated are **P* < 0.001 and †*P* = 0.001.

nucleotides (see section 3.3.4) becomes the most probable. Considerable further work will be required to identify the factor(s) that interact with this 15 nucleotide fragment in each cell type and to elucidate the mechanism that prevents the *UGT1A4* promoter from being activated in HepG2 cells. Further work is also required to characterise the second element that participates in the HNF1 α -response of these promoters and to determine whether it is the same element as site 2 of the *UGT1A3* promoter, as defined in HepG2 cells.

3.4 General discussion and summary

3.4.1. Achievement of aims

This chapter describes the cloning of the human UGT1A3 and UGT1A4 promoters to approximately 3.4 kilobases and the UGT1A5 promoter to 1.5 kb. Deletion, mutation and HNF1 α/β -over-expression analyses established that the putative HNF1-binding sites of all three promoters are functional *in vitro*; while comparative studies revealed that although HNF1 factors are critical for UGT1A3 and UGT1A4 promoter activity they are also insufficient to drive high levels of transcription.

Further investigations into the function of the *UGT1A3* promoter in HepG2 cells revealed two promoter regions between nucleotides -200 and -150 that were required for maximal promoter activity. Interestingly, these two functionally active regions of the *UGT1A3* promoter identified in this series of experiments are considerably more conserved between *UGT1A3* and *UGT1A4* (site 1: 14/15 nucleotides, 93%; site 2: 10/11 nucleotides, 91%) than the intervening region (9/14 nucleotides, 64%) (Figure 3.1). Conservation of elements between promoters is often used as an indicator for functional importance (Cartharius *et al.*, 2005); thus, despite the apparent lack of activity of these regions in the *UGT1A4* promoter in HepG2 cells, it was not surprising to find that at least one of these regions is utilised in common in an alternative cell line. Finally, the results presented strongly suggest that there is at least one element in the proximal 150 bp of the *UGT1A3* and *UGT1A4* promoters that causes their activity to differ *in vitro*.

3.4.2. Future directions

Two major studies that would add value to the presented work are highlighted here. Firstly, it would be of interest to confirm the functional importance of the two newly identified transcriptional elements of the *UGT1A3* promoter and determine the factors that bind them. The former could be achieved by mutation of the nucleotides hypothesised to be involved, followed by comparison to the wild-type sequence in promoter-reporter assays or EMSA; however, the latter task would be more complicated. Super-shift EMSA studies could be used to assess whether any of the factors predicted by MatInspector can bind site 2, but as there are no satisfactory predictions associated with site 1, a screening method with higher throughput and much less bias would also need to be employed. One feasible option would be to use the predicted binding site in a yeast one hybrid screen against HepG2 cDNA, similar to that described by Catlow *et al.* (2007), to identify potential transcription factors.

Secondly, it would be worthwhile to elucidate the element(s) in the proximal 150 nucleotides of the *UGT1A3* and *UGT1A4* promoters that cause their differential regulation. Because the HNF1-binding site at nucleotides -137 to -149 is crucial for promoter activity, further deletions could not be used to identify potential promoter elements; however, the construction of longer *UGT1A3/UGT1A4* chimeric promoters through reciprocal nucleotide substitutions or overlap PCRs could be used instead. The latter would be useful for assessing groups of nucleotides for function, as the 14 bases that the *UGT1A3* and *UGT1A4* promoters differ by in this region are sufficiently well spread that producing combinatorial mutations by site-directed mutagenesis would be arduous. Additional information may also be gleaned by including the *UGT1A5* promoter in these analyses, as the *UGT1A4* promoter shares functional similarities with both *UGT1A3* and *UGT1A5*, suggesting that it may represent a mixture of the elements that functionally define these highly related sequences.

3.4.3. Relevance to pharmacogenetics

Because UGT1A3 and UGT1A4 are purported to be key contributors to the metabolism of many pharmaceuticals and endogenous molecules, it is of interest to understand the environmental and genetic conditions that control their expression and activity. This study has added a significant body of information to the current understanding of the *UGT1A3* and *UGT1A4* proximal promoters that, in combination with the suggested experiments in section 3.4.2, will be useful for understanding the interindividual and tissue-specific expression profiles of these proteins. The promoter regions identified as important can now be specifically screened for allelic variants with functional consequences and, as they are identified, the factors that bind these elements can also be screened for variation in activity or expression level. Work towards these ends is presented in Chapter 6.

Because it is expected that proteins bound at distal enhancer regions of a promoter exert their effects through interaction with the transcriptional complexes formed over the core and proximal promoter regions, differences in the proximal promoters between the *UGT1A3* and *UGT1A4* genes, and between individuals in the same gene, can be expected to affect both basal and inducible gene expression. A recent study (Senekeo-Effenberger *et al.*, 2007) highlights the need to elucidate the basic functions of the *UGT1A3* and *UGT1A4* promoters before their response to liganded nuclear receptors can be fully understood. In the humanised mouse line Tg-*UGT1*, *UGT1A3* and *UGT1A4* were both found to be up-regulated by activated PPAR α , yet the *UGT1A4* hepatic response was only a small fraction of that seen for *UGT1A3*, even though the *UGT1A4* response in the small intestine was substantial. The authors' conclusions that PPAR α receptor abundance was not the sole factor dictating the PPAR α -response (Senekeo-Effenberger *et al.*, 2007) is in agreement with the findings of this study that regulatory pathways in common to these two genes can still diverge functionally due to the influence of other promoter elements.

3.4.4. Summary

This chapter describes the cloning and *in vitro* analysis of the *UGT1A3*, *UGT1A4* and *UGT1A5* proximal promoters. The results obtained highlight a critical role for HNF1 factors in *UGT1A3* and *UGT1A4* promoter activity and suggest a basis for the poor expression of UGT1A5 in humans. Two additional elements required for the maximal activity of the *UGT1A3* promoter in liver-derived cells were also identified, at least one of which appears to be shared by the *UGT1A4* promoter, but is only active in the context of *UGT1A3*. Further work suggested to elucidate the cause of this phenomenon may uncover an important fundamental difference between the *UGT1A3* and *UGT1A4* promoters that, at least in part, allows these highly related genes to be independently regulated in the liver.

CHAPTER FOUR HNF1 TRANSCRIPTION FACTORS ARE ESSENTIAL FOR THE *UGT1A9* PROMOTER RESPONSE TO HNF4α

Published in part as: Gardner-Stephen, D.A. and Mackenzie, P.I. (2007) Hepatocyte Nuclear Factor 1 transcription factors are essential for the UDPglucuronosyltransferase 1A9 promoter response to Hepatocyte Nuclear Factor 4α. *Pharmacogenetics and Genomics* 17: 25-36.

4.1 Introduction

4.1.1. The UGT1A7, UGT1A8, UGT1A9 and UGT1A10 gene cluster

The UGT1A7, UGT1A8, UGT1A9 and UGT1A10 exon 1 sequences, known as the UGT1A7-1A10 cluster, share > 70% identity (Gong *et al.*, 2001) (see Chapter 1, Figure 1.3). In addition, the proximal promoters of UGT1A7-1A10 share over 78% homology to approximately 400 bp upstream of their initiation codons (Figure 4.1); at which point the UGT1A7 promoter sequence diverges from the remainder due to the replacement of a 28 bp segment with 305 bp of Alu-like sequence. However, the UGT1A8-1A10 promoters continue to share > 75% identity to over 1 kb upstream of their TSSs (Gregory *et al.*, 2003). Despite these similarities and much like the UGT1A3-1A5 cluster of the same UGT1A locus (Chapter 3), each enzyme of the UGT1A7-1A10 subfamily has its own unique, but overlapping, set of substrates and a gene-specific expression profile. Of particular interest, the expression patterns of UGT1A7, UGT1A8 and UGT1A10 are all strictly extrahepatic, in stark contrast to UGT1A9, which has a strong presence in the liver (Strassburg *et al.*, 1997b; Mojarrabi and Mackenzie, 1998). This chapter presents a study designed to further



Figure 4.1: Alignment of the -391 bp UGT1A9 proximal promoter with the corresponding regions of UGT1A7, UGT1A8 and UGT1A10. The nucleotide sequence immediately upstream of human UGT1A9 exon 1 was aligned with the equivalent regions from the UGT1A7, UGT1A8 and UGT1A10 promoters using ClustalX software (Thompson *et al.*, 1997) and GenBank record AF297093 (Gong *et al.*, 2001). Periods indicate identity of each sequence with UGT1A9, while nucleotide substitutions are designated by the appropriate letter and deletions/insertions are indicated by hyphens. The UGT1A9 transcription start site is highlighted in bold underlined italic text, the initiation codon is indicated by bold italics and the 5' boundaries of the UGT1A9-321bp and UGT1A9-184bp promoter inserts are shown with arrows. Mutations introduced into the UGT1A9 promoter constructs are denoted in bold under the wild-type sequence.

the understanding of the *UGT1A9* proximal promoter function *in vitro*, with particular regard to mechanisms that may contribute to the unique hepatic expression of this enzyme among its most closely related family members.

4.1.2. Expression and substrates of UGT1A9

Human UGT1A9 is expressed primarily in the liver and kidneys, but has also been reported in the gastrointestinal tract (oesophagus, stomach and colon) and steroid responsive tissues such as prostate, breast, ovary and testes (Strassburg *et al.*, 1997b; McGurk *et al.*, 1998; Strassburg *et al.*, 1998a; Albert *et al.*, 1999; Strassburg *et al.*, 1999). Substantial interindividual variation in the hepatic expression of UGT1A9 has also been unequivocally demonstrated, with expression in the liver varying at the levels of mRNA concentration (Congiu *et al.*, 2002), protein concentration (Girard *et al.*, 2004) and protein activity (Court *et al.*, 2001; Bernard and Guillemette, 2004; Girard *et al.*, 2004). In addition, conflicting reports have been published regarding the presence of UGT1A9 mRNA in gastric tissue, suggesting that expression in this organ also varies between individuals, or that UGT1A9 expression is not uniform throughout the stomach and that the location of biopsy is important (Strassburg *et al.*, 1997b; Albert *et al.*, 1999).

UGT1A9 contributes to the glucuronidation of a large variety of chemicals. These include planar and bulky phenols (such as 4-methyl phenol and 4-*t*-butyl phenol respectively (Albert *et al.*, 1999)), coumarins and flavonoids (including 4-methylumbelliferone, scopoletin, 5,7-dihydroxyflavone and naringenin (Albert *et al.*, 1999)), polychlorinated dibenzo-*p*-dioxins (*e.g.* 8-OH-2,3,7-trichloro-dibenzo-*p*-dioxin (Kasai *et al.*, 2004)), mycophenolic acid (an immunosuppressive prodrug (Bernard and Guillemette, 2004)), the anaesthetic propofol (Soars *et al.*, 2004), the

analgesic paracetamol (Court et al., 2001), NSAIDs such as etodolac (Tougou et al., 2004), anticancer agents such as SN-38, flavopiridol and bropirimine (Gagne et al., 2002; Ramirez et al., 2002; Wynalda et al., 2003; Yamanaka et al., 2004), fibrate hypolipidaemics (e.g. gemfibrozil, fenofibric acid, clofibric acid and ciprofibric acid (Barbier et al., 2003c)), catechol oestrogens (including 4-hydroxyoestradiol and 4-hydroxyoestrone (Albert et al., 1999)), fatty acid metabolites (such as 20-carboxyleukotriene B₄ (Turgeon *et al.*, 2003b)) and dietary/tobacco procarcinogens (N-OH-PhIP and NNAL (Ren et al., 2000; Malfatti and Felton, 2001)). As these substances are bioactive and/or toxic at the concentrations readily encountered in everyday life, genetic variations in the coding or regulatory regions of UGT1A9 that result in a less active enzyme or lower protein expression levels are hypothesised to have clinically relevant outcomes (Court et al., 2001; Congiu et al., 2002; Gagne et al., 2002; Bernard and Guillemette, 2004; Girard et al., 2004). Therefore, it is desirable to understand more fully the mechanisms that contribute to the expression of this gene in order to understand its contribution to drug toxicity and efficacy, as well as its possible relevance to issues such as cancer risk.

4.1.3. Regulatory controls of UGT1A9

Published studies addressing the function of the *UGT1A9* gene and the elements that distinguish it from the extrahepatic members of the *UGT1A7-1A10* cluster are still limited in number, but include several reporter-promoter experiments, two SNP analyses and a clinical study involving patients receiving chemotherapy for treatment of metastatic colorectal cancer. Among the first studies investigating the molecular function of the *UGT1A9* promoter was work showing that the *UGT1A8* and *UGT1A10* proximal promoters are up to 8 times more active than the equivalent *UGT1A9* promoter construct when tested in the Caco-2 colon-derived cell line. This

difference was found to be partly dependent on a Sp1/Inr-like region conserved between *UGT1A8* and *UGT1A10*, but altered in *UGT1A9* at two nucleotide positions (Gregory *et al.*, 2003). In the same year, another study also linked the liver-enriched nuclear receptor PPAR α with hepatic expression of human UGT1A9. PPAR α agonists moderately increase the expression of UGT1A9 mRNA transcripts in human hepatocytes and HepG2 cells, an effect mediated by a PPAR α -responsive element located at nucleotides -719 to -706 of the *UGT1A9* promoter (Barbier *et al.*, 2003c). Subsequently, evidence that the *UGT1A8*, *UGT1A9* and *UGT1A10* genes share two promoter elements that may co-ordinately regulate the expression of each in the gastrointestinal tract was published. In each case, Cdx2 up-regulated their respective proximal promoters in transient assays, but only in the presence of a weak but functional HNF1 binding element (Gregory *et al.*, 2004). This HNF1 site, which was able to facilitate a 2 to 3-fold up-regulation of the *UGT1A8-1A10* genes in the presence of over-expressed murine HNF1 α , but not murine HNF1 β , is also present in the promoter of the *UGT1A7* gene (Gardner-Stephen and Mackenzie, 2005).

Another *UGT1A9* promoter-reporter study published in 2004 reported that the length of the poly-deoxythymidine tract that starts at nucleotide -118 of the *UGT1A9* promoter affects promoter activity *in vitro*. In HepG2 cells, a 158 bp *UGT1A9* promoter construct containing ten consecutive thymine residues was 2.6 times more active than the equivalent promoter with nine (Yamanaka *et al.*, 2004). However, the function of this polymorphism *in vivo* is still under debate. In support of the findings of Yamanaka *et al.* (2004), Carlini and colleagues found that the more prevalent *UGT1A9* -118 (dT)_{9/9} genotype was associated with less diarrhoea and better response than the *UGT1A9* -118 (dT)_{10/10} genotype in patients with metastatic colorectal cancer after treatment with irinotecan/capecitabine combination therapy (Carlini et al., 2005). The authors hypothesised that the lower activity UGT1A9 allele allows a higher systemic concentration of SN-38 to be achieved, while limiting the delivery of SN-38 to the gastrointestinal tract through biliary excretion of the SN-38glucuronide and subsequent hydrolysis. However, an analysis correlating UGT1A9 promoter SNPs and hepatic UGT1A9 concentration by Girard and co-workers found no effect of the UGT1A9 -118 (dT)_n polymorphism on UGT1A9 expression levels in human liver microsomes, but that the identities of several other promoter nucleotides (at positions -275, -331/-440, -665 and -2152) predicted hepatic UGT1A9 expression (Girard *et al.*, 2004). In their hands, the UGT1A9 -118 (dT)_n polymorphism also only altered promoter activity in vitro by 1.4-fold (Girard et al., 2006). Yet the same authors found that while the UGT1A9 -118 (dT)_n SNP did not predict UGT1A9 expression levels, it was associated with the glucuronidation rate of SN-38 (Girard et al., 2006). Another UGT1A9 SNP, the intronic polymorphism I339, was also associated with SN-38 glucuronidation rate and hepatic UGT1A9 and UGT1A1 content (Girard et al., 2006). The UGT1A9 promoter polymorphisms at positions -275 and -2152 have also been associated with mycophenolic acid exposure in renal transplant patients (Kuypers et al., 2005). To date, no mechanisms have been suggested for any of the reported associations of promoter sequence and activity.

Finally, in a recent study of UGT1A9 regulation, Barbier and co-workers have shown that HNF4 α regulates the human UGT1A9 proximal promoter *in vitro* (Barbier *et al.*, 2005). The ability to positively influence transcription from the UGT1A9 promoter in their experimental system required a weak HNF4 α -response element positioned at nucleotides -372 to -360 that is, importantly, absent in the promoters of the other UGT1A7-1A10 cluster genes due to single or dual nucleotide substitutions. However, mutating the UGT1A8 promoter to create a UGT1A9-equivalent HNF4 α -binding element only partially bestowed HNF4 α -responsiveness to the *UGT1A8* promoter construct, indicating that further elements within the *UGT1A9* promoter are also important for optimal HNF4 α -mediated expression. This was not entirely unexpected, as detailed investigations of liver-specific regulatory elements reveal that hepatic expression of a gene generally requires association of numerous LETFs with the promoter, many of which may have multiple binding sites (Schrem *et al.*, 2002; Costa *et al.*, 2003). Therefore, as the HNF4 α -binding site was the first and only element to be identified that exclusively activated the *UGT1A9* gene of the *UGT1A7-1A10* cluster, and because at least one other sequence difference between the promoters of these genes must co-operate in the HNF4 α -mediated induction of *UGT1A9*, it was decided that the regulation of the *UGT1A9* promoter by HNF4 α warranted further investigation.

4.1.4. The hepatocyte nuclear factor 4 transcription factor family

4.1.4.1. Physical attributes of HNF4 proteins

Hepatocyte nuclear factor 4α is a 465 amino acid zinc-finger transcription factor belonging to the HNF4 subfamily of the nuclear receptor superfamily (Schrem *et al.*, 2002). Other known members of the HNF4 subfamily are HNF4 β , HNF4 γ and DHNF4 (Zhong *et al.*, 1993; Holewa *et al.*, 1997), of which HNF4 γ is the only other protein known to be found in mammals (Drewes *et al.*, 1996). However, the function of HNF4 γ remains poorly characterised, and only HNF4 α was considered for the following investigations into the function of the *UGT1A9* promoter. Yet it should be noted that HNF4 γ is also a potential regulator of *UGT1A9*.

The known structural domains of HNF4 α include an N-terminal region required for transactivation (activation function (AF)1; amino acids 1-49), a C-terminal repressor

domain (also known as the F domain; amino acids 368-465), and two domains conserved among nuclear receptors: a zinc-finger DNA-binding domain (amino acids 50-115) and a large hydrophobic region (amino acids 135-367) that functions as a second activation domain (AF-2) as well as the ligand-binding and dimerisation domains (Ryffel, 2001; Schrem et al., 2002). Like many nuclear receptors, HNF4a homodimers bind to DNA sequences that loosely match a pair of AGGTCA-like NRREs. In the case of HNF4 α , these NRREs are arranged as direct repeats, separated by either one or two deoxyadenylate molecules, but with a very strong preference for only one spacer nucleotide. Thus the consensus sequence for known HNF4a-binding elements is RGGNCAAAGKTCR, where the "CAAAG" motif forms the essential core of the binding site, and R = A or G, K = G or T and N = A, C, G or T (Fraser et al., 1998). However, of the nuclear receptors that form dimers, HNF4 α is among the minority in that it is not known to heterodimerise with any other members of the nuclear receptor superfamily and has been specifically shown not to heterodimerise with the most common nuclear receptor heterodimerisation partner, the retinoid X receptor $(RXR)\alpha$, or with other nuclear receptors including RXR β , RXR γ , retinoid acid receptor (RAR) α , RAR β 2, RAR γ , vitamin D receptor, PPAR α and thyroid hormone receptor α (Jiang *et al.*, 1995). HNF4 α homodimers are stable in solution in the absence of DNA (Jiang et al., 1995).

Although human $HNF4\gamma$ is encoded on a separate chromosome to $HNF4\alpha$ (chromosomes 8 and 20, respectively (Drewes *et al.*, 1996)), the two genes are very similar. Overall, HNF4 α 1 (a splice variant of HNF4 α ; see section 4.1.4.2) and HNF4 γ share 70% identity (Plengvidhya *et al.*, 1999), and for the amino acids required for ligand binding, dimerisation and DNA recognition, this identity increases to 96%, 91% and 95% respectively (Drewes *et al.*, 1996; Wisely *et al.*,

2002). Thus, it has been suggested that heterodimerisation between HNF4 α and HNF4 γ may be the one exception to the observation that these proteins only form homodimers (Wisely *et al.*, 2002). Support for this hypothesis comes from functional studies showing that excess HNF4 γ in transient transfections can decrease the activity of co-transfected HNF4 α to the levels achieved by HNF4 γ alone (Drewes *et al.*, 1996).

The search for ligands of HNF4 α and HNF4 γ has caused much controversy. Originally, HNF4 α was considered an orphan nuclear receptor, as HNF4 α activates transcription in the absence of exogenously added ligand (Sladek *et al.*, 1990) and no activity-modulating ligands could be identified. However, in 1998, Hertz and colleagues showed that fatty acyl-coenzyme A (CoA) thioesters with chain lengths of 12 to 22 carbon atoms can be bound by amino acids 96 to 455 of HNF4 α (the HNF4 α ligand-binding and F domain), whereas free fatty acids are not (Hertz et al., 1998). They also showed that fatty acyl-CoA thioesters behave as HNF4 α transactivational agonists or antagonists, depending on their length and degree of saturation. However, in 2002, two research groups crystallised the HNF4 α (amino acid 132-382 or amino acid 103-465) and HNF4 γ (amino acids 103-408) ligand-binding domains, finding that in crystallised protein the ligand-binding pocket is occupied by free fatty acids, is too small to accommodate the suggested thioesters, and that the bound fatty acids are held so tightly that they cannot be stripped out or exchanged for radiolabelled counterparts (Dhe-Paganon et al., 2002; Wisely et al., 2002; Duda et al., 2004). Therefore, they concluded that HNF4 α and HNF4 γ are constitutively bound and activated by fatty acids, which are most likely irreversibly incorporated into the protein structure during synthesis and folding. Despite this work, Hertz and colleagues continued to show that endogenous and xenobiotic acyl-CoAs alter the

activity of HNF4a in cell culture and that intracellular long-chain acyl-CoA synthases are required for fatty acids to exert an effect on HNF4a-mediated transcription (Hertz et al., 2001). Furthermore, collaborations with Petrescu and coworkers showed that fatty acyl-CoA binding to the HNF4α ligand-binding domain significantly alters its secondary structure, in a manner that is opposite between compounds known to be agonists and those that are antagonists (Petrescu *et al.*, 2002). Finally, in 2005, the apparent discrepancies were resolved by publications showing that the C-terminal F domain of HNF4 α has thioesterase activity and specifically binds and hydrolyses fatty acyl-CoA molecules, and that fatty acids trapped in the ligand-binding domain of the full length protein are exchangeable, but with a very strong preference for fatty acid molecules generated from hydrolysable fatty acyl-CoAs by the C-terminal thioesterase (Hertz et al., 2005). Thus, the fatty acids found in the ligand-binding pocket of HNF4 α reflect the cellular pool of fatty acyl-CoA thioesters rather than the local free fatty acid constituents (Hertz et al., 2005; Schroeder et al., 2005). Therefore, the balance of current evidence suggests that HNF4 α and HNF4 γ bind the same or similar ligands, and that fatty acids are endogenous modulators of HNF4 activity via a pathway that involves esterification and hydrolysis, and an atypical relationship between the ligand-binding domain and receptor activation.

4.1.4.2. Expression profile of HNF4α

In humans, HNF4 α expression has been demonstrated in the liver, kidney, pancreas, small intestine and colon (Sladek *et al.*, 1990; Drewes *et al.*, 1996). Liver and kidney are also major sites of expression in the mouse (Zhong *et al.*, 1994). The HNF4 α gene consists of 13 exons (Furuta *et al.*, 1997; Torres-Padilla *et al.*, 2001) that give rise to numerous splice variants in both humans and mice. Of particular note are

splice variants HNF4 α 1, HNF4 α 2 and HNF4 α 7. HNF4 α 1 is the originally identified HNF4 α mRNA transcript (Sladek *et al.*, 1990); however, HNF4 α 2 is actually the most abundant HNF4a mRNA in adult liver and HepG2 cells, and is a stronger transactivator than HNF4 α 1 in some circumstances due to enhanced interactions with co-activators. The HNF4 α 1 and HNF4 α 2 proteins differ by a 10 amino acid insertion present in the F domain of HNF4 α 2 relative to HNF4 α 1 (Hata *et al.*, 1992; Chartier et al., 1994; Hata et al., 1995; Drewes et al., 1996; Sladek et al., 1999). On the other hand, HNF4 α 7 is an abundant splice variant in foetal liver, intestine, stomach and pancreas and, although also present in adult stomach, is very low in adult liver. This splice variant originates from an alternative promoter to HNF4 α 1/HNF4 α 2 and gives rise to a protein that has a different N-terminal domain to that of HNF4 α 1 (Nakhei *et* al., 1998; Torres-Padilla et al., 2001; Torres-Padilla et al., 2002; Briancon et al., 2004). HNF4 α 7 is a better transactivator of several promoters from genes known to be expressed early in development than HNF4 α 1 or HNF4 α 2 (Torres-Padilla *et al.*, 2001). HNF4 α 8 is the HNF4 α 2-equivalent splice variant of HNF4 α 7, having the same C-terminal sequence as HNF4 α 2, and the same N-terminal region as HNF4 α 7 (Torres-Padilla et al., 2001). The remaining HNF4α splice variants are minor species and are yet to be properly characterised, although it is known that HNF4 α 3, which has a completely different, shorter C-terminal domain than HNF4 α 1 and HNF4 α 2, has similar activity to HNF4 α 1 *in vitro* (Kritis *et al.*, 1996), and that HNF4 α 4, which has a 30 amino acid insertion in the N-terminal domain relative to HNF4a1, has very little activity in transfection assays (Drewes et al., 1996). HNF4 α 5 and HNF4 α 6 have the same N-terminal sequence as HNF4 α 4, but C-terminal ends equivalent to HNF4 α 2 and HNF4 α 3 respectively (Furuta *et al.*, 1997).

Several external and endogenous factors or circumstances are known to influence either the expression or transactivational activity of HNF4 α in humans or rodents. These include bile acids, cytokines, hypoxia, diet and exposure to drugs such as clofibric acid analogues (Viollet *et al.*, 1997; Hertz *et al.*, 2001; Mazure *et al.*, 2001; Zhang and Chiang, 2001; Li *et al.*, 2006). Transcription factors that interact with the human *HNF4\alpha* promoter include HNF1 α , HNF1 β , HNF6 and GATA6 (Hatzis and Talianidis, 2001), while HNF1 β and FXR have been implicated in the expression of HNF4 α in the mouse (Zhang and Chiang, 2001; Wang *et al.*, 2004a).

 $HNF4\gamma$ was first identified in humans, and the corresponding mRNA is found in human liver, kidney, pancreas, small intestine, colon, testes, brain and lung (Plengvidhya *et al.*, 1999). Although it has an overlapping expression and activity profile with HNF4 α , HNF4 γ cannot substitute for this transcription factor, as homozygous HNF4 α -knockout mice die early in embyrogenesis (Chen *et al.*, 1994).

4.1.4.3. Gene targets of HNF4α

The number of potential HNF4 α gene targets in humans is enormous. The proportion of proximal promoters shown to be bound by HNF4 α in human hepatocytes and pancreatic islet cells is 11-12% of the Hu13K DNA microarray; far greater than the equivalent results for HNF1 α or HNF6 (approximately 1.6%) and any other transcription factor previously tested by the same research group (maximum 2.5%) (Odom *et al.*, 2004). HNF4 α is known to act as both a positive and negative regulator of gene expression, and is particularly important in determining the hepatic phenotype during development (Li *et al.*, 2000).

A small selection of the many genes thought to be positively regulated by HNF4 α in humans includes important examples involved in fatty acid, lipoprotein and lipid

metabolism (e.g. apolipoprotein (apo)AII, apoB, apoCIII, medium chain acyl-CoA dehydrogenase and fatty acid-binding protein (fabp)2 (Ladias et al., 1992; Carter et al., 1993; Klapper et al., 2007)), amino acid and protein metabolism (for example, al-antitrypsin (Hu and Perlmutter, 1999)), haematopoiesis (erythropoietin and transferrin (Schaeffer et al., 1993; Galson et al., 1995)), blood coagulation (factors VII, IX, X and XI (Reijnen et al., 1992; Hung and High, 1996; Pollak et al., 1996; Tarumi et al., 2002)), biotransformation (CYP2A6, CYP2B6, CYP2C9, CYP2D6, CYP3A4, CYP3A5, CYP7A1, CYP8B1, UGT1A9, UGT2B11 and UGT2B15 (Jover et al., 2001; Zhang and Chiang, 2001; Odom et al., 2004; Barbier et al., 2005; Li et al., 2006) and this thesis) and organic ion transport (e.g. organic cation transporter 1 and OAT2 (Popowski *et al.*, 2005; Saborowski *et al.*, 2006)). HNF4 α is also thought to be a central regulator of carbohydrate metabolism, as mutations in the human $HNF4\alpha$ gene cause the inherited autosomal dominant disease, mature onset diabetes of the young subtype 1 (MODY1) (Ryffel, 2001). However, most studies linking HNF4 α with individual genes involved in carbohydrate metabolism have so far only investigated rodent homologues. Examples include insulin, aldolase B. phosphoenolpyruvate carboxykinase and human glucose-6-phosphatase (Hall et al., 1995; Gregori et al., 1998; Li et al., 2000; Wang et al., 2000; Bartoov-Shifman et al., 2002; Hirota *et al.*, 2005). Finally, HNF4 α is known to have a positive influence on the expression of several other transcription factors in humans and/or rodents, including HNF1α (rat and human) (Miura and Tanaka, 1993; Gragnoli et al., 1997; Li et al., 2000; Odom et al., 2004), PXR (mouse) (Li et al., 2000; Kamiya et al., 2003), HNF6 (rat) (Lahuna et al., 2000) and the small heterodimer partner (SHP; human) (Lai et al., 2003).

Conversely, a small number of genes are known to be repressed by HNF4 α , including both alternative promoters of the *HNF4\alpha* gene in humans and mice (Briancon *et al.*, 2004; Magenheim *et al.*, 2005). Other genes known to be repressed by HNF4 α include rat mitochondrial acetyl-CoA synthase (Rodriguez *et al.*, 1998), rat arginase (Chowdhury *et al.*, 1996) and rat peroxisomal acyl-CoA oxidase (Nishiyama *et al.*, 1998). Rat *apoAI*, which is activated by HNF4 α through one DNA element is also subject to negative regulation by HNF4 α through another (Murao *et al.*, 1997). Repression may be caused directly by competition with other nuclear receptors that bind the same site but have a greater transactivational potential for that gene (Nakshatri and Chambon, 1994; Nishiyama *et al.*, 1998; Rodriguez *et al.*, 1998), or indirectly by inhibiting the function of other transcription factors, such as Sp1, without necessarily binding DNA (Chowdhury *et al.*, 1996; Magenheim *et al.*, 2005).

4.1.4.4. HNF4α interacts with numerous co-factors and other proteins

HNF4 α has been shown to physically interact with numerous proteins that do not bind NRREs, including acyl-CoA-binding protein (Petrescu *et al.*, 2003), c-Jun (Li *et al.*, 2006) and Sp1 (Magenheim *et al.*, 2005). In addition, HNF4 α has been shown to be able to synergistically regulate genes in combination with other transcription factors such as the androgen receptor (Chen *et al.*, 2005b), HNF6 (Beaudry *et al.*, 2006), the glucocorticoid receptor (GR) (Nitsch *et al.*, 1993) and C/EBP α (Pitarque *et al.*, 2005). Conversely, FoxA factors bound to the α -1-microglobulin/bikunin precursor gene promoter inhibit the positive effects of HNF4 α (Rouet *et al.*, 1995), while PXR and HNF1 α have both been shown to agonise or antagonise HNF4 α mediated transactivation, depending on the promoter context (Ktistaki and Talianidis, 1997; Hu and Perlmutter, 1999; Ozeki *et al.*, 2001; Divine *et al.*, 2003; Bhalla *et al.*, 2004; Chen *et al.*, 2005b). In these listed instances, repression is caused by direct protein interactions or competition for co-factors; however, nuclear receptors such as chicken ovalbumin upstream promoter-transcription factors and apoAI regulatory protein 1 can also repress HNF4 α -driven genes by competing for access to NRREs (Ladias *et al.*, 1992; Mietus-Snyder *et al.*, 1992).

Multiple transcriptional co-factors and components of the general transcription machinery have also been shown to interact with HNF4 α . In particular, the HNF4 α N-terminal acidic activation domain interacts with the general transcription factors TBP, TAF_{II}31, TAF_{II}80, TFIIB, TFIIH-p62, the co-activators CBP/p300 and PC4, and the transcriptional adaptor ADA2 (Yoshida et al., 1997; Green et al., 1998b). Functional interactions between HNF4 α AF-2 and CBP, as well as between HNF4 α AF-2 and p160 nuclear receptor co-factors containing LXXLL motifs have also been established. HNF4 α activity is stimulated by p160 co-activators such as SRC-1, PGC-1, glucocorticoid receptor interacting protein (GRIP)-1 and amplified in breast cancer 1 (Wang et al., 1998; Lee et al., 2000; Yoon et al., 2001). Other nuclear receptors, such as PXR and CAR can also bind these co-activators and thereby decrease HNF4 α activity by sequestering its required co-factors (Bhalla *et al.*, 2004; Miao et al., 2006). CBP/p300 and SRC-1 mediate target gene expression by recruiting and activating the basal transcriptional apparatus, and also by overcoming the inhibitory effects of chromatin structure through their intrinsic HAT activities (Dallas et al., 1997; Soutoglou et al., 2000b; Schrem et al., 2002) (see also Chapter 3, section 3.1.5.4). In addition, CBP is known to acetylate HNF4 α , a modification that is crucial for its nuclear retention and also important for DNA-binding (Soutoglou *et al.*, 2000a). On the other hand, HNF4 α activity is repressed by the SMRT co-repressor, presumably through recruitment of HDACs to HNF4a/SMRT

complexes and/or by blocking positive interactions with GRIP1, CBP, or p300 (Ruse *et al.*, 2002; Torres-Padilla *et al.*, 2002), and by Prox1, which causes repression by competing for binding to the AF-2 domain with the co-activator PGC-1 (Song *et al.*, 2006). Finally, SHP, which belongs to the nuclear receptor superfamily but lacks a DNA-binding domain (Seol *et al.*, 1996), also interacts with the N-terminal AF-1 region, as well as the DNA-binding and AF-2 domains of HNF4 α . Thereby SHP inhibits HNF4 α -mediated transactivation of promoters through multiple mechanisms: by competing with p160 co-activator binding to HNF4 α ; by direct repression of HNF4 α function; and by preventing HNF4 α from binding to its cognate DNA elements (Lee *et al.*, 2000; Shimamoto *et al.*, 2004).

4.1.5. Aims

While work by Barbier and colleagues showed that HNF4 α is a major contributor to the activity of the *UGT1A9* promoter (Barbier *et al.*, 2005), the identified element did not fully explain the difference between the response of the *UGT1A9* promoter and the other members of the *UGT1A7-1A10* cluster (section 4.1.3). Therefore, the aims of this study were to:

- 1. Further characterise the role of HNF4 α in the hepatic expression of UGT1A9;
- 2. Identify additional promoter elements that contribute to the UGT1A9 gene response to HNF4 α .

4.2 Methods

4.2.1. Construction of the pGL3+ reporter plasmid

In order to provide appropriate cloning sites for the *UGT1A9*-2k promoter in the pGL3-basic reporter vector, pGL3+ was created. A sticky-ended, double-stranded

DNA fragment containing the original sequence of the pGL3-basic multiple-cloning site plus additional *Pst*I and *Eco*RI restriction sites was generated by kinase treatment and subsequent annealing of two oligonucleotides, pGL3insertF and pGL3insertR (see Table 4.1 for oligonucleotide sequences). The kinase reaction consisted of 10 µM each oligonucleotide, 70 mM Tris-HCl, 10 mM MgCl₂, 5 mM DTT, 1 mM ATP and 10 units T4 polynucleotide kinase (pH 7.6) and was incubated at 37°C for one hour before heating to 95°C for 5 minutes in a dry block heater. The heating element was then switched off and the oligonucleotides annealed by leaving the sample in the block, allowing it to return to room temperature without assistance.

The newly annealed insert was then ligated into *MluI/XhoI* restricted pGL3-basic to generate pGL3+, and sequenced (see Chapter 2, section 2.2.7 for details of endonuclease restriction and ligation methods, and Appendix 2 for restriction map of pGL3+).

4.2.2. Generation of the *UGT1A9*-2k reporter construct

pGL3+ vector was prepared for insertion of the proximal 2 kilobases of the *UGT1A9* promoter by restriction with *Pst*I and CIP treatment (see Chapter 2, section 2.2.7). The *UGT1A9*-2kb promoter fragment transferred into pGL3+ was released from pBS-2P by *Pst*I digestion. (pBS-2P is a pBlueScript II (pBSII) clone containing a 2 kb fragment of the human *UGT1A9* promoter, constructed by Dr. Kim Duncliffe (Department of Clinical Pharmacology, Flinders University). This *UGT1A9* promoter insert was originally shuttled into pBSII from a lambda clone, which was isolated from a human placenta lambda library (Clontech, Mountain View, CA) by Dr. Duncliffe (Gregory *et al.*, 2003)). The desired *UGT1A9*-2kb DNA fragment was separated from the pBSII vector fragment by agarose gel electrophoresis, recovered

Oligonucleotide	Nucleotide Sequence (5'→3')	Nucleotide Position on Target Gene	RE
pGL3 insertF	CGCGTCTGCAGGAATTCGCTAGCCCGGGC	NA	NA
pGL3insertR	TCGAGCCCGGGCTAGCGAATTCCTGCAGA	NA	NA
apoCIII-810for	AGCCAT <u>GCTAGC</u> ACGAGAGAATCAGTCCTGGT	apoCIII: -810 to -791	NheI
apoCIII+23rev	AGCCAT <u>CTCGAG</u> CTGCCTCTAGGGATGAACT	apoCIII: +23 to +3	XhoI
1A7prom-5kb	AGCCAT <u>AAGCTT</u> TTCAACTGTAGAAAAACAGAAC	UGT1A7: -5133 to -5112	HindIII
1A7rev	AGCCAT <u>AAGCTT</u> TTCGCAATGGTGCCGTCCAGC	UGT1A7: +310 to +290	HindIII
1A7rev2	AGCCAT <u>AAGCTT</u> ACATATAGTGGAAGGAGGCCA	UGT1A7: +41 to +21	HindIII
1A7prom-2.5kb	AGCCAT <u>AAGCTT</u> AAACTGTCAGTGAGTGTTGAT	UGT1A7: -2585 to -2565	HindIII
1A9prom-184bp	TTTT <u>GGTACC</u> TCAGCAAAAGCTACTC	UGT1A9: -184 to -169	KpnI
1A9prom-321bp	AGCCAT <u>GGTACC</u> TTTCTGAACCTTCAAGGTCCA	UGT1A9: -321 to -300	KpnI
1A9prom-1kb	ACGCATCTGCAGGTTCTTGCCGAAGCCTTC	UGT1A9: -1038 to -1021	PstI
1A9UTRrevMluI	AGCCAT <u>ACGCGT</u> CAGAGAACTGCAGCTGAGAGC	UGT1A9: -1 to -21	MluI
1A9UTRrev	AGCCAT <u>CTCGAG</u> CAGAAGTGCAGCTGAGAGC	UGT1A9: -1 to -21	PstI
QC1A9-HNF4s1mt	CTAAATTTTGCTCTGGGACGGGCCTTGAAAAAAAATTAGC	UGT1A9: -386 to -348	NA

Oligonucleotide	Nucleotide Sequence (5'→3')	Nucleotide Position on Target Gene	RE
QC1A9-HNF4s2mt	GTTCTGCCCCAAGGCGGCCTTGTAAGCTACTGTTGTC	UGT1A9: -246 to -208	NA
QC1A9-HNF4s2mt8	GTTCTGCCCCCAAGGCAGAGTATAAGCTACTGTTGTC	UGT1A9: -246 to -208	NA
QC1A9-HNF4s2cons	GTTCTGCCCCCAGGGCAAAGGTCATAAGCTACTGTTGTC	UGT1A9: -246 to -208	NA
QC1A9-HNF1s1mt	CTTGTTCTTTTGGGTCGCTCATTGTCAGTGACTG	UGT1A9: -161 to -128	NA
QC1A9-HNF1s2mt	GGTCCAAAAGCATTGGTCGCTAATTCTGCTTCTAAAC	UGT1A9: -305 to -269	NA
QC1A9-HNF1s2mt8	CTTCAAGGTCCAAAAGCATTGCTTAGTAATTTTGCTTCTAAACTTAAC	UGT1A9: -311 to -264	NA
QC1A9-mtT-275A	GGTTAATAATTCTGCTACTAAACTTAACATTGCAGCACAGGGC	UGT1A9: -291 to -249	NA
QC1A9-mtC-62T	GTGCTGGTATTTCTTCCACCTACTGTATCATAGGAGC	UGT1A9: -76 to -40	NA
QC1A9-mtA-59G	GTGCTGGTATTTCTCCCGCCTACTGTATCATAGGAGC	UGT1A9: -76 to -40	NA
QC1A9-mtC-62T/A- 59G	GTGCTGGTATTTCTTCCGCCTACTGTATCATAGGAGC	UGT1A9: -76 to -40	NA
QC1A8-HNF4s2mt9	GGCATGATCTGTCCAAGGCAAAGACCATAAGCTACTCTTATAG	UGT1A8: -264 to -222	NA
QC1A8-HNF1s2mt9	CAAGGTCCTAAAGCATTGGTTAATAATTCTGTTTCTAAACTCACG	UGT1A8: -320 to -276	NA
T7	TAATACGACTCACTATAGGGGAGA	T7 promoter of pCMX plasmid	NA
HNF1α-546Xrev	AGCCATGGATCCTTAGTCTGAGGTGAAGACCTGCTT	HNF1 α : +1636 to +1616	BamHI

Table 4.1 continued.

148

continued.
Table 4.1

QCHNF4-mtR127W CAGAATGAGCGGGACT		Nucleotide Position on Larget Gene	RE
	ACTGGATCAGCACTCGAAG	HNF4a: +391 to +422	NA
QCHNF4-mtE276Q GCAGATCGATGACAATC	ATCAGTATGCCTACCTCAAAGCC	HNF4α: +927 to +873	NA
QCHNF4- GGTGCCTCGAGCTGTGC mtDG69/70AA	TGCCGCCTGCAAGGGCTTCTTC	HNF4α: +217 to +252	NA
QCHNF4- CTGCAAGGGCTTCTTCG mtRR76/77AE	TCGCGGAGGCGTGCGGAAGAAC	HNF4α: +237 to +273	NA

RE: restriction endonuclease site, as underlined. NA: not applicable. Mutations are highlighted in bold. Nucleotide positions are numbered relative to the "A" of the initiation codon.

using the Qiagen gel extraction kit (see Chapter 2, section 2.2.8), and ligated into the *Pst*I cut pGL3+ vector. Because the cloning of this promoter fragment was nondirectional, clones were screened for insert orientation by *XhoI/Spe*I digestion and a representative from both the forward and reverse orientation clones was chosen and sequenced.

4.2.3. Generation of UGT1A9 deletion constructs and mutants

A construct containing the UGT1A9-184bp proximal promoter was generated by PCR amplification from the UGT1A9 promoter lambda clone described above. Amplification was achieved using *Taq* DNA polymerase as described in Chapter 2, section 2.2.6.4 with oligonucleotides 1A9prom-184bp and 1A9UTRrevMluI (Table 4.1). Cycling conditions were 95°C for 4 minutes, followed by 30 cycles of 95°C for 30 seconds, 50°C for 30 seconds and 72°C for one minute, and the reaction was finished with a final extension step of 72°C for 10 minutes. To amplify the UGT1A9-321bp fragment, the 1A9prom-321bp sense primer (Table 4.1) was combined with 1A9UTRrevMluI in a PfuTurbo PCR reaction on the pGL3+1A9-2k plasmid as template of: 95°C for 4 minutes, 30 cycles of 95°C for 30 seconds, 55°C for 30 seconds and 72°C for 1 minute, and a final extension step of 72°C for 5 minutes. Both PCR products were sequentially digested with MluI and KpnI and cloned into a similarly digested sample of pGL3-basic vector. Both inserts were sequenced in full to ensure that no PCR-generated errors had been introduced. To amplify the UGT1A9-1kb fragment, the primers 1A9prom-1kb and 1A9UTRrevPstI were used in a Taq DNA polymerase reaction (94°C for 4 minutes; 30 cycles of 94°C for 30 seconds, 55°C for 30 seconds and 72°C for 90 seconds; and 72°C for 10 minutes) using pGL3+1A9-2k as the template. The *Pst*I-digested 1 kb PCR product was then

cloned into the *Pst*I site of pGL3+, screened for insert direction by *Pvu*II and *Xho*I digest, and sequenced in full.

To generate *UGT1A9*-2kb and *UGT1A9*-1kb proximal promoter-reporter clones carrying mutations as indicated by Figure 4.1, site-directed mutagenesis was performed as described in Chapter 2, section 2.2.6.5. The sense sequences for each oligonucleotide pair used for site-directed mutagenesis PCRs are listed in Table 4.1 with the mutated nucleotides highlighted in bold. *UGT1A9*-specific nucleotides were also substituted into the *UGT1A8*-2kb proximal promoter HNF1s2 and HNF4s2 regions in the same way.

4.2.4. Isolation of the *UGT1A7*-5k promoter and generation of pGL3-1A7-2.5k To create the *UGT1A7* promoter constructs, approximately 5 kb of the *UGT1A7* promoter were cloned from human genomic DNA by nested PCR. *Not*I-digested HEK293T genomic DNA was used as template for two successive rounds of PCR using *Taq* DNA polymerase with *Taq* Extender PCR additive (see Chapter 2, sections 2.2.6.3 and 2.2.6.4). Firstly, the 1A7prom-5kb primer was combined with 1A7rev (Table 4.1) for the external PCR reaction, which contained 0.125 Units/µl *Taq* DNA polymerase, 0.125 Units/µl *Taq* Extender and 2.5 ng/µl *Not*I-digested gDNA, and was cycled through a single stage of 95°C for 4 minutes; 35 cycles of 95°C for 30 seconds, 55°C for 30 seconds and 72°C for 10 minutes; and finished with an additional extension step of 72°C for 10 minutes. One microlitre of the resulting PCR product was then used as template for a second round of amplification using primers 1A7prom-5kb and 1A7rev2 under the same conditions as the previous PCR. The ensuing 5 kb PCR product was cloned non-directionally into the *Hind*III site of pSII, shuttled into the *Hind*III site of pGL3-basic and screened for direction of

insertion with *Eco*RI. Chosen clones were sequenced from each end to confirm the identity of the insert as the *UGT1A7* promoter.

The pGL3-1A7-5k construct was then used as template to amplify approximately 2.5 kb of the *UGT1A7* proximal promoter using the same *Taq* DNA polymerase/*Taq* Extender mix and PCR conditions as the 5 kb PCR reaction. The oligonucleotides used to amplify this region of the *UGT1A7* promoter were 1A7prom-2.5kb and 1A7rev2 (Table 4.1). The resulting product was cloned directly into the *Hind*III site of pGL3-basic, and also screened for direction of insertion with *Eco*RI. Finally, chosen clones were screened for PCR errors in the proximal promoter region by sequencing.

4.2.5. UGT1A8 and UGT1A10 reporter constructs

The *UGT1A8*-2k and *UGT1A10*-2k constructs were created in our laboratory by Drs. Philip Gregory and Rikke Lewinsky respectively, as previously described (Gregory *et al.*, 2003; Gregory *et al.*, 2004). The pGL3-1A8-1k promoter-reporter vector was also created by Dr. Gregory (Gregory *et al.*, 2003).

4.2.6. Isolation of the *apoCIII*-810/+23 promoter

To generate the pGL3-apoCIII-810/+23 reporter construct, the corresponding region was amplified from HEK293T genomic DNA using primers apoCIII-810for and apoCIII+23rev (Table 4.1), adapted from Fraser *et al.* (1998). *Taq* DNA polymerase was used for the amplification of the *apoCIII* promoter from human (HEK293T) genomic DNA with the cycling parameters: 95°C for 4 minutes; 35 cycles of 95°C for 30 seconds, 59°C for 1 minute and 72°C for 2 minutes; and 72°C for 5 minutes. The resulting PCR product was digested with *Nhe*I and *Xho*I and cloned into the same two respective sites of pGL3-basic.

4.2.7. Transcription factor expression vectors

The construction or acquisition of all expression plasmids used in this chapter, except pCMX-HNF1 α 546X (which is described below), is detailed in Chapter 2, sections 2.1.4, 2.3.1 and 2.3.2.

The cDNAs encoded by the pCMX-HNF1 α and pCMX-HNF1 β expression vectors are human in origin and express the A-variants of each transcription factor. The human HNF1 α 546X truncation mutant was generated by PCR amplification from the pCMX-HNF1 α vector template with primers T7 and HNF1 α -546Xrev (Table 4.1). The PCR was performed with *PfuTurbo* and the conditions used were: 95°C for 4 minutes; followed by 30 cycles of 95°C for 30 seconds, 55°C for 30 seconds and 72°C for 4 minutes; and finished with a final extension step of 72°C for 5 minutes. The resulting PCR product was restricted with *Hind*III and *Bam*HI and cloned into these sites of pCMX-PL2.

cDNAs encoding the human HNF4 α 1, human HNF4 α 2 and rat HNF4 α 1 variants were also cloned into pCMX-PL2. Mutations in the human HNF4 α 2 coding region were introduced by site-directed mutagenesis as described in Chapter 2, section 2.2.6.5. The sense sequences of each oligonucleotide pair used for site-directed mutagenesis of the human HNF4 α 2 cDNA are listed in Table 4.1, with the mutated nucleotides highlighted in bold.

4.2.8. Transient transfection and luciferase reporter assay

HepG2 and HEK293T cells were seeded into 24-well plates and transfected as described in Chapter 2, section 2.2.10. Each well was transfected with either 0.5 μg of empty pGL3-basic or a reporter vector carrying the indicated *UGT1A7*, *UGT1A8*, *UGT1A9*, *UGT1A10* or a*poCIII* promoter sequences. For induction studies, 0.25 μg

of each pCMX expression vector, as specified in the results, was co-transfected with the pGL3 reporter constructs. Empty pCMX-PL2 expression vector was added as required to maintain a total of 1 µg DNA per transfection and pRL-null (0.025µg) was added to all transfections as an internal control for transfection efficiency. After 48 hours, cells were lysed in passive lysis buffer and analysed for firefly and *renilla* luciferase activity using the Dual-Luciferase Reporter Assay System as detailed in Chapter 2, section 2.2.11.

4.2.9. Electrophoretic mobility-shift and super-shift assays

HepG2 nuclear extract, prepared as described in Chapter 2, section 2.2.12, or *in vitro* synthesised HNF1 α and HNF4 α protein was used to perform EMSAs and super-shift assays as detailed in Chapter 2, sections 2.2.13, 2.2.14 and 2.2.15. The sense sequences of the double-stranded DNA probes used are listed in Table 4.2. The anti-HNF1 α and anti-HNF4 α antibodies used in the super-shift assays were sourced from Santa Cruz Biotechnologies (sc-6547 and sc-6556).

4.2.10. In vitro synthesis of transcription factor proteins

In vitro synthesis of HNF1 α and HNF4 α protein was performed using the TNT Quick Coupled Transcription/Translation kit (Promega) according to the manufacturer's instructions. Briefly, pCMX-HNF1 α or pCMX-HNF4 α expression plasmid was included at a concentration of 20 ng/µl in 50 µl reaction mixes that also consisted of 40 µl TNT Quick master mix (containing reticulocyte lysate, RNA polymerase, nucleotides, salts and RNasin ribonuclease inhibitor) and 20 mM methionine. The reaction was incubated at 37°C for 90 minutes to generate the desired protein and stored in aliquots at -20°C.

Oligonucleotide	Nucleotide Sequence (5'→3')
apoCIII-HNF4 [§]	CAGCAGG <u>TGACCTTTGCCCA</u> GCGCCC
UGT1A9-HNF4s1 [§]	TTTGCTCT <u>GGGACAAATTCCA</u> AAAAAAATTAG
UGT1A9-HNF4s2	GCCCCCAAGGCAAAGACCATAAGCT
UGT1A9-HNF1s1	TTTGG <u>GTAAATCATTGTC</u> AGTGACTGA
UGT1A9-HNF1s1mt	CTTGTTCTTTTGG <u>GTCGCTCATTGTC</u> AGTGACTG
UGT1A7-HNF1s2	GCATTG <u>CTTAATAATTTTG</u> TTTCTA
UGT1A8-HNF1s2	GCATTG <u>CTTAGTAATTTTG</u> TTTCTA
UGT1A9-HNF1s2	GCATTG <u>GTTAATAATTCTG</u> CTTCTA
UGT1A9-HNF1s2mt	GCATTG <u>GTCGCTAATTCTG</u> CTTCTA
UGT1A10-HNF1s2	GCATTG <u>CTGAATAATTCTG</u> TTTCTA

 Table 4.2: Oligonucleotides used for EMSA and super-shift experiments.

Underlined text indicates the extent of the putative HNF1 or HNF4-binding sites in each probe and deliberate mutations are highlighted in bold. NB: Only the sense strand of each oligonucleotide pair is shown. [§]Probe sequences are the same as those used by Barbier *et al.* (2005).

4.2.11. Western blot

Joanna Treloar performed the Western blots presented in Figures 4.3B and 4.10B, using total cell lysates from HepG2 and HEK293T cells, as well as *in vitro* synthesised HNF1 α and HNF4 α proteins. The protocol used is detailed in Chapter 2, section 2.2.16.

4.2.12. Statistical analysis

Statistical treatment of all reporter-promoter assay data was performed as described

in Chapter 2, section 2.2.17 using independent-samples t-tests.

4.3 **Results and discussion**

4.3.1. The *UGT1A9* proximal promoter responds strongly to over-expression of human HNF4α2 in HepG2 cells

In a recent paper by Barbier and colleagues it was shown that the human UGT1A9 promoter was responsive to the presence of over-expressed HNF4 α protein *in vitro* and, most interestingly, that the remaining UGT1A7-IA10 gene promoters were not stimulated by HNF4 α (Barbier *et al.*, 2005). To determine whether these findings could be replicated with my reporter constructs, the human UGT1A7-IA10 proximal promoters were co-transfected with a human HNF4 α 2 expression vector into the human hepatocellular carcinoma-derived cell line, HepG2. Figure 4.2 shows that, in these cells, over-expression of human HNF4 α 2 increased the UGT1A9-2kb promoter activity by 45-fold (P < 0.001), a response considerably larger than the 5 to 8-fold response reported by Barbier *et al.* (2005). In accordance with their work, however, the UGT1A7 and UGT1A8 promoter reporter constructs were unaffected (P = 0.804 and P = 0.167 respectively) and UGT1A10 was slightly inhibited (P = 0.005) by co-transfection with pCMX-HNF4 α . Therefore, further experiments were designed to characterise in more detail the unique regulation of the UGT1A9 promoter by HNF4 α .

4.3.2. The *UGT1A9* promoter -372 to -360 HNF4α-binding site is not necessary for the *UGT1A9* promoter response to human HNF4α2

The recent publication linking HNF4 α expression with *UGT1A9* transcription indicated that this was due to a weak but functional HNF4 α -response element found only in the *UGT1A9* promoter, located at nucleotide position -372 to -360 (HNF4 site 1, Figure 4.1) (Barbier *et al.*, 2005). An oligonucleotide probe over this region bound HNF4 α weakly in EMSA, and mutation of this site abolished pSG5-HNF4 α -



Figure 4.2: Over-expression of human HNF4a in HepG2 cells strongly upregulates the human *UGT1A9* proximal promoter. HepG2 cells were transfected in triplicate with 0.5 µg of pGL3-based reporter vector, 25 ng pRL-Null and 0.25 µg pCMX-PL2 or pCMX-HNF4a2 as described under "Methods". Mutations (HNF4s1mt, HNF4s2mt and HNF4s2cons) in the UGT1A9 HNF4 sites as defined by Figure 4.1 were introduced by site-directed mutagenesis. Cells were harvested 48 hours post-transfection and assayed for both firefly and *renilla* luciferase activity. Results shown are the mean firefly luciferase activities relative to the internal *renilla* controls, expressed as a fold induction over the promoterless pGL3-basic vector (set at a value of 1). Error bars represent one standard deviation. P values for the indicated comparisons are *P < 0.001, $\ddagger P = 0.005$, $\ddagger P = 0.013$ and #P > 0.05 (not significant). Also, *UGT1A9* promoter deletion constructs have significantly greater activity than empty pGL3-basic vector (**P < 0.001, $\ddagger P = 0.008$) and pGL3-1A9-2kb HNF4s2cons has greater basal and inducible activity than pGL3-1A9-2kb (††P = 0.002).

mediated activation in HepG2 transfections. Therefore, I first targeted this region in my desire to further understand the *UGT1A9* response to HNF4 α . Figure 4.2 shows the results of human HNF4 α 2 co-transfections in HepG2 cells with *UGT1A9* promoter constructs where this element (HNF4 site 1) is either mutated (pGL3-1A9-2k HNF4s1mt) or deleted (pGL3-1A9-321bp). It was found that, in the context of the 2 kb proximal promoter, mutation of the HNF4 α -binding site caused only a modest reduction of less than 2-fold in the activation by human HNF4 α 2 (P < 0.001). In addition, the construct containing only the proximal 321 bp of the *UGT1A9* promoter, where the putative HNF4 α site is excluded, was activated to an even greater extent by HNF4 α than the original 2 kb construct (P < 0.001). In contrast, activation of the *UGT1A9* promoter by human HNF4 α 2 could be abolished by further 5' deletion of the *UGT1A9* sequence, leaving only 184 bp upstream of the initiation codon (P = 0.472) (Figure 4.2). Thus, I established that there was least one important element in the -321 to -184 region of the *UGT1A9* promoter for HNF4 α -responsiveness.

4.3.3. The UGT1A9 response to HNF4 α in vitro is dependent on a second HNF4 α -binding element

Examination of the nucleotide sequence of the *UGT1A9* promoter between bases -321 and -184 revealed a CAAAG motif, known to be the core sequence required for HNF4 α binding to DNA (Fraser *et al.*, 1998). Therefore, I postulated that this element (nucleotides -235 to -223) may serve as the core of a second binding site for HNF4 α , and may contribute to the HNF4 α -responsiveness of the *UGT1A9* promoter. Mutation of this putative binding site (HNF4 site 2, Figure 4.1) almost completely abolished the response of the *UGT1A9*-2kb promoter construct to over-expression of human HNF4 α 2 in HepG2 cells, although the 1.5-fold residual increase in luciferase expression that was retained in co-transfections was statistically significant (*P* = 0.013) (Figure 4.2). Furthermore, an oligonucleotide probe encompassing this site was able to bind HNF4 α from HepG2 nuclear extracts (Figure 4.3A). It was also confirmed that HNF4 site 1 could bind HNF4 α ; however, under the assay conditions


Figure 4.3: Assessment of HNF4a binding to UGT1A9 promoter sequences in vitro. A. HNF4a binds weakly to both HNF4 sites of the UGT1A9 promoter in mobility-shift assavs. ^{32}P end-labelled electrophoretic double-stranded oligonucleotide probes (50,000 cpm) containing HNF4 sites of the human apoCIII or UGT1A9 promoters were incubated with 5 µg HepG2 nuclear extract and subjected to electrophoresis on a 4% (w/v) non-denaturing polyacrylamide gel. Binding of HNF4 α to each probe was demonstrated by super-shift of the relevant complexes with 2 μ g of anti-HNF4 α antibody added immediately after the labelled oligonucleotides. The complexes pertinent to this study are indicated with arrows. B. HNF4a and HNF1a protein can be detected in HepG2 but not HEK293T total cell lysates. HepG2 and HEK293T total cell lysates were probed for the presence of HNF4 α and HNF1 α proteins by Western blot as described in Chapter 2, section 2.2.16. In vitro translated HNF4a and HNF1a proteins were utilised as positive controls.

used, this probe also bound a considerable quantity of protein that could not be super-shifted with anti-HNF4 α antibody. Both *UGT1A9* HNF4 α sites were found to be extremely weak binders of HNF4 α in comparison to a probe containing the most proximal HNF4 α -binding element of the *apoCIII* gene (Figure 4.3A).

4.3.4. The UGT1A9 proximal promoter response to HNF4 requires HNF1a

When the response of the *UGT1A9* proximal promoter to human HNF4 α 2 in HEK293T cells (which do not express endogenous HNF4 α (Maeda *et al.*, 2002) (Figure 4.3B)) was investigated, absolutely no increase in reporter activity was observed (P = 0.054) (Figure 4.4A). In contrast, the *apoCIII*-810/+23 promoter, which is also regulated by HNF4 α (Fraser *et al.*, 1998), was increased by over 100-fold under the same conditions (P < 0.001). This indicated that at least one additional component, missing from the transcription factor or co-factor profile of HEK293T cells, was required for up-regulation of the *UGT1A9* promoter by HNF4 α *in vitro*. Furthermore, this factor was not required for the *apoCIII* promoter response to HNF4 α .

The homeodomain transcription factor HNF1 α has been reported to operate synergistically with HNF4 α on several promoters (Hu and Perlmutter, 1999; Ozeki *et al.*, 2001; Eeckhoute *et al.*, 2004). In addition, HEK293T cells do not express HNF1 factors (Bernard *et al.*, 1999) (Figure 4.3B), the *UGT1A9* promoter is known to possess a functional HNF1 site located between nucleotides -148 and -136 (HNF1 site 1, Figure 4.1) (Gregory *et al.*, 2004), and the *apoCIII* promoter does not contain any functional HNF1 sites (Kritis *et al.*, 1993). Therefore, it was decided to investigate whether human HNF1 α was able to affect the HNF4 α -mediated regulation of the *UGT1A9* promoter. When an HNF1 α expression vector was

Figure 4.4: HNF1a synergistically regulates the UGT1A9 promoter with HNF4a through a second HNF1-binding site. A. Human HNF4a and HNF1a synergistically up-regulate the UGT1A9 promoter in HEK293T cells. HEK293T cells were transfected, according to the methods described in section 4.2.8, with 0.5 ug pGL3-basic or reporter vector containing the indicated lengths of the human UGT1A9 or apoCIII promoters. Each transfection also included 25 ng pRL-Null and 0.25 µg of the human HNF4 α 2 and/or HNF1 α expression vectors. The DNA concentration in each transfection was kept constant by addition of empty pCMX-PL2 vector as necessary. B. A second HNF1-response element is a major requirement for the action of HNF4a on the UGT1A9 promoter. Mutations were introduced into the known HNF1 response element (HNF1s1mt) and a putative HNF1 binding site (HNF1s2mt, HNF1s2mt8) in the UGT1A9-2k promoter by site directed mutagenesis (see Figure 1). HepG2 cells were transfected with the UGT1A9 wild-type and mutant promoter reporter constructs in the presence of pCMX-PL2 or either of the HNF4α or HNF1α expression plasmids, as described for HEK293T cells. The results of both the HEK293T and HepG2 experiments are the means of triplicate samples, expressed as a relative value of firefly luciferase activity to the internal renilla control, compared to the pGL3-basic control (set to 1). The error bars indicate one standard deviation. ND = not done. P values for the indicated comparisons are *P < 0.001, $\ddagger P = 0.036$ and #P > 0.05 (not significant). Also, pGL3-1A9-2k HNF1s1mt and pGL3-1A9-2k HNF1s2mt constructs are statistically less responsive to HNF1 α than the wild-type UGT1A9 promoter in HepG2 cells ($\dagger \dagger \dagger P$ = 0.003 and $\ddagger P = 0.038$).



included in HEK293T transfections of the UGT1A9 promoter, a modest increase in luciferase reporter activity of 8-fold (P < 0.001) was observed. Interestingly however, when co-transfections of pGL3-1A9-2k, pCMX-HNF4a2 and pCMX-HNF1 α were performed, a synergistic response of 45-fold was obtained (P < 0.001) (Figure 4.4A). A similar response was also seen with the deletion construct pGL3-1A9-321 (P < 0.001). In stark contrast, HNF1 α over-expression did not activate the apoCIII construct in HEK293T cells, but rather served to diminish the promoter's response to the presence of excess human HNF4 $\alpha 2$ (P = 0.036), as previously reported (Kritis et al., 1993). Interestingly, simultaneous over-expression of both HNF4 α and HNF1 α in HepG2 cells transfected with the UGT1A9-2kb promoter did not greatly increase luciferase expression (less that 2-fold) relative to that achieved by HNF4 α alone (P = 0.026) (Figure 4.5). However, HepG2 cells are known to express both HNF1α and HNF1β (Song et al., 1998) (Figure 4.3B). Therefore, I postulated that the expression levels of endogenous HNF1 factors in these cells, particularly HNF1 α , was adequate to support the observed HNF4 α -response, and that mutation of the HNF1 α binding site(s) involved would reveal the synergistic nature of the interaction.

Further evidence corroborating the importance of HNF1 α in the HNF4 α -mediated regulation of UGT1A9 expression was obtained by substituting the weak *UGT1A9* HNF4 site 2 element with an optimal HNF4 consensus sequence (HNF4s2cons). Rowley and co-workers recently showed that HNF1 α and HNF4 α worked to increase the promoter occupancy rate of each other on the *fabp1* gene (Rowley *et al.*, 2006). As both putative HNF4-binding sites of the *UGT1A9* promoter perform poorly in EMSA, the possibility that the requirement for HNF1 α in HNF4 α -mediated up-



Figure 4.5: HNF1a only moderately increases activation of the UGT1A9 proximal promoter by HNF4a in HepG2 cells. HepG2 cells were transfected with 25 ng pRL-Null, 0.5 µg pGL3-basic or pGL3+1A9-2k reporter plasmid and 0.25 µg of the human HNF4a2 and/or HNF1a expression vectors. The DNA concentration in each transfection was kept constant by addition of empty pCMX-PL2 vector as necessary. The results presented are the means of triplicate samples, expressed as a relative value of firefly luciferase activity to the internal *renilla* control, compared to the pGL3-basic control (set to 1). The error bars represent one standard deviation. $\ddagger P = 0.026$.

regulation of the *UGT1A9* promoter may have been simply due to its ability to recruit HNF4 α was considered. However, it was found that substituting *UGT1A9* HNF4s2 for HNF4s2cons did not substantially alter the response of the *UGT1A9* promoter to either HNF1 α or HNF4 α . While the basal activity of the *UGT1A9* promoter and the magnitude of the response to over-expressed HNF4 α in HepG2 cells were both marginally increased (*P* = 0.002 for both) (Figure 4.2), the absolute requirement for HNF1 α was completely preserved (Figure 4.4A). Although this result by no means eliminates the possibility that HNF1 α is involved in recruitment of HNF4 α to the *UGT1A9* promoter, it does indicate that this factor has a true transactivational role in the observed partnership.

4.3.5. A second HNF1 site is essential for synergistic regulation of the UGT1A9 promoter by HNF1α and HNF4α

To test whether the known HNF1-binding element was responsible for the synergistic response of the *UGT1A9* promoter to HNF4 α and HNF1 α , this region was mutated using site-directed mutagenesis. Interestingly, disruption of the *UGT1A9* HNF1 site 1 element decreased the HNF1 α /HNF4 α synergism in HEK293T cells by 60% (P < 0.001) (Figure 4.6), but only reduced the HNF4 α -responsiveness of the *UGT1A9* promoter in HepG2 cells by 26% (not significant, P = 0.179) (Figure 4.4B). However, it did completely prevent HNF1 α -responsiveness of the promoter in both HepG2 and HEK293T cells, indicating that the mutagenesis was successful (Figure 4.4B, P = 0.003 and Figure 4.6, P < 0.001). Therefore, I reasoned that there was at least one additional element in the *UGT1A9* proximal promoter that was involved in its regulation by HNF1 α and HNF4 α .

Examination of the *UGT1A9* promoter DNA sequence between nucleotides -321 and -184 revealed a second possible HNF1-binding site at position -290 to -278 bp (HNF1 site 2, Figure 4.1, also mentioned in Girard *et al.* (2004)). Mutation of this site reduced the ability of HNF4 α to activate the *UGT1A9*-2kb reporter construct in HepG2 cells by nearly 90% (*P* < 0.001) whilst only decreasing the activation seen with HNF1 α alone by 35% (*P* = 0.006) (Figure 4.4B). In HEK293T cells, mutation of this putative HNF1-binding site also had a significant negative impact on the synergistic regulation of the *UGT1A9* promoter by HNF1 α and HNF4 α (*P* < 0.001), which was greater than the effect of mutating the HNF1 site 1 sequence (*P* = 0.001) (Figure 4.6). In addition, EMSA analyses established that a probe over the HNF1 site2 element was similarly capable of binding *in vitro* synthesised HNF1 α protein as the previously defined *UGT1A9* HNF1-binding site 1 (Gregory *et al.*, 2003) (Figure



Figure 4.6: Mutation of a second putative HNF1-binding site decreases the cooperative activation of the *UGT1A9* proximal promoter by HNF1a and HNF4a in HEK293T cells. Two HNF1-binding sequences in the pGL3-1A9-2k reporter vector were inactivated individually and together by site-directed mutagenesis. The resulting constructs, as well as the indicated wild-type *UGT1A7*, *UGT1A8*, *UGT1A9* and *UGT1A10* promoter-reporters (0.5 µg of each), were transfected into HEK293T cells with 25 ng of pRL-Null and 0.25 µg pCMX-HNF1a or pCMX-HNF4a2 as indicated, using empty pCMX-PL2 vector to equalise the DNA concentration between the combinations as required. Cells were harvested 48 hours posttransfection and assayed for both firefly and *renilla* luciferase activity. Results shown are the mean firefly luciferase activities relative to the internal *renilla* controls, expressed as a fold induction over the promoterless pGL3-basic vector (set at a value of 1). Error bars indicate one standard deviation. *P* values for the indicated comparisons are **P* < 0.001 and †*P* = 0.001.

4.7). Importantly, when mutations were incorporated into both newly discovered transcription factor binding sites, HNF4 site 2 and HNF1 site 2, the statistically significant residual responses to HNF4 α by either of the singly mutated promoters in HepG2 cells (Figures 4.2, P = 0.012 and 4.4B, P < 0.001) were completely abolished (P = 0.219) (Figure 4.4B). Furthermore, the 14.2-fold and 12.8-fold HNF1 α /HNF4 α synergistic responses of the 1A9-2k HNF1s1mt (P < 0.001) and 1A9-2k HNF1s2mt (P < 0.001) constructs in HEK293T cells were also inhibited by mutation of both sites in the one vector (pGL3-1A9-2k HNF1s1&s2mt). Although the residual activation of the 1A9-2k HNF1s1&s2mt promoter by co-expressed HNF1 α and



Figure 4.7: The second UGT1A9 HNF1-response element binds HNF1a in vitro. Electrophoretic mobility-shift assays were performed using 50,000 cpm of a ³²P end-labelled oligonucleotide probe encompassing the known UGT1A9 HNF1 site (HNF1s1) and *in vitro* synthesised HNF1a protein (see "Methods"). HNF1a protein was also incubated with probes containing UGT1A9 HNF1s2 or the corresponding regions of the UGT1A7, UGT1A8 and UGT1A10 promoters. For the HNF1a super-shifts, 2 µg of anti-HNF1a antibody were added to the incubations immediately following the labelled probe. Complexes were resolved on a 4% (w/v) non-denaturing polyacrylamide gel. Arrows indicate the positions of free probe and complexes containing HNF1a. Wt, wild-type; Mt, mutant. Note: the right-hand panel showing the super-shifted Wt UGT1A7 HNF1s2 probe-HNF1a complexes is vastly over-exposed relative to the left-hand panel.

HNF4 α was still significant in HEK293T cells, it was less than 2-fold (P < 0.001) (Figure 4.6).

4.3.6. Additional evidence for a direct interaction of HNF4a with UGT1A9

As mentioned in the introduction to this chapter (section 4.1.4.4), co-operative interaction between the HNF1 α and HNF4 α transcription factors on gene regulatory promoters other than UGT1A9 has been observed previously. In fact, it is becoming increasingly evident that the transcriptional relationship between HNF1 α and HNF4 α is extremely complex. Examples of genes with binding sites for both factors found to be co-operatively regulated by HNF4 α and HNF1 α include the rat *fabp1*, and human dihydrodiol dehydrogenase (DD)4 and insulin genes (Ozeki et al., 2001; Bartoov-Shifman et al., 2002; Divine et al., 2003). Yet not all promoters subject to HNF4 α /HNF1 α co-regulation are HNF1 α dependent as UGT1A9 is. The fabp1 gene promoter responds to HNF4 α in HeLa cells in the absence of HNF1 α (Rowley *et al.*, 2006), while the insulin gene is independently up-regulated by HNF4 α in HEK293 cells (Bartoov-Shifman et al., 2002). Furthermore, it has been found that it not always necessary for a promoter to possess binding sites for both HNF1a and HNF4a for regulatory interactions to occur between these two factors. For instance, HNF1a can interfere with HNF4a mediated transcription of certain promoters that contain binding sites for HNF4 α , but not HNF1, such as the upstream regulatory region of the apoCIII gene (Kritis et al., 1993) (Figure 4.4A). Of particular relevance to this project, HNF4 α has also been observed to act as a co-activator for HNF1 α on a liver pyruvate kinase promoter fragment that contains an HNF1 binding site, but no HNF4 elements (Eeckhoute et al., 2004). Further to the myriad of possible physical interactions between these two transcription factors, the HNF1 α /HNF4 α relationship is complicated by considerable cross-talk between their respective genes (Odom et

al., 2004). In liver, both transcription factors are able to positively regulate the other's expression level through interactions with response elements in their respective promoters, while HNF1 α can repress its own expression (Kuo *et al.*, 1992; Ktistaki and Talianidis, 1997; Hatzis and Talianidis, 2001).

Given that there are multiple possible transcriptional interactions between HNF4 α and HNF1 α , and that the demonstrated binding of HNF4 α to the HNF4s1 and HNF4s2 elements of the *UGT1A9* promoter in EMSA is weak and seems somewhat incongruent with the large effect that this factor has on *UGT1A9* promoter activity, it is reasonable to question whether HNF4 α genuinely affects the *UGT1A9* promoter by binding to the identified element, or through another, indirect mechanism. The latter could possibly occur through alteration of the cellular gene expression profile by HNF4 α or by over-expressed HNF4 α behaving as a co-factor for HNF1 α .

To test whether the DNA binding function of HNF4 α is necessary for its ability to regulate the *UGT1A9* promoter, two separate approaches were used. Firstly, four HNF4 α mutants were assessed for their ability to regulate *UGT1A9* – two MODY mutants and two artificial mutants. MODY mutant HNF4 α R127W has previously been shown to have a lower DNA-binding potential than wild-type HNF4 α , while MODY mutant HNF4 α E276Q has a normal affinity for the HNF4-binding site of the *HNF1\alpha* gene but has impaired function when required to perform as a co-factor for HNF1 α (Suaud *et al.*, 1999; Yang *et al.*, 2000; Eeckhoute *et al.*, 2004). The artificial mutants, HNF4 α DG69/70AA and HNF4 α RR76/77AE, were created based on homology with the oestrogen receptor β (ER β). Within the zinc-finger DNA-binding domain of the ER β , amino acid pairs 167/168 and 174/175 have been specifically shown to interact directly with DNA and can be mutated to disrupt DNA binding without affecting a known protein-protein interaction with Stat5b also mediated by this domain (Schwabe *et al.*, 1993; Bjornstrom and Sjoberg, 2002). When the zincfinger domain of HNF4 α was aligned with the same region of the ER β , the HNF4 α nuclear receptor amino acid pairs homologous with ER β positions 167/168 and 174/175 were 69/70 and 76/77 (Figure 4.8). Therefore, HNF4 α amino acid pairs 69/70 and 76/77 were mutated to match the DNA-binding-deficient ER β mutant (Bjornstrom and Sjoberg, 2002).

Α



Figure 4.8: Alignment of the oestrogen receptor β and HNF4 α DNA-binding domains. A. Linear alignment of the oestrogen receptor β (ER β) and HNF4 α DNA-binding domains using Clustal X software (Thompson *et al.*, 1997). B. Schematic illustration of the ER β DNA-binding domain adapted from Bjornstrom and Sjoberg (2002) and corresponding representation of the HNF4 α sequence. Amino acids mutated to specifically interrupt DNA-binding are marked in bold.

When the two HNF4 α 2 mutants HNF4 α R127W and HNF4 α E276Q were tested for their ability to up-regulate the *UGT1A9*-2k promoter in HepG2 cells, it was found that the activity of HNF4 α R127W was impaired by 32% (P < 0.001), but that HNF4 α E276Q was more transcriptionally competent than wild-type HNF4 α (P =0.001) (Figure 4.9A). Similarly, both mutants could co-operate with HNF1 α to

Figure 4.9: The HNF4a R127W, DG69/70AA and RR76/77AE mutants have decreased ability to up-regulate the *UGT1A9* promoter *in vitro*. A. and C. HepG2 and B. HEK293T cells were transfected in triplicate with 0.5 µg of pGL3-based reporter vector, 25 ng pRL-Null and 0.25 µg pCMX plasmids expressing wild-type or mutant HNF4a2 and/or wild-type HNF1a as described under "Methods". The DNA concentration in each transfection was kept constant by addition of empty pCMX-PL2 vector as necessary. Cells were harvested 48 hours post-transfection and assayed for both firefly and *renilla* luciferase activity. Results shown are the mean firefly luciferase activities relative to the internal *renilla* controls, expressed as a fold induction over the promoterless pGL3-basic vector (set at a value of 1). Error bars represent one standard deviation. *P* values for the indicated comparisons are **P* < 0.001, $\dagger P = 0.001$, $\ddagger P = 0.023$ and $\ddagger P = 0.039$. Also, ** pCMX-HNF4a DG69/70AA increases transcription from the *UGT1A9*-2k promoter to a greater extent than pCMX in HepG2 cells (*P* < 0.001) and #pCMX-HNF4a RR76/77AE has no effect on the *UGT1A9*-2k promoter (*P* > 0.05, not significant).



regulate the *UGT1A9* promoter in HEK293T cells, but HNF4 α R127W was less active than wild-type HNF4 α (*P* = 0.039) (Figure 4.9B). In contrast to the MODY HNF4 α mutants, HNF4 α DG69/70AA and HNF4 α RR76/77AE were almost or completely inactive against the *UGT1A9* promoter, the former only increasing reporter expression by 6% of wild-type HNF4 α (*P* = 0.001) (Figure 4.9C). These results correlate with the ability of these mutants to bind the consensus HNF4binding site probe in EMSA (Figure 4.10A) and are not due to poor expression of the HNF4 α R127W mutant in transfection (Figure 4.10C) or poor expression of any mutants in the *in vitro* synthesised samples (Figure 4.10C). Since the HNF4 α E276Q mutant is known to be defective as an HNF1 α co-factor (Eeckhoute *et al.*, 2004) but has wild-type activity towards the *UGT1A9* promoter, it is unlikely that HNF4 α behaves as an HNF1 α co-factor in *UGT1A9* regulation.

For the second investigation, transfected cells were treated with *S*-nitroso-*N*-acetyl penicillamine (SNAP), a nitric oxide donor known to decrease the ability of HNF4 α to bind DNA in HepG2 cells without causing HNF4 α degradation (de Lucas *et al.*, 2004). It was found that HNF4 α -mediated regulation of the *UGT1A9* promoter was significantly decreased relative to the solvent control (*P* < 0.001) in HepG2 cells treated with 500 μ M SNAP (Figure 4.11). Furthermore, the detrimental effect of SNAP treatment on the activation of the *UGT1A9* promoter by HNF4 α was more profound than its effect on the *apoCIII* promoter, which did not reach statistical significance (*P* = 0.086) (Figure 4.11). This result is consistent with the affinity of HNF4 α for its binding site in the *UGT1A9* promoter being significantly less than that of the *apoCIII* binding site, and therefore easier to disrupt. Taken together with the HNF4 α mutant data, these results strongly suggest that the DNA-binding function of HNF4 α is important for the *UGT1A9* response.

Figure 4.10: Expression and binding capacity of wild-type and mutant HNF4a proteins in vitro. A and B. The HNF4a R127W, DG69/70AA and RR76/77AE mutants have decreased DNA binding in EMSA. Wild-type and mutant HNF4 α proteins were generated by in vitro transcription/translation and tested for DNAbinding activity to 50,000 cpm of *apoCIII* ³²P end-labelled oligonucleotide probe as described under "Methods". Complexes were resolved on a 4% non-denaturing polyacrylamide gel; arrows indicate the positions of free probe and complexes containing HNF4 α . Note that panel B is the same experiment as panel A, but with a longer exposure. No binding of HNF4a RR76/77AE was detected against the apoCIII probe after either exposure period. C. Expression of wild-type and mutant HNF4a proteins in in vitro synthesised samples as well as in transfected HEK293T cells was confirmed by Western blot. One microlitre of in vitro translated protein mix or 25 µg of HEK293T total cell lysate was probed for the presence of HNF4 α protein by Western blot as described in Chapter 2, section 2.2.16. The same HNF4a E276Q TNT reaction was included in both blots as a crossreference.





Figure 4.11: S-nitroso-N-acetyl penicillamine (SNAP) treatment decreases the ability of HNF4a to up-regulate the *UGT1A9* promoter in HepG2 cells. HepG2 cells were transfected with 25 ng pRL-Null, 0.5 µg promoter-reporter vector (pGL3-basic, pGL3-1A9-2k or pGL3-apoCIII-810/+23) and 0.25 µg HNF4a2 expression vector according to section 4.2.8. Six hours after transfection, cells were treated with 500 µM SNAP for an additional 42 hours, and harvested by lysis. Total cell lysates were analysed for firefly and *renilla* luciferase activity, and the mean (n = 3) firefly to *renilla* luciferase ratios (relative to pGL3-basic, which is set arbitrarily to 1) are reported. Error bars represent one standard deviation, and *P* values for the indicated comparisons are **P* < 0.001 and #*P* > 0.05 (not significant).

Since it seems that the ability of HNF4 α to bind DNA is required for regulation of the *UGT1A9* promoter, the remaining question is whether HNF4 α binds the *UGT1A9* promoter directly, or up-regulates the gene promoter of another factor that subsequently binds the region labelled *UGT1A9* HNF4s2. There are several reasons to believe that the former is the most likely scenario. Firstly, chromatin immunoprecipitation experiments performed by Barbier *et al.* (2005) show that the *UGT1A9* promoter can be enriched for by immunoprecipitation using anti-HNF4 α

antibody. Secondly, the interactions observed in this chapter in HepG2 cells were confirmed in HEK293T cells. HEK293 cells have no endogenous HNF1 or HNF4a expression (Bernard et al., 1999; Maeda et al., 2002) (Figure 4.3B), and overexpression of HNF4 α in HEK293 cells only up-regulates 57 of the 18,400 genes represented by the Affymetrix HG U133A DNA oligonucleotide microarray. Only 3 of these 57 up-regulated genes are known to be transcriptional regulators (Lucas et al., 2005). Thirdly, mutation of the UGT1A9 HNF4s2 element to the consensus HNF4-binding sequence either did not change (Figure 4.4A) or strengthened (Figure 4.2) the HNF4 α -response, while mutating this element away from the HNF4-binding site consensus abolished the observed phenomenon (Figure 4.2). Thus the factor(s) that bind to the UGT1A9 HNF4s2 region have a preference for HNF4-binding sitelike elements. Given that the HEK293 transcriptome is hardly affected by HNF4 α over-expression (Lucas et al., 2005), and that the factor that binds the HNF4s2 region prefers HNF4-like sites and co-operates with HNF1 α , the likelihood that HNF4 α is not interacting directly with the UGT1A9 promoter seems slim. Finally, apparent lack of, or weak binding of a factor under EMSA conditions is not always a good reflection of in vivo binding and function (Mitchelmore et al., 2000). If the HNF4 α -UGT1A9 interaction is stabilised by other factors brought into close proximity by other response elements surrounding HNF4s2 (*i.e.* interactions not replicable in an EMSA reaction), this would account for the apparent discrepancy between the EMSA and transfection data. Thus, in the absence of evidence to the contrary, it seems highly likely that HNF4 α activates the UGT1A9 promoter through direct binding, the most common mechanism for HNF4a-mediated transcriptional activation. This hypothesis could be tested further using *in vitro* chromatin assays, where naked DNA templates are assembled into chromatin and transcribed in vitro.

If addition of HNF4 α to such a system, where it cannot alter the expression of any other components, also improves transcription from the *UGT1A9* promoter, then direct interaction with the *UGT1A9* promoter would be strongly implicated.

4.3.7. Nucleotide substitutions in the second HNF4 and HNF1 sites may partly explain the uniqueness of the *UGT1A9* promoter response to HNF4α

As the original UGT1A9 HNF4 α -binding element (HNF4 site 1) was insufficient to explain the differing responses of the UGT1A7-IA10 cluster to human HNF4 α in my hands, I examined whether the second HNF1 and HNF4 α -response elements could be responsible. When the appropriate regions of the UGT1A7-IA10 promoters were aligned it was evident that there was at least one nucleotide substitution in each of the three non-responsive promoters that decreased the homology to the HNF1 α consensus binding site GTTAATNATTAAC (Tronche *et al.*, 1997). Likewise, the UGT1A9 HNF4s2 region was not completely conserved in any of the three non-responsive promoters (Figure 4.1). Of particular interest was that the UGT1A9 sequence within the core region of the HNF4 α -binding sequence.

Substitution of the second UGT1A9 HNF1 α site with the corresponding UGT1A8 sequence had a similar effect on promoter responsiveness as the mutation designed to totally abolish HNF1 α binding. The UGT1A9-2kb HNF1s2mt8 promoter construct could only be activated by HNF4 α in HepG2 cells to approximately 7-fold; considerably less than the 34-fold increase in reporter activity obtained from the wild-type promoter under the same conditions (P < 0.001) (Figure 4.4B). In addition, binding of *in vitro* synthesised HNF1 α to UGT1A8 or UGT1A10 probes corresponding to the second HNF1 α -response element of the UGT1A9 promoter could not be demonstrated (Figure 4.7). Furthermore, an equivalent probe from the

UGT1A7 promoter bound only a small fraction of the HNF1 α protein retarded by either of the UGT1A9 HNF1-binding site probes, which was confirmed to be HNF1 α by super-shift (Figure 4.7).

Exchange of the *UGT1A9*-HNF4s2 element for the equivalent *UGT1A8* sequence was also extremely effective in reducing the HNF4 α -responsiveness of the *UGT1A9* promoter. The very low basal activity of the pGL3-1A9-2kb HNF4s2mt8 reporter vector could only be increased 3-fold by over-expression of HNF4 α 2 (P < 0.001) (Figure 4.2). In addition, binding of HNF4 α to the *UGT1A8* HNF4s2 probe in EMSA could not be demonstrated (Figure 4.3A).

To investigate whether the *UGT1A9* HNF1s2 and HNF4s2 elements were sufficient to confer HNF4 α -responsiveness to the *UGT1A8* promoter, the relevant *UGT1A9* nucleotides were introduced into the pGL3-1A8-2kb construct by site-directed mutagenesis. The HNF4s2 element was introduced into the *UGT1A8* promoter alone, and in combination with the HNF1s2 element. However, neither construct could respond positively to over-expressed HNF4 α when co-transfected into HepG2 cells. Conversely, the pGL3-1A8-2k reporter vectors carrying the HNF1s2mt9 and HNF4s2mt9 sequence changes were repressed by HNF4 α over-expression (P < 0.001and P = 0.002, respectively), while the double mutant was unaffected (P = 0.134) (Figure 4.12). This result shows that at least three elements of the *UGT1A9* promoter are involved in its unique regulation by HNF4 α .

4.3.8. The UGT1A9 initiator-like region contributes to its HNF4α-response

Since it is evident that additional, unidentified differences between the UGT1A9 and UGT1A8 promoters control whether these genes respond to HNF4 α or not, I began the search for extra transcriptional elements that may be involved. In the past, my



Figure 4.12: Introduction of the *UGT1A9* HNF1s2 and HNF4s2 elements into the *UGT1A8* promoter does not confer HNF4a-responsiveness to pGL3-1A8-2k. The *UGT1A9* HNF1s2 and HNF4s2 sequences were substituted into the *UGT1A8*-2k promoter by site-directed mutagenesis. HepG2 cells were then transfected in triplicate with 0.5 µg each promoter-reporter vector, with 25 ng pRL-Null and 0.25 µg pCMX-PL2 or pCMX-HNF4a2 as indicated. Transfections were harvested after 48 hours and analysed for firefly luciferase and *renilla* luciferase activities. Results are presented as the mean luciferase:*renilla* activity ratio, relative to pGL3-basic (set at one). The error bars indicate one standard deviation. *P* values for the indicated comparisons are **P* < 0.001, ††*P* = 0.002 and #*P* > 0.05 (not significant).

colleagues and I have established that the *UGT1A8* and *UGT1A9* genes are differentially regulated through their Inr-like regions (Gregory *et al.*, 2003); therefore, I hypothesised that LETFs bound to the *UGT1A9* HNF4s2 and HNF1s2 sites can only influence transcription when combined with proteins that specifically bind the *UGT1A9* Inr-like sequence. Using the *UGT1A9*-1kb promoter-reporter constructs available from our published study (Gregory *et al.*, 2003), it was found that altering nucleotide -59 to the corresponding *UGT1A8* sequence (*UGT1A9*-1k A-59G) decreased the *UGT1A9* response to HNF4 α (*P* = 0.013) (Figure 4.13). This mutation has been shown to allow more Sp1-containing protein complexes to bind to



Figure 4.13: The Inr-like region of *UGT1A9* is more supportive of HNF4 α -responsiveness than the equivalent region of *UGT1A8*. HepG2 cells were transfected with 0.5 µg reporter constructs carrying the proximal kilobase of the *UGT1A9* promoter, 25 ng pRL-Null and 0.25 µg empty pCMX-PL2 or pCMX-HNF4 α 2 expression plasmid. The HNF4 α -response of the wild-type promoter was compared to the response of similar constructs carrying *UGT1A8* nucleotides at promoter positions -59, -62 or both. All results are the mean ratios of the firefly to *renilla* luciferase activity of three replicates, presented relative to the pGL3-basic control. The standard deviation of each triplicate is indicated by the error bars. NB: This experiment was only performed once. *P* values for the comparisons indicated are $\ddagger P = 0.013$ and #P > 0.05 (not significant).

the Inr-like region than occurs over the wild-type site (Gregory *et al.*, 2003). Mutation of the other nucleotide difference that was important for *UGT1A8* and *UGT1A10* activity (at position -62), had no effect on the HNF4 α -responsiveness of the *UGT1A9* promoter alone (P = 0.876) (Figure 4.13) or when used in combination with the A-59G mutation (P = 0.060). Although these results show that the *UGT1A9* Inr-like region is optimised for interaction with HNF4 α relative to the equivalent *UGT1A8* sequence, there is no absolute requirement for this nucleotide combination in the *UGT1A9* HNF4 α -responsiveness. Thus, the nucleotide difference(s) between the *UGT1A8* and *UGT1A9* promoters that prevent *UGT1A8* from being HNF4 α -responsive, other than those that reside within HNF1s2 and HNF4s2, remain unknown. Clearly, further work is required to fully elucidate the mechanisms that drive the hepatic expression of *UGT1A9*.

4.3.9. HNF1β can support activation of the UGT1A9 promoter by HNF4α

Although encoded by separate genes, HNF1 α and HNF1 β share highly homologous dimerisation and DNA binding domains. As a consequence, these factors readily heterodimerise and recognise the same DNA response elements (Rey-Campos et al., 1991). Numerous genes that are regulated by HNF1 α have also been shown to be responsive to HNF1B, albeit in many cases, to a lesser extent. However, I was unable to find any reports that demonstrated that HNF1 β is able to co-operate in any way with HNF4 α to increase gene transcription. Therefore, the ability of HNF1 β to synergistically up-regulate the UGT1A9 promoter in conjunction with HNF4a was investigated. When HEK293T cells were co-transfected with the UGT1A9-2kb reporter construct and an HNF1 β expression vector, the UGT1A9 promoter was activated 3.6-fold (P < 0.001), approximately half the increase achieved by HNF1a (Figure 4.14A). In addition, a synergistic response was observed between HNF1β and HNF4 α ; while these factors were individually able to increase reporter expression by 3.6-fold (P < 0.001) and 1.3-fold (P = 0.005) respectively in HEK293T cells, a combined response of 15-fold (P < 0.001) was obtained. Furthermore, HNF1 β did not interfere with the HNF4 α -response of the UGT1A9 promoter in HepG2 cells. Over-expression of HNF1ß and HNF4a in HepG2 cells gave 123% of the UGT1A9 promoter response observed with HNF4 α alone (not significantly different, P = 0.411) (Figure 4.14B). Conversely, a truncated HNF1 α

Figure 4.14: HNF1^β facilitates regulation of the UGT1A9 proximal promoter by HNF4a. A and C. HEK293T or B. HepG2 cells were transfected in triplicate with 0.5 µg pGL3-basic, pGL3-UGT1A9-2k or pGL3-1A9-2k HNF1s1&s2mt reporter vectors, 25 ng pRL-Null and the appropriate combinations of 0.25 µg pCMX-HNF4a2, pCMX-HNF1a, pCMX-HNF1B or pCMX-HNF1a 546X. All methods are described in section 4.2 and the DNA concentration in each transfection was standardised by addition of empty pCMX vector as necessary. Transfected cells were harvested 48 hours post-transfection and assayed for firefly and renilla luciferase activity. The results are presented as the mean firefly: renilla luciferase ratio, relative to the pGL3-basic control which is set to 1. Errors bars indicate one standard deviation. P values for the indicated comparisons are: *P < 0.001, $\dagger \dagger P = 0.002$ and #P > 0.05 (not significant). Also, activity of ** marked transfections differs from pGL3-1A9-2kb/pCMX co-transfections (P < 0.001). ¥Activity of the pGL3-1A9-2k + pCMX-HNF1α 546X co-transfection differs from the pGL3-basic + pCMX-HNF1α 546X, pGL3-1A9-2k HNF1s1&2mt + pCMX-HNF1α 546X and pGL3-1A9-2k + pCMX-HNF1 α co-transfections (P < 0.001 for all three comparisons).



construct, HNF1 α 546X that possesses DNA-binding and dimerisation domains but no activation domain I, and is not able to co-operate strongly with HNF4 α on the *UGT1A9* promoter in HEK293T cells (Figure 4.14C), significantly inhibits the HNF4 α -response of the *UGT1A9* promoter in HepG2 cells (Figure 4.14B) (*P* = 0.002). Therefore, it seems likely that endogenous HNF1 α /exogenous HNF1 β heterodimers and/or HNF1 β homodimers are also capable of interacting with HNF4 α to synergistically activate the *UGT1A9* promoter in HepG2 cells.

This work shows for the first time that HNF1 β is able to synergistically up-regulate a promoter in conjunction with HNF4 α . In contrast to the interactions between HNF1 α and HNF4 α , the functional relationship between HNF4 α and HNF1 β has had little attention. The few studies that have investigated interactions of HNF1 β with HNF4 α have shown that HNF1 β either has no role in co-operative initiation of transcription with HNF4 α (Hu and Perlmutter, 1999; Hatzis and Talianidis, 2001), or is detrimental to HNF4 α -mediated promoter activity, even though HNF1 α exhibits synergy (Bartoov-Shifman *et al.*, 2002). Although not well expressed in the adult liver, HNF1 β binds to HNF1 sites with an affinity equal to that of HNF1 α , and is highly expressed in the kidney (Rey-Campos *et al.*, 1991), an organ that also contains high levels of UGT1A9. These findings that HNF1 β can co-operatively regulate the *UGT1A9* promoter with HNF4 α in HEK293T cells, and does not interfere with HNF4 α -mediated regulation in cells that express HNF1 α are consistent with HNF4 α also having a co-operative role in renal expression of *UGT1A9*.

4.3.10. Differences between published work and this study are not explained by usage of HNF4α1 or the *UGT1A9* T-275A polymorphism

In the course of this investigation, several important mechanistic differences between this work and that previously published (Barbier *et al.*, 2005) were found. In my hands, the HNF4 site 1 element was not necessary for the response of the UGT1A9 promoter to HNF4 α , and a much greater response of the UGT1A9 promoter to HNF4 α was also observed. This was despite the experimental systems in the two studies being very similar, using the same cell line and the same reporter vector system.

Several possible explanations exist for the conflicting results of the two studies. Firstly, differences in the HepG2 cells used or culture conditions could be to blame. Alternatively, the use of different HNF4 α isoforms could be responsible, as the authors did not specify the origin or identity of their HNF4 α cDNA. In this study the most abundant HNF4 α transcript from the human liver, HNF4 α 2, was used, although much of the literature describes studies done with the rat HNF4 α 1 splice variant, now known to exhibit lower activity than HNF4 α 2 in some circumstances (Sladek *et al.*, 1999). A final possibility is that the work of Barbier *et al.* (2005) was performed with a *UGT1A9* promoter clone that possesses one or more polymorphic differences relative to our construct, resulting in the changed response. Since the prospect of different HNF4 α isoforms or *UGT1A9* alleles resulting in an alternative outcome has interesting ramifications for understanding interindividual variation in UGT1A9 expression, it was decided to investigate these avenues further.

To explore whether other HNF4 α isoforms commonly used in other regulatory studies performed similarly to the human HNF4 α 2 variant used in my work, human HNF4 α 1 and rat HNF4 α 1 were chosen and cloned into pCMX-PL2. When these HNF4 α expression vectors were tested for their ability to regulate the *UGT1A9* promoter in HepG2 cells, it was found that both human and rat HNF4 α 1 were similar to human HNF4 α 2 (P = 0.893 and P = 0.252 respectively) in their ability to activate the *UGT1A9* promoter (Figure 4.15A and B). In addition, in HEK293T cells, the rat



Figure 4.15: Comparison of human HNF4 α 2, human HNF4 α 1 and rat HNF4 α 1 pCMX expression vectors for effectiveness against the *UGT1A9* promoter. Human HNF4 α 1 and rat HNF4 α 1 variants were cloned into pCMX-PL2 and compared with human HNF4 α 2 for their ability to regulate the pGL3-1A9-2k reporter construct. A and B. HepG2 cells and C. HEK293T cells were transfected in triplicate with 0.5 µg pGL3-basic or pGL3-1A9-2k, 25 ng pRL-Null and 0.25 µg each HNF expression vector as appropriate. The DNA concentration in each transfection was kept constant by addition of empty pCMX-PL2 vector as necessary. Forty eight hours post-transfection, total cell lysates were assayed for firefly and *renilla* luciferase activities. Each column represents the mean firefly:*renilla* ratio achieved for each treatment, relative to the pGL3-basic control, which is set to 1. Error bars represent one standard deviation and the *P* values for each indicated statistical comparison are $\dagger \dagger \dagger P = 0.003$ and #P > 0.05 (not significant).

HNF4 α 1 expression construct had greater activity than the original human HNF4 α 2

expression vector against pGL3-1A9-2k (P = 0.003) (Figure 4.15C). Thus, the

difference in activity between the human HNF4 α 2 and rat HNF4 α 1 constructs was

insufficient to explain the discrepancies between my results and those previously published. Since the rat HNF4 α 1 cDNA was obtained from one of the authors of the conflicting paper, and is presumably the variant used in their work, it was concluded that investigation of other possible HNF4 α variants was unlikely be meaningful, and I therefore next considered the possibility that *UGT1A9* genotype may be the determining factor.

In humans, the hepatic concentration of UGT1A9 has been correlated with promoter genotype by Girard et al. (2004). Furthermore, in addition to the obvious differences between the HNF4 α -reponses of Dr. Barbier's and my UGT1A9 promoter constructs, I noticed that the basal activity of my UGT1A9 promoter construct also differed from that of Barbier et al. (2005). In contrast to the results of Barbier and co-workers, which show no basal activity for the UGT1A9 promoter in HepG2 cells above that of the empty reporter vector (Barbier et al., 2005), my 2 kb, 321 bp and 184 bp UGT1A9 promoters increased luciferase expression over background by 2 to 5-fold $(P \le 0.008$, see Figure 4.2). Therefore, I hypothesised that promoter polymorphisms may account for the different promoter behaviours. In the work by Girard et al. (2004), polymorphisms at positions -275, -331/-440, -665 and -2152 predicted hepatic UGT1A9 expression. Of these, I considered the deoxythymidine to deoxyadenosine substitution at nucleotide -275 the most likely candidate, as it was: a) one of the SNPs most strongly associated with gene expression (Girard et al., 2004); b) the only one to be covered by the UGTIA9-321bp reporter construct; and c) positioned only three nucleotides downstream of the HNF1s2 element required for the HNF4α-response. If the nucleotide substitution at this position creates/strengthens a binding site for a transcription factor that restricts access of HNF1 factors to the HNF1s2 element, it was conceivable that this could adversely

affect the HNF4 α -response of the *UGT1A9* promoter. Therefore, deoxyadenosine was substituted into the *UGT1A9*-2k promoter at position -275, and the alternative sequence tested for any effects on basal or inducible activity.

In HepG2 cells, it was found that the UGT1A9-2k promoter had the same basal activity regardless whether the -275 base was thymine or adenine (P = 0.438) (Figure 4.16). Therefore, this base substitution alone cannot explain why Dr. Barbier's UGT1A9 promoter constructs have no basal activity (Barbier et al., 2005). Interestingly however, the HNF4 α 2-response of the UGT1A9-2k T-275A construct was only 55% that of the reference promoter (P = 0.001) (Figure 4.16). Again, this altered response is insufficient to explain the differences between my results and that of Barbier and colleagues. Nonetheless, the reduced HNF4α-responsiveness of the UGT1A9-2k T-275A construct is of interest. This polymorphism occurs in Caucasian populations at a frequency of 0.06 to 0.07, and is associated with a higher level of UGT1A9 expression in the liver (Girard et al., 2004). Yet, it was also noted by the authors that this polymorphism is in strong association with another SNP (-2152) and that it is impossible to differentiate the causative effect of these two polymorphisms on UGT1A9 levels. Thus, it may transpire that: a) the -2152 SNP can overcompensate for any detrimental effect of the -275 SNP on HNF4a-mediated regulation; b) in cells that express high levels of UGT1A9 (HepG2 cells have relatively poor UGT1A9 expression: see Chapter 5) other available factors make the HNF1s2 site less vital for HNF4 α -mediated regulation; and/or c) HNF4 α is not as important in the *in vivo* expression of UGT1A9 as Barbier, myself and our respective colleagues have postulated. It will be of interest to further investigate the relationship between the function of the UGT1A9 promoter in vivo versus in vitro, its cooperative regulation by HNF4 α and HNF1 α , and any involvement of these two SNPs.



Figure 4.16: The UGT1A9 -275 SNP decreases stimulation of the UGT1A9 promoter by HNF4a2 in vitro. The UGT1A9 T to A SNP at promoter position -275 was introduced into the UGT1A9-2k promoter by site-directed mutagenesis as described in "Methods". HepG2 cells were co-transfected with 25 ng pRL-Null, 0.25 μ g pCMX-PL2 or pCMX-HNF4a2 and 0.5 μ g pGL3-basic, pGL3-1A9-2k or pGL3-1A9-2k T-275A, and analysed 48 hours later for firefly and *renilla* luciferase activity. The mean (n = 3) firefly activity of each experimental group, relative to the internal *renilla* control and expressed as a proportion of basal pGL3-basic activity (set to 1), is presented. Error bars indicate one standard deviation. The *P* value for the indicated statistical comparison is $\dagger P = 0.001$. NB: This experiment was performed only once.

4.4 General discussion and summary

4.4.1. Achievement of aims

It has long been recognised that UGT1A9 is the only hepatically expressed enzyme of the UGT1A7-1A10 gene cluster. Barbier *et al.* (2005) recently showed that, while the human UGT1A9 promoter is governed by HNF4 α , similar regions of the UGT1A7, UGT1A8 and UGT1A10 promoters are unaffected by HNF4 α over-

expression. Since the four promoters are highly conserved, it was suggested that HNF4 α may contribute to the unique liver-specific expression of UGT1A9, either independently or in conjunction with other transcription factors. The aims of this work were to further investigate the relationship between HNF4 α and UGT1A9 promoter activity, and to identify key elements that distinguish UGT1A9 from the remaining UGT1A7-1A10 genes. Accordingly, a major element through which HNF4 α interacts with the UGT1A9 promoter was identified. It was also shown that the HNF4 α -response of the UGT1A9 promoter is completely dependent on the presence of HNF1 factors, and that there are at least three major functional differences between the UGT1A8 and UGT1A9 promoters that allow HNF1 and HNF4 α to co-operatively up-regulate only the latter in hepatocyte-derived cells. Additional elements specific to the UGT1A9 promoter, such as the HNF4s1 and the Inr-like regions also support the UGT1A9 promoter response.

4.4.2. Future directions

This work has helped identify several possibilities for further investigation into the function of the UGT1A7-1A10 gene promoters. Firstly, the nature of the HNF4 α -HNF1 α interaction on UGT1A9 remains largely uncharacterised. Synergistic regulation by HNF4 α and HNF1 α operates on a number of hepatic genes in humans and rodents; yet it is clear that there is not just one standard mechanism by which this occurs. As previously highlighted, HNF4 α -HNF1 α synergy is not necessarily coincident with HNF1 α -dependence of HNF4 α -mediated activation, or the presence of HNF4-binding sites. Furthermore, different HNF4 α and HNF1 α mutations have different effects in different systems. The HNF4 α E276Q mutant is known to have decreased physical interaction with HNF1 α (Eeckhoute *et al.*, 2004), yet is at least as active as wild-type on the UGT1A9 promoter. On the other hand, HNF4 α R127W has

wild-type-like activity on some promoters, but not all (Navas *et al.*, 1999; Lausen *et al.*, 2000; Yang *et al.*, 2000), including *UGT1A9*. Indeed, the only human gene that I can ascertain as having a similar response profile to HNF4 α and HNF1 α as *UGT1A9* encodes DD4, which interestingly is another biotransformation enzyme. DD4 is involved in the bioactivation of polycyclic aromatic hydrocarbons and metabolism of pharmaceutical drugs (Ozeki *et al.*, 2001). However, very little has been reported on the mechanistic aspect of DD4 gene regulation by HNF4 α and HNF1 α , and the HNF4 and HNF1-binding sites of the DD4 and *UGT1A9* are very differently located (relative to each other and the TSS) - so it is reasonable to expect that detailed investigations will uncover regulatory differences between these genes. Such mechanistic differences are important for understanding how genes that are apparently controlled by the same factors are independently regulated and not affected in a blanket manner by environmental and genetic influences.

One of the ways that genes regulated by the same factors are thought to be individually controlled is through recruitment of specific co-factor combinations (Torres-Padilla and Weiss, 2003). It may transpire that the interaction between HNF4 α and HNF1 α on the *UGT1A9* promoter is mediated by co-factor bridges between these two transcription factors, as direct interaction seems to be insufficient to explain the observed synergy. Not only is the HNF4 α E276Q mutant (with decreased direct interaction with HNF1 α) 100% active towards the *UGT1A9* promoter, the less active HNF4 α R127W mutant was found to be able to bind HNF1 α at least as well as wild-type HNF4 α (Rowley *et al.*, 2006), and the HNF1 α 546X mutant cannot co-operate with HNF1 α 546X, possesses all of the domains currently known to interact directly with HNF4 α (Ktistaki and Talianidis,

1997; Rowley *et al.*, 2006). Further investigation into the transcriptional relationship between HNF4 α and HNF1 α on the *UGT1A9* promoter compared to other similarly regulated promoters will allow a better understanding of how they remain independently regulated, and which changes in conditions are likely to most greatly affect which genes.

Another aspect of the role of HNF4 α and HNF1 α in the regulation of *UGT1A9* requiring further attention is its importance *in vivo*. Approaches that could be taken include the use of a cell line/primary cells that express high levels of UGT1A9, or extracts from liver tissue. In the former, it would be hoped that the *UGT1A9*-2k promoter would have much higher basal activity than in HepG2 cells, making it possible to determine the role of the HNF4s2 and HNF1s2 elements in the basal activity of the *UGT1A9* promoter by mutation. This is an ideal complementary method to over-expression of transcription factors, as it does not rely on changes to the cellular environment to produce results, and thus removes one level of possible artefacts. In the latter, it would be possible to measure the levels of HNF4 α and HNF1 α mRNA or protein expression and investigate whether levels of either factor correlate with the expression of UGT1A9.

Apart from further characterisation of the HNF4 α -HNF1 α -UGT1A9 relationship, there are several other worthwhile avenues for investigation of UGT1A9 proximal promoter function highlighted by this work. Of particular interest would be studies that continue to elucidate the functional differences between the UGT1A7, UGT1A8, UGT1A9 and UGT1A10 promoters. As discussed earlier, the HNF4s2 and HNF1s2 sites are not present in UGT1A8, but when introduced, are still insufficient to allow HNF4 α -mediated regulation of the UGT1A8 promoter. This observation may be the result of either a third important UGT1A9 element, or active repression of the UGT1A8 promoter. In addition, whilst the UGT1A7 HNF1s2 is much weaker than the corresponding UGT1A9 element, it can still interact with HNF1 α to some extent, and the UGT1A7 HNF4s2 element does not have a nucleotide substitution in the core binding region as is the case for UGT1A8 and UGT1A10 (Figure 4.1). Therefore, it will be of interest to see whether either UGT1A7 sequence can function in the context of the UGT1A9 promoter. If so, unidentified important element(s) of UGT1A9 may turn out to be the "master switch" determining the uniqueness of the UGT1A9 HNF4 α -response, and may be of relatively high importance. Like the HNF1s2 site, this putative element may not appear to have any function when not required for the HNF4 α -response, so may be easier to find using cells that have high basal UGT1A9 expression.

Another interesting difference between the UGT1A7-IA10 proximal promoters that could be pursued is the nature of their HNF1 α -only-responses. This is of interest because UGT1A7, UGT1A8 and UGT1A10 all share the same HNF1s1, which is one nucleotide different to the corresponding site of UGT1A9 (Figure 4.1); yet these promoters clearly fall into two groups when classified by HNF1 α -only-response (Figure 4.6). When HNF1 α is over-expressed in HEK293T cells, the UGT1A7 and UGT1A9 promoters respond similarly, while UGT1A8 and UGT1A10 do not respond at all. Mutation of UGT1A9 HNF1s1 abolishes the HNF1 α -response of this promoter. Presumably the HNF1 α -response of the UGT1A7 and UGT1A9 promoters is reliant on an unidentified factor that does not interact with the UGT1A8 and UGT1A10genes, at least in the same way.

Further work also needs to be done to understand the effect of genotype on UGT1A9 expression, and whether this is at all related to the observed HNF4 α /HNF1 α regulation. The *UGT1A9* T-275A SNP seems to decrease the HNF4 α -response of the
UGT1A9 promoter, even though it is associated with higher UGT1A9 protein concentration. However, whilst this is counterintuitive, as previously discussed there are several possible explanations for such a result. To see whether the T-275A nucleotide substitution does adversely affect HNF1 α binding to HNF1s2, an EMSA using a probe that encompasses both the HNF1s2 and -275 regions could be used to see whether there is an unidentified factor that can competitively exclude HNF1 α by binding over -275 and surrounding nucleotides.

Finally, it would be of interest to determine whether HNF4 γ can also participate in the regulation of *UGT1A9*, either as a substitute for HNF4 α , or as a competitor. The most similarly regulated human gene known, *DD4*, can be regulated by either HNF4 α or HNF4 γ (Ozeki *et al.*, 2001). Furthermore, use of HepG2 extracts in EMSA showed protein complexes formed on the HNF4-binding site of *DD4* contained HNF4 γ and had the same apparent mobility as those with HNF4 α . Presumably the complexes on the probe were largely, but not exclusively HNF4 α -HNF4 γ heterodimers, as antibody against HNF4 α super-shifted all complexes, but antibody against HNF4 γ only super-shifted a majority of the retarded probe (Ozeki *et al.*, 2001). Thus, it is possible that the complexes that are present on the *UGT1A9* HNF4-site probes in the Figure 4.3 EMSA also contain HNF4 γ .

4.4.3. Relevance to glucuronidation in humans

Although a major detoxification pathway, glucuronidation in humans is subject to considerable interindividual variation, even within organs where UGT expression is considered to be both strong and constitutive. Variable UGT activity between independent liver samples has been noted for numerous chemicals (Court *et al.*, 2001; Wiener *et al.*, 2004; Girard *et al.*, 2005) and quantitative PCR techniques have

shown that this may be partly due to altered levels of UGT transcripts (Congiu *et al.*, 2002). An increasing number of studies suggest that such interindividual disparities may have important clinical implications, determining personal therapeutic outcome or disease development in certain contexts (Ramirez *et al.*, 2002; Girard *et al.*, 2004; Kuypers *et al.*, 2005). Many of the known substrates of UGT1A9 are toxins, carcinogens/procarcinogens or pharmaceutical agents with relatively narrow therapeutic windows (see section 4.1.2), thus interindividual variation in UGT1A9 activity or expression may be a determinant of drug toxicity/efficacy or a risk factor for developing cancer (Albert *et al.*, 1999; Ren *et al.*, 2000; Malfatti and Felton, 2001; Gagne *et al.*, 2002; Ramirez *et al.*, 2002; Bernard and Guillemette, 2004). Furthermore, co-expression of UGT1A9 with other UGT1A family members alters the glucuronidation kinetics of substrates that are specific for the latter (Fujiwara *et al.*, 2007b). Thus, changes in the expression levels of UGT1A9 relative to other UGT1As may also affect the clearance and detoxification of substances that are not UGT1A9 substrates.

Interindividual differences in UGT1A9 expression, particularly in the liver, are well established (Court *et al.*, 2001; Congiu *et al.*, 2002; Bernard and Guillemette, 2004; Girard *et al.*, 2004). However, the mechanisms that give rise to the observed variability are largely uncharacterised. A number of *UGT1A9* allelic variants have been discovered (Girard *et al.*, 2004), but those that are strongly associated with altered UGT1A9 protein levels are relatively rare and insufficient to fully explain the variability observed. Apart from genetic variation in the target gene, interindividual variation in expression can arise through variation in the levels or activity of important transcription factors or their co-factors. In turn, this variation can arise from genetic diversity in transcription factor gene promoters or coding regions, in

exposure to stimuli that alter the activity of transcription factors, or variation in the transcription factors that control expression of the transcription factors (a neverending interconnected web of possibilities). Functional genetic variants of HNF4 α are known. Rare mutations, such as HNF4a R127W and HNF4a E276Q, are associated with MODY1 (Ryffel, 2001), while several common variants have been postulated to be risk factors for type 2 diabetes mellitus, high serum lipid levels and metabolic syndrome (Love-Gregory et al., 2004; Silander et al., 2004; Weedon et al., 2004; Ek et al., 2006; Hara et al., 2006; Weissglas-Volkov et al., 2006; Lehman et al., 2007). Similarly, several rare mutations in the HNF1 α and HNF1 β genes have been associated with MODY3 and MODY5 respectively (Ryffel, 2001), while common HNF1 α variants have been associated with type 2 diabetes mellitus, insulin resistance, insulin and serum C-peptide secretion during oral glucose tolerance tests, and high-density lipoprotein cholesterol levels (Urhammer et al., 1997; Urhammer et al., 1998a; Chiu et al., 2000; Babaya et al., 2003; Chiu et al., 2003; Holmkvist et al., 2006). Since common functional variants exist, it is feasible that HNF4 α or HNF1 α variants (coding or regulatory region) may partly determine the level of UGT1A9 expression in human liver.

In addition, there are a number of stimuli and conditions that have been associated with altered HNF4 α levels or activity. Negative regulators of HNF4 α levels and/or activity include: polyunsaturated fatty acyl-CoAs, the levels of which can be modulated in the liver by diet (Hertz *et al.*, 1998; Kalderon *et al.*, 2002); acyl-CoA hypolipodemic pharmaceuticals such as fibrates (interestingly, these are also UGT1A9 substrates) (Hertz *et al.*, 2001; Kalderon *et al.*, 2002); bile acids such as CDCA (Popowski *et al.*, 2005; Li *et al.*, 2006); the inflammatory mediator nitric oxide, and cytokines IL-1 β and transforming growth factor β 1 (de Lucas *et al.*, 2004;

Li *et al.*, 2006); pharmaceuticals that are PXR ligands (for example rifampicinliganded PXR competes with HNF4 α for the co-activator PGC-1) (Bhalla *et al.*, 2004); hypoxia (Mazure *et al.*, 2001); and phosphorylation (Viollet *et al.*, 1997; Leclerc *et al.*, 2001). Notably, phosphorylation of HNF4 α is increased by fasting (Viollet *et al.*, 1997). Known positive regulators of HNF4 α levels and/or activity include: hepatic saturated fatty acyl-CoAs of length C14-C16, the levels of which again are related to diet (Hertz *et al.*, 1998; Kalderon *et al.*, 2002); differentiation, particularly in intestinal cells as they move from the crypt to villus (Stegmann *et al.*, 2006); the transcription factor HNF1 β (Wang *et al.*, 2004a), hepatitis C virus (HCV) infection (Qadri *et al.*, 2006); and increasing availability of acyl-CoA-binding protein (Petrescu *et al.*, 2003).

A number of the stimuli and conditions that affect HNF4 α also influence HNF1 α . For instance, HCV infection increases HNF1 α expression (Qadri *et al.*, 2006), as does differentiation in Caco-2 enterocytes (Hu and Perlmutter, 1999); while the inflammatory cytokine IL-1 β (Geier *et al.*, 2003) and bile acids such as CDCA adversely affect HNF1 α (Jung and Kullak-Ublick, 2003). The latter occurs through repression of HNF4 α -mediated activation of the *HNF1\alpha* gene promoter (Jung and Kullak-Ublick, 2003). Ceramide also reduces available active HNF1 α protein (Park *et al.*, 2004a), while lipopolysaccharide (Roe *et al.*, 2001) and tumour necrosis factor α (Geier *et al.*, 2003) decrease HNF1 α -mediated transcription in treated cells.

Whether any of the listed factors affect UGT1A9 expression through modulation of HNF4 α and HNF1 α -mediated regulation remains unknown; however, this will depend on whether any affect HNF4 α activity by mechanisms relevant to *UGT1A9* transcription, and what effect they have on other transcription factors that also control the *UGT1A9* promoter. For example, the inhibition of HNF4 α by CDCA and

IL-1 β is mediated through disrupted recruitment of PGC-1 (Li *et al.*, 2006); so if PGC-1 is important in HNF4 α -mediated regulation of *UGT1A9*, then UGT1A9 expression may be affected by changes in CDCA and IL-1 β levels. Also, alternative cell types may be differentially affected by each stimulus, as genes expressed in more than one cell type are not necessarily driven by identical transcriptional complexes in each (Schaeffer *et al.*, 1993; Navalon-Garcia *et al.*, 2006).

Interesting possibilities for determinants of UGT1A9 expression via the HNF4 α /HNF1 α interaction include fasting and inflammation. Fasting not only decreases HNF4 α activity, but also increases the risk of hepatotoxicity after paracetamol ingestion (Whitcomb and Block, 1994). Fasting depletes the availability of glutathione and UDP-glucuronic acid, required for the oxidation and glucuronidation of paracetamol (Whitcomb and Block, 1994; Zimmerman and Maddrey, 1995). However, it is conceivable that the effect of prolonged fasting on the hepatotoxicity of paracetamol may also be enhanced by decreased expression of relevant UGTs through phosphorylation of HNF4 α . The three most active UGTs towards paracetamol are UGT1A9, UGT1A1 and UGT1A6 (Court *et al.*, 2001), and all are potential HNF4 α -target genes (see Chapter 5 for UGT1A1 and UGT1A6). On the other hand, inflammation decreases the hepatic expression of some human UGTs. In liver biopsies with increased IL-1ß mRNA, UGT1A9 mRNAs tended to be reduced (although the trend failed to reach significance for the 5 samples tested, 4 of which were HCV infected) (Congiu et al., 2002). Given that HCV infection may partly compensate for the effect of inflammatory mediators on HNF4a and HNF1a by increasing their expression, it would be interesting to determine whether the likely loss of UGT1A9 expression is: a) related to the effects of inflammation on HNF4a and HNF1 α ; and b) greater in instances where hepatitis is not HCV-related.

4.4.4. Summary

This study highlights the synergistic role of HNF1 α and HNF4 α in regulation of the *UGT1A9* promoter *in vitro*, and illustrates the potential for both promoter and transcription factor variants to alter UGT1A9 expression. The discovery of two new response elements of the *UGT1A9* promoter enhances our understanding of the mechanisms that may contribute to hepatic, and possibly renal, expression of UGT1A9. Evidence presented in this chapter also points to the existence of at least one additional crucial element for hepatic *UGT1A9* regulation that remains unidentified. Finally, this study shows that HNF1 β can be substituted for HNF1 α in the synergistic regulation of *UGT1A9*, the first time a positive HNF4 α -HNF1 β interaction has been reported.

HNF4 and HNF1 transcription factors are both expressed in tissues where UGT1A9 is found: in the liver, kidney and gastrointestinal tract (Sladek *et al.*, 1990; Rey-Campos *et al.*, 1991). Interestingly, HNF1 α levels have already been correlated with UGT2B7 mRNA, while both HNF4 α and HNF1 α have been correlated with DD4 transcript abundance in human liver (Toide *et al.*, 2002; Ozeki *et al.*, 2003). Therefore, it seems feasible that a *UGT1A9* promoter-specific interaction between HNF4 α and HNF1 could contribute to the unique expression pattern of this protein amongst the *UGT1A7-1A10* cluster, and that variation in the levels or activity of these factors could influence UGT1A9 expression.

CHAPTER FIVE REGULATION OF ENDOGENOUS UGT EXPRESSION IN HEPG2 CELLS BY LIVER-ENRICHED TRANSCRIPTION FACTORS

5.1 Introduction

5.1.1. Promoter-reporter assays versus endogenous gene expression

The use of synthetic promoter-reporter constructs in transient-transfection cell-based assays is a common approach for investigating the function of eukaryotic gene promoters. As evidenced by earlier chapters, this methodology can yield many useful insights into the relationships between a promoter nucleotide sequence, the proteins that interact with this promoter, and the resulting mRNA synthesis - its major advantage being that the experimental system is relatively simple and amenable to manipulation. Not only can putative transcription factors be over-expressed in the host cell to assess their effect on reporter gene expression, but mutations and bona fide genetic variations can also be introduced into the promoter or the over-expressed transcription factors to further elucidate function. However, there are also a number of significant limitations to this approach. Noteworthy disadvantages of promoterreporter assays include the following: a) each promoter must be cloned before any hypotheses can be tested (which can become quite arduous if multiple genes are to be investigated); b) the reporter plasmid is not incorporated into the genome and is therefore presumably not subject to the same degree of epigenetic control as the endogenous gene; and c) important enhancer and silencer elements may reside many kilobases upstream or downstream of the proximal promoter, or within introns, and therefore may be unintentionally excluded from analysis.

To help negate these concerns, changes in endogenous gene expression caused by chemical ligand exposure or exogenous transcription factor expression can also be studied. Quantification of endogenous gene expression can be achieved through a variety of methods, including biochemical enzyme assays, Western analysis of protein expression, or PCR detection of mRNA transcripts. This chapter describes the transcriptional response of human *UGT1A* and *UGT2B* genes to over-expression of ten LETFs in the human hepatocellular carcinoma cell line, HepG2. These experiments were performed in untreated cells, but also in cells treated with a histone deactetylase inhibitor, trichostatin A (TSA), to investigate any association of histone deacetylation with *UGT* gene repression in HepG2 cells. Finally, transient transfection assays were used to further investigate several of the interesting interactions between LETFs and *UGT* promoters discovered in this way, as there is still no convenient way of manipulating promoter sequences *in vivo*. Such comparisons are also useful to determine whether results obtained in the easily manipulated *in vitro* systems are likely to bear any relevance to the *in vivo* situation.

5.1.2. Aims

The aims of the work presented in this chapter were two-fold. These were to:

- 1. Identify new regulatory factors for the human UGT genes;
- Further investigate known interactions between transcription factors and UGT promoters in a setting that may represent the *in vivo* situation better than transient plasmid transfections. Of particular interest were the HNF1 interactions with the UGT1A1 (Bernard *et al.*, 1999), UGT1A3, UGT1A4 (Gardner-Stephen and Mackenzie, 2007b), UGT1A9 (Gregory *et al.*, 2004)

and UGT2B7 (Ishii *et al.*, 2000) promoters, and the HNF4 α interaction with UGT1A9 (Barbier *et al.*, 2005; Gardner-Stephen and Mackenzie, 2007a).

There were several good reasons to expect to find new LETF-binding sites in at least some human *UGT* promoters. Firstly, human UGTs are not expressed ubiquitously, but most are expressed at moderate to high levels in the human liver (with the exception of UGT1A5, UGT1A7, UGT1A8 and UGT1A10). Of these exceptions, UGT1A7, UGT1A8 and UGT1A10 are expressed to high levels in at least one section of the gastrointestinal tract, an organ where many of the transcription factors known as LETFs are also expressed. In general, genes that have been found to be controlled by one or more LETFs generally possess multiple LETF-binding sites in their promoters. In contrast, only one or two LETF-binding sites have been discovered for most *UGT* genes; therefore, it seemed likely that a thorough investigation of human *UGT* genes would confirm them as targets of further LETFs.

The developmental profile of hepatic UGT expression (see Chapter 1, section 1.8.1) also indicates that the current understanding of *UGT* regulation is lacking. Much of the past work on human *UGT* promoters has focused on the role of HNF1 factors in driving transcription; however, HNF1 factors are expressed early in liver development (De Simone *et al.*, 1991; Cereghini *et al.*, 1992), while UGTs are not detected in early foetal liver and are not well expressed until after birth, even though most have been shown to possess HNF1-binding sites. This suggests that additional LETFs, expressed later in hepatic development, are also required for UGT expression.

5.1.3. Rationale

The transcription factors used in this experiment were PXR, HNF1a, HNF1b, HNF4a, HNF6, FoxA1, FoxA2, FoxA3, C/EBPa and C/EBPB. Since the most important organ for UGT expression and activity is the liver, these proteins were chosen on the basis that they are all LETFs, and represent five separate protein families or superfamilies: zinc-finger nuclear receptors (HNF4 and PXR), POU homeodomain proteins (HNF1), onecut homeodomain proteins (HNF6), forkhead box proteins (FoxA) and basic-region leucine-zipper proteins (C/EBP) (Schrem et al., 2002; Schrem et al., 2004). Further considerations made in choosing these transcription factors were that PXR, HNF1a, HNF1b, HNF4a and C/EBPa have previously been identified as having a role in the expression of at least one UGT in humans or rodents (Hansen et al., 1998; Bernard et al., 1999; Ishii et al., 2000; Rae et al., 2001; Gardner-Stephen et al., 2004; Barbier et al., 2005; Gardner-Stephen and Mackenzie, 2007b; Gardner-Stephen and Mackenzie, 2007a), while HNF6, C/EBPβ and the FoxA family members, have been identified as regulators of other drugmetabolising enzymes in humans and rodents (Jover et al., 1998; Delesque-Touchard et al., 2000; Jover et al., 2002; Rodriguez-Antona et al., 2003; Bort et al., 2004). In particular, HNF1 α , HNF1 β and HNF4 α were chosen because they have been shown to regulate the human UGT1A3, UGT1A4 and/or UGT1A9 promoters in vitro in work presented earlier in this thesis. The structural and functional features of HNF1 α , HNF1 β and HNF4 α have been discussed at length in Chapters 3 and 4; therefore, the following literature review will only address these aspects in detail for the newly introduced transcription factors.

5.1.3.1. Pregnane X receptor

The pregnane X receptor (NR112) is a ligand-regulated nuclear receptor that contains two zinc fingers (constituting the DNA-binding domain) and a large ligand-binding domain of 293 amino acids. This ligand-binding domain is also required for dimerisation and transcriptional activation (Bertilsson *et al.*, 1998; Lehmann *et al.*, 1998; Kliewer *et al.*, 2002). PXR is expressed in the liver, and to a lesser extent, the colon and small intestine (Bertilsson *et al.*, 1998; Lehmann *et al.*, 1998), and heterodimerises with RXR α (Kliewer *et al.*, 1998) to activate genes through binding sites consisting of dual AGGTCA-like NRREs. These NRREs are arranged as direct or everted repeats, separated by three to eight nucleotides (Kliewer *et al.*, 2002).

On ligand binding, PXR undergoes a conformational change that facilitates interaction with p160 co-activators, such as SRC-1, PGC-1 and the receptorassociated co-activator 3 (Itoh *et al.*, 2006; Johnson *et al.*, 2006; Li and Chiang, 2006), while disrupting associations with co-repressors such as SMRT (Johnson *et al.*, 2006). Transcriptional activation of target genes by liganded PXR/co-activator complexes is then brought about through both histone acetylation and direct interactions with the basal transcription machinery. In addition to its associations with p160 proteins, PXR is also known to interact directly with other nuclear receptors and transcription factors to influence target gene transcription. PXR activity can be inhibited by the presence of the SHP nuclear receptor that lacks a conventional DNA-binding domain (Ourlin *et al.*, 2003). PXR has also been shown to recruit HNF4 α to the *CYP3A4* promoter but interferes with HNF4 α regulation of *CYP3A7*. Both of these latter interactions may be mediated through a co-activator common to PXR and HNF4 α , PGC-1 (Bhalla *et al.*, 2004; Li and Chiang, 2006). Interestingly, binding of alternative ligands to PXR may change its interactions with other transcription factors: PXR-mediated induction of the *CYP3A4* promoter is variably reliant on interactions with Sp1, FoxA proteins or C/EBP α , depending on the ligand bound (Bombail *et al.*, 2004).

The known ligands of PXR are a structurally diverse set of compounds that include several pregnenolone and progesterone derivatives, lithocolic acid, and many pharmacologically active substances such as paclitaxel, hyperforin, clotrimazole and rifampicin (Bertilsson *et al.*, 1998; Lehmann *et al.*, 1998; Moore *et al.*, 2000; Staudinger *et al.*, 2001; Synold *et al.*, 2001). Whilst the murine homologue of PXR shares 96% identity with the human protein within the DNA-binding domain, there is only 77% homology within their ligand-binding domains. Thus, while murine PXR binds the same DNA targets as human PXR, the two homologues respond to a different, but overlapping, set of ligands (Bertilsson *et al.*, 1998; Lehmann *et al.*, 1998; Staudinger *et al.*, 2001).

It has been proposed that PXR operates as a master regulator in the elimination of xenobiotics from the body, as liganded PXR regulates the expression of numerous biotransformation enzymes and transporter proteins, including CYP3A4, CYP3A7 and CYP2C9, (Bertilsson *et al.*, 1998; Pascussi *et al.*, 1999; Chen *et al.*, 2004), UGT1A1, UGT1A3, UGT1A4, UGT1A6 and UGT1A9 (Rae *et al.*, 2001; Gardner-Stephen *et al.*, 2004; Soars *et al.*, 2004; Sugatani *et al.*, 2004; Bock and Kohle, 2005; Chen *et al.*, 2005a), as well as MRP and perhaps glutathione-*S*-transferase family members (Kast *et al.*, 2002; Maglich *et al.*, 2002). PXR expression is itself subject to control by other ligand-activated nuclear receptors such as PPARα (Aouabdi *et al.*, 2006) and possibly the oestrogen and glucocorticoid receptors (Gibson *et al.*, 2006).

PXR may also be involved in a negative feedback loop that inhibits its own expression (Gibson *et al.*, 2006).

As PXR has already been shown to up-regulate transcription of endogenous UGTs in HepG2 cells (Gardner-Stephen *et al.*, 2004), this transcription factor was chosen to serve as the positive control for this study. Furthermore, whilst *UGT1A1*, *UGT1A3*, *UGT1A4*, *UGT1A6*, *UGT1A9* and *UGT2B7* have all been identified as potential PXR targets by various research groups, their transcriptional responses have not been quantitatively assessed with methods as sensitive as real-time PCR. Finally, of the PXR-responsive *UGT* genes, a NRRE has only been identified for *UGT1A1*. Confirmation of other responsive genes would justify a search for further PXR-responsive elements within the *UGT1A* locus.

5.1.3.2. The hepatocyte nuclear factor 1 family

Of the transcription factors thought to be important in the constitutive expression of UGTs, HNF1 α is, by far, the most thoroughly studied. Putative HNF1-binding sites have been identified in the proximal promoters of all human *UGT* genes (Figure 5.1) (Auyeung *et al.*, 2001; Gardner-Stephen and Mackenzie, 2005), and function has been demonstrated for sites in the *UGT1A1* (Bernard *et al.*, 1999), *UGT1A3*, *UGT1A4* (Gardner-Stephen and Mackenzie, 2007b and this thesis), *UGT1A8*, *UGT1A9*, *UGT1A10* (Gregory *et al.*, 2004; Gardner-Stephen and Mackenzie, 2007a and this thesis), *UGT2B7* (Ishii *et al.*, 2000) and *UGT2B17* (Gregory *et al.*, 2000) promoters. There is also substantial circumstantial evidence pointing to HNF1 α as an important regulator of UGT expression: the expression of UGTs and HNF1 α in adult tissues overlaps considerably (compare Chapter 1, Table 1.2 to Chapter 3, section 3.1.5.2); HNF1 α knockout mice have reduced UGT expression (Pontoglio *et al.*, 1996; Shih *et al.*, 2001); many of the identified human *UGT* HNF1 sites are highly

	-113 -54
1A1	CTTTTTATAGTCACGTGACACA <mark>GTCAAACATTAAC</mark> TTGGTGTATCGATTGGTTTTGCCA
	-91 -79
	_171112
1 2 3	
145	
1A4	GAAAGGCAGTTATAGATTAATG GGTAATAAGTAAC TGGAGGAGGGCACTTTGTCTTCCAA
	* ** ** * ******** ** ****** **********
	-149 -137
	101 120
176	-191 — 192 — по и и о со со сталовата со сталовата со сталовата со со сталовата со со сталовата со со сталовата со со стал
IAO	-169 -157
	-312 -253
1A8	TCTTCAAGGTCCTAAAGCATTGCTTAGTAATTTTGTTTCTAAACTCA-CGTTACAGCACT
1A10	TCTTCAAGGTCCAAAAGCATTGCTGAATAATTCTGTTTCTAAACTCA-C-TTGCAGC
1A9	CCTTCAAGGTCCAAAAGCATTG GTTAATAATTCTG CTTCTAAACTTAACATTGCAGCACA
IA/	
	_200 _279
	-290 -270
	-170 -116
1A8	TCATGTATTCCTGTTCTTATGA <mark>GTAAATCATTGGC</mark> AGTGAGTGTGATTTTTTTTTTTTTT
1A10	TCGTGTGTTATCGTTCTTATGA <mark>GTAAATCATTGGC</mark> AGTGAGTGTGATTTTTTTTTTTTT
1A9	TCATATATTCTTGTTCTTTTGG <mark>GTAAATCATTGTC</mark> AGTGACTGATTTTTTTTAT
1A7	TCATGTATTATTATGAGTAAATCATTGGCAGTGAATGTGAATTTTTTTTT
	** * * ** ** ** ***********************
	-148 -136
	101 (0)
	- 12 1 - Ͳል ሞሞ ካራ ሞም ምር አል ራምር ሞ አል አል <mark>ራም ም አስ አምም ም ም አስ ለ</mark> ሞም ም ምር አል ምም እም አስ አስ አስ የምር ም ምር አል የምር እም ምር እም ምር እም ም
UGT2B7	
UGT2B17	TCAAATTTTAGCA GTTATATTTTTAAC TTGATTGATTTTTCCTCAGATA
UGT2B15	-TATTTACTTCAAATTTTAGCAGTTATATTTTAACTTGATTGATTTTTCCTCAGATA
UGT2B10	-TATTTACTTTGAATTGAAGGAGTTATGTTTTAACTTGATTGATTTATCTCTGTATA
UGT2B28	-TATTTACTTTGGATTGAAGGAGTTATGTTTTAACTTGATTGATTTATCTCTGTATA
UGT2B11	-TATTTACTTTGGATTGGAGGAGTTATATTTTAACTTGATTGATTTATCTCTGTATA
	* * * **** ********
	- 99 - 87

Figure 5.1: All known human UGT1A and UGT2B promoters contain putative HNF1-binding sites. The UGT1A3-1A5, UGT1A7-1A10 and UGT2B proximal promoter regions containing HNF1-binding elements are aligned. The putative HNF1-binding sites are boxed, with the sites that have been confirmed experimentally indicated in bold. Numbering is relative to the initiation codon of UGT1A1, UGT1A3, UGT1A6, UGT1A9 and UGT2B7 as appropriate. This figure was adapted from Auyeung *et al.* (2001), Bernard *et al.* (1999), Gardner-Stephen and Mackenzie (2005), Gardner-Stephen and Mackenzie (2007a).

conserved within UGT genetic sub-clusters (Figure 5.1), indicating likely functional

importance; UGT2B7 expression levels have been correlated with HNF1a mRNA in

human liver samples (Toide et al., 2002); and an extensive microarray study detected

occupation of the *UGT1A1*, *UGT2B11* and *UGT2B15* promoters by HNF1 α in primary hepatocytes (Odom *et al.*, 2004).

However, despite all this evidence, there is still some doubt about the relative importance of HNF1 α in the expression of each UGT enzyme *in vivo*. Reasons for this include; lack of correlation of UGT2B15 with HNF1 α levels (Toide *et al.*, 2002), residual bilirubin glucuronidation in HNF1 α -knockout mice (Pontoglio *et al.*, 1996), the aforementioned inability of either HNF1 α or HNF1 β to turn on UGT expression during early liver development, and the possibility that transcription factor sites found to be crucial for the activity of short promoter constructs may be less important in longer sections of the same promoter, as is the case for the proximal C/EBP α sites in the *CYP3A4* gene (Martinez-Jimenez *et al.*, 2005). Therefore, it was hoped that inclusion of HNF1 α in this study would shed further light on the relative importance of this transcription factor for expression of each UGT enzyme.

Although HNF1 β binds the same DNA response elements as HNF1 α , the relevance of HNF1 β to human UGT expression is not as well established. Potentially, HNF1 β could also up-regulate UGT expression through the same sites as HNF1 α . Alternatively, HNF1 β may modulate the effects of HNF1 α through competition or dimerisation, as it is generally accepted that HNF1 β has a lower transactivation potential than HNF1 α . However, functional interaction of HNF1 β with human UGT promoters has, so far, only been demonstrated for *UGT1A1* (Bernard *et al.*, 1999) and *UGT1A9* (Gardner-Stephen and Mackenzie, 2007a and this thesis).

5.1.3.3. Hepatocyte nuclear factor 4α

HNF4 α is a zinc-finger nuclear receptor involved in the expression of numerous human enzymes involved in biotransformation, including CYP3A4, CYP3A5,

CYP2A6, CYP2B6, CYP2C9, CYP2D6 (Jover *et al.*, 2001; Li and Chiang, 2006) and UGT1A9 (Barbier *et al.*, 2005; Gardner-Stephen and Mackenzie, 2007a and this thesis, Chapter 4). Naiki and colleagues have also found that the UGT1A family of mRNA transcripts are increased in HepG2 cells infected with adenovirus expressing rat HNF4 α (Naiki *et al.*, 2004). Therefore, it seemed likely that HNF4 α plays a role in regulating *UGT* genes other than *UGT1A9*. Strengthening this hypothesis, HNF4 α was found to occupy the *UGT2B11* and *UGT2B15* promoters in human hepatocytes (Odom *et al.*, 2004). Furthermore, in the mouse, HNF4 α has been implicated in the regulation of both *Ugt1a1* and *Ugt1a9*. Interestingly though, knockout of HNF4 α expression in the mouse (Hayhurst *et al.*, 2001) resulted in a decrease of murine Ugt1a9 transcripts by 73% (Barbier *et al.*, 2005), while, in contrast, HNF4 $\alpha^{-/-}$ mice had 14-fold higher levels of Ugt1a1 mRNA than their wild-type litter mates (Ding *et al.*, 2006).

A second reason for studying the effect of HNF4 α on UGT expression comes from developmental observations. Although HNF4 α is expressed early in hepatic development (Duncan *et al.*, 1994), the ratio of the various HNF4 α splice variants (see Chapter 4, section 4.1.4.2) present in embryonic liver changes as the hepatocytes mature. HNF4 α 7 transcripts are found at high levels in embryonic hepatocytes, but decrease around the time of birth to become only a minor fraction of the total liver HNF4 α mRNA (Nakhei *et al.*, 1998). In contrast, HNF4 α 1 and HNF4 α 2 are also expressed in embryonic liver, but are increased perinatally (Torres-Padilla *et al.*, 2001). HNF4 α 1, HNF4 α 2 and HNF4 α 7 all share the same DNA-binding domain; however, HNF4 α 7 has a distinct N-terminal domain that allows it to interact differently with co-activators and co-repressors from HNF4 α 1 and HNF4 α 2, and thus regulate different genes (Torres-Padilla *et al.*, 2001; Torres-Padilla *et al.*, 2002). One of the major determinants thought to be responsible for the perinatal increase in HNF4 α 1 and HNF4 α 2 expression is the concomitant rise in glucocorticoids, which also induces several other hepatic genes at birth (Berger *et al.*, 1996; Bailly *et al.*, 2001). Moreover, GR and HNF4 α have been shown to co-operatively regulate the hepatic tyrosine aminotransferase gene (Nitsch *et al.*, 1993). Since expression of most human UGTs is initiated or increased around birth (see Chapter 1, section 1.8.1), it is possible that birth-related changes in LETF expression, such as the switch in HNF4 α isoforms, are involved. Interestingly, UGT1A1 has already been shown to be a direct target of GR (Sugatani *et al.*, 2005a), so the potential for a synergistic activation of UGT1A1 by GR and HNF4 α also exists.

5.1.3.4. Hepatocyte nuclear factor 6

The transcription factor hepatocyte nuclear factor 6, also known as Onecut 1, possesses a single cut-domain and a divergent homeodomain, and binds to DNA sequences of consensus DRRTCVATND where D = A, G or T, R = A or G and V = A, G or C (Lemaigre *et al.*, 1996; Lannoy *et al.*, 1998; Jacquemin *et al.*, 1999). The second, and only other known, human member of this transcription factor class is Onecut 2, which is expressed in liver and skin (Jacquemin *et al.*, 1999). HNF6 is expressed in human liver and pancreas (Rausa *et al.*, 1997) and regulates the hepatic expression of a number of proteins such as glucokinase (Lannoy *et al.*, 2002) and the transcription factors FoxA2 and HNF4 α (Samadani and Costa, 1996; Landry *et al.*, 1997; Rausa *et al.*, 1997; Hatzis and Talianidis, 2001; Briancon *et al.*, 2004; Odom *et al.*, 2004). In turn, the *hnf6* gene is under the transcriptional control of HNF4 α and C/EBP α , at least in the rat (Lahuna *et al.*, 2000; Rastegar *et al.*, 2000). HNF6 has been reported to physically interact with FoxA2 to both positively and negatively regulate gene promoters, depending on the promoter configuration (Delesque-

Touchard *et al.*, 2000; Rausa *et al.*, 2003; Rubins *et al.*, 2005), while synergistic interactions have been identified with HNF1 α , C/EBP α , HNF4 α and another nuclear receptor, the retinoic-acid-receptor-related orphan receptor α (Hatzis and Talianidis, 2001; Nacer-Cherif *et al.*, 2003; Beaudry *et al.*, 2006; Yoshida *et al.*, 2006). The co-activators recruited to a target gene by HNF6 depend on the target promoter and include CBP, P/CAF and PGC-1 α (Lannoy *et al.*, 2000; Rausa *et al.*, 2003; Beaudry *et al.*, 2006).

HNF6 has previously been reported to be bound to the *UGT1A1*, *UGT2B11* and *UGT2B15* promoters in liver when assessed by a chromatin immunoprecipitation experiment coupled with an extensive promoter microarray (Odom *et al.*, 2004). However, no functional studies have yet been performed for these interactions.

5.1.3.5. The forkhead box A (FoxA, HNF3) family

The human FoxA transcription factor family, also known as the HNF3 family, is comprised of three isoforms: FoxA1 (HNF3 α), FoxA2 (HNF3 β) and FoxA3 (HNF3 γ). Encoded by separate genes, these three proteins share a winged helix/forkhead box DNA-binding domain of 93-95% identity, through which they bind as monomers to DNA sequences of consensus VAWTRTTKRYTY (where V = A, G or C, W = A or T, R = A or G and K = G or T) (Overdier *et al.*, 1994). FoxA1, FoxA2 and FoxA3 are able to interact with nucleosome core histones H3 and H4 to open compacted chromatin and alter nucleosome positioning, thereby facilitating the binding of other transcription factors to target promoters (McPherson *et al.*, 1993; Shim *et al.*, 1998; Chaya *et al.*, 2001; Cirillo *et al.*, 2002). In situations where this occurs, FoxA1 binding to its DNA elements is more stable on nucleosome-bound DNA than free DNA (Cirillo and Zaret, 1999). In addition, FoxA factors may also contribute to target gene expression by enhancing the stability of the pre-initiation complex (Crowe *et al.*, 1999).

Of the FoxA proteins, FoxA3 is the most widely and highly expressed. In the adult mouse, FoxA proteins are found in liver, lung, stomach, small intestine and colon. FoxA3 is the predominant FoxA species in these tissues; except the lung, from which it is absent. In particular, hepatic FoxA3 expression is approximately treble that of either FoxA1 or FoxA2, and may be a negative regulator of the latter two in the liver (Kaestner *et al.*, 1998). FoxA3 expression is also detectable in mouse heart, adipose tissue, thymus, ovary and testes. In contrast, FoxA1 is found in the brain, pancreas, kidney and prostate, while FoxA2 is most highly expressed in the pancreas (Kaestner *et al.*, 1994; Rausa *et al.*, 1997; Besnard *et al.*, 2004). FoxA factors are involved in the regulation of many hepatic genes, including gluconeogenic enzymes such as phospho*enol*pyruvate carboxykinase and aldolase B (Vallet *et al.*, 1995; Friedman and Kaestner, 2006), and a number of human *CYPs*, including *CYP3A4*, *CYP2C8*, *CYP2C18* and *CYP2C19* (Bombail *et al.*, 2004; Bort *et al.*, 2004).

Like most other LETFs, FoxA proteins regulate, and are regulated by, other LETFs. *Hnf4a* is a FoxA3 target gene (Bailly *et al.*, 2001), and FoxA proteins have a weak, but positive effect on the *hnf1a* promoter (Kuo *et al.*, 1992) – at least in rodents. In turn, *foxA1* and *foxA2* are subject to auto- and cross-regulation through FoxA-binding sites in their promoters (Pani *et al.*, 1992; Peterson *et al.*, 1997; Zhang *et al.*, 2005), *foxA2* is a target of HNF6 and C/EBP factors (Samadani *et al.*, 1995; Samadani and Costa, 1996; Lahuna *et al.*, 2000), while *foxA3* is a target gene of HNF1 (Hiemisch *et al.*, 1997). In addition, many functionally relevant, physical interactions between FoxA and other transcription factors or co-activators have been reported. For example, FoxA factors bound to the α -1-microglobulin/bikunin

precursor gene promoter support HNF1 driven transcription, yet inhibit the effects of HNF4α (Rouet et al., 1995); FoxA factors compete with HNF1 for binding on the rat aldolase B gene (Gregori et al., 1994); HNF1a, HNF6 and C/EBPa each interact synergistically with FoxA family members to activate transcription of various target genes (Cha et al., 2000; Delesque-Touchard et al., 2000; Rausa et al., 2003; Rodriguez-Antona et al., 2003); FoxA2 inhibits HNF6 stimulation of the murine glut2 promoter (Rausa et al., 2003); and SHP physically interacts with all three FoxA proteins to repress their DNA-binding activity and therefore, the activation of their target genes (Kim et al., 2004). The FoxA proteins have also been postulated to behave as "pioneer factors" for nuclear receptors such as the glucocorticoid, oestrogen and androgen receptors (Gao et al., 2003; Laganiere et al., 2005; Zhang et al., 2005) - proceeding and facilitating binding of the nuclear receptors to their respective NRREs (Friedman and Kaestner, 2006). Of particular interest is a recent report that small interfering RNA (siRNA)-mediated knockdown of FoxA1 transcripts in MCF-7 breast cancer cells results in a reduction of oestrogen-induced recruitment of the oestrogen receptor to the UGT2B17 promoter (Laganiere et al., 2005).

Despite the many similarities and apparent redundancies that have been observed for FoxA1, FoxA2 and FoxA3 in hepatic gene regulation, functionally distinct roles for the different proteins are emerging, particularly in embryonic development (Friedman and Kaestner, 2006). In addition, FoxA2 is the only isoform to interact with HNF6 to co-operatively regulate the *foxA2* promoter (Rausa *et al.*, 2003) although FoxA1 and FoxA2 can both synergistically enhance HNF6-mediated regulation of the rat *CYP2C12* promoter (Delesque-Touchard *et al.*, 2000). Furthermore, the co-activator PGC-1 only interacts with FoxA3 (Kim *et al.*, 2004),

while FoxA1 expression is specifically diminished in hepatocytes during the acutephase response (a condition that also results in reduced CYP activity) (Shedlofsky *et al.*, 1994; Qian *et al.*, 1995).

FoxA1, FoxA2 and FoxA3 were included in this study because there is substantial evidence for their involvement in regulating genes of biotransformation pathways, especially CYPs. Many transcription factors identified as CYP regulators have subsequently been shown to also regulate genes with conjugative functions; therefore, the FoxAs were considered important LETFs to test for involvement in *UGT* regulation. In addition, FoxA1 has been indirectly implicated in the expression of UGT2B17, at least in breast cells (Laganiere et al., 2005). A second reason to test these factors was the uncertainty that surrounds the ability of transient transfection systems to adequately assess the function of FoxA proteins. Although chromatinmediated repression of transcription from transiently transfected promoter-reporter plasmids is relieved by FoxA expression in certain cases (Crowe et al., 1999), most data suggests that the importance of transcription factors with chromatin-altering functions may be underestimated in plasmid-based assays. Transiently transfected plasmids do not always acquire the same higher-order chromatin structure as observed with genomic DNA and may be inappropriately accessed by ubiquitous factors (Smith and Hager, 1997). In addition, inconsistent results between plasmidreporter and endogenous responses to FoxA genes have been previously reported (Bort *et al.*, 2004); therefore, it was considered prudent to assess the effects of FoxA over-expression on genomic UGT promoters.

5.1.3.6. The CCATT/enhancer binding protein family

The C/EBP transcription factor family is one subset of the basic region leucine zipper (bZIP) transcription factor family that also includes c-jun, c-fos and CREB.

Members of the bZIP family of proteins are characterised by a bipartite bZIP domain, consisting of a basic region for DNA binding and a leucine zipper region for dimerisation (Landschulz *et al.*, 1989; Ramji and Foka, 2002). Sequential binding of two monomers to the palindromic repeat RTTGCGYAAY (where R = A or G and Y = C or T) in target DNA allows the formation of a dimer, which then stabilises the DNA-protein complex (Hsu *et al.*, 1994; Osada *et al.*, 1996; Kohler *et al.*, 1999). C/EBP factors can homodimerise, heterodimerise with each other, or heterodimerise with other bZIP proteins, such as those of the CREB protein family. C/EBP α and C/EBP β heterodimers bind DNA with the same specificity as their respective homodimers; however, C/EBP heterodimers with other bZIP proteins generally bind asymmetric sequences composed of the consensus half-sites for each monomer (Williams *et al.*, 1991; Hsu *et al.*, 1994; Shuman *et al.*, 1997).

The C/EBP transcription factor family consists of six members: C/EBP α , C/EBP β , C/EBP β , C/EBP δ , C/EBP ϵ and C/EBP ζ , although only C/EBP α and C/EBP β are considered true LETFs (Lekstrom-Himes and Xanthopoulos, 1998; Schrem *et al.*, 2004). In humans, C/EBP α is expressed in placenta, liver, intestine, lung, peripheral blood leukocytes, skeletal muscle, pancreas, heart, spleen, prostate and adipose tissue (Antonson and Xanthopoulos, 1995; Harp *et al.*, 2001). Likewise, at least in rodents, C/EBP β is expressed in liver, intestine, adipose tissue, lung, heart and spleen, but also kidney (Cao *et al.*, 1991; Williams *et al.*, 1991). Expression of C/EBP β in human liver has been confirmed, although protein levels appear to be subject to considerable interindividual variation (Ferrini *et al.*, 2001; Tomizawa *et al.*, 2003). C/EBP γ and C/EBP ζ are ubiquitously expressed (Roman *et al.*, 1990; Ron and Habener, 1992; Thomassin *et al.*, 1992), while C/EBP δ is constitutively expressed in a limited number of tissues (intestine, adipose tissue and lung), yet is strongly

induced in liver and many other tissues during the acute-phase response (Cao *et al.*, 1991; Alam *et al.*, 1992; Kinoshita *et al.*, 1992; Yamada *et al.*, 1997). C/EBPE is expressed in myeloid and lymphoid cell lineages, but not liver (Antonson *et al.*, 1996).

The study of gene regulation by C/EBP α and C/EBP β is complicated by the fact that both proteins are expressed as multiple isoforms, generated through use of internal methionines as alternative initiation codons. C/EBP α has two functionally distinct isoforms (p42 and p30) while C/EBP β has up to four (p38, p35: also known as liverenriched transcriptional activator protein (LAP), p21: also known as liver-enriched transcriptional inhibitory protein (LIP), and p14). Full-length and LAP-C/EBP β can also be proteolytically cleaved in newborn liver to generate LIP-C/EBP β . (Descombes and Schibler, 1991; Lin *et al.*, 1993; Ossipow *et al.*, 1993; Welm *et al.*, 1999; Xiong *et al.*, 2001; Ramji and Foka, 2002). The truncated isoforms p30-C/EBP α and LIP-C/EBP β both possess full DNA-binding and dimerisation capabilities, but lack most or all of the N-terminal activation domains of their fulllength counterparts (Ramji and Foka, 2002).

While full-length p42-C/EBP α acts as a transactivator for many hepatic genes and inhibits cell proliferation, the N-terminally truncated p30-C/EBP α lacks antimitotic activity and has attenuated transcription activation potential. p30-C/EBP α was found to activate the murine C/EBP α promoter; however, very little activity was demonstrated towards another known C/EBP α target, the mouse albumin promoter. p30-C/EBP α was also able to dramatically reduce p42-C/EBP α -driven transcription from the albumin promoter, but only when present in vast excess. At the p30/p42 ratios normally seen in mammalian liver, only a small degree of inhibition was observed compared to activation by p42-C/EBPα alone (Lin *et al.*, 1993; Ossipow *et al.*, 1993).

Similarly, LIP-C/EBPB does not possess the same activation potential as the fulllength and LAP-C/EBP^β proteins. However, in contrast to the C/EBP^α isoforms, LIP-C/EBPB usually behaves as a dominant-negative inhibitor of LAP-C/EBPB activity, inhibiting LAP-C/EBPβ-mediated transactivation at low LIP/LAP ratios. It has been proposed that this is due to inactivation of LAP through heterodimerisation, combined with a higher affinity of the LIP homodimers and LIP-LAP heterodimers for the common DNA recognition sequences than LAP homodimers. The ratio of LAP-C/EBPB to LIP-C/EBPB in the hepatocyte is important in maintaining differentiation, and for controlling proliferation in response to liver damage. Pertinent to this, LIP-C/EBPB production can be regulated independently of LAP-C/EBP^β translation through binding of the CUG repeat binding protein CUGBP1 to C/EBPß mRNA; favouring translation of the low-molecular-weight isoforms (Descombes and Schibler, 1991; Timchenko et al., 1999; Welm et al., 2000; Luedde et al., 2004; Timchenko et al., 2005). It should also be noted that, because the shorter C/EBP isoforms are produced by leaky ribosomal scanning, transient transfection with plasmids carrying the full-length C/EBPa or C/EBPB coding regions results in the expression of the truncated as well as the full-length isoforms (Descombes and Schibler, 1991; Ossipow et al., 1993; Xiong et al., 2001).

In humans, C/EBPα has been identified as a regulator of numerous hepatic genes including albumin, *ADH2*, insulin-like growth factors (van Dijk *et al.*, 1992; van Ooij *et al.*, 1992; Jover *et al.*, 1998), and *CYP3A4*, *CYP3A5*, *CYP3A7*, *CYP2B6*, *CYP2C9*, *CYP2D6* and *CYP2A6* (Jover *et al.*, 1998; Rodriguez-Antona *et al.*, 2003; Bombail *et al.*, 2004). In mice, many additional hepatic targets of C/EBPα have been

identified, but one of particular interest is Ugt1a1 (Lee *et al.*, 1997). Deletion of the $C/EBP\alpha$ gene in mice leads to severe jaundice due to an increase in unconjugated serum bilirubin. The rat UGT2B1 promoter has also been proposed to be a $C/EBP\alpha$ target gene (Hansen *et al.*, 1998).

In regards to its participation in the LETF network, C/EBP α regulates its own expression in both humans and mice, although the mechanism involved varies between the two species. In humans, autoregulation of the human C/EBP α promoter is mediated indirectly through the ubiquitously expressed upstream stimulating factor, rather than through a direct C/EBP-binding site as found in mice (Timchenko *et al.*, 1995; Schrem *et al.*, 2004). C/EBP α also positively influences the rate of cleavage of C/EBP β to LIP-C/EBP β in mice and in human cells (Welm *et al.*, 1999), while other LETF target genes of C/EBP α include the murine *hnf4a1* promoter (Bailly *et al.*, 2001), the rat *hnf* δ gene (Rastegar *et al.*, 2000) and rat *foxA2* (Samadani *et al.*, 1995; Yoshida *et al.*, 2006). As highlighted previously, C/EBP α is also known to be able to synergistically activate some of its target genes with other transcription factors such as PXR (Bombail *et al.*, 2004), HNF6 (Yoshida *et al.*, 2005) and FoxA3 (Rodriguez-Antona *et al.*, 2003). HNF4 α (Pitarque *et al.*, 2005) and LAP-C/EBP β (van Ooij *et al.*, 1992) are also able to co-operatively activate target genes with C/EBP α .

As is the case for C/EBPα, much more work has been done on the regulation of rodent genes by C/EBPβ than human genes. However, known human hepatic target genes of C/EBPβ include *MDR1*, *ADH1*, *ADH2*, *ADH3*, *CYP2A6* and *CYP3A4* (van Ooij *et al.*, 1992; Combates *et al.*, 1994; Martinez-Jimenez *et al.*, 2005; Pitarque *et al.*, 2005).

As C/EBP α and C/EBP β bind the same DNA elements, there is substantial over-lap in C/EBP α and C/EBP β target genes. However, not all genes with C/EBP-binding sites are responsive to both. For example the C/EBP α gene target *UGT2B1* is not transactivated by C/EBP β , whereas the *ADH1* promoter responds specifically to C/EBP β (van Ooij *et al.*, 1992; Hansen *et al.*, 1998). Similarly, synergistic interactions between C/EBP α and other LETFs, such as HNF6, are not necessarily supported by C/EBP β (Yoshida *et al.*, 2006), or may even be repressed. The normally activating C/EBP β isoform, LAP-C/EBP β , interferes with HNF4 α activation of the *CYP2A6* promoter, even though C/EBP α co-operates with HNF4 α on the same promoter (Pitarque *et al.*, 2005).

In mice and rats, C/EBP β regulates its own promoter and that of *C/EBP* α (Chang *et al.*, 1995; Welm *et al.*, 2000). The *C/EBP\beta* promoter is also responsive to proinflammatory cytokines such as IL-6 and IL-1. Consequently, hepatic C/EBP β expression is increased in inflammation, the acute-phase response and in response to mechanical liver damage. In addition, translation of the LIP-C/EBP β isoform is increased under these conditions, and appears to cause the concomitant decrease in C/EBP α expression seen during the acute-phase response (Alam *et al.*, 1992; Welm *et al.*, 2000; Jover *et al.*, 2002). Increases in C/EBP β expression have also been associated with maintenance of FoxA2 expression during the acute-phase response in rats (Samadani *et al.*, 1995). Synergistic interactions of C/EBP β with other LETFs have been identified for HNF1 α and HNF1 β (Divine *et al.*, 2003), C/EBP α (van Ooij *et al.*, 1992) and at least one of the rat FoxA proteins (Pani *et al.*, 1992).

C/EBP α and C/EBP β were chosen for inclusion in this study for several reasons. Firstly, C/EBP α has a role in the expression of two non-homologous rodent *UGTs*, while both transcription factors regulate multiple human enzymes important for metabolism and excretion of lipophilic compounds. Although there is no evidence for C/EBPβ-mediated regulation of mammalian UGTs as yet, these two proteins bind the same recognition sequences, heterodimerise, and regulate each other's expression. Therefore, it was considered appropriate to include C/EBP β as well as C/EBP α in this study. Secondly, the expression of C/EBP α and C/EBP β during development is similar to that seen for many UGTs. Glucuronidation of most UGT substrates is absent or substantially lower in human foetal liver relative to adult, but increases directly after birth and reaches adult levels in the first few months or years of life (see Chapter 1, section 1.8.1). In rodents, C/EBPa expression is detectable in early liver development (Westmacott et al., 2006), but increases substantially late in foetal liver development, spikes around birth and is only found at high levels in fully differentiated cells (Birkenmeier *et al.*, 1989). Likewise, C/EBP β is more strongly expressed in late than early foetal liver development with a transient increase around birth, but importantly, the LAP/LIP ratio gradually increases by 5-fold between the period just before birth and adulthood (Descombes and Schibler, 1991). This relative increase in LAP-C/EBPB has been associated with the gradual postnatal increase in expression of certain hepatic genes (van Ooij et al., 1992) and could potentially affect UGTs. Thirdly, expression of C/EBP proteins is altered in disease states that also affect human UGTs. During inflammation, both the levels of C/EBP α and the ratio of LAP/LIP-C/EBPB in hepatocytes decrease, along with the expression of numerous CYP enzymes and, although to a lesser extent, several UGTs (Alam et al., 1992; Welm et al., 2000; Congiu et al., 2002; Aitken et al., 2006). C/EBPa levels have also been reported to be very low in human hepatocellular carcinoma, a condition that has also been associated with decreased expression of all hepatic UGT1A forms except UGT1A6 (Strassburg et al., 1997a; Xu et al., 2001). Given

that expression of C/EBP proteins is altered in several circumstances that are also associated with changes in UGT expression, it was of interest to determine whether either C/EBP α or C/EBP β levels in HepG2 cells could affect the transcription of any endogenous *UGT* genes.

5.1.3.7. Choice of method

The effect of a given treatment on the expression of endogenous genes can be measured at several molecular levels. Ideally, it is best to measure final functional protein activity, as this is the outcome of most biological and clinical significance; however, protein and mRNA levels are two other common reference points that are used to assess changes in gene expression. The risk associated with measuring gene expression at points other than protein activity is that many genes are regulated through post-transcriptional and post-translational mechanisms. For example, proteins that are incorrectly folded or erroneously modified/unmodified (*e.g.* by leader sequence cleavage, phosphorylation or glycosylation) may be inactive, but will still usually be detected by Western blot. Furthermore, mRNA levels do not always correlate well with protein; for example, the tissue distribution of C/EBP α and C/EBP β mRNA is far broader than that of detectable proteins (Williams *et al.*, 1991).

However, for technical reasons, when it is of interest to study all known human UGTs, it is currently most appropriate to measure changes in UGT expression at the mRNA level. Assessment of the levels of each individual UGT by function would require unique probe substrates for each, and whilst substrates that are glucuronidated solely by one UGT have been identified, the substantial overlap in substrates between UGT proteins still precludes the measurement of all family members in this way. Furthermore, the effects of heterodimerisation between UGT

forms on function are poorly characterised, and may affect the interpretation of results in cells expressing multiple UGTs. Likewise, because of the high similarity in amino acid sequence between UGT family members, especially within the genetic sub-clusters, it is not possible to distinguish between all UGT forms with currently available antibodies. In contrast, there are sufficient dissimilarities between the coding sequences of each UGT to allow the design of oligonucleotide pairs that only amplify one target gene. Therefore, the current methods of choice for measuring UGT expression changes are PCR-based. In the case of UGTs, it is generally assumed that the resulting data is meaningful, as multiple studies imply that increases in UGT mRNA correspond with increases in UGT protein, and increases in UGT protein correspond with increases in glucuronidation (Girard *et al.*, 2004; Sugatani *et al.*, 2004; Harrington *et al.*, 2006).

5.2 Methods

5.2.1. Generation of liver-enriched transcription factor expression plasmids

Each LETF was cloned into the pCMX-PL2 expression vector as described in detail in Chapter 2, section 2.3. In brief, PXR, HNF1 α , HNF1 β , HNF4 α and FoxA3 were cloned directly from human cDNA, while the remaining transcription factor sequences were sourced from either rat or mouse orthologues, as attempts to procure them from human material were unsuccessful. HNF6 was cloned from mouse liver cDNA and has 99% homology to human HNF6 at the amino acid level. FoxA1 and FoxA2 were sub-cloned from vectors containing the rat sequences for each gene, which have 92% and 96% homology with the respective human amino acid sequences. Rat C/EBP α and C/EBP β were also sub-cloned from previously constructed vectors, and have 93% and 71% identity to their orthologous human amino acid sequences.

5.2.2. Transfection of HepG2 cells and extraction of total RNA

HepG2 cells cultured as described in Chapter 2.2.1 were plated in 6-well plates at a density of 1×10^6 cells per well, 24 hours before transfection. Transfection with 5 µg of expression plasmid was achieved using Lipofectamine 2000 as described in section 2.2.10. All cells were harvested for total RNA at 72 hours post-plating (48 hours post-transfection, 42 hours after addition of rifampicin or 24 hours after addition of TSA). RNeasy Mini spin-columns were used to purify the total RNA according to the manufacturer's instructions as detailed in Chapter 2, section 2.2.3.1. All transfection/treatment combinations described were performed at least twice in independent experiments, with the exception of the 300 nM TSA-treated transfections, which were only performed once.

5.2.3. Trichostatin A treatment of HepG2 cells

Twenty-four hours post-transfection, the culture medium of each well was replaced with fresh medium containing 300 nM TSA, 3 μ M TSA or vehicle (1:1000 diluted ethanol). Non-transfected cells were also treated with TSA or vehicle, 48 hours post-plating.

5.2.4. Rifampicin treatment of HepG2 cells

Six hours after transfection of cells with pCMX-PXR, the culture medium of each transfected or control well was replaced with fresh medium containing vehicle (DMSO diluted 1:1000) or 10 μ M rifampicin. To ensure exposure to 10 μ M rifampicin for the whole incubation period, the culture medium and DMSO/rifampicin treatments were replaced again at 30 hours post-transfection.

5.2.5. Reverse transcription and quantitative real-time PCR

Column-purified total RNA was treated with DNase I, and reverse-transcribed into cDNA using the random hexamer method detailed in Chapter 2, sections 2.2.3.3 and 2.2.3.4. After RNase H treatment, one 25th of each cDNA sample was used for quantification of UGT transcripts, or one 800th for quantification of 18S rRNA. The generic methods for QPCR can be found in Chapter 2, section 2.2.6.6, while the specific primers, annealing temperatures and templates used to generate standard curves are detailed in Table 5.1. Each pair of PCR primers was validated for specificity towards only the desired gene by visual inspection of the PCR product after gel electrophoresis and by sequencing, prior to the commencement of this PhD candidature. This work was performed by Anne Rogers, Dr. Takahito Nishiyama and me. Likewise, the plasmid templates used to generate standard curves for each gene, as listed below, were constructed before the commencement of this PhD candidature by Dr. Takahito Nishiyama and Anne Rogers. The pEF-IRES and pBS derived vectors, as well as pCR-blunt-2B28, contain the entire coding regions of the indicated genes. The remaining vectors each contain a single copy of the PCR product generated when the listed primers are used against the corresponding gene.

It should be noted that this application of the QPCR method determines the relative, not the absolute, concentration of target transcripts. This is due to at least three reasons. Firstly, the apparent mRNA copy number is dependent on whether cDNA synthesis is primed with gene-specific or random hexamer primers (Zhang and Byrne, 1999). Secondly, cDNA synthesis of the target and reference templates may not occur at the same efficiencies (Zhang and Byrne, 1997), and thirdly, the PCR amplification efficiencies of plasmid standards and their corresponding cDNA

Fable 5.1:	Primers and annealing conditions used for qua	ntification of human UGT and	I 18S RNA trai	nscripts.	
Gene	Nucleotide Sequence (5'→3')	Nucleotide Position on Target RNA	Amplicon (bp)	Annealing temp (°C)	Standard
18S rRNA	F: CGATGCTCTTAGCTGAGTGT* R: GGTCCAAGAATTTCACCTCT*	<i>18S rRNA</i> : +761 to +780 <i>18S rRNA</i> : +974 to +955	214	58	pCR-blunt-18S
HNF1α	F: CCAACACAGGTGCCTCCACCCTGGT R: CCGTGTGGGGTGTACTGGGGCCACCT	<i>HNF1</i> α : +1280 to +1304 <i>HNF1</i> α : +1546 to +1523	267	58	pCMX-HNF1α
UGTIAI	F: TTTTGTCTGGCTGTTCCCACT* R: GAAGGTCATGTGATCTGAATGAGA*	<i>UGT1A1</i> : +368 to +388 <i>UGT1A1</i> : +618 to +595	251	58	pCR-blunt-1A1
UGT1A3	F: ATGGCAATGTTGAACAATATG R: GGTCTGAATTGGTTGTTAGTAATC	<i>UGT1A3</i> : +340 to +360 <i>UGT1A3</i> : +610 to +587	271	54	pEF-IRES-1A3
UGT1A4	F: ACGCTGGGCTACACTCAAGG* R: TCATTATGCAGTAGCTCCACACAA*	<i>UGT1A4</i> : +277 to +296 <i>UGT1A4</i> : +404 to +381	128	58	pBS-1A4
UGT1A6	F: CTTTTCACAGACCCAGCCTTAC* R: TATCCACATCTCTTGAGGACAG*	<i>UGT1A6</i> : +439 to +460 <i>UGT1A6</i> : +727 to +704	289	58	pCR-blunt-1A6
UGT1A7	F: TGGCTCGTGCAGGGTGGACTG R: TTCGCAATGGTGCCGTCCAGC	<i>UGT1A7</i> : +2 to +22 <i>UGT1A7</i> : +310 to +290	309	58	pEF-IRES-1A7

lable 5.1 col	ntinued.				
Gene	Nucleotide Sequence (5'→3')	Nucleotide Position on Target RNA	Amplicon (bp)	Annealing temp (°C)	Standard
UGT1A8	F: CTGCTGACCTGTGGCTTTGCT R: CCATTGAGCATCGGCGAAAT	UGT1A8: +49 to +69 UGT1A8: +294 to +275	246	58	pEF-IRES-1A8
UGT1A9	F: GAGGAACATTTATTATGCCACCG* R: GCAACAACCAAATTGATGTGTG*	UGT1A9: +643 to +665 UGT1A9: +760 to +739	118	58	pCR-blunt-1A9
UGT1A10	F: CCTCTTTCCTATGTCCCCAATGA# R: GCAACAACCAAATTGATGTGTG*	<i>UGT1A10</i> : +556 to +578 <i>UGT1A10</i> : +760 to +739	205	58	pEF-IRES-1A10
UGT2B4	F: TCTACTCTTAAATTTGAAGTTTATCCTGT* R: TCAGCCAGCAGCTCACCACAGGG*	<i>UGT2B4</i> : +208 to +236 <i>UGT2B4</i> : +485 to +463	278	58	pCR-blunt-2B4
UGT2B7	F: AGTTGGAGAATTTCATCATGCAACAGA* R: TCAGCCAGCAGCTCACCACAGGG*	<i>UGT2B7</i> : +254 to +280 <i>UGT2B7</i> : +485 to +463	232	58	pCR-blunt-2B7
UGT2B10	F: TGACATCGTTTTTGCAGATGCTTA* R: CAGGTACATAGGAAGGAGGGAA*	<i>UGT2B10</i> : +432 to +455 <i>UGT2B10</i> : +583 to +562	152	58	pCR-blunt-2B10
UGT2B11	F: CTTCCATTCTTTTTGATCCCAATGATG* R: TCAGCCAGCAGCTCACCACAGGG*	<i>UGT2B11</i> : +179 to +205 <i>UGT2B11</i> : +485 to +463	307	58	pCR-blunt-2B11

Table 21

223

Lable 5.1 co	ntinued.				
Gene	Nucleotide Sequence (5'→3')	Nucleotide Position on Target RNA	Amplicon (bp)	Annealing temp (°C)	Standard
UGT2B15	F: GTGTTGGGAATATTATGACTACAGTAAC* R: TCAGCCAGCAGCTCACCACAGGGG*	<i>UGT2B15</i> : +348 to +375 <i>UGT2B15</i> : +488 to +466	141	58	pCR-blunt-2B15
UGT2B17	F: GTGTTGGGAATATTCTGACTATAATATA* R: CAGGTACATAGGAAGGAGGGAA*	UGT2B17: +348 to +375 UGT2B17: +589 to +568	242	58	pCR-blunt-2B17
UGT2B28	F: TCTTTTGATCCCAATGACGCATT R: TCAGCCAGCAGCTCACCACAGGG*	UGT2B28: +186 to +209 UGT2B28: +485 to +463	300	58	pCR-blunt-2B28
*Primer sequen	ices obtained from Congui et al. (2002). #Primer sequence of	btained from Strassburg et al. (1997b	b). Mismatches betv	ween mRNA and pr	imers are indicated in

bold text. F: forward primer, R: reverse primer.

targets are unlikely to be the same (due to differences in secondary or tertiary structure, as well as differences in the sensitivity of double versus single stranded standards (Overbergh *et al.*, 1999)). To obtain precise mRNA copy numbers for each gene, it is necessary to spike the RNA preparations with synthetic RNA standards before cDNA synthesis (Bustin, 2000). Ideally, the target and reference genes should also be measured in the same tube to account for loading errors. However, as the purpose of this study was to measure change in UGT mRNA levels rather than the absolute concentration, it was deemed that the relative values this protocol would provide were both adequate and appropriate. Indeed, QPCR, as performed in this study, is currently the most common method used to measure the relative levels of specific mRNA transcripts between samples.

5.2.6. Cloning of the UGT1A3-9.4k and UGT1A4-5k promoters

Attempts to amplify the entire 9.4 kb DNA region separating the *UGT1A3* and *UGT1A4* first exons in one piece were unsuccessful. This was primarily due to the high similarity of the UGT1A3 primers to regions flanking the *UGT1A4* promoter. Because the entire *UGT1A4* promoter is only 5 kb, this sequence always amplified in preference to the longer *UGT1A3* sequence. Therefore, the following strategy was used to obtain a pGL3-1A3-9.4k clone.

Firstly, 50 ng of BAC clone 1308M2 DNA was used as template to amplify the 6.4 kb immediately upstream of the *UGT1A3* first exon using the ExpandTM Long Template PCR System, which utilises a mixture of *Taq* and *Tgo* DNA polymerases to optimise amplification range and fidelity (4.8×10^{-6} mutations per base pair per duplication). The primer site 6.4 kb downstream of the *UGT1A3* initiation codon was chosen to avoid repetitive DNA sequences in the *UGT1A3* promoter as well as

regions of homology with the *UGT1A4* promoter. The PCR reaction was performed in 1× "ExpandTM Buffer 3" with 500 μ M dNTPs, 300 nM each primer (1A3prom-6.4kNhe1: 5' AGCCAT<u>GCTAGC</u>TCATTAAGTGGAAGTGGATCA 3', and 1A3UTRNhe1: 5' AGCCAT<u>GCTAGC</u>CTCAGCAGAAGACACGGACA 3') and 3.75 Units ExpandTM DNA polymerase mix. The PCR parameters were: an initial step of 94°C for 2 minutes; followed by 10 cycles of 94°C for 10 seconds, 60°C for 30 seconds and 68°C for 10 minutes; followed by a further 25 cycles of 94°C for 10 seconds, 60°C for 30 seconds and 68°C for 10 minutes plus 15 seconds for each successive cycle; and a final extension at 68°C for seven minutes. The resulting PCR product was ligated into the TA cloning vector pCR-2.1 and transformed into TOP10 *E. coli* to generate pCR-1A3-6.4k.

The identity of the cloned PCR product was confirmed as the *UGT1A3*-6.4kb promoter by sequencing of both ends. The *UGT1A3*-6.4k insert was then excised from the vector backbone by *Nhe*I restriction digest (after the pCR plasmid sequence had been cut with *Xho*I and *Spe*I and CIP-treated to prevent it from participating in any downstream ligation reactions), ligated into *Nhe*I/CIP treated pGL3-basic and transformed into DH5 α *E. coli*. This generated reporter vectors with inserts in the forwards and reverse orientations: pGL3-1A3-6.4kfor and pGL3-1A3-6.4krev.

To obtain the remaining 3 kb of the full *UGT1A3* promoter, the primers 1A3-9.4XbaINheI (5' AGCCAT<u>TCTAGAGCTAGCGTCTGTATTGGTGCCTTC</u> 3') and 1A3-6.4revBsrG1 (5' GGAAGAAGAAGTTGGTAC<u>TGTACAG</u> 3') were used in another ExpandTM PCR reaction, using the same reaction conditions as described for the *UGT1A3*-6.4kb product, but with 55°C annealing. The resulting PCR product was restricted with *Xba*I and *Bsr*GI, ligated into *XbaI/Bsr*GI-cut pCR-1A3-6.4k and transformed into DH5 α *E. coli* to generate pCR-1A3-9.4k. The full *UGT1A3*
promoter sequence was then excised from *XbaI/SpeI/*CIP-treated pCR-1A3-9.4k using *Nhe*I, ligated non-directionally into *Nhe*I/CIP-treated pGL3-basic and transformed into DH5α *E. coli*. To confirm the integrity of the constructed *UGT1A3*-9.4k promoters, forward and reverse orientation clones were tested for the expected restriction patterns after digestion with *Nhe*I, *Hind*III or *Pst*I, and the ends of the inserts were sequenced.

To obtain a clone of the whole DNA sequence spanning the region between the *UGT1A4* and *UGT1A5* first exons, BAC DNA from clone 1308M2 was digested with *XbaI* and *NheI*, and subjected to electrophoresis through a 0.8% (w/v) agarose gel. All DNA fragments of length 5 to 6 kb were excised, purified using the QIAquick Gel Extraction kit and ligated into *XbaI*/CIP-treated pBSII. Clones containing BAC fragments that encompassed the *UGT1A4* promoter were detected by PCR, using primers 1A3/4rev-common and 1A3/4prom-0.5k (Table 3.1) as described in section 2.2.6.4, with cycling conditions of 95°C for 4 minutes, followed by 30 cycles of 95°C for 30 seconds, 50°C for 30 seconds and 72°C for 1 minute, and a final extension at 72°C for 5 minutes. A pBS-1A4-6k clone was then used as template for the following cloning PCR, rather than the 1308M2 BAC DNA, due to better yield and lack of non-specific amplification.

To clone the full *UGT1A4* promoter without the extra sequences still present in pBS-1A4-6k, 50 ng of pBS-1A4-6k was used as template in a *PfuTurbo* PCR reaction using primers 1A4UTRXhoI (Table 3.1) and 1A4prom-5kNheI (5' AGCCAT<u>GCTAGCGTCTGTATTGGTGCCTTT</u> 3'), set up as per Chapter 2, section 2.2.6.3. The cycling conditions were: 95°C for 1 minute; 35 cycles of 95°C for 1 minute, 55°C for 1 minute and 68°C for 15 minutes; and a final extension at 68°C for 10 minutes. The resulting PCR product was then cloned directly into pCR-

Blunt and shuttled to pGL3-basic utilising the *Xho*I and *Nhe*I sites engineered into the primers. Finally, both ends of the cloned fragment were sequenced to confirm its identity as the region immediately upstream of the *UGT1A4* first exon.

5.2.7. Transfection of HepG2 cells and dual-luciferase assay

Transient co-transfections of 0.5 μ g of firefly luciferase reporter plasmid, 0.25 μ g of transcription factor expression plasmid and 25 ng pRL-Null were performed as described in Chapter 2, section 2.2.10, using HepG2 cells seeded at a density of 2 × 10^5 cells per well in 24-well plates. Cells were lysed 48-hours post-transfection using passive lysis buffer, and the lysates assayed for firefly and *renilla* luciferase activity as described in Chapter 2, section 2.2.11.

5.2.8. Hepatocyte nuclear factor 1a Western blot

Joanna Treloar performed the HNF1 α Western blot presented in Figure 5.7, using lysates from untreated and 3 μ M TSA-treated HepG2 cells. The protocol used is detailed in Chapter 2, section 2.2.16.

5.3 Results and discussion

5.3.1. Validation of 18S rRNA as a suitable internal control for HepG2 cells over-expressing liver-enriched transcription factors and/or treated with trichostatin A or rifampicin

When performing quantification of mRNA by reverse transcription and QPCR, it is necessary to control for any inter-sample differences in amplifiable cDNA that are inadvertently introduced during preparation. Significant sources of error are reported to include variable quality of RNA after extraction (which can then significantly affect both the determination of RNA concentration by spectrophotometric methods and the efficiency of cDNA synthesis), and compounded operator error due to multistep protocols (Bustin, 2000; Bustin, 2002; Huggett *et al.*, 2005). These errors are typically accounted for by normalising the gene(s) of interest against a gene whose expression (ideally) remains static between samples regardless of treatment, a so-called "house-keeping" gene. This can be effective because the reference transcript is subjected to all the steps of RNA extraction, quantification and reverse transcription as the genes to be investigated. However, the three most popular candidate reference genes, β -actin, GAPDH and 18S rRNA, have all been reported to vary significantly under certain experimental conditions and it is therefore imperative to determine the suitability of a chosen reference under the conditions used.

In our laboratory in the past, we have found that 18S rRNA typically varies less than β -actin or GAPDH mRNA in human cell lines subjected to transfection with transcription factor expression plasmids or treated with various chemicals. This observation is supported by several papers that have explored the effect of various experimental conditions on the expression of these three genes (Schmittgen and Zakrajsek, 2000; Selvey *et al.*, 2001; Bas *et al.*, 2004). Also, cell-cycle phase affects GAPDH expression (Mansur *et al.*, 1993) and two factors used in this study, C/EBPa and C/EBP β , can have profound effects on cell-cycle progression (Schrem *et al.*, 2004). Furthermore, C/EBPa has been found to bind to the human GAPDH promoter (Claeyssens *et al.*, 2003). With these factors in mind, 18S was chosen as the most likely suitable reference for the following LETF and TSA experiments.

To validate the choice of 18S rRNA as a suitable reference, 18S transcript levels were compared to total RNA concentration for all transfection, rifampicin-treatment and TSA-treatment combinations used in this study (Figure 5.2). When 18S rRNA







Figure 5.2: 18S rRNA levels in treated HepG2 cells as a proportion of total extracted RNA. A. HepG2 cells were treated with: 1:1000 ethanol, 300 nM trichostatin A (TSA) or 3 μ M TSA for 24 hours; 1:1000 DMSO or 10 μ M rifampicin for 42 hours; or transfected with plasmid expressing pregnane X receptor and subsequently treated with rifampicin or DMSO. B. HepG2 cells were transfected with control plasmid or one of nine vectors expressing the indicated liver-enriched transcription factors and subsequently incubated in the presence of 1:1000 ethanol, 300 nM or 3 μ M TSA. Total RNA harvested from cells exposed to each treatment was analysed by quantitative PCR for 18S rRNA content and the results expressed as the mean 18S content/ μ g RNA relative to untreated cells (n = 3). Error bars indicate one standard deviation.

levels were analysed as a function of total RNA compared to untreated cells, it was found that there was a maximum of 12% change in apparent 18S rRNA levels in response to 300 nM or 3 µM TSA, or any combination of 10 µM rifampicin and PXR treatment (Figure 5.2A). Furthermore, it was found that the maximum change in 18S transcripts/µg total RNA, relative to untreated control cells, was less than 1.6-fold for LETF-transfected cells, regardless of TSA treatment (Figure 5.2B). In addition, no transcription factor or TSA treatment consistently altered 18S levels across all samples, leading to the assumption that 18S rRNA is not truly regulated by any of the experimental conditions tested and the small differences observed are most likely the result of genuine experimental variation. Similar levels of variation in the purity of RNA extracted with Qiagen RNeasy columns have been previously reported (Bustin, 2002); thus, the variation seen is within the known experimental error for similar systems. Therefore, 18S was shown to be an acceptable reference gene for this experimental system. However, it should be noted that these results do not indicate whether 18S would be a suitable reference gene for the same experiment in cell lines other than HepG2, as TSA was been shown to affect 18S rRNA levels in LNCaP cells and mouse prostate (Mogal and Abdulkadir, 2006).

5.3.2. Basal levels of UGT1A and UGT2B mRNA transcripts in HepG2 cells

HepG2 cells are considered to be one of the most differentiated human liver celllines available (Knowles *et al.*, 1980; Ishiyama *et al.*, 2003). However, the UGT content of these cells has not been well defined. Since immortalised cells never possess the exact same gene expression profile as the primary cells from which they were derived, (indeed HepG2s are known to be poor expressers of CYP enzymes (Rodriguez-Antona *et al.*, 2002)) and the expression profile of HepG2 cells is likely to be dependent on culture conditions (as there is considerable disagreement in the literature about the expression of certain genes in these cells), HepG2 RNA was assessed for UGT mRNA content under basal conditions. All UGT1A and UGT2B forms, barring UGT1A5, were tested for expression in HepG2 cells grown under the conditions prescribed by the ATCC (see Chapter 2, section 2.2.1). It was found that UGT2B7 and UGT2B10 were by far the most highly transcribed UGTs in HepG2 cells (approximately 5,000 mRNA copies per 1×10^9 18S rRNA molecules), followed by UGT2B11, UGT1A1, UGT2B4 and UGT1A6, which were all readily detectable (200-800 mRNA copies per 1×10^9 18S rRNA molecules). UGT1A9, UGT2B15, UGT2B17 and UGT1A3 were also detectable at low levels (20-50 mRNA copies per 1×10^9 18S rRNA molecules), however, the basal levels of UGT1A4, UGT1A7, UGT1A8, UGT1A10 and UGT2B28 were below the reliable detectable limit of this assay at less than 10 mRNA copies per 1×10^9 18S rRNA molecules (Figure 5.3). Three of these genes, UGT1A7, UGT1A8 and UGT1A10 are considered to be strictly extrahepatic in their expression (Strassburg et al., 1997b; Mojarrabi and Mackenzie, 1998; Strassburg et al., 1998a). Therefore, it is likely that the low levels detected by this sensitive technique reflect illegitimate transcription; the principle that transcription of any gene can be detected in any cell given sufficient sensitivity of detection (Chelly et al., 1989; Chelly et al., 1991). Consequently, it would seem that UGT1A4 and UGT2B28 are also not truly expressed in HepG2 cells under the basal culture conditions used.

This study shows that the UGT expression profile of UGTs in HepG2 cells differs significantly from that of primary hepatocytes. A similar study using primary tissue showed that UGT2B4, UGT2B7, UGT1A4 and UGT1A9 are the most prevalent UGTs in liver, followed by UGT2B15, UGT2B10, UGT1A1, UGT1A6 and UGT1A3. UGT2B11 and UGT2B17 were the most difficult hepatic UGTs to detect



Figure 5.3: Expression of UGT mRNA in HepG2 cells under basal culture conditions. Total RNA, extracted from untreated HepG2 cells, was analysed by QPCR for UGT1A and UGT2B mRNA content. Results are expressed as mean mRNA levels relative to 18S rRNA (n = 3) plus one standard deviation.

in primary hepatocytes (Congiu *et al.*, 2002). UGT2B28 was not tested in the work of Congiu *et al.* (2002), but is also expressed in the liver (Levesque *et al.*, 2001). It is not known how the absolute levels of UGT transcripts compare between HepG2 and hepatocytes. However, for the primer pairs and PCR conditions that are the same between the HepG2 study and the work of Congiu *et al.* (2002) (*i.e.* UGT1A1, UGT1A4, UGT1A9 and all UGT2Bs except UGT2B28), the relative efficiency of each PCR should be the same, making it possible to directly compare the rank orders of UGT expression in the two sample sets. Using UGT2B7 as the reference, HepG2 cells have lost substantial relative expression of all UGTs except UGT2B10 and

possibly UGT2B11, with the greatest losses suffered by UGT2B4, UGT1A4 and UGT1A9. Reasons for these changes to the ratios of UGT forms expressed could include changes in the transcription factor profile in the HepG2 hepatoma cells, or loss of exposure to hormones or other chemical signals that would normally be present in the whole organ/organism and may up-regulate UGT transcription *in vivo*. However, this work shows that most hepatic *UGT* genes are transcriptionally active in HepG2 cells, at least to some extent, with the exception of *UGT1A4* and *UGT2B28*. Therefore, it is reasonable to expect that most of the hepatic *UGT* promoters are not situated in closed chromatin in HepG2 cells and are therefore likely to be accessible to over-expressed transcription factors.

5.3.3. Effect of rifampicin-liganded pregnane X receptor on endogenous UGT expression in HepG2 cells

To study the effect of rifampicin-liganded PXR expression on hepatic *UGT* expression, HepG2 cells were transfected with either empty control plasmid or PXR-expressing plasmid, and subsequently treated with either vehicle or 10 μ M rifampicin. QPCR analysis of total RNA from treated cells showed that UGT1A1 and UGT1A3 mRNA transcripts were increased 13-fold relative to 18S rRNA in cells transfected with PXR and then treated with rifampicin, when compared to untreated controls (*P* < 0.001). Rifampicin alone also increased the level of both of these transcripts in HepG2 cells by more than 2-fold (UGT1A1: *P* = 0.004 and UGT1A3: *P* = 0.039) (Figure 5.4). This is consistent with the presence of endogenous PXR expression in HepG2 cells (Aninat *et al.*, 2006). In contrast, the mRNA levels of all other hepatic *UGT* genes were only increased 2-fold or less by the PXR/rifampicin combined treatment, with the only significant response being from *UGT1A6* (2.0-



Figure 5.4: Endogenous UGT mRNA levels in HepG2 cells treated with 10 μ M rifampicin and/or over-expressed pregnane X receptor. Total RNA, extracted from HepG2 cells treated with 1:1000 ethanol, 10 μ M rifampicin and/or over-expressed PXR, was analysed by reverse transcription QPCR for altered mRNA levels of all hepatic members of the human *UGT1A* and *UGT2B* families. Results have been normalised to 18S content and are expressed as the mean change in mRNA level (n = 3), relative to untreated cells. Error bars indicate one standard deviation. *P* values for the indicated comparisons are *P < 0.001, $\ddagger P = 0.004$, $\ddagger P = 0.011$ and $\ddagger P = 0.039$.

fold, P = 0.011) (Figure 5.4). The mRNA levels of UGT1A7, UGT1A8 and UGT1A10 also did not change (data not shown). This confirmed previous semiquantitative PCR work in our laboratory that showed that UGT1A1 and UGT1A3 mRNAs were the most substantially increased UGT transcripts in HepG2 cells in the presence of over-expressed, rifampicin-liganded PXR (Gardner-Stephen *et al.*, 2004). It also showed that the transfection, RNA extraction, reverse transcription and QPCR protocols had all been performed successfully.

As a model for the human hepatocyte, the HepG2 cells used herein are likely to be useful only for studying the well established effect of PXR on the regulation of *UGT1A1* and *UGT1A3*. Although, increases in mRNA level or UGT activity have been reported for UGT1A4 (4.2-fold in primary hepatocytes, and present but not quantified in Tg-UGT1 transgenic mice and another HepG2 line), UGT1A6 (1 to 4fold in primary hepatocytes, and present but not quantified in Tg-UGT1 transgenic mice), UGT1A9 (1 to 1.7-fold in primary hepatocytes, and present but not quantified in Tg-UGT1 transgenic mice), UGT2B7 (0.8 to 4-fold in primary hepatocytes), UGT2B15 (1.2 to 1.5-fold in liver) and UGT2B28 (1.6-fold in liver) in the presence of PXR ligands, particularly rifampicin and carbamazepine (Rae et al., 2001; Gardner-Stephen et al., 2004; Soars et al., 2004; Chen et al., 2005a; Oscarson et al., 2006), there was no change in these genes, except UGT1A6, in this study. This was despite the presence of excess PXR protein, which exaggerates the UGT1A1 and UGT1A3 responses (compare 5.2 and 3.8-fold in primary hepatocytes respectively (Rae et al., 2001), to 13-fold in PXR-transfected HepG2 cells (Figure 5.4)). One reason for the lack of response in HepG2 cells may include a deficiency in basal transcription factors required to support PXR-mediated activation, especially in the case of UGT1A4 and UGT2B28, since both are essentially unexpressed in HepG2 cells. Alternatively, genotype may play a role in the PXR-responsiveness of some UGT genes. The UGT1A9 and UGT2B7 PXR-responses of primary cells have been reported to be highly dependent on the individual from whom the hepatocytes were isolated, with some hepatocyte preparations being completely unresponsive (Soars et al., 2004). Therefore, it is possible that HepG2 cells are genetically incapable of a full UGT-response to PXR.

To further study the molecular mechanisms that confer PXR-responsiveness to *UGT* genes other than *UGT1A1* and *UGT1A3*, it will be necessary to develop additional models. Possibilities include other hepatic cell lines that better express UGTs (especially UGT1A4), or exhibit PXR-responsiveness for more *UGT* genes. One

possibility is the newly characterised human hepatoma cell line HepaRG, which exhibits hepatocyte-like expression of several CYPs, UGT1A1 and PXR when cultured in the presence of DMSO (Aninat *et al.*, 2006). Methods that improve the differentiation state of HepG2 cells, such as exposure to retinoic acid (Falasca *et al.*, 1999), may also induce more hepatocyte-like behaviour in response to PXR ligands. However, these results suggest that HepG2 cells are a suitable system for identifying and characterising the molecular determinants of the *UGT1A3* PXR-response, in much the same manner as has been achieved for *UGT1A1* (Sugatani *et al.*, 2004). Further work towards identifying the mechanism by which PXR regulates *UGT1A3* is presented below.

5.3.4. UGT1A3 promoter-reporter constructs lack ligand-dependent PXR responsiveness in HepG2 cells

To locate the PXR NRRE responsible for the PXR-responsiveness of the human UGT1A3 gene, the 3.3 kb UGT1A3 promoter-reporter construct described in Chapter 3 was initially tested for increased expression of the reporter gene in the combined presence of over-expressed PXR and 10 μ M rifampicin. However, although a 3.5-fold increase in relative luciferase activity (P < 0.001, Figure 5.5) was observed when the UGT1A3-3.3k promoter was treated with PXR and rifampicin, a 2.6-fold (P < 0.001) increase was also observed with unliganded PXR. Therefore, the ligand-mediated PXR-response obtained for the UGT1A3-3.3k promoter was only 1.4-fold, which although statistically significant (P < 0.001), was insufficient to explain the response seen for the endogenous gene and was deemed unlikely to be biologically relevant.



Figure 5.5: PXR up-regulates the UGT1A3 and UGT1A4 promoters in vitro in a ligand-independent manner. HepG2 cells were transfected with 25 ng pRL-Null plus 0.5 µg reporter plasmids carrying 3.3 kb, 6.4 kb or 9.4 kb of the UGT1A3 promoter, or 5 kb of the UGT1A4 promoter, with or without co-transfection of 0.25 µg pCMX-PXR. Transfected cells were also treated with solvent (1:1000 DMSO), or 10 µM rifampicin. Results presented are the mean (n = 3) increase in luciferase reporter activity, relative to the *renilla* internal control, over concurrent pGL3-basic transfections. Error bars indicate one standard deviation. P values for the indicated comparisons are *P < 0.001 and $\dagger P = 0.001$.

Since the *UGT1A1* PXR NRRE is located over 3285 nucleotides upstream of the *UGT1A1* start codon (Xie *et al.*, 2003), and PXR-binding sites had been predicted at -6930 and -8040 bp of the *UGT1A3* promoter (Vyhlidal *et al.*, 2004), it was reasoned that the *UGT1A3* PXR NRRE may lie upstream of the tested section of promoter. Therefore, two longer fragments of the DNA upstream of the UGT1A3 promoter were cloned: the proximal 6.4 kb, and the entire 9.4 kb region between the *UGT1A3* initiation codon and the *UGT1A4* first exon 3' splice site. The entire 5 kb region between the *UGT1A4* initiation codon and the *UGT1A5* first exon was also cloned for comparative purposes. However, no ligand-dependent PXR response was

obtained for either gene, regardless of the construct tested (Figure 5.5). Although some increase in luciferase activity was observed in response to the presence of overexpressed PXR without exogenously supplied ligand for pGL3-1A3-3.3k (2.6-fold, P< 0.001), pGL3-1A3-6.4kfor (5.5-fold, P < 0.001), pGL3-1A3-9.4kfor (6.2-fold, P <0.001) and pGL3-1A4-5k (3.6-fold, P < 0.001), the vectors containing reverseorientation inserts had similar fold increases in reporter gene activity when cotransfected with pCMX-PXR (pGL3-1A3-6.4krev: 3.4-fold, P = 0.001 and pGL3-1A3-9.4krev: 4.5-fold, P = 0.001) (Figure 5.5). Therefore, it would appear that this response is most likely an artefact of the experimental system.

The somewhat surprising absence of authentic response by the UGT1A3 promoter to PXR and rifampicin was consistent over several experiments and plasmid preparations. Because of good basal expression of the luciferase reporter, and the response of the endogenous gene (section 5.3.3), it is reasonable to assume that the transfections were effective, that no experimental components were degraded and that endogenous PXR ligands in HepG2 cell culture are at insufficient levels to explain the observed results. However, the unlikely possibility of experimental failure could, in future, be controlled for by the inclusion of a promoter known to be responsive in the same assay, such as the published UGT1A1 (Xie et al., 2003; Sugatani et al., 2005a) or CYP3A4 (Hustert et al., 2001) constructs. Interestingly, all three of these studies used either constitutively active PXR, or PXR-responsive enhancers ligated directly to proximal promoters, to demonstrate PXRresponsiveness. Thus, it may transpire that the combination of liganded PXR and full UGT1A3 promoter-reporter plasmid is unsuitable for detecting the expected PXRresponse. Nevertheless, more interesting possible explanations for the lack of PXRresponse by the UGT1A3 promoter constructs include that the PXR NRRE for UGT1A3: a) is genuinely inactive in the cloned UGT1A3 allele (or has been inactivated by PCR mutation); b) requires an appropriate chromatin structural context to be effective; or c) still lies outside the cloned promoter region. The first scenario could simply be addressed by re-cloning the UGT1A3 promoter inserts from a different template (*e.g.* HepG2 genomic DNA) and repeating the experiment. The second possibility, as to whether the chromatin structure of the UGT1A3 promoter contributes to its regulation by PXR, could be further investigated by generating a stable HepG2 cell pool where the pGL3-1A3-9.4k reporter construct is integrated into the genome. Examination of the final option, however, would be more involved. Additional portions of the UGT1A locus could be screened for PXR-binding elements by the method of Xie *et al.* (2003) without the need for generating large promoter clones. However, if a PXR NRRE cannot be identified upstream of the UGT1A3 first exon, or immediately downstream, it may be necessary to consider whether the UGT1A1 PXR NRRE is also able to affect the UGT1A3 gene.

Enhancer elements, such as the *UGT1A1* PBREM cluster of transcription factor binding sites, in which the *UGT1A1* PXR NRRE resides, can exist up to tens of thousands of nucleotides upstream or downstream of their target promoter. They can also be present in either orientation in the DNA, and are sometimes shared between neighbouring genes. Enhancers can be prevented from promiscuous regulation of inappropriately receptive promoters in their vicinity by insulator and silencer elements – however, no studies have been done to determine whether the *UGT1A1* PBREM functions in a uni- or bi-directional manner (Hatzis and Talianidis, 2002; de Laat and Grosveld, 2003; West and Fraser, 2005). If the *UGT1A1* PBREM is necessary for the regulation of *UGT1A3* by PXR, it presumably must interact physically with the *UGT1A3* proximal promoter in PXR/rifampicin treated HepG2 cells. Chromosome conformation capture analyses could be used to determine whether the spatial proximity of the *UGT1A1* PBREM and *UGT1A3* promoter is affected by PXR/rifampicin treatment (Dekker, 2006), and hence, whether a functional interaction is likely. If it was found that the *UGT1A1* PBREM has a role in the expression of UGT1A3 as well as UGT1A1, then functional polymorphisms in this region (Sugatani *et al.*, 2002) may also be relevant to interindividual variation in UGT1A3 expression.

5.3.5. Effect of trichostatin A treatment on endogenous UGT expression in HepG2 cells

Histone acetylation, regulated by the opposing activities of HATs and HDACs, plays an important role in the control of gene transcription. Hyperacetylation of histones appears to create a permissive environment for gene expression by relaxing chromatin structure and making the DNA more accessible to modifying enzymes, transcription factors and RNA polymerase. Highly acetylated histones are generally associated with actively transcribed genes and genes poised for transcription, but not silent genes (Grunstein, 1997; Marks *et al.*, 2000; Schrem *et al.*, 2002).

Genes repressed by histone deacetylation are resistant to the influence of certain transcription factors that would otherwise activate their promoters (Honda *et al.*, 2006). In tissue-specific expression of genes, this can be a legitimate mechanism to prevent their improper expression in the wrong cell types or at the wrong developmental stage. However, in immortilised cell lines, where some genes are inappropriately repressed, aberrant histone acetylation may interfere with what would otherwise be a genuine response to an exogenously supplied transcription factor. To assess whether any *UGT* genes are likely to be repressed by histone deacetylation in HepG2 cells, cultures were treated with TSA and checked for

changes in the levels of each UGT mRNA, except UGT1A5. TSA is a specific and potent HDAC inhibitor (Yoshida *et al.*, 1990), which has previously been shown to change the expression profile of HepG2 cells (although no *UGT* data was reported) (Chiba *et al.*, 2004). Initially, TSA was added at a concentration of 3 μ M, to emulate the work of Bort and colleagues, which showed that TSA was necessary for FoxA3 to access the HepG2 endogenous *CYP2C8* and *CYP2C19* promoters (Bort *et al.*, 2004). However, considerable cytotoxicity was observed at this TSA concentration after 24 hours, so the experimental protocol was altered to also include a lower TSA concentration of 300 nM.

Exposure of HepG2 cells to 300 nM TSA greatly increased the mRNA levels of three UGTs: UGT1A1 (18.7-fold, P < 0.001), UGT1A3 (13.3-fold, P = 0.006) and UGT2B17 (9.1-fold, P < 0.001) (Figure 5.6). Other UGTs that had small responses (but greater than 2-fold) were: UGT1A4 (increased 2.2-fold, P = 0.001), UGT1A9 (decreased 2.5-fold, P = 0.42), UGT2B4 (increased 2.2-fold, P = 0.004), UGT2B11 (increased 2.1-fold, P = 0.015) and UGT2B15 (increased 2.1-fold, P = 0.001).

Exposure to 3 μ M TSA had a stronger positive effect on the levels of UGT1A1 and UGT1A3 mRNA than 300 nM TSA, giving rise to 46.4-fold (P < 0.001) and 33.9-fold (P = 0.003) increases respectively (Figure 5.6). Conversely, the increase in UGT2B17 transcripts seen after exposure to 300 nM TSA was muted for 3 μ M TSA, dropping to only a 2.7-fold increase (P = 0.027). Other UGT transcripts that were increased in HepG2 cells after treatment with 3 μ M TSA were UGT2B11 (3.2-fold, P = 0.009) and interestingly, three genes barely detectable in untreated HepG2 cells. UGT1A4 was increased 6.9-fold (P = 0.036), UGT1A10 was increased 4.5-fold (P = 0.002), and UGT2B28 was increased 5.8-fold (P = 0.024). In all three cases, treatment with 3 μ M TSA caused expression to exceed 30 mRNA copies per 1 × 10⁹



Figure 5.6: Effects of trichostatin A on UGT mRNA levels in HepG2 cells. HepG2 cells were treated for 24 hours with either 1:1000 diluted ethanol, 300 nM TSA or 3 μ M TSA and harvested for total RNA. The levels of all human UGT1A and UGT2B mRNAs, except UGT1A5, were assessed by QPCR, normalised to 18S rRNA content, and expressed as the mean change in UGT mRNA (n = 3) when compared to untreated cells. Error bars indicate one standard deviation. *P* values for the indicated comparisons are **P* < 0.001, †*P* = 0.001, ††*P* = 0.002, †††*P* = 0.003, ‡‡*P* = 0.006, ‡‡‡*P* = 0.009 and §*P* > 0.02 but < 0.05.

18S rRNA molecules, which is comparable to the basal levels of UGT1A9, UGT2B15, UGT2B17 and UGT1A3 in HepG2 cells. There were also several significant losses of UGT expression in 3 μ M TSA-treated HepG2 cells: UGT1A6 was decreased 6.7-fold (P = 0.003), UGT1A9 was decreased 5-fold (P = 0.001), UGT2B7 was decreased 3.4-fold (P = 0.034) and UGT2B10 was decreased 30.3-fold (P < 0.001).

The effects of TSA on UGT transcript levels in HepG2 cells indicate that several *UGTs* are repressed by the activity of HDACs. The most likely mechanism for this observation is through histone deacetylation, although it should be noted that HDACs can also directly deacetylate a few select transcription factor proteins such as GATA-1 and p53, decreasing their activity (Huo and Zhang, 2005). More importantly however, even assuming the involvement of histone acetylation, it must

be recognised that this data does not distinguish whether the substantial derepression of several UGT genes is a direct or indirect effect of TSA. TSA treatment relieves chromatin-mediated repression across the entire genome, altering the expression of many genes. One study using 670 nM TSA showed that of 4608 liver and gastric genes tested, 187 were increased and 63 were decreased in HepG2 cells (Chiba et al., 2004). Therefore, it is reasonable to expect that some affected genes will encode transcription factors, and that their downstream promoter targets could consequently also be altered. In support of this, cultured primary hepatocytes treated with 25 µM TSA retained expression of HNF4 α and C/EBP α , two transcription factors that are usually lost over time in untreated hepatocyte cultures (Henkens et al., 2007). Furthermore, HepG2 cells treated with 250 nM TSA had modest increases (approximately 2-fold) in HNF4 α and FoxA1 mRNAs, while HNF1 α and C/EBP α levels were essentially unchanged (Yamashita et al., 2003). However, despite not knowing the exact mechanism by which TSA alters UGT transcript levels, this experiment showed that TSA treatment is a potentially useful tool for exploring the interaction of LETFs with endogenous UGT genes in HepG2 cells. Both concentrations tested created a more permissive environment for the transcription of at least three UGTs. Thus, TSA treatment may allow transcription factors better access to these UGT promoters, change the activity of important transcription factors, and/or cause the expression of one or more factors/co-factors required to cooperate in the regulation of UGT genes. Therefore, in the following experiments, LETFs were expressed both in the absence and presence of TSA to assess their potential as UGT regulators.

Although it could not be covered by the scope of this thesis, it would be valuable to further investigate the mechanisms behind the profound increase in UGT1A1,

UGT1A3, and UGT2B17 mRNA levels by TSA in HepG2 cells. Such a study would be likely to provide further insights into the transcriptional controls operating on these genes. Worthwhile experiments would include tests designed to separate the contributions of the local effects of chromatin deacetylation at the *UGT* loci from the effects caused by changes in transcription factor expression and activity, such as:

- a) Nuclease sensitivity mapping (Aronow *et al.*, 1995; Caslini *et al.*, 2006) and anti-acetylated histone chromatin immunoprecipitation assays (Chiba *et al.*, 2004) to assess the chromatin structure of HepG2 *UGT* promoters with and without TSA treatment. The former assesses the accessibility of DNA to endonucleases and is a measure of the degree of chromatin condensation; the latter tests the acetylation status of the nucleosomes integrated into the promoter of interest.
- b) TSA treatment of HepG2 cells transiently transfected with UGT promoterreporter plasmids (Kwon *et al.*, 2006; see also section 5.3.7.1). A plasmid response to TSA that changed abruptly in a progressive promoter-deletion experiment would indicate specific target gene sequences that were required for the response. Successful site-directed mutagenesis of the responsible nucleotides would strengthen such a hypothesis. Identification of the target elements would help facilitate discovery of the required mediators, which could then be tested for changes in expression or activity.
- c) DNase I footprinting to detect TSA-induced changes in the binding patterns of nuclear proteins to PCR-amplified *UGT* promoter segments. The nucleotide sequence of newly protected elements may convey sufficient information to identify altered transcription factor(s).

- d) Comparison of untreated and TSA-treated HepG2 nuclear extracts for their ability to drive transcription in *in vitro* chromatin assays. Our laboratory is currently developing *in vitro* chromatin assays to study *UGT* promoter function: this system could also be used to study the effects of TSA-mediated changes in nuclear proteins on transcription from untreated chromatin. TSAinhibition of HDACs has been shown to be reversible on dialysis (Kijima *et al.*, 1993); thus it would be possible to completely uncouple the direct and indirect effects of TSA treatment on UGT transcription.
- e) QPCR or microarray studies to identify genes with altered mRNA levels in TSA-treated HepG2 cells. Although correlation does not indicate causality, transcription factors and co-factors that have altered expression in TSA-treated cells would make logical candidates for further investigation as *UGT* regulators. While QPCR is cheaper, and generally more sensitive and specific than microarrays (Lucas *et al.*, 2005; Senkel *et al.*, 2005), it is also a less efficient method of screening potential regulators of a gene, and is necessarily biased towards better characterised genes. To begin the process of investigating the effect of TSA on the expression of transcription factors in HepG2 cells, HNF1a levels in treated and untreated cells were investigated by quantitative PCR (see section 5.3.6).

In addition to these mechanistic studies, it would also be prudent to investigate whether TSA and other HDAC inhibitors are able to alter UGT expression in primary hepatocytes and hepatocellular carcinomas. Only a subset of genes induced by TSA in HepG2 cells is also up-regulated in human hepatocytes (Chiba *et al.*, 2004). However, the responses of *UGT1A1* and *UGT1A3* are approximately three times greater than any other response reported for HepG2 cells, and a response *in*

vivo could have clinical significance. Valproic acid, a relatively weak HDAC inhibitor (Phiel et al., 2001; Furchert et al., 2007), does not significantly increase UGT1A1 glucuronidation of the irinotecan metabolite SN-38 in patients receiving both pharmaceuticals (Raymond et al., 2003). However, more potent histone deacetylase inhibitors, such as suberoylanilide hydroxamic acid and MS-275 (a benzamide), are currently in clinical trials as anti-cancer therapeutics (Hess-Stump, 2005; Kelly et al., 2005; Duvic et al., 2007; Furchert et al., 2007), and suberoylanilide hydroxamic acid has been suggested for combined therapy with irinotecan for hepatocellular carcinoma (Ocker et al., 2005). So, the potential for clinically relevant HDAC effects on glucuronidation exists. It is also possible that HDACs could alter the local pharmacological properties of drugs within tumours. UGT1A1 mRNA is repressed in the majority of colorectal cancers, but is restored by treatment with HDAC inhibitors (Gagnon et al., 2006). Therefore, HDAC treatment could potentially increase the resistance of malignant cells to therapeutics that are inactivated by glucuronidation, such as irinotecan. UGT1A1 and UGT1A3 mRNA levels are also reduced in hepatocellular carcinoma (Strassburg et al., 1997a), and in light of the presented HepG2 data, may respond to HDACs. Therefore, further and more deliberate investigation is warranted into the effects of HDAC inhibitors on glucuronidation in humans.

5.3.6. Effect of trichostatin A treatment on endogenous expression of HNF1α in HepG2 cells

It has previously been shown, by Western blot and EMSA, that HepG2 cells cultured in our laboratory under the conditions described in Chapter 2, section 2.2.1 express HNF1 α (Gardner-Stephen and Mackenzie, 2005; Gardner-Stephen and Mackenzie, 2007a and Chapter 4, section 4.3.3). Since HNF1 α has been proposed to bind to the proximal promoters of all human UGT genes, it was investigated whether TSA treatment had any effect on endogenous HNF1 α expression.

Initially, the effect of 3 µM TSA treatment on the levels of HNF1a protein in HepG2 cells was investigated by Western blot (Figure 5.7A). As expected, it was found that HNF1a protein was readily detectable in lysates from untreated HepG2 cells. However, HNF1a levels in lysates from 3 µM TSA-treated cells were below the limit of detection. Although no attempt was made to demonstrate that the observed loss of HNF1α was not due to a general degradation of protein in TSA-treated HepG2 cells, it was shown that the loss of HNF1 α expression in 3 μ M TSA-treated HepG2 cells was also observable at the mRNA level. When HepG2 HNF1 α mRNA concentrations were investigated by quantitative reverse transcription PCR in the same manner as discussed for UGT mRNA transcripts, and it was found that the solvent carrier, ethanol, had no effect on HNF1a mRNA levels (Figure 5.7B). Similarly, 300 nM TSA had no significant effect on HNF1 α (1.4-fold increase, P =0.059). In contrast, HNF1 α mRNA levels were severely repressed in 3 μ M TSAtreated cells (46.8-fold decrease, P < 0.001). Therefore, it would seem that the loss of HNF1 α expression in 3 μ M TSA-treated HepG2 cells was caused by either a specific increase in HNF1 α mRNA degradation relative to 18S rRNA, or a decrease in $HNF1\alpha$ promoter activity, rather than degradation of protein. A TSA-induced decrease in $HNF1\alpha$ promoter activity would be the scenario most consistent with the known action of TSA.

Transfection of HepG2 cells with pCMX-HNF1 α greatly enhanced HNF1 α levels (17.4-fold, P < 0.001, Figure 5.7C). Co-treatment with either 300 nM or 3 μ M TSA did not statistically affect HNF1 α mRNA levels in pCMX-HNF1 α transfected cells (*P* = 0.131 and 0.916, respectively; Figure 5.7C). Although this does not represent

Figure 5.7: Effect of ethanol, trichostatin A and pCMX-HNF1a on the levels of HNF1a in HepG2 cells. A. 25 µg protein lysate from untreated and 3 µM TSAtreated HepG2 cells were probed for HNF1a protein by Western blot, as described in Chapter 2, section 2.2.16. *In vitro* synthesised HNF1a (1 µl) was included in the Western blot as a positive control. B. and C. HepG2 cells were treated for 24 hours with 1:1000 diluted ethanol, 300 nM TSA or 3 µM TSA, with or without prior transfection with 0.5 µg pCMX-PL2 or pCMX-HNF1a as per "Methods". The mean levels of HNF1a mRNA (n = 3) were assessed by QPCR and normalised to 18S rRNA content. Error bars indicate one standard deviation. *P* values for the indicated comparisons are **P* < 0.001 and #*P* > 0.05 (not significant).



conclusive proof that TSA does not cause degradation of HNF1 α transcripts, it is certainly consistent with the hypothesis that TSA affects HNF1 α expression by specifically altering its regulation. Furthermore, these results show that treatment of HepG2 cells with TSA after transfection does not prevent mRNA synthesis from the CMV promoter present in pCMX-derived expression vectors, as no decreases in HNF1 α mRNA levels were observed when pCMX-HNF1 α -transfected cells were concurrently treated with TSA.

5.3.7. Effect of over-expressed liver-enriched transcription factors on endogenous UGT expression in HepG2 cells

Although UGT1A7, UGT1A8 and UGT1A10 are not expressed in hepatocytes, these genes were included in this study because many of the tested LETFs are also expressed in tissues where these UGTs are found, such as the intestine. In the presence of the chromatin-relaxing agent TSA, the mechanisms that normally silence these genes in hepatocytes might be bypassed and a response to LETFs may be seen if the UGT1A7, UGT1A8 or UGT1A10 promoters contain LETF binding sites that are normally functional in another cell type. Although any responses obtained for these genes would be biologically irrelevant for the liver, they would indicate potentially important interactions for further analysis in more appropriate cell lines. Furthermore, it was hoped that differences in the responses of UGT1A7, UGT1A8 and UGT1A10 to that of UGT1A9 would reveal important information about the mechanisms that cause UGT1A9 to be the only hepatically-expressed enzyme from the closely related UGT1A7-UGT1A10 gene cluster. However, since no responses to LETFs were obtained, with or without TSA treatment, for UGT1A8 or UGT1A10, only UGT1A7 data is included in the results presented below. The following figure (Figure 5.8A-M) summarises the responses of the HepG2 endogenous UGT genes to

Figure 5.8A-C: Effect of over-expressed liver-enriched transcription factors on endogenous UGT1A1, UGT1A3 and UGT1A4 expression in HepG2 cells. HepG2 cells were transfected with 0.5 µg empty pCMX-PL2 or pCMX-derived vectors expressing HNF1 α , HNF1 β , HNF4 α , FoxA1, FoxA2, FoxA3, C/EBP α or C/EBP β . Twenty-four hours after transfection, cells were also treated with 300 nM TSA or 3 µM TSA. After a further 24 hours, cells were harvested for total RNA and analysed for A. UGT1A1, B. UGT1A3 or C. UGT1A4 mRNA content by reverse transcription QPCR. Levels of each UGT were normalised against 18S rRNA and the results are presented as mean fold changes (n = 3) relative to the untreated, untransfected control (set arbitarily to a value of one). Error bars indicate one standard deviation. *P* values for the indicated comparisons are **P* < 0.001, †*P* = 0.001, †††*P* = 0.003, ‡*P* = 0.005, ‡‡*P* = 0.006 and §*P* > 0.02 but < 0.05.



Figure 5.8D-F: Effect of over-expressed liver-enriched transcription factors on endogenous UGT1A6, UGT1A7 and UGT1A9 expression in HepG2 cells. HepG2 cells were transfected with 0.5 µg empty pCMX-PL2 or pCMX-derived vectors expressing HNF1 α , HNF1 β , HNF4 α , FoxA1, FoxA2, FoxA3, C/EBP α or C/EBP β . Twenty-four hours after transfection, cells were also treated with 300 nM TSA or 3 µM TSA. After a further 24 hours, cells were harvested for total RNA and analysed for D. UGT1A6, E. UGT1A7 or F. UGT1A9 mRNA content by reverse transcription QPCR. Levels of each UGT were normalised against 18S rRNA and the results are presented as mean fold changes (n = 3) relative to the untreated, untransfected control (set arbitarily to a value of one). Error bars indicate one standard deviation. *P* values for the indicated comparisons are **P* < 0.001, †*P* = 0.001, †††*P* = 0.003, ‡‡*P* = 0.006, ‡‡‡*P* = 0.009 and §*P* > 0.02 but < 0.05.



Figure 5.8G-I: Effect of over-expressed liver-enriched transcription factors on endogenous UGT2B4, UGT2B7 and UGT2B10 expression in HepG2 cells. HepG2 cells were transfected with 0.5 µg empty pCMX-PL2 or pCMX-derived vectors expressing HNF1 α , HNF1 β , HNF4 α , FoxA1, FoxA2, FoxA3, C/EBP α or C/EBP β . Twenty-four hours after transfection, cells were also treated with 300 nM TSA or 3 µM TSA. After a further 24 hours, cells were harvested for total RNA and analysed for G. UGT2B4, H. UGT2B7 or I. UGT2B10 mRNA content by reverse transcription QPCR. Levels of each UGT were normalised against 18S rRNA and the results are presented as mean fold changes (n = 3) relative to the untreated, untransfected control (set arbitarily to a value of one). Error bars indicate one standard deviation. *P* values for the indicated comparisons are **P* < 0.001, †*P* = 0.001 and §*P* > 0.02 but < 0.05.



Figure 5.8J-L: Effect of over-expressed liver-enriched transcription factors on endogenous UGT2B11, UGT2B15 and UGT2B17 expression in HepG2 cells. HepG2 cells were transfected with 0.5 µg empty pCMX-PL2 or pCMX-derived vectors expressing HNF1 α , HNF1 β , HNF4 α , FoxA1, FoxA2, FoxA3, C/EBP α or C/EBP β . Twenty-four hours after transfection, cells were also treated with 300 nM TSA or 3 µM TSA. After a further 24 hours, cells were harvested for total RNA and analysed for J. UGT2B11, K. UGT2B15 or L. UGT2B17 mRNA content by reverse transcription QPCR. Levels of each UGT were normalised against 18S rRNA and the results are presented as mean fold changes (n = 3) relative to the untreated, untransfected control (set arbitarily to a value of one). Error bars indicate one standard deviation. *P* values for the indicated comparisons are **P* < 0.001, †*P* = 0.001, ††*P* = 0.002, ‡‡*P* = 0.006, ‡‡‡*P* = 0.009, ¥*P* = 0.013, ¥¥*P* = 0.014, ¥¥¥*P* = 0.015 and §*P* > 0.02 but < 0.05.





Figure 5.8M: Effect of over-expressed liver-enriched transcription factors on endogenous UGT2B28 expression in HepG2 cells. HepG2 cells were transfected with 0.5 µg empty pCMX-PL2 or pCMX-derived vectors expressing HNF1 α , HNF1 β , HNF4 α , FoxA1, FoxA2, FoxA3, C/EBP α or C/EBP β . Twenty-four hours after transfection, cells were also treated with 300 nM TSA or 3 µM TSA. After a further 24 hours, cells were harvested for total RNA and analysed for UGT2B28 mRNA content by reverse transcription QPCR. Levels of each UGT were normalised against 18S rRNA and the results are presented as mean fold changes (n = 3) relative to the untreated, untransfected control (set arbitarily to a value of one). Error bars indicate one standard deviation. P values for the indicated comparisons are *P <0.001, †P = 0.001, \$P = 0.007, \$P = 0.012 and §P > 0.02 but < 0.05.

each transcription factor tested (HNF1 α , HNF1 β , HNF4 α , FoxA1, FoxA2, FoxA3, C/EBP α or C/EBP β), as measured by changes in mRNA concentrations. Changes in each UGT mRNA concentration was only considered to be of likely biologically relevance if the change was greater than 2-fold and reached statistical significance (*P* < 0.05).

5.3.7.1. Hepatocyte nuclear factor 1a

When HNF1 α was over-expressed in HepG2 cells, no effect was seen on the mRNA levels of the following *UGT* genes: *UGT1A1* (Figure 5.8A), *UGT1A3* (Figure 5.8B), *UGT1A4* (Figure 5.8C), and all of the *UGT2B* forms (Figure 5.8G-M). In contrast, UGT1A6 mRNA was increased 5.6-fold (*P* = 0.003) (Figure 5.8D), UGT1A7 mRNA

was increased by 9.1-fold (P < 0.001) (Figure 5.8E), and UGT1A9 mRNA was increased 3.1-fold (P = 0.03) (Figure 5.8F).

In the presence of 300 nM TSA, the HNF1 α response of *UGT1A6* was increased from 5.6-fold to 10.6-fold (*P* = 0.002) (Figure 5.8D), while the *UGT1A7* and *UGT1A9* responses were both diminished such that they lost statistical significance. The remaining *UGT* genes were also not significantly altered by combined HNF1 α /300 nM TSA treatment, relative to their expression in pCMX-PL2transfected cells treated with 300 nM TSA.

In contrast, exogenous over-expression of HNF1 α in the presence of 3 μ M TSA caused a great deal of change in UGT mRNA levels in HepG2 cells. Statistically significant increases in mRNA, greater than 2-fold relative to pCMX-PL2-transfected cells treated with 3 μ M TSA, were observed for UGT1A4 (8.6-fold, *P* < 0.001), UGT1A6 (26.8-fold, *P* < 0.001), UGT1A7 (45.5-fold, *P* < 0.001), UGT1A9 (16.0-fold, *P* < 0.001), UGT2B4 (18.3-fold, *P* = 0.001), UGT2B11 (3.3-fold, *P* = 0.002), UGT2B15 (4.9-fold, *P* = 0.006), UGT2B17 (3.4-fold, *P* = 0.014) and UGT2B28 (12.0-fold, *P* = 0.001) transcripts (Figure 5.8C-G and J-M).

At least one potential HNF1-binding site has been identified in the proximal promoter of every known human UGT1A and UGT2B gene (Figure 5.1). As a result, HNF1 α is the most thoroughly studied transcriptional regulator of human UGTs to date. However, until now, no attempts have been made to compare the effects of HNF1 α on each individual gene in one experimental system. Therefore, the relative importance of this transcription factor in the expression of each UGT has remained unexplored. This experiment begins to address this issue, highlighting the variety of functional interactions that HNF1 α may have with different UGT genes, and the

importance of not only identifying relevant transcription factors, but also understanding how they fit into the hierarchy of a gene's transcriptional control.

The first notable observation from this experiment is that the endogenous levels of UGT1A1, UGT1A3 and UGT1A4 mRNAs do not change with HNF1a overexpression in HepG2 cells. Furthermore, UGT1A1 and UGT1A3 mRNA levels are highly increased in the presence of 3 µM TSA, a condition that has been shown to cause dramatic loss of HNF1 α expression (Figure 5.7). Yet, all three corresponding genes have previously been identified as HNF1 α target genes in reporter-plasmid assays, and each contains an experimentally confirmed HNF1 α -binding site in its proximal promoter (10/12, 12/12 and 10/12 consensus nucleotides respectively, although the UGT1A4 element does not contain any perfect half-sites). There are a least two explanations that would reconcile these seemingly contradictory results. The first is that HNF1 α is not required for the expression of these UGT1A forms in *vivo*, and that the dependence of the naked proximal promoters on this transcription factor is an artefact of studying short promoter sequences. However, the weight of evidence suggests that a more complex reason lies behind these observations. Excess HNF1 α does not greatly increase the activity of the UGT1A3 or UGT1A4 promoters in vitro in HepG2 cells (the effect of HNF1 α on the naked UGT1A1 promoter in HepG2 cells is unreported), and the importance of HNF1 α for the function of the UGT1A3 and UGT1A4 promoters is only evident if their respective HNF1-binding sites are destroyed or if HNF1-factors are initially absent in the host cell and subsequently supplied (see Chapter 3). In regards to these latter properties, the UGT1A1 promoter behaves similarly (Bernard et al., 1999). This has led to the hypothesis that the strong HNF1-binding elements in the UGT1A1, UGT1A3 and UGT1A4 promoters are fully occupied, or nearly so, by HNF1 factors at the
concentrations at which they naturally occur in HepG2 cells (Chapter 3). If this premise is true, and holds true for the endogenous genes also, then excess HNF1a would not be expected to greatly increase endogenous gene expression either. Therefore, a lack of response does not preclude a regulatory role for HNF1a. However, a loss of HNF1 α , as seen after the 3 μ M TSA exposure would be expected to decrease expression of HNF1 α -dependent genes, while instead, the UGT1A1 and UGT1A3 promoter activities were greatly increased. Yet, this unexpected response could also be explained if the presence of TSA makes the function of HNF1a redundant for these genes. As reviewed in Chapter 3, section 3.1.5.4, HNF1 α has a dual purpose in gene regulation; chromatin remodelling and recruitment of the transcription machinery. The former is achieved through recruitment and activation of proteins with HAT activity such as p300/CBP and P/CAF (Pontoglio et al., 1997; Rollini et al., 1999; Soutoglou et al., 2000b; Parrizas et al., 2001; Soutoglou et al., 2001), and HNF1 α has been proposed to only play an obligate role in transcriptional activation when it is required to recruit HAT activity (Parrizas et al., 2001). Since treatment with HDAC inhibitors has a similar net effect on gene acetylation as HAT recruitment, TSA may be able to replace the function of HNF1 factors on the UGT1A1 and UGT1A3 promoters by counteracting any recruitment of HDACs that would normally occur. Interestingly, unliganded and antagonist-loaded nuclear receptors can recruit HDAC activity to their target genes through co-repressors such as SMRT and NCoR (Ng and Bird, 2000; Karvonen et al., 2006), and UGT1A1 is a known target for numerous nuclear receptors (PXR, CAR, GR, AhR and PPARa) (Sugatani et al., 2004; Usui et al., 2006a; Senekeo-Effenberger et al., 2007). UGT1A3 is also a target for PPARa and PXR (Gardner-Stephen et al., 2004; Senekeo-Effenberger et al., 2007). Furthermore, TSA-mediated relief of nuclear

receptor-mediated gene repression, through inhibition of nuclear receptor-associated HDACs, is a recently recognised phenomenon (Huang and Hung, 2006; Karvonen *et al.*, 2006; Kwon *et al.*, 2006; Qi and Ratnam, 2006). Therefore, if the literature and this experiment are considered collectively, the data suggests that HNF1 α may be more important in directing histone acetylation than recruitment of the transcription machinery to *UGT1A1* and *UGT1A3* in the hepatic chromosomal setting, as was found for the murine *glut2* and L-type pyruvate kinase genes in pancreatic cells (Parrizas *et al.*, 2001).

Since increasing HNF1 α in HepG2 cells did not have a corresponding effect on the *UGT1A1* or *UGT1A3* genes, the importance of HNF1 α in maintaining their basal expression could, in the future, possibly be demonstrated through knock-down experiments. Indeed, unpublished studies performed by Anne Rogers in our laboratory show that siRNA-mediated knock-down of HNF1 α mRNA in Caco-2 cells results in similar fold decreases in HNF1 α and UGT1A1 mRNA levels (personal communication, results not shown). However, HepG2 cells would not be a suitable experimental system for such an investigation of the *UGT1A3* gene, due to insufficient basal expression of UGT1A3 mRNA. Instead, primary hepatocytes could be used to study the effect of HNF1 α -knock-down on *UGT1A3* (and all other hepatic *UGTs*), but are resistant to conventional transfection methods and would require treatment with an adenoviral-mediated HNF1 α -antisense targeting vector system like that described for HNF4 α knock-down by Jover *et al.* (2001), or similar. Several donors would also be required to account for interindividual differences in response.

Evidence that genomic UGT1A4 may also be a genuine HNF1 α -target gene, as previously suggested by transient transfection assays (Gardner-Stephen and Mackenzie, 2007b and Chapter 3), was only obtained when cells were co-treated with 3 μM TSA and pCMX-HNF1α. This combined treatment resulted in an increase in UGT1A4 transcripts of 8.7-fold (Figure 5.8C). The simplest explanation of these results is that HNF1 α plays a role in recruiting, or positioning, the transcription machinery to the UGT1A4 gene, but only if permitted by other more important chromatin remodelling factors, and/or in co-operation with other transactivator(s) normally unavailable in HepG2 cells. Treatment of HNF1 α with 1 μ M TSA has been shown to inhibit its interaction with HDAC-1, allowing constitutive activation through its co-factors p300/CBP and P/CAF (Soutoglou et al., 2001). If the HNF1a dimers that normally reside on the integrated UGT1A4 promoter require the assistance of other factors to facilitate a switch in bound co-factors from NCoR/HDAC to p300/CBP and P/CAF, this would explain why only the high concentration of TSA had a profoundly positive effect on UGT1A4 transcription by HNF1 α . Additional evidence that HNF1 α may be relatively low in the hierarchy of proteins required for UGT1A4 expression comes from transfections in HEK293T cells that showed that the UGT1A4 proximal promoter has some basal activity in the complete absence of HNF1 factors (Chapter 3, Figure 3.3A). What is certain, however, is that HNF1 α and the other factors normally expressed in HepG2 cells are insufficient to drive UGT1A4 promoter activity when this gene is chromatin bound.

The stark contrast between the responses of the *UGT1A3* and *UGT1A4* genes to HNF1 α , with or without TSA exposure, was not anticipated due to the high similarity in primary sequence of their proximal promoters (over 88% identity over 1kb) and their comparable responses to HNF1 α and HNF1-site mutation in transient transfection assays (Chapter 3). Thus, this new data illustrates the value of performing studies on endogenous genes, and suggests that distal enhancer elements

may play a significant role in the differential expression of UGT1A3 and UGT1A4. However, this work also raises questions regarding how many of these differences would be apparent if the UGT1A4 gene was already expressed at a more appropriate basal level. Comparison of the results obtained in HepG2 cells with a cell line that expresses UGT1A4 may be useful for further characterising the role of HNF1 α in UGT1A4 promoter activity. However, since such a model was unavailable, the full UGT1A3 and UGT1A4 promoters, which are both active in HepG2 transient transfection assays, were tested for responsiveness to HNF1 α in the presence of 3 μ M TSA to determine whether any further information could be gleaned regarding the differential expression of these genes.

The result of treating cells transfected transiently with the full *UGT1A3* and *UGT1A4* promoters with 3 μ M TSA (Figure 5.9) was surprisingly similar to the effect seen with the endogenous genes (Figure 5.8B and C). The *UGT1A3*-9.4kb promoter was highly responsive to the presence of TSA, giving an 11.4-fold increase in luciferase reporter activity over solvent treated cells (P < 0.001). In contrast, the response of the *UGT1A4*-5kb promoter was only 1.3-fold (P = 0.014) (Figure 5.9). This observation reflects the much greater increase in UGT1A3 mRNA levels in 3 μ M TSA-treated HepG2 cells over UGT1A4 (Figure 5.6). Furthermore, HNF1 α over-expression in TSA-treated cells only had a mild effect on the *UGT1A3* promoter, resulting in a further 2.4-fold increase in reporter activity (P < 0.001), but a drastic effect on the *UGT1A4* promoter (Figure 5.9), as seen for the endogenous gene (Figure 5.8C). Reporter gene expression from the *UGT1A4*-5kb promoter in the presence of both HNF1 α and 3 μ M TSA was increased 58.6-fold over TSA alone (P < 0.001). Interestingly, the combined effects of HNF1 α -over expression and 3 μ M

TSA-treatment on the *UGT1A3* and *UGT1A4* promoters resulted in similar overall reporter gene expression (Figure 5.9).



Figure 5.9: The *UGT1A3* and *UGT1A4* promoter responses to TSA and HNF1a. HepG2 cells were transiently transfected with 25 ng pRL-Null, 0.5 µg pGL3-basic, pGL3-1A3-9.4k or pGL3-1A4-5k and 0.25 µg pCMX-PL2 or pCMX-HNF1a. After 24 hours, each transfection was exposed to 1:1000 diluted ethanol, or 3 µM TSA for a further 24 hour period. The luciferase reporter activity of each cell lysate was then normalised to *renilla* activity and mean ratios (n = 3) are presented relative to ethanol-treated pGL3-basic transfections. Error bars indicate one standard deviation. *P* values for the indicated comparisons are **P* < 0.001 and ¥¥*P* = 0.014. NB: this experiment was only performed once.

The discovery that the transiently transfected *UGT1A3* promoter is specifically activated by TSA in HepG2 cells will allow some of the questions raised earlier in this thesis to be investigated. From the current data, it is still impossible to determine whether the effect of TSA on *UGT1A3* is due to direct changes in histone acetylation or a secondary effect of changed protein expression in HepG2 cells. However, TSA-treatment of promoter deletion constructs, TSA treatment of promoter constructs with specific mutations (particularly in the known HNF1-binding site), *in vitro* chromatin assays and histone acetylation studies could all be combined to determine the mechanisms responsible for the differences in the *UGT1A3* and *UGT1A4* TSA response. Similar studies would also determine whether the synergistic effect of

HNF1 α and TSA on the *UGT1A4* promoter is the result of HDAC-inhibition, an interaction of HNF1 α with TSA-induced factors, or possibly, both.

Interestingly, of all the human *UGT1A* and *UGT2B* genes, the *UGT1A6*, *UGT1A7* and *UGT1A9* promoters were the only sequences to respond to excess HNF1 α in HepG2 cells without TSA. This result is consistent with the presence of putative HNF1 sites in these sequences (10/12, 9/12 and 9/12 consensus nucleotides respectively, with no perfect half-sites), and with the functional data published for the *UGT1A9* promoter (Gregory *et al.*, 2004; Gardner-Stephen and Mackenzie, 2007a). Elements that diverge significantly from the HNF1-binding consensus, whilst still remaining capable of binding HNF1 factors, could be expected to mediate a response to increased levels of HNF1 α . Such sites are unlikely to compete efficiently for HNF1 factors at physiological levels; therefore, increasing HNF1 α levels may increase binding-site occupancy. To my knowledge, this is also the first actual report of HNF1 α -responsiveness for human *UGT1A6* and *UGT1A7*, although both outcomes have been predicted (Auyeung *et al.*, 2003; Gardner-Stephen and Mackenzie, 2005).

In addition to their increased expression in response to increased HNF1 α , UGT1A6 and UGT1A9 mRNA levels are both severely diminished by 3 μ M TSA treatment. Since their expression can be restored by exogenous HNF1 α in the presence of TSA, but only to similar levels as seen for pCMX-HNF1 α -transfected cells not treated with TSA, it appears that the relationship between HNF1 α and expression of UGT1A6 and UGT1A9 is one of complete dependence. Furthermore, the response is independent of any other intracellular changes TSA may cause. Since histone hyperacetylation by 3 μ M TSA does not functionally compensate for the loss of HNF1 α that it triggers, the predominant role of HNF1 α in the expression of UGT1A6 and UGT1A9 appears to be the recruitment or activation of other transcription factors or the transcription machinery. This observation is consistent with the discovery that the HNF4 α response of the *UGT1A9* promoter is completely reliant on HNF1 factors (Gardner-Stephen and Mackenzie, 2007a and Chapter 4). Interestingly however, HNF1 α does not stimulate transcription from the *UGT1A6*-3kb promoter in transient transfections (Figure 5.10), even though this construct contains the predicted HNF1-binding site. This is in direct contrast with the effect of HNF1 α on the proximal *UGT1A9* promoter (Chapter 4, Figure 4.4B) and again highlights the value of testing endogenous gene responses. As an explanation, it is possible that the HNF1-binding site of the transiently transfected *UGT1A6*-3k promoter is fully occupied by HNF1 factors under basal conditions, as hypothesised for the *UGT1A4* promoter, which likewise has a HNF1-binding site with 10/12 matches with



Figure 5.10: Response of the *UGT1A6* proximal promoter to HNF1a and HNF4a in HepG2 cells. HepG2 cells were transfected with 25 ng pRL-Null, 0.5 µg pGL3-basic or pGL3-1A6-3k and 0.25 µg pCMX-PL2, pCMX-HNF1a or pCMX-HNF4a. After 48 hours, cells were lysed and assayed for luciferase and *renilla* activity. Mean luciferase:*renilla* ratios (n = 3) are presented relative to the result for pGL3-basic (set arbitrarily to one). Error bars represent one standard deviation. The *P* value for the indicated comparison is **P* < 0.001.

consensus but no perfect half-site. However, the data from the endogenous gene belies this suggestion, unless the structural chromatin environment reduces the affinity of the *UGT1A6* HNF1-binding site for HNF1 factors without completely precluding HNF1 binding. Other possible explanations are that the observed *UGT1A6* response to HNF1 α is not mediated through the published HNF1-binding site, but through another site not included in the cloned fragment; or that the transfected promoter fragment is sufficiently more accessible than the endogenous gene to cause the (still theoretical) role of HNF1 α in recruiting additional transcription factors or the transcriptional machinery to the *UGT1A6* promoter to become superfluous. Longer clones of the *UGT1A6* promoter and site-directed mutagenesis of the putative HNF1-binding site will be required to test these hypotheses.

Although the UGT1A7 gene is expected to be HNF1 α -responsive in the appropriate context, as its proximal promoter structure is most like that of UGT1A8, UGT1A9 and UGT1A10, the response of UGT1A7 to over-expressed HNF1 α in HepG2 cells in the absence of chromatin-altering agents was unforeseen. Like UGT1A8 and UGT1A10, which did not respond to HNF1 α in HepG2 cells (data not shown), UGT1A7 is only extrahepatically expressed. Yet, the unaided response of UGT1A7 to HNF1 α indicates that this gene is poised for transcription in liver-derived HepG2 cells. Whether this is true in human hepatocytes, or is a function of the slightly dedifferentiated state of HepG2 cells is unknown, but an interesting question. Given that UGT1A7 has a weak HNF1 site, perhaps this gene is not expressed in hepatocytes partly because in these cells it cannot recruit enough HNF1 α when this factor is only available at physiological levels. On the other hand, UGT1A7 mRNA transcription may be driven by HNF1 α in non-hepatic tissues where different factors

are available to bind to the *UGT1A7* promoter, thereby recruiting HNF1 α to its binding site. Over-expression of HNF1 α in primary hepatocytes would help determine whether *UGT1A7* is normally more tightly suppressed in the liver than in HepG2 cells. Promoter studies combined with TSA treatment, as suggested for the *UGT1A3* and *UGT1A4* promoters, may also be worthwhile for comparing the promoters of *UGT1A7* and *UGT1A9*. Despite having very similar proximal promoters and basal responses to HNF1 α , HNF1 β and HNF4 α over-expression (see also sections 5.3.7.2 and 5.3.7.3), *UGT1A7* and *UGT1A9* have very different reactions to the combined TSA/HNF1 α treatments. Elucidation of the cause may reveal important functional elements that differ between these genes, and contribute to our understanding of the mechanisms causing their distinct tissue-specific expression patterns.

Interestingly, the *UGT2B* genes all have one known HNF1-binding site (9/12 consensus nucleotides each, no perfect half site) in their proximal promoters (Figure 5.1), yet none of these genes were up-regulated by excess HNF1 α alone in HepG2 cells. However, all but two, *UGT2B7* and *UGT2B10*, were responsive to HNF1 α in the presence of 3 μ M TSA, identifying their promoters as potential HNF1 α targets. Why no HNF1 α -response was seen for the *UGT2B7* and *UGT2B10* genes under any conditions is a mystery, particularly as both genes are transcriptionally active in HepG2 cells (Figure 5.3), the *UGT2B7* promoter has been shown to be HNF1 α -responsive *in vitro* (Ishii *et al.*, 2000; Gregory *et al.*, 2006) and UGT2B7 mRNA expression levels have been correlated with HNF1 α mRNA levels in human liver (Toide *et al.*, 2002). Perhaps the HNF1-binding elements of these genes are fully occupied at physiological levels, even though they are expected to be of relatively low affinity, through recruitment or stabilisation by other proteins. HNF1 α -mediated

regulation of the *UGT2B7* gene is already known to be enhanced by co-operation with Oct-1 in HepG2 cells and Cdx2 in Caco-2 cells (Ishii *et al.*, 2000; Gregory *et al.*, 2006). Another possibility is that a factor required for co-operation with HNF1 α is limited in HepG2 cells, and the system is already relatively saturated with HNF1 α . Either way, the hypothetical need for at least one other factor for HNF1 α to effectively regulate the *UGT2B7* and *UGT2B10* promoters is supported by the observation that over-expression of HNF1 α did not rescue their expression after repression by 3 μ M TSA, and thus the profound loss of UGT2B7 and UGT2B10 expression in 3 μ M TSA-treated cells is not entirely due to the physical absence of HNF1 α . A third possibility is that HNF1 α has no real role in the expression of the endogenous *UGT2B7* and *UGT2B10* genes in hepatocytes, and the relationship noted between HNF1 α and UGT2B7 mRNAs by Toide *et al.* (2002) is not a causative association. Once again, siRNA-mediated knockdown of HNF1 α in HepG2 cells would add valuable information to the apparently contradictory observations between this experiment and previous results.

The responses of the remaining UGT2B genes to over-expressed HNF1 α and/or TSA treatment were very similar in pattern to UGT1A4. Therefore, the arguments presented earlier for the involvement of HNF1 α in the expression of UGT1A4 are also pertinent for these genes, except that it is unknown whether any of these promoters have residual activity in HNF1-negative cells. One additional piece of information we do have however, is that siRNA-mediated knockdown of HNF1 α in Caco-2 cells results in a decrease in UGT2B15 transcripts (Anne Rogers, unpublished observations), providing further support for the hypothesis that HNF1 α genuinely interacts with this gene in at least one cell type.

5.3.7.2. Hepatocyte nuclear factor 1β

Similarly to HNF1 α , HNF1 β over-expression had significant positive impact on the levels of UGT1A6, UGT1A7 and UGT1A9 mRNA in HepG2 cells, increasing them by 2.5-fold (P = 0.036), 8.1-fold (P < 0.001) and 2.9-fold (P = 0.044) respectively (Figure 5.8D-F). However, on co-treatment with 300 nM TSA, none of these responses to HNF1 β over-expression were retained. On the other hand, UGT2B28 mRNA levels, which were unaffected by HNF1 β alone, or in combination with 300 nM TSA, were increased 4.8-fold (P = 0.007) by HNF1 β over-expression in the presence of 3 μ M TSA.

Although HNF1B is not expressed to high levels in the adult liver (Rey-Campos et al., 1991), it was of interest to compare its effect with that of HNF1 α because these two proteins recognise the same nucleotide sequences and heterodimerise readily, but possess functionally divergent activation domains (see Chapter 3, section 3.1.5.1). Of the three genes transactivated by HNF1 α alone, UGT1A7 and UGT1A9 were equally responsive to HNF1 β as HNF1 α (P > 0.05) (Figure 5.8E and F), whereas the third HNF1 α -responsive gene, UGT1A6, was also increased by HNF1 β , but only to half the extent (P = 0.014). In general, HNF1 β has been considered to be a less potent transactivator than HNF1a (Senkel et al., 2005), and this assumption has held true for transient transfections against the human UGT1A3 (Chapter 3). UGT1A8 (Gregory et al., 2004), UGT1A9 (Gardner-Stephen and Mackenzie, 2007a and Chapter 4, section 4.3.9), UGT2B7 (Ishii et al., 2000), UGT2B17 (Gregory et al., 2000), rat UGT1A6 (Auyeung et al., 2003) and rat UGT2B1 (Hansen et al., 1997) promoters; with human and rat UGT1A1 and rat UGT1A7 being the only known exceptions (Bernard et al., 1999; Metz et al., 2000). However, Senkel and colleagues have recently postulated that HNF1 β may be a more effective transactivator in the

chromosomal context than in transient transfections (Senkel *et al.*, 2005), a hypothesis that appears relevant to at least *UGT1A9*. The discovery that HNF1 β may be as effective as HNF1 α in regulating the *UGT1A7* and *UGT1A9* genes suggests that HNF1 β may have a more important role in non-hepatic UGT expression than previously thought.

In the presence of TSA, the functional differences between HNF1 α and HNF1 β were more apparent. *UGT1A6*, the only gene to show a further increase in response to HNF1 α in the presence of 300 nM TSA, was not affected by the HNF1 β /300 nM TSA combination (in comparison to HNF1 β alone, *P* > 0.05). Furthermore, the *UGT* gene responses to HNF1 α observed in the presence of 3 μ M TSA were either significantly reduced or completely absent for HNF1 β . This result indicates that the mechanism responsible for the widespread synergistic effect of HNF1 α and 3 μ M TSA on the expression of many HepG2 endogenous *UGT* genes is not simply the loss of HDAC-1 recruitment by HNF1 α , because HNF1 β also interacts with HDAC-1 and has greatly increased activity in the presence of 1 μ M TSA (Barbacci *et al.*, 2004). Therefore, the specific up-regulation of UGT transcription by HNF1 α and TSA is likely to also involve interactions with proteins that have increased expression and/or activity after TSA exposure, and that differ in their ability to cooperate with the two HNF1 proteins.

Interestingly, HNF1 β over-expression did not inhibit the transcription of any *UGT* genes, even those that have been shown to be exclusively responsive to HNF1 α in HepG2 transient transfections, such as *UGT2B7* (Ishii *et al.*, 2000) and *UGT2B17* (Gregory *et al.*, 2000). In the event of HNF1 β over-expression, HepG2 endogenous HNF1 α protein would be expected to exist largely in heterodimers with HNF1 β , and would be competing with HNF1 β homodimers for DNA-binding sites. As a

consequence, it is expected that where HNF1 β cannot substitute functionally for HNF1 α , HNF1 β over-expression would prevent gene activation by HNF1 α . Thus, it seems that either: a) HNF1 α is not supporting basal expression of the human *UGT* genes; or b) HNF1 α /HNF1 β heterodimers have similar activity towards the *UGT* promoters as HNF1 α homodimers and specific protein-protein interactions between HNF1 α and other promoter-bound proteins cause only HNF1 α -containing heterodimers to be recruited.

5.3.7.3. Hepatocyte nuclear factor 4α

Over-expression of HNF4 α in HepG2 cells had no effect on any UGT2B forms, whether expressed in the presence or absence of TSA. In contrast, HNF4 α alone increased the mRNA levels of UGT1A1 by 7.0-fold, (*P* < 0.001), UGT1A6 by 5.9-fold (*P* = 0.001), UGT1A7 by 12.7-fold (*P* < 0.001) and UGT1A9 by 21.8-fold (*P* < 0.001) (Figure 5.8A and D-F).

Addition of 300 nM TSA to HNF4 α over-expressing cells enhanced the UGT1A9 response (74.0-fold, P < 0.001) and decreased the apparent response of *UGT1A1* (2.4-fold, P = 0.001), *UGT1A6* (2.6-fold, P = 0.049) and *UGT1A7* (9.3-fold, P = 0.006) (Figure 5.8A and D-F). However, in all three latter cases, the decrease in fold change was caused by increased UGT expression in the 300 nM TSA/pCMX controls, rather than a decrease in the total transcripts induced by HNF4 α .

Addition of 3 μ M TSA to HNF4 α over-expressing cells had a variable effect, depending on the UGT studied. UGT1A1 expression continued to be enhanced by HNF4 α at this TSA concentration (2.8-fold, P = 0.005), however, the *UGT1A6* and *UGT1A7* responses were completely inhibited, while the *UGT1A9* response was severely suppressed, being reduced to 7.6-fold (P = 0.001) (Figure 5.8A and D-F).

As was found for the HNF1 factors, over-expressed HNF4 α had multiple UGT targets and at least two distinct activation pathways in HepG2 cells. Of the four UGT1A genes that responded to HNF4 α , UGT1A9 was the only previously known target, although the expression patterns of UGT1A1 and UGT1A6 (Table 1.2) are both consistent with possible HNF4a-mediated regulation. The discovery of an unaided response of UGT1A7 to another LETF in HepG2 cells was, once again, surprising, as the only currently known overlap in HNF4 α and UGT1A7 expression occurs in the stomach. Other major sites of HNF4 α expression - the liver, kidney and intestine - are all free of UGT1A7 mRNA. Furthermore, in transfections, the UGT1A7 proximal promoter is completely unresponsive to HNF4 α (Barbier et al., 2005; Gardner-Stephen and Mackenzie, 2007a and Chapter 4). These observations previously founded the hypothesis that HNF4a-responsiveness is one mechanism that affords UGT1A9 its unique hepatic status among the UGT1A7-1A10 genetic cluster; however this theory must now be questioned. Clearly there is more work required, both to characterise the UGT1A7 promoter elements that allow HNF4α to increase UGT1A7 transcript levels, and to identify further DNA elements that cause UGT1A9 to be specifically expressed in hepatocytes. None of the remaining LETFs investigated in this experiment provided any leads towards the second challenge.

The four HNF4 α -responsive *UGT1A* genes can be divided on the basis of the combined effect of TSA and excess HNF4 α on their transcript levels. 300 nM TSA synergistically increased the transactivational activity of HNF4 α towards *UGT1A1* (Figure 5.8A), but had little or no effect for *UGT1A6*, *UGT1A7* and *UGT1A9* (Figure 5.8D, E and F) over HNF4 α alone. Exposure to 3 μ M TSA further accentuated the response of *UGT1A1* to HNF4 α , but abolished the responses of *UGT1A6* and

UGT1A7, and reduced the *UGT1A9* response by 90%. Therefore, it would be reasonable to postulate that the mechanisms that drive HNF4α-mediated transcription from *UGT1A1* are different from those of the other three genes. HNF4α activation of the *UGT1A9* proximal promoter in HepG2 cells is dependent on the co-expression of at least one of the HNF1 factors (Gardner-Stephen and Mackenzie, 2007a and Chapter 4). Since 3 μ M, but not 300 nM, TSA exposure is known to cause a loss of HNF1α mRNA in these cells, this mechanism alone could explain the significant loss of *UGT1A9* activity, and by association, also that of *UGT1A6* and *UGT1A7*. Once the HNF4α-responsive elements of the *UGT1A6* and *UGT1A7* promoters have been identified, this hypothesis could be tested by transient transfection in HEK293 cells as was demonstrated for *UGT1A9*. Towards this end, the *UGT1A6*-3kb promoter has been shown to be HNF4α-responsive in HepG2 cells (*P* < 0.001) (Figure 5.10).

The residual effect of HNF4 α on the *UGT1A9* gene in the presence of 3 μ M TSA, and hence in the absence of HNF1 α , is also interesting. This weaker activation could represent co-operation with HNF1 β , as the effect of 3 μ M TSA on expression of HNF1 β is unknown, or with residual levels of HNF1 α too low to be detected by Western blot. Alternatively, the residual activation observed may represent a minor HNF1-independent effect of HNF4 α on *UGT1A9*. The latter possibility may be mediated through mechanisms that do not operate on transiently transfected DNA, or may be the result of HNF4 α -binding site(s) positioned beyond the currently tested promoter fragments.

The synergistic increase in UGT1A1 mRNA levels at both tested concentrations of TSA indicates that HNF4 α probably does not require HNF1 α for recruitment to the *UGT1A1* promoter or interaction with the transcriptional machinery. However, if these two factors would ordinarily synergistically affect acetylation of the *UGT1A1*

promoter, this effect would be masked by the TSA treatment, so cannot be ruled out. How TSA increases the effect of HNF4 α on the *UGT1A1* promoter is unknown, but possibilities include: a) increased accessibility of the promoter through inhibition of HDACs recruited by other factors; b) increased acetylation of the HNF4 α protein itself, which increases its nuclear retention and DNA binding (Soutoglou *et al.*, 2000a); c) increased activity of HNF4 α through prevention of its known associations with HDAC through SMRT (Torres-Padilla *et al.*, 2002); or d) increased cooperation with other transcription factors or co-factors normally limited/unavailable in untreated HepG2 cells. Further work, as described for HNF1 α in section 5.3.7.1, will be necessary to further characterise the role of HNF4 α in *UGT1A1* expression. In particular, it will be interesting to determine the relative importance of HNF4 α in the constitutive and inducible expression of UGT1A1, as HNF4 α was found to be necessary for the CAR-response of the mouse *Ugt1a1* gene, but not basal expression (Ding *et al.*, 2006).

5.3.7.4. Hepatocyte nuclear factor 6

HNF6 was unable to affect the expression of any human *UGT* gene in HepG2 cells without the assistance of TSA. In combination with 300 nM or 3 μ M TSA however, UGT1A4 mRNA transcript levels were increased by 5.7-fold (*P* = 0.010) and 4.4-fold (*P* < 0.001) respectively (Figure 5.8C). An increase in UGT2B11 mRNA levels (2.9-fold, *P* < 0.001) was also observed at the higher TSA concentration (Figure 5.6J).

Acetylation of HNF6 protein by CBP has been shown to increase its half-life, allowing it to accumulate to higher levels in HepG2 cells co-transfected with HNF6 and CBP, than those transfected with HNF6 alone (Rausa *et al.*, 2004). Therefore, although it is unknown whether HDACs can deacetylate HNF6, thereby reducing the

protein's stability, it is conceivable that TSA could protect HNF6 from degradation, and hence increase its apparent activity. However, it seems unlikely that this is the mechanism responsible for the TSA-mediated HNF6-responses of the *UGT1A4* gene. The first reason for this conclusion is that expression of HNF6 from a transiently transfected, CMV-promoter-driven construct has previously been shown to saturate an HNF6-responsive reporter-promoter without exogenously supplied mediators of acetylation (Rausa *et al.*, 2004). Secondly, while HNF6 is only able to transactivate the endogenous *UGT1A4* gene in the presence of TSA, the transiently transfected *UGT1A4* promoter is increased 10.3-fold (P < 0.001) by HNF6 in TSA-naive cells (Figure 5.11). Moreover, 3 μ M TSA is inhibitory to this interaction, reducing the ability of HNF6 to activate the exogenous promoter by two thirds (P = 0.001). In



Figure 5.11: HNF6 regulates the transiently transfected *UGT1A4* promoter in HepG2 cells. HepG2 cells seeded into 24-well plates were transiently transfected with 25 ng pRL-Null, 0.5 µg pGL3-basic, pGL3-1A4-5k and 0.25 µg pCMX-PL2 or pCMX-HNF6 per well. Twenty-four hours post-transfection, cells were treated with 1:1000 diluted ethanol or 3 µM TSA. After a further 24 hours, cell lysates were assayed for luciferase and *renilla* activity. Mean luciferase:*renilla* ratios (n = 3) are presented relative to the result for pGL3-basic (set arbitrarily to one). Error bars represent one standard deviation. *P* values for the indicated comparisons are **P* < 0.001 and †*P* = 0.001. NB: the ethanol and TSA-treated triplicates of this experiment were only performed once.

light of these results, it seems unlikely that TSA is increasing HNF6 activity, or causing the expression of co-factors, or other transcription factors, required for HNF6 to have a functional impact on the *UGT1A4* promoter. Therefore, the most likely mechanism through which HNF6 increases the levels of UGT1A4 mRNA in HepG2 cells is through improved accessibility of the promoter. Whether a similar argument will hold true for the *UGT2B11* promoter is uncertain, as this gene is already relatively active in HepG2 cells (Figure 5.3). However, TSA alone had a similar effect on *UGT2B11* as *UGT1A4* (Figure 5.6), so it is both conceivable and of interest to investigate.

5.3.7.5. Forkhead box proteins FoxA1, FoxA2 and FoxA3

When FoxA1 was over-expressed in HepG2 cells, the most responsive gene was UGT2B15, with an increase in mRNA levels of 4.0-fold (P = 0.002) (Figure 5.8K). An increase in UGT1A1 transcripts of 2.9-fold (P = 0.032) was also observed (Figure 5.8A). In the presence of 300 nM TSA, the UGT2B15 mRNA response to FoxA1 was still apparent (2.2-fold, P = 0.034), although it was stronger when the cells were co-treated with 3 μ M TSA (10.1-fold, P = 0.013). Concomitant 3 μ M TSA exposure also allowed FoxA1 to increase UGT2B11 transcripts by 8.9-fold (P = 0.001) and UGT2B28 mRNA by 8.4-fold (P = 0.001) (Figure 5.8J and M). No *UGT1A* responses were observed to FoxA1 in the presence of TSA.

No changes in HepG2 UGT mRNA levels greater than 2-fold were recorded for cells transfected with FoxA2 expression plasmid alone, or cells that received the combined FoxA2 expression/300 nM TSA treatment. Furthermore, the only response to FoxA2 over-expression in the presence of 3 μ M TSA that was greater than two-fold and reached statistical significance was from *UGT2B28* (2.7-fold, *P* = 0.046).

The effect of FoxA3 over-expression on the UGT mRNA profile of HepG2 cells was most similar to that of FoxA1. Alone, FoxA3 over-expression increased levels of UGT2B15 mRNA by 4.2-fold (P < 0.001) (Figure 5.8K). In addition, although the increase of UGT2B15 transcripts dropped to less than two-fold with simultaneous 300 nM TSA treatment, 3 μ M TSA restored UGT2B15 activation by FoxA3 to 5.4fold (P < 0.001). The results obtained with FoxA3 further resembled the effects of FoxA1 in that FoxA3 was able to cause the accumulation of UGT2B11 (8.9-fold increase, *P* < 0.001) and UGT2B28 (9.9-fold increase, *P* = 0.001) transcripts in HepG2 cells treated with 3 μ M TSA (Figure 5.8 J and M). Because the responses of the human *UGT* genes to FoxA1 and FoxA3 are essentially the same, they will be discussed together.

Two studies that have collectively investigated FoxA3 regulation of five human *CYP* genes have identified five independent patterns of response (Rodriguez-Antona *et al.*, 2003; Bort *et al.*, 2004). Similarly, the study presented in this thesis has identified four human *UGTs* as potential FoxA1 and/or FoxA3 gene targets, with three distinguishable response patterns. *UGT1A1* most resembled *CYP2C9* with a weak response to FoxA1 that was not synergistically increased by TSA. The *UGT2B11* and *UGT2B28* genes only responded to FoxA1/FoxA3 in the presence of 3 μ M TSA, much like *CYP2C8* and *CYP3A4*. And akin to the *CYP2C19* gene, the endogenous *UGT2B15* responses to FoxA1 and FoxA3 were both strengthened by the addition of TSA. Thus, the observation that *UGT* gene family members are regulated by many of the same transcription factors, yet independently of each other through different mechanisms, is a recurring theme that also extends to the regulation of other biotransformation-enzyme superfamilies. Furthermore, comparing the current literature and the work presented in this chapter, there appear to be more similarities

in the FoxA regulation of genes from different enzyme superfamilies than within genetic clusters, suggesting a potential mechanism for the co-ordinate regulation of enzymes required for different stages of chemical metabolism and elimination that would be worthy of further investigation.

FoxA proteins can stably bind to their target sequences within compacted, hypoacetylated chromatin to promote the assembly of enhancer complexes, and the acetylation status of histones does not affect DNA binding by FoxA1 (Cirillo and Zaret, 1999). Furthermore, to date, FoxA factors have not been reported to physically associate with HDACs. Therefore, it is unlikely that TSA greatly enhances the accessibility of FoxA target sites or increases the inherent activity of these proteins. Yet, addition of 3 μ M TSA improved the UGT responses to FoxA1 and FoxA3 in all but one example. One possible explanation for these results is that although overexpressed FoxA proteins bind to the identified genes, and perhaps even remodel their respective promoters, these interactions are insufficient to efficiently recruit and activate the basal transcription machinery. Other transcription factors are presumably required. These may be limited in HepG2 cells and supplied or replaced by TSA; or in the case of repressors, removed or inactivated by TSA. In support of this hypothesis, Rodriguez-Antona and colleagues found that FoxA3 alone did not affect CYP3A4 promoter activity in the context of either plasmid or genomic DNA in HepG2 cells (Rodriguez-Antona et al., 2003). However, FoxA3 was highly cooperative with C/EBP α , activating the endogenous promoter in a hepatocyte-specific manner.

A logical extension of this theory is that it is possible for any of the *UGT* genes that were completely unresponsive to FoxA factors in this assay to still be FoxA targets, either in liver or other tissues; provided that none of the conditions tested supplied all of the factors required for their FoxA-mediated activation. Indeed, it is not expected that FoxA factors are independently capable of driving UGT expression, as FoxA factors are expressed early in development (Friedman and Kaestner, 2006), but UGTs are not. One potential example is UGT2B17, a gene that requires FoxA1 for its oestrogen-responsiveness in MCF-7 cells (Laganiere et al., 2005), but did not respond to FoxA in this experiment. It will be relatively difficult to identify such genes, if they exist, especially if FoxA factors are only required for the inducible component of their expression. siRNA-mediated knockdown of FoxA1 did not affect the basal levels of UGT2B17 transcripts (Laganiere et al., 2005), and it is unknown whether this is because FoxA factors are entirely unnecessary for basal expression, or whether FoxA2 or FoxA3 can compensate for FoxA1 in constitutive transcription but cannot interact with the oestrogen receptor. Certainly, for the currently identified UGT targets of FoxA, FoxA1 and FoxA3 have very similar abilities to drive expression. Expression of a truncated FoxA protein in hepatocyte-derived cells has been shown to decrease expression of certain FoxA-dependent genes (Vallet et al., 1995), and presumably competes with all three FoxA factors by occupying all binding sites. Consequently, this mutant may be a better tool for assessing the requirement of human UGT genes for FoxA factors in HepG2 or primary cells than siRNA knockdown studies.

The lack of responses to the FoxA2 expression plasmid in this experiment may have several causes, including a genuine lack of involvement of FoxA2 in *UGT* expression, deficiencies in the HepG2 proteome such that FoxA2 cannot support transcription from its target genes, insufficient homology of rat FoxA2 with the human protein to allow functional equivalence (unlikely, but possible at 96% identity), or a vector error that prevents FoxA2 expression. Since no sensible

conclusions can be drawn from the presented data if FoxA2 protein was not expressed, the FoxA2 expression vector was validated against a published gene target (see section 5.3.8). The vector was found to be functional; therefore, it is reasonable to conclude that FoxA2 is sufficiently different from the remaining two FoxA family members that it does not participate in *UGT* regulation, at least in the same way. This is an interesting observation, as FoxA1 and FoxA2 are the more closely related in sequence, being 39% identical and 51% similar outside the highly conserved forkhead domain, while FoxA3 is only weakly similar to FoxA1 and FoxA2 (Friedman and Kaestner, 2006).

5.3.7.6. CCATT/enhancer binding protein α

C/EBPa over-expression in HepG2 cells did not alter UGT expression in the absence of TSA, or when performed in conjunction with 300 nM TSA treatment. However, addition of 3 μ M TSA to C/EBPa-expressing cells increased the levels of UGT1A7, UGT2B17 and UGT2B28 mRNA. UGT1A7 mRNA levels increased in pCMX-CEBPa/3 μ M TSA-treated cells to 8.5 times the levels in pCMX/3 μ M TSA-treated controls (P = 0.009) (Figure 5.8E). Likewise, UGT2B17 mRNA levels were increased by 3.3-fold (P = 0.015) and UGT2B28 mRNA levels by 2.4-fold (P =0.027) (Figure 5.8L and M).

Of the *UGT* genes that responded to C/EBP α in HepG2 cells, the most responsive was an extrahepatic gene. C/EBP α was the fourth LETF found to affect UGT1A7 mRNA levels in HepG2 cells, but the only one requiring 3 μ M TSA to do so. The action of this LETF on *UGT1A7* was somewhat surprising, as there is no overlap in the known expression patterns of UGT1A7 and C/EBP α (see Table 1.2 and section 5.1.3.6). It seems therefore, that either C/EBP α does not normally access the *UGT1A7* promoter in tissues such as the liver, lung and intestine, where C/EBP α is

highly expressed, or it does not drive transcription from the UGT1A7 promoter in these tissues when it does bind. This raises the question as to whether C/EBPa could actually be a repressor of UGT1A7 promoter activity. It is possible that C/EBPa can genuinely bind to the UGT1A7 promoter, but normally does so in a repressive capacity. If TSA treatment interferes with co-repressor recruitment by C/EBPa on the UGT1A7 gene, allowing inappropriate association with co-activators, it is conceivable that this experimental system could return a false positive signal for a bound repressor. Generally, C/EBPa is regarded as a positive transcription factor, which can mediate gene transcription through direct interactions with the basal transcription machinery and by recruiting p300/CBP and other chromatin remodelling factors to target promoters. However, a small number of promoters that are repressed by C/EBP α have been reported, including the rat *hnf* δ gene (Rastegar *et* al., 2000). Most recently, a report by McFie and colleagues has shown that the human transcriptional elongation factor CA150 can physically interact with C/EBP α , but only on target genes negatively regulated by C/EBPa (McFie et al., 2006). CA150 is expressed in liver and lung (while gastrointestinal tissues remain untested for CA150 content), so it is feasible that such an interaction could be relevant for the UGT1A7 promoter in these tissues. C/EBP α can also repress target gene expression through recruitment of HDAC-1 (Di-Poi et al., 2005), a mechanism that is clearly vulnerable to TSA treatment. *PPAR* β is one such target gene, and its expression pattern in keratinocytes was found to be mutually exclusive with that of the C/EBPs (Di-Poi et al., 2005). Interestingly, despite all its other regulatory similarities with UGT1A7, UGT1A9 did not respond to C/EBPa, with or without TSA treatment. If UGT1A7 is indeed normally repressed by C/EBPa, this difference could partially explain why UGT1A9 is hepatically expressed, while UGT1A7 is not. To investigate

whether C/EBP α is involved in the regulation of the *UGT1A7* gene (and if so, how), further cloning and *in vitro* experiments will be necessary. In addition, it would be useful to ascertain whether C/EBP α occupies the *UGT1A7* promoter in hepatocytes and other primary cells through ChIP analyses.

UGT2B17 and UGT2B28 are both hepatic genes, yet respond to C/EBP α in a very similar manner to the non-hepatic UGT1A7. Whether this is coincidental or implies functional similarity remains to be seen. It is possible that C/EBP α represses the expression of these genes, despite their hepatic nature, as rat hn/6 is a hepatic gene negatively regulated by C/EBP α . Alternatively, C/EBP α may genuinely positively affect these genes, but in a manner that requires TSA, either to provide gene access or additional transcription factors/co-factors absent from the HepG2 nuclear environment. This is also feasible, as synergistic activation of another gene, human CYP3A4, by C/EBP α and 3 μ M TSA has been previously reported (Rodriguez-Antona *et al.*, 2003). In the published case, TSA was postulated to replace the role of FoxA3 in relaxing the chromatin surrounding the *CYP3A4* C/EBP-binding site, allowing stronger activation by C/EBP α .

UGT2B17 is unique among the human hepatic *UGT* genes in that UGT2B17 mRNA levels showed a tendency to decrease with increasing fibrosis/cirrhosis scores in liver biopsies (Congiu *et al.*, 2002) (it should be noted that *UGT2B28* had not been discovered at this time and was not included in the study). Therefore, transcription factors that exclusively affect *UGT2B17* (or *UGT2B17* and *UGT2B28* only) are potential mediators of this effect. Since C/EBPa levels decrease when significant hepatocyte proliferation is occurring (Mischoulon *et al.*, 1992), which is the case in one third of cirrhotic livers (Donato *et al.*, 2001), it would be of interest to explore whether there is any relationship between C/EBPa and UGT2B17 or UGT2B28

levels in damaged liver tissue. If C/EBP α does control UGT2B17 expression in human liver, it may also play important roles in other tissues where these proteins are co-expressed, such as prostate and lung (Antonson and Xanthopoulos, 1995; Beaulieu *et al.*, 1996)

5.3.7.7. CCATT/enhancer binding protein β

C/EBP β was ineffective as a transactivator for endogenous *UGT* genes in HepG2 cells. Like FoxA2, no significant responses over two-fold were observed for any *UGT* genes exposed to C/EBP β over-expression in the absence of TSA, or after treatment with 300 nM TSA. Combined C/EBP β expression and 3 μ M TSA treatment did cause a 2.9-fold increase in UGT2B28 transcripts (*P* = 0.012) (Figure 5.8M), but no other responses were observed.

Since the only response to C/EBP β was weak, and from *UGT2B28*, a gene that exhibited marginal responses to nearly all of the tested LETFs in the presence of 3 μ M TSA, it seemed wise to test the integrity of the C/EBP β expression vector. Coexpression with a *fabp1* promoter-reporter plasmid (see section 5.3.8) gave a weak, but positive response. Therefore, it was concluded that over-expressed rat C/EBP β has no effect on human UGT mRNA levels in HepG2 cells under the experimental conditions used. However, some further technical points are worthy of consideration before C/EBP β is completely discarded as a potential regulator of human *UGTs*. Although rat C/EBP β is often used to study human promoters, whether or not this result can be directly extrapolated to human C/EBP β requires further investigation as the human and rat C/EBP β proteins share only 71% identity. Also, it would be of interest to determine the ratios of LAP-C/EBP β to LIP-C/EBP β produced in the transfected cells. If LIP-C/EBP β is produced to significant levels from the pCMX-C/EBP β expression vector, this could potentially decrease or completely mask any positive effects that C/EBP β may otherwise have on endogenous target genes (Descombes and Schibler, 1991). Repetition of the experiment with an expression vector designed to express only the human LAP-C/EBP β protein, by substituting the human gene in the vector described in Descombes and Schibler (1991), would confirm the inability of C/EBP β to independently increase UGT expression in HepG2 cells.

The failure of human UGTs to respond in any substantial way to C/EBP β , and for the most part, also C/EBP α , implies that the changes seen in UGT expression around the time of birth, and during inflammation are driven by factors other than C/EBPs. In addition, these observations offer a possible explanation as to why conditions such as cirrhosis and inflammation affect human *CYPs* to a much greater extent than *UGTs*, resulting in acute, clinically relevant, losses of CYP-mediated drug metabolism. Unlike most human *UGT* genes, many human *CYP* genes have been identified as targets of C/EBP α , C/EBP β or both (Jover *et al.*, 1998; Rodriguez-Antona *et al.*, 2003; Bombail *et al.*, 2004; Martinez-Jimenez *et al.*, 2005; Pitarque *et al.*, 2005). Thus, inflammatory stimuli may be able to preferentially affect CYP expression through C/EBP-mediated regulatory pathways.

5.3.8. Validation of the FoxA2 and C/EBPβ expression vectors

To assess whether an expression vector is functional, host cells can be tested for the presence of relevant mRNA by PCR or protein by Western blot; however, neither positive result guarantees that the exogenous protein is folded correctly or that it is capable of normal function. Therefore, the functionality of the pCMX-FoxA2 and pCMX-C/EBP β plasmids was assessed by testing for FoxA2 and C/EBP β activity against an appropriate reporter-promoter construct. The reporter vector chosen,

pTS388, contains 617 nucleotides of the proximal rat *fabp1* gene promoter, inserted into the *Bgl*II and *Kpn*I sites of pGL3-basic (Rowley *et al.*, 2006). This promoter was previously shown to be responsive to HNF1 α , FoxA2 and C/EBP β in HepG2 cells (Divine *et al.*, 2003).

When pTS388 was co-transfected into HepG2 cells with pCMX-HNF1 α or pCMX-FoxA2, increases in firefly luciferase expression relative to the *renilla* luciferase control were observed of 9.1-fold and 4.9-fold respectively (Figure 5.12), confirming that active FoxA2 was expressed from the pCMX-FoxA2 plasmid. On the other hand, no apparent increase in reporter gene activity was detected for pTS388 and pCMX-C/EBP β co-transfections. However, a closer inspection of the raw data revealed that firefly luciferase expression was actually increased from the *fabp1*



Figure 5.12: The pCMX-FoxA2 and pCMX-C/EBP β expression plasmids are active in HepG2 cells. Transient co-transfections of 0.25 µg of pCMX-PL2, pCMX-HNF1 α , pCMX-FoxA2 or pCMX-C/EBP β expression plasmids with 0.5 µg of pTS388 and 25 ng pRL-Null were performed as described in Chapter 2, section 2.2.10. Results are presented as fold increases over the pTS388/pCMX transfections plus one standard deviation. Firefly:*renilla* luciferase ratios, as well as individual changes in firefly and *renilla* luciferase expression are included. NB: This experiment was only performed once.

promoter by pCMX-C/EBPβ, but that *renilla* luciferase expression from the pRL-Null transfection-efficiency control was also increased, and to an even greater extent (Figure 5.12). Therefore, it appears that the pCMX-C/EBPβ plasmid is indeed active, but that pRL- Null is an inappropriate vector to control for transcription efficiency in this situation because it is also C/EBPβ-responsive. In future experiments involving C/EBPβ, better results may be obtained using phRL-Null (Promega) as the internal control, a pRL-Null-derived vector that has had numerous transcription factor binding sites removed in an effort to prevent outcomes such as the one presented. The transcription factor binding sites removed from pRL-Null included multiple C/EBP binding elements (Zhuang *et al.*, 2001).

5.4 General discussion and summary

5.4.1. Achievement of aims

This study achieved both of its aims: to identify new potential regulators of the human hepatic *UGT* genes; and to further investigate the effects of HNF1 and HNF4 α on endogenous *UGT* expression in a liver cell line. Previously unreported interactions between *UGTs* and LETFs identified by this work and warranting further investigation were HNF4 α with *UGT1A1* and *UGT1A6*, HNF6 with *UGT1A4* and *UGT2B11*, FoxA1 and FoxA3 with *UGT2B11*, *UGT2B15* and *UGT2B28* and C/EBP α with *UGT2B17*. In addition, HNF1 α , HNF1 β , HNF4 α and C/EBP α were identified as potential regulators of the *UGT1A7* gene, and these possibilities should be pursued in a more appropriate cell line.

A significant number of observations were also made regarding different patterns of interaction between the LETFs and each *UGT* target, especially for HNF1 α and HNF4 α . Such differences are likely to: a) stem from differing combinations of LETF

interactions with co-factors and other transcription factors; and b) be related to the body's ability to independently regulate expression of each UGT despite high nucleotide sequence similarities and shared transcription factors.

Finally, it was observed that no two UGT genes were identically regulated by the combinations of LETFs and TSA used in this study, and that the UGTs with the most similar LETF responses were not necessarily those considered most closely related. Regardless of whether the human UGTs are grouped based on their amino acid sequences or the nucleotide sequences of their proximal promoters, several gene clusters are evident. These are: UGT1A3, UGT1A4 and UGT1A5; UGT1A7, UGT1A8, UGT1A9 and UGT1A10; UGT2B15 and UGT2B17; and UGT2B11 and UG2B28. Yet these groupings were poorly predictive of promoter function. UGT2B11 and UGT2B28 are closely related (94% amino acid identity) and behaved similarly; yet the only response observed in common for UGT2B15 and UGT2B17, which are also 94% identical, was to HNF1a. There were also some very striking functional similarities between UGT1A1 and UGT1A3, but no overlap between the responses of UGT1A3 and UGT1A4, even though the latter belong to the same UGT cluster. Many of the UGT proteins encoded by the genes of each cluster have substantial overlap in substrate specificity, leading to a considerable level of redundancy within the glucuronidation system. The independent regulation of such genes may be a further safe-guard that ensures continued glucuronidation, at least to some extent, in the event that one gene is adversely affected at the regulatory level.

5.4.2. Broader directions for future investigations

Apart from the various experiments already suggested in section 5.3, there are a number of other investigations that would also add value to the presented results.

These include:

a) A detailed characterisation of the chosen experimental system. To further understand the mechanisms behind the changes in UGT expression in response to LETF over-expression and TSA treatment, it would be beneficial to know what the basal levels of each LETF are under the conditions in which HepG2 cells are grown in our laboratory, and how they are affected by TSA treatment. Additional experiments incorporating TSA concentrations between 300 nM and 3 μ M are also recommended, as these two TSA concentrations produced opposite effects for a number of UGTs. It is necessary to specifically characterise the HepG2 cells under the conditions used in this experiment, because although the expression levels of each LETF used have been previously reported in the literature for HepG2 cells, many of the accounts are conflicting. For example: HNF1B was found in HepG2 cells by Auyeung et al. (2003), but not Kikuchi et al. (2006); FoxA1 was expressed in HepG2 cells when grown by Qian et al. (1995), but not in HepG2 cells cultured by Rodriguez-Antona et al. (2002); C/EBP α was found to be absent in HepG2 cells by Buck et al. (1994) but present by Ishiyama et al. (2003); and both major C/EBPB isoforms were found to be absent from HepG2 cells in a report by Descombes and Schibler (1991), yet Rodriguez-Antona et al. (2002) found that LAP-C/EBPB but not LIP-C/EBPB was expressed in HepG2 cells. It is likely that different culture conditions used between laboratories, or inadvertent selection of HepG2 sub-populations, accounts for many of these inconsistencies. In particular, insulin is a popular additive for HepG2 cultures, although not used in our laboratory, and has been found to alter expression of numerous genes in HepG2 cells and other hepatoma cell lines, including albumin, phospho*enol*pyruvate carboxykinase, LIP-C/EBPβ, the co-factor PGC-1, and several *CYPs* (Campos and Baumann, 1992; Duong *et al.*, 2002; Martinez-Jimenez *et al.*, 2006a; Martinez-Jimenez *et al.*, 2006b). Different detection methods may also account for some differences in perceived expression of a gene, particularly for the C/EBPs where transcription is not well correlated with translation (Williams *et al.*, 1991).

- b) An investigation addressing the relationship between UGT mRNA levels and UGT protein expression in treated cells. Where possible, specific antibodies or substrates could be used to determine whether the LETF/TSA-induced increases in UGT mRNA are translated into corresponding increases in protein. In light of the extremely recent discovery that UGT1A1 and presumably other UGT1A family members can be alternatively spliced into active and inactive forms (Levesque *et al.*, 2007b), it would be pertinent to determine whether the balance of full-length (isoform 1, i1) to the shorter i2 isoform transcripts is altered, particularly with TSA treatment. The real-time assays used in this work will have detected both the UGT1A i1 and i2 mRNAs.
- c) Repetition of the experiment in cell lines derived from other human tissuetypes. Differences in the *UGT* gene responses to LETFs between cell types may give further insight into the mechanisms behind the tissue-specific patterns of UGT expression seen in humans. In particular, it will be of interest to determine whether *UGT1A8* and *UGT1A10* are responsive to the LETFs that are also expressed in the gastrointestinal tract when tested in a more appropriate cell line, such as Caco-2.

d) Experiments designed to test further transcription factors and co-factors, or combinations of LETFs for involvement in UGT regulation. Other regulatory proteins that have potential to regulate UGT promoters include members of a sixth family of LETFs, the proline and acidic amino acid residue-rich family of bZIP transcription factors, especially D-site binding protein (DBP). DBP is expressed only after birth (Mueller et al., 1990), and is important in the expression of several hepatic genes including albumin, CYP7A genes and CYP2C6 (Schrem et al., 2004). DBP can also synergise with HNF1 on target promoters (Babajko and Groyer, 1993). Other potential transcription factors include GATA family members, GR, ubiquitous proteins such as Sp1 and CCAAT box binding factors, and co-factors such as PCG-1. These factors are either known to synergise with LETFs identified as UGT regulators, or have been shown to be involved in TSA-mediated responses of other genes. Therefore, they are logical choices to pursue. GATA-4 occupancy of the albumin gene enhancer requires FoxA1 (Cirillo and Zaret, 1999), and several UGT genes were found to be FoxA1 targets. GR has already been shown to affect UGT1A1 (Sugatani et al., 2005a) and may synergise with HNF4 α on this or other UGT genes (Nitsch et al., 1993). Sp1 and CCAAT box binding factor binding to promoters are commonly found to be enhanced in the presence of TSA (Kwon et al., 2006; Qi and Ratnam, 2006). PCG-1 has been found to be low in HepG2 cells compared to liver, and its relative absence retards the expression of a subset of HNF4 α target genes in these cells (Oberkofler et al., 2004; Martinez-Jimenez et al., 2006b). Therefore, expression of PGC-1 in HepG2 cells may enhance HNF4α-mediated transcription of UGT1A1, UGT1A6, UGT1A7 or UGT1A9, but more

importantly, may reveal more UGT genes that are HNF4 α -responsive, but have different co-factors requirements to those already discovered. PGC-1 is also a co-factor for HNF6, so potentially affects the expression of HNF6target UGT genes as well (Beaudry *et al.*, 2006).

Since many LETFs have been shown to interact synergistically when recruited to the same promoter, it would be astute to test them in combination for their effects on endogenous gene expression in HepG2 cells. Known combinations of LETFs that produce synergistic interactions in the correct context, which could be tested with our current clones, include: HNF1 with HNF4α; HNF4α with PXR, HNF6 or C/EBPα; HNF6 with HNF1, FoxA1, FoxA2 or C/EBPα; FoxA factors with HNF1 or C/EBPα; and C/EBPα with PXR or C/EBPβ.

- e) An assessment of the effects of DNA methylation on UGT expression. Methylation is another common mechanism whereby genes are silenced (Schrem *et al.*, 2002) and may prevent genuine interactions of the LETFs with their target *UGTs*. Furthermore, chromatin acetylation and DNA methylation are dynamically linked, and HDAC inhibitors such as TSA and valproate can trigger the demethylation of a small subset of genes (Cervoni and Szyf, 2001; Milutinovic *et al.*, 2007). The direct effect of methylation on gene expression can be assessed through bisulphite mapping and treatment of cells with DNA demethylating agents such as 5-aza-2'-deoxy-cytidine.
- f) Use of additional models to assess the importance of the newly identified transcription factor interactions with UGT promoters. Although primary human hepatocytes are subject to variability in patient drug history and

genetics, as well as being difficult to procure and transfect, they may be required to detect some LETF interactions with human *UGT* genes. For example, C/EBP α barely increased expression from an aldolase B gene promoter-reporter construct in HepG2 cells, but strongly enhanced reporter expression in primary hepatocytes (Gregori *et al.*, 1993).

Another option is the construction of stable cell lines to test LETF interactions with *UGT* promoters. If synthetic *UGT* promoter fragments integrated into the HepG2 genome behave similarly to the endogenous gene, this system has the advantage that the promoters can be manipulated as required before integration, allowing study of the relative importance of various promoter elements. Also, reporter gene assays are less arduous and costly than real-time PCR. Integrated promoters may be more appropriate than transient transfections when chromatin structure is important in gene regulation: for example, the human β -globin gene is only expressed correctly in transgenic mice when all four DNase-1 hypersensitive sites are included, even though transient transfections only reveal enhancer function for one of these sites (Ellis *et al.*, 1996); and a stably integrated *c-jun* promoter is 20-times more responsive to retinoic acid treatment than the same construct in transfections, while activation proceeds through entirely different elements (Kitabayashi *et al.*, 1992).

Finally, rodent models of human UGT expression will be increasingly useful for testing the role of each LETF in UGT expression. A transgenic mouse strain, Tg-*UGT1*, harbouring the entire known human *UGT1A* locus except much of the *UGT1A8* promoter and the *UGT1A11* and *UGT1A12p* first exons is now available. These mice exhibit a tissue-restricted expression

pattern of human UGT mRNAs that bears many similarities to bona fide human UGT expression (Chen et al., 2005a). A mouse model of the UGT1A8 promoter is also being developed by our laboratory. Cross-breeding of either mouse model with LETF-knockout animals has the potential to clarify which factors are indispensable for UGT expression. Despite some cross-species limitations, enough functional similarities exist between the human and murine orthologues for the LETFs investigated in this chapter that such experiments should be extremely informative. Constitutive or conditional knockout animals are available for hepatic expression of HNF1α (Pontoglio et al., 1996; Lee et al., 1998), HNF4a (Hayhurst et al., 2001), HNF6 (Jacquemin et al., 2000), FoxA1 (although animals die within 10-14 days of birth) (Kaestner et al., 1999), FoxA2 (Sund et al., 2000), FoxA3 (Kaestner et al., 1998), C/EBPa (Lee et al., 1997) and C/EBP_β (Tanaka et al., 1995) but not HNF1B. In particular, it would be of interest to determine whether *UGT1A1* expression is decreased in Tg-*UGT1*/HNF4 $\alpha^{-/-}$ mice. Expression of the mouse Ugtlal homologue is increased when HNF4 α is absent, indicating that HNF4 α is not a crucial factor for transcription of this gene. It was postulated that an increase in bile acid accumulation in HNF4 $\alpha^{-/-}$ mice caused the increased Ugt1a1 expression, through stimulation of PXR and FXR rather than HNF4 α being a repressor of Ugtlal (Ding et al., 2006). If a similar scenario occurs for the human UGT1A1 gene, comparisons with the UGT1A3, UGT1A4, UGT1A6 and UGT1A9 responses will make it possible to refine this hypothesis, as they have different combinations of PXR- and HNF4aresponsive behaviours.

5.4.3. Relevance to pharmacogenetics and disease

As the mechanisms that control the expression of human UGTs become better understood, it will be possible to identify more of the factors that lead to interindividual variation in glucuronidation. The work presented in this chapter identifies a number of transcription factors that may be instrumental in controlling human UGT expression, and accordingly, should be further investigated in the context of interindividual variation. Polymorphisms in the genes coding for these transcription factors or their co-factors, or in their cognate binding sites, may affect UGT expression. Indeed, there are several non-coding polymorphisms in human UGT genes that have been associated with altered promoter activity, UGT expression or adverse drug events (Acuna et al., 2002; Girard et al., 2004), but have not yet been allocated any function. Potentially, these SNPs could alter transcription factor binding sites. Furthermore, stimuli that alter the expression or activity of these transcription factors may also change the balance of UGTs relative to each other, and to other metabolic pathways. As discussed in Chapter 1, all of these parameters may influence drug efficacy or toxicity in an individual, or their vulnerability to diseases caused by xenobiotic exposure or protracted disturbances in homeostasis.

Mutations and polymorphisms exist in the $HNF1\alpha$, $HNF1\beta$ and $HNF4\alpha$ genes, as discussed in Chapter 4, and vary in consequential severity from being primary causes of MODY, to being a potential risk factor for atherosclerosis or type II diabetes, or having no apparent effect (Ryffel, 2001; Babaya *et al.*, 2003; Love-Gregory *et al.*, 2004; Holmkvist *et al.*, 2006). In addition, it is known that mutations in HNF1 α affect different genes to different extents, presumably due to the diversity of roles that HNF1 α can have in driving transcription from different promoters (Soutoglou *et al.*, 2001). Different functional variants of PXR affect the interactions of PXR with
the *CYP3A4* promoter and the *MDR1* gene differently (Hustert *et al.*, 2001; Zhang *et al.*, 2001) and variation in PXR expression has been linked to inflammatory bowel disease (Dring *et al.*, 2006). Therefore, it is conceivable that variants of some LETFs could affect the expression of UGT proteins, and in a manner that preferentially affects a subset of their *UGT* targets according to the mechanisms by which they interact. In contrast, *C/EBPa* appears to be free of frequent functional polymorphisms, but mutations in this gene are associated with haematologic cancers (Gombart *et al.*, 2002). Very little work has been done to identify polymorphisms of the other LETFs that elicited UGT responses, although multiple alleles are known to exist for *HNF6*, *FoxA1* and *FoxA3* (Vaisse *et al.*, 1997).

Transcription factor levels are also known to vary between individuals. For example, the level of HNF1α mRNA in human liver varies up to 10-fold (Toide et al., 2002). Accordingly, target genes may be expressed at levels that are directly related to the concentrations of their most important transcription factors (Toide *et al.*, 2002), or alternatively, may be subject to threshold effects, meaning that relatively small changes in transcription factor concentration can result in relatively large changes in target gene expression (Beaudry et al., 2006). Examples of compounds known to alter transcription factors in human tissues include chenodeoxycholate (decreases HNF1a and HNF4α expression liver) (Jung al., 2007). in et lipopolysaccharide/proinflammatory cytokines (decrease CAR, PXR, RXR, and PGC-1a expression in kidney-derived cells or liver) (Assenat et al., 2004; Wang et al., 2005), insulin (increases LIP-C/EBPß and represses PGC-1a expression in liverderived cells) (Duong et al., 2002; Martinez-Jimenez et al., 2006b), retinoic acid (increases HNF4 α expression in liver-derived cells) (Hatzis and Talianidis, 2001) and genistein (increases HNF4α activity in liver-derived cells) (Ktistaki et al., 1995).

In addition, diet and signalling molecules such as thyroid hormones, growth hormones, retinoic acid, glucocorticoids, insulin, ceramide and cytokines modulate HNF1 α , HNF4 α , HNF6, C/EBP α and/or C/EBP β expression or activity in rodents (Viollet *et al.*, 1997; Lahuna *et al.*, 2000; Park *et al.*, 2004a; Park *et al.*, 2004b; Schrem *et al.*, 2004). Thus, there is plenty of scope for investigating the role of LETF variability in interindividual variation in glucuronidation.

5.4.4. Summary

In short, the work presented in this chapter has identified a number of LETFs that may be instrumental in the expression of human hepatic UGTs, and shown that not all *UGT* targets of a particular transcription factor are co-regulated. Identification of polymorphisms and mutations in the genes encoding these LETFs, as well as in their *UGT* gene binding sites, will improve our understanding of the mechanisms that cause interindividual variation in UGT expression. Therefore, further investigations into the effect of common transcription factor variants on UGT expression are warranted, as are studies that aim to identify more SNPs in human *UGT* promoters. Both types of studies are represented in Chapter 6, where potential causes of interindividual variation in UGT1A3 expression are explored.

CHAPTER SIX GENETIC DETERMINANTS OF HUMAN UGT1A3 EXPRESSION

6.1 Introduction

6.1.1. Genetic variation in human UGT genes

There are a large number of genetic variations known to be present in human *UGT* genes, some of which are already known to alter UGT function or expression, and/or to occur in association with increased risk of developing disease, particularly cancer. As of July 2007, there were 62 officially recognised alleles of *UGT1A1*, 7 of *UGT1A3*, 17 of *UGT1A4*, 7 of *UGT1A5*, 22 of *UGT1A6*, 10 of *UGT1A7*, 4 of *UGT1A8*, 19 of *UGT1A9*, 13 of *UGT1A10*, 24 of *UGT2B4*, 20 of *UGT2B7*, 6 of *UGT2B15*, 2 of *UGT2B17* and 3 of *UGT2B28* (UGT Nomenclature Committee, 2005). The vast majority of these variants are located in the coding region of the *UGT* genes, although some promoter and intronic variants are also included in this list. However, not all distinct alleles with coding region polymorphisms lead to changes in UGT protein, as some represent silent mutations. The known associations between *UGT* polymorphisms, disease and metabolism of pharmaceuticals are detailed in Chapter 1, section 1.8.6.

In addition to the officially recognised *UGT* alleles, there are a growing number of polymorphisms identified in *UGT* promoter regions, which have not all been allocated allele designations because their linkage with coding region variants has not been fully determined. In particular, a thorough study of the *UGT1A9* proximal promoter has identified 15 promoter alleles, several of which are associated with altered expression levels of UGT1A9 protein (Girard *et al.*, 2004). In contrast, at the

time the study reported in this chapter was performed, there was very little information regarding the presence of polymorphisms in the promoters of the UGT1A3-1A5 cluster. Specifically, the only information available for UGT1A3 was that five promoter SNPs had been identified from the human genome sequencing project and had been allocated reference SNP (rs) identification numbers. However, information was available regarding their linkage or functionality. Yet, no interindividual hepatic UGT1A3 mRNA levels have been reported to be among the most variable for human liver UGTs (Congiu et al., 2002). Furthermore, UGT1A3 mRNA has been reported to be expressed in a polymorphic fashion in the human small intestine, being undetectable in a subset of the population screened (Strassburg et al., 2000). Therefore, I decided to investigate whether polymorphisms in the UGT1A3 proximal promoter could be at least partly responsible for the variation observed in UGT1A3 mRNA levels between individuals. In addition, since I had previously identified an HNF1-binding site in the UGT1A3 promoter that is essential for activity in vitro (Chapter 3), the possibility that polymorphisms in the $HNF1\alpha$ gene coding region could affect UGT1A3 promoter activity was also explored.

6.1.2. *HNF1α* gene polymorphisms in humans

Ever since it was established that mutations in the human $HNF1\alpha$ gene lead to an autosomal dominant form of diabetes mellitus known as MODY3, genetic variation in this gene has been of interest (Ryffel, 2001). It is now known that, apart from rare MODY3-causing $HNF1\alpha$ mutations, there are also several relatively common polymorphisms of $HNF1\alpha$ that lead to changes in the HNF1 α amino acid sequence and have more subtle functional consequences. These include polymorphisms that produce the HNF1 α variants HNF1 α I27L, HNF1 α A98V and HNF1 α S487N, chosen for inclusion in this study. The leucine amino acid substitution in HNF1 α I27L occurs in the dimension domain, and has a frequency of 32.3% in healthy Danish Caucasians (Urhammer et al., 1997) and 48.2% in healthy Japanese men (Babaya et al., 2003). The valine amino acid substitution in HNF1 α A98V lays two amino acids to the N-terminal side of the POU domain, and has an allelic frequency of 4.2% in healthy Danish Caucasians (Urhammer et al., 1997). The asparagine amino acid substitution in HNF1α S487N is positioned in the C-terminal HNF1α activation domain, and has an allelic frequency of 29.3% in healthy Danish Caucasians (Urhammer et al., 1997). Two of these HNF1a variants may be associated with disease risk in humans. Two studies have suggested that the I27L HNF1 α variant is associated with insulin resistance (Urhammer et al., 1997; Chiu et al., 2000), although this result could not be replicated in a third, larger group of subjects (Urhammer *et al.*, 1997). In contrast, the I27L HNF1a variant was found to be associated with high levels of serum highdensity lipoprotein-cholesterol, and therefore may be protective against atherosclerosis (Babaya et al., 2003). The HNF1a A98V variation is also associated with disease in humans, being a risk factor for poor pancreatic β -cell function during glucose challenge (Urhammer et al., 1997; Urhammer et al., 1998a). However, no significant functional changes have been attributed to the S487N HNF1α amino acid substitution (Urhammer et al., 1998b).

The remaining two HNF1 α variants used in this study were HNF1 α P291fsinsC and HNF1 α WT+21. HNF1 α P291fsinsC, an HNF1 α MODY3 mutant that occurs in several independent family lines, was included in this study as a negative control, as it has been shown to be a dominant negative inhibitor of HNF1 α function (Yamagata *et al.*, 1998; Ryffel, 2001). On the other hand, HNF1 α WT+21 arises from an mRNA transcript discovered in HepG2 cells by Tamara Height in the process of cloning

HNF1 α from these cells. This transcript contains an additional 21 nucleotides (AGGCTGCTCTGCTCCCCAGG) - derived from intron 8 of *HNF1\alpha* and inserted at position +1647, resulting in a 7-amino acid insertion in the carboxyl-terminal activation domain (Ryffel, 2001). It was unknown whether the resulting protein would have altered activity, but since HepG2 cells were derived from a human hepatocellular carcinoma (suggesting that this transcript may be present in this or other cancer types) it was of interest to investigate.

6.1.3. Aims

The work presented in this chapter was designed to investigate whether genetic variations in the *UGT1A3* proximal promoter or the *HNF1a* coding region were significant contributors to interindividual differences in UGT1A3 expression in humans. Therefore, the aims were to:

- 1. To sequence the proximal promoter of the *UGT1A3* gene from human genomic DNA samples in order to identify *UGT1A3* promoter nucleotide polymorphisms and investigate their segregation into alleles;
- 2. To clone the distinct *UGT1A3* promoter alleles identified in Aim 1 and to investigate their transcriptional activities *in vitro*;
- To compare the chosen HNF1α variants HNF1α WT+21, HNF1α I27L, HNF1α A98V, HNF1α S487N and HNF1α P291fsinsC with the reference HNF1α protein for the ability to transactivate the UGT1A3 promoter *in vitro*.

6.2 Methods

6.2.1. Amplification of UGT1A3 sequences from genomic DNA

Genomic DNA samples isolated from unrelated patients of German descent with colon cancer (n = 50) and matched controls (n = 51) were kindly provided by Prof. J. Abel (Medical Institute of Environmental Hygiene, Dusseldorf, Germany). The *UGT1A3*-883bp promoter was amplified from 100 ng of genomic DNA in a two-round nested PCR using 0.5 Units *Taq* DNA polymerase in a reaction volume of 20 µl (see Chapter 2, section 2.2.6.4). The first round of PCR was performed using primers 1A3/4prom-1.5k and 1A3/1A4/1A5+619rev (see Table 6.1), designed to amplify all three promoters of the *UGT1A3-1A5* cluster. This was a deliberate design feature, intended to allow sequence analysis of all three promoters without requiring addition genomic DNA material for each; thus preserving a limited resource. The PCR conditions were: initial denaturation at 95°C for 4 minutes; 35 cycles of 95°C for 30 seconds, annealing at 50°C for 30 seconds and extension at 72°C for 1 minute; and a final extension step at 72°C for 5 minutes.

The second round of amplification was specific for the *UGT1A3* promoter and utilised primers 1A3prom-884bp and 1A3CDS+352 (Table 6.1). The PCR regime was as for the first round of amplification, with two exceptions: an annealing temperature of 60°C, and only 32 cycles of amplification. After sequencing, identified alleles were reamplified from the first round PCR product with primers 1A3prom-884NheI and 1A3UTRXho1 and cloned into the *Nhe*I and *Xho*I sites of pGL3-basic.

Oligonucleotide	Nucleotide Sequence (5'→3')	Nucleotide Position on Target Gene(s)	RE
1A3/4prom-1.5k	AGCCAT <u>GCTAGC</u> TAAGGGGGGGTTGGAGGAATAGT	<i>UGTIA3</i> : -1539 to -1519; <i>UGTIA4</i> : -1574 to -1554; <i>UGTIA5</i> : -1550 to -1530	NheI
1A3/1A4/1A5+619rev	ATGTCATGTGGTCTGAATTGG	<i>UGT1A3</i> : +619 to +599; <i>UGT1A4</i> : +619 to +599; <i>UGT1A5</i> : +619 to +599	NA
1A3prom-884bp	GCCTGGATGACTGAAATAAAG	<i>UGT1A3</i> : -884 to -864	NA
1A3CDS+352	TCAACATTGCCATACTTCTGA	<i>UGT1A3</i> : +352 to +332	NA
1A3prom-884NheI	AGCCATGCTAGCGCCTGGATGACTGAAATAAAG	<i>UGT1A3</i> : -884 to -864	NheI
1A3UTRXho1	AGCCAT <u>CTCGAG</u> CTCAGCAGAAGACACGGACA	UGT1A3: -1 to -20	XhoI
1A3prom-108bp	CACGTTGATTTGCTAAGTGG	UGT1A3: -108 to -89	NA
1A3intron1rev	TGGATGAAGGCACCAATACA	<i>UGT1A3</i> : +890 to +871 (intronic)	NA
RE: restriction endonucleas	e site, as underlined. NA: not applicable.		

Amplification of the *UGT1A3* exon 1 sequence from genomic DNA samples was achieved using the PCR primers and conditions reported by Iwai *et al.* (2004). These were primers 1A3prom-108bp and 1A3intron1rev (Table 6.1) and initial denaturation at 95°C for 2 minutes; 32 cycles of 95°C for 1 minute, annealing at 62°C for 1 minute and extension at 72°C for 2 minutes; and a final extension step at 72°C for 8 minutes.

6.2.2. HNF1a expression vectors

The wild-type and variant HNF1 α expression vectors used in this study are detailed in Chapter 2, section 2.1.4.

6.2.3. Transient transfection and luciferase reporter assay

Transient co-transfections of 0.5 μ g of pGL3-1A3-promoter-reporter plasmid, 0.25 μ g of HNF1 α expression plasmid and 25 ng pRL-Null were performed as described in Chapter 2, section 2.2.10, using HepG2 cells seeded at a density of 2 × 10⁵ cells per well in 24-well plates. Cells were lysed 48-hours post-transfection using passive lysis buffer, and the lysates assayed for firefly and *renilla* luciferase activity as described in Chapter 2, section 2.2.11.

6.2.4. Statistical analysis

Statistical analysis of *UGT1A3* promoter variant frequencies and compliance with Hardy-Weinberg equilibrium (tested by Pearson χ^2) were performed using Microsoft Office Excel 2003 (Microsoft Corporation, WA) and SPSS (SPSS Inc., IL) software.

6.3 **Results and Discussion**

6.3.1. The UGT1A3 promoter contains multiple SNPs, resulting in five alleles

Sequence analysis of the proximal 884 nucleotides of the *UGT1A3* promoter in 101 individuals uncovered 10 SNPs and one insertion/deletion event, as detailed in Table

6.2. By comparison to the published UGT1A3 promoter sequence, I deduced that seven of these SNPs segregated into five alleles that were found to be in Hardy-Weinberg equilibrium (df(10) χ^2 = 13.19, P = 0.213) in the tested population. Five of these seven SNPs had been previously reported by the HapMap project (see Table 6.2), but the two rarest (-148Y and -553R) were novel. The remaining SNPs occurred only as heterozygous changes in a single sample each, and were therefore designated mutations, as their allelic frequencies were $\leq 0.5\%$. It was not determined whether these mutations were present in the original genomic template or were a product of the first round of PCR amplification, but they were certainly not introduced in the second PCR amplification reaction, as all were retrieved by re-amplification and cloning from first-round PCR products. Of the five genuine UGT1A3 promoter alleles, the most common (frequency = 0.55) was the reference sequence. The next most regularly observed allele was UGT1A3 Promoter 2 (-66C/-204G/-581T/-751C/ -758G), with a frequency of 0.32. The remaining three alleles were rare, found at frequencies of less than 0.05 (5%). The functional integrity of the UGT1A3 Promoter 3 allele (-66C/-148C/-204G/-581T/-751C/-758G) was of particular interest, as the T-148C SNP is located within the HNF1-binding site of the UGT1A3 promoter.

After the completion of this project and during preparation of this thesis, Caillier *et al.* (2007) published a study of the *UGT1A3* promoter that produced similar findings to those reported here. Caillier's study of 249 Caucasians from the Québec Family Study identified the exact same seven polymorphisms in the *UGT1A3* promoter, although they found that these segregated into six alleles rather than five. Five of the *UGT1A3* promoter alleles identified by Caillier *et al.* (2007) were identical to those described in this chapter, with similar frequencies as found in the German population (Reference sequence = 55% in both studies; Promoter 2/H2 = 32% in Germans

Alleles				Nucleotide p	osition			и	Frequency
	-66 ^a	-148	-204 ^b	-553	-581 ^c	-751 ^d	-758°		
Reference	Т	Т	Α	IJ	C	Т	А	111	0.55
Promoter 2	C	Т	Ċ	Ċ	Т	C	IJ	65	0.32
Promoter 3	C	C	Ċ	Ċ	Т	C	IJ	٢	0.035
Promoter 4	C	Т	Ċ	А	Т	C	IJ	٢	0.035
Promoter 5	С	Т	G	G	Т	Т	G	8	0.04
Mutations				Descript	ion				Frequency
Promoter A	Reference	allele with C to G	i mutation at posit	tion -48				1	0.005
Promoter B	Reference	allele with T to C	mutation at posit	100 - 99				1	0.005
Promoter C	Promoter	2 with G to C mut	ation at position -	207				1	0.005
Promoter D	10 bp dele	tion with 13 bp in	sertion. Replace:	UGTIA3-200 to	-191 (TCGGTCT	TTT) with "AAA	ACTGTGGGGCC		0.005

versus 28.7% for Cailler *et al.* (2007); Promoter 3/H3 = 3.5% versus 4.6%; Promoter 4/H5 = 3.5% versus 4.0%; and Promoter 5/H4 = 4% versus 6%, respectively). The sixth allele, not present in the German population studied, was formed by the -581T polymorphism occurring in isolation from the remaining polymorphisms, and was only present in the Québec Family Study at a frequency of 1.6%. Thus, it remains unclear whether this allele is not present in people of German descent, or whether sequencing a larger cohort of German individuals would reveal its occurrence at a low frequency. Although the sequences identified by Caillier *et al.* (2007) have since been allocated *UGT1A3** allele names, unfortunately these names cannot be adopted in this work because the linkage between the promoter alleles identified herein and *UGT1A3* coding region variants was not determined.

6.3.2. Activities of the newly defined UGT1A3 promoter alleles in vitro

The four new *UGT1A3* promoter alleles and four mutated sequences (Table 6.2) were re-amplified, cloned into the pGL3 reporter vector and used in transient transfections to establish whether their activities were altered, relative to the published sequence. This experiment revealed a small amount of variation between the basal activities of the *UGT1A3* promoter alleles in HepG2 and Caco-2 cells, the extent of which was less than 1.4-fold (Figure 6.1A and B). Of the differences observed, only the mutant *UGT1A3* Promoter A and D alleles had statistically altered activity relative to the reference allele in HepG2 cells (P = 0.009 and P = 0.003 respectively). In Caco-2 cells, the *UGT1A3* Promoter 3, B, C and D alleles also had statistically altered activity relative to the reference allele (P = 0.016, P = 0.019, P = 0.003 and P =0.018 respectively). The alternative *UGT1A3* promoter sequences also all responded



Figure 6.1: Alternative UGT1A3 promoter alleles vary less than 1.4-fold in basal activity. Five alternative UGT1A3-884bp promoter alleles, including the reference sequence, were cloned into the pGL3 reporter vector and transfected into A. HepG2 or B. Caco-2 cells. A further four promoter constructs containing the mutations reported in Table 6.2 were also tested. All transfections contained 25 ng of the pRL-Null control vector, were performed in triplicate and were assayed for luciferase and *renilla* activity 48 hours post-transfection as described in "Methods". Results are presented as the mean firefly luciferase activities relative to the internal *renilla* control, plus one standard deviation. The values of the promoter-less pGL3-basic control transfections are set to 1 (indicated by the white bar). Dark grey bars indicate the activities of genuine UGT1A3 promoter alleles; light grey bars indicate the activities of the *UGT1A3* reference promoter sequence containing observed mutations. *P* values for the indicated comparisons are $\dagger \dagger \dagger P = 0.003$, $\ddagger \ddagger P = 0.009$ and $\$P \ge 0.016$ but ≤ 0.019 .

strongly to over-expression of HNF1 α in HEK293T cells, in a similar manner to the reference construct. None of the effects of HNF1 α on the variant promoters differed to the wild-type response by more than 2-fold (Figure 6.2), although all but *UGT1A3* Promoter allele B were statistically increased relative to the reference promoter sequence. In particular, the *UGT1A3* Promoter 3 allele (-66C/-148C/-204G/-581T/



Figure 6.2: Alternative UGT1A3 promoter alleles are not decreased in their ability to respond to over-expressed HNF1a. Five alternative UGT1A3-884bp promoter alleles, including the reference sequence, were cloned into the pGL3 reporter vector and transfected into HEK293T cells. A further four promoter constructs containing the mutations reported in Table 6.2 were also tested. All transfections contained 25 ng of the pRL-Null control vector and 0.25 µg pCMX-PL2 or pCMX-HNF1a, were performed in triplicate and were assayed for luciferase and *renilla* activity 48 hours post-transfection as described in "Methods". Results are presented as the mean firefly luciferase activities relative to the internal *renilla* control, plus one standard deviation. The values of the promoter-less pGL3-basic control transfections are set to 1. Dark grey bars indicate the activities of genuine *UGT1A3* promoter alleles with HNF1a. *P* values for the indicated comparisons are **P* < 0.001, $\ddagger P = 0.001$, $\ddagger P = 0.006$ and \$ P = 0.01.

-751C/-758G) that contains the 148C polymorphism within the HNF1 α -binding site was increased by 1.3-fold (P = 0.006), even though this nucleotide change theoretically slightly decreases the integrity of the HNF1-binding element. Therefore, it seemed that genetic differences in the *UGT1A3* proximal promoter are unlikely to be a major cause of interindividual variation in UGT1A3, although it cannot be ruled out that the identified polymorphisms may have a greater effect in the genomic context.

Interestingly, the results of this study disagree with those reported in the recent publication by Caillier et al. (2007). While I found that there were no statistically significant differences in the activities of any of the four genuine promoter variants (Promoters 2-5) relative to the activity of the reference UGT1A3 promoter sequence in HepG2 cells (Figure 6.1A), Caillier et al. (2007) found that the activity of these same four promoter alleles (H2-H5) were decreased by 2 to 2.5-fold, and the promoter not identified in my study (H6) was also decreased by 30%. The cause of these differing results is unknown, but may be due to differences in cell culture conditions for the HepG2 cells used, or the length of the promoter constructs used. The constructs used in the Québec study contained 1144 bp of UGT1A3 promoter, rather than 884 bp. Yet, Caillier and colleagues found that there were no further polymorphisms in the UGT1A3 promoter region between nucleotides -1144 and -884 (Caillier *et al.*, 2007): so, if it transpires that promoter length is important, this would indicate that the relationship between the allelic variants and promoter activity is more complex than just the altered binding of one or more transcription factors over the polymorphic regions, and it would be of importance to test much greater lengths of the UGT1A3 promoter for allele-determined function.

6.3.3. Regulation of the UGT1A3 promoter by HNF1a variants

HNF1 α is subject to polymorphic variation in humans and is an important transcription factor for the UGT1A3 promoter in vitro. Therefore, because the differences in basal activities of the UGT1A3 promoter alleles were insufficient to explain the extent of UGT1A3 mRNA variation observed in human tissues, it was investigated whether variants of HNF1a could affect the rate of transcription of this gene. The UGT1A3-500bp promoter was co-transfected into HEK293T cells with pCMX vectors expressing three polymorphic HNF1a variants discovered in humans (I27L, A98V and S487N), HNF1a WT+21, and the HNF1a mutant P291fsinsC that is associated with MODY3. It was found that the three constructs encoding proteins with single amino acid substitutions all had 20-25% lower activity towards the UGT1A3 promoter than wild-type HNF1a (Figure 6.3). In contrast, the P291fsinsC mutant could not support any transcription from the reporter construct in HEK293T cells and the HNF1 α WT+21 variant was slightly more active than the reference HNF1α construct (Figure 6.3A). In HepG2 cells, the P291fsinsC HNF1α mutant was found to behave in a dominant negative manner (Figure 6.3B) as previously reported (Yamagata et al., 1998). These results suggest that the functional effects of the HNF1a I27L, A98V and S487N polymorphisms are unlikely to account for much of the observed interindividual variation of UGT1A3 expression *in vivo*. However, this experiment was limited in scope in that the HNF1 α variants could only be tested as over-expressed protein; lower concentrations of the less active HNF1a variants may cause non-linear losses of promoter transcription. Furthermore, no adjustments were made for any possible variance in the levels of HNF1 α expressed from the alternative expression constructs.



Figure 6.3: Known HNF1a protein variants are insufficient to explain the variability of UGT1A3 mRNA levels in humans. HNF1a variants as previously described by Mackenzie and colleagues (2005a) were tested for their ability to regulate the reference *UGT1A3*-500bp promoter in A. HEK293T cells or B. HepG2 cells. Transfections were performed as per "Methods" and contained 0.5 µg of pGL3 or pGL3-1A3-500, 0.25 µg of empty pCMX-PL2 or pCMX vectors encoding the HNF1a variants and 25 ng pRL-Null. Results are the means obtained from triplicate samples, expressed as a relative value of firefly luciferase activity to the internal *renilla* control, compared to the pGL3-basic control (set to 1). Error bars indicate one standard deviation. WT: wild-type. *P* values for the indicated comparisons are **P* < 0.001, $\dagger\dagger\uparrow$ *P* = 0.003, $\ddagger P = 0.005$, $\ddagger P = 0.007$ and \$ P = 0.024.

The *UGT1A3* Promoter 2 and Promoter 3 alleles were also tested for activity in combination with the HNF1 α variants in HEK293T cells, but the results were as found for the reference *UGT1A3* promoter, with no *UGT1A3* promoter/HNF1 α variant combinations resulting in changes in reporter gene expression of greater than 1.4-fold (Figure 6.4). Promoter 2 was chosen because it is the most prevalent allele other than the *UGT1A3* reference sequence, while Promoter 3 was chosen due to the presence of the nucleotide difference within the identified HNF1-binding site.



Figure 6.4: Two UGT1A3 promoter variants interact with the tested HNF1a protein variants similarly to the UGT1A3 reference promoter sequence. HNF1a variants as previously described by Mackenzie et al. (2005a) were tested for their ability to regulate the UGT1A3-883bp Promoter 2 (P2) and Promoter 3 (P3) reporter constructs in HEK293T cells, relative to the UGT1A3-883bp reference promoter (Ref)/wild-type (WT) HNF1a combination. Transfections were performed as per "Methods" and contained 0.5 µg of pGL3 or pGL3-1A3-883 reporter constructs, 0.25 µg of empty pCMX-PL2 or pCMX vectors encoding the HNF1a variants and 25 ng pRL-Null. Results are the means obtained from triplicate samples, expressed as a relative value of firefly luciferase activity to the internal renilla control, compared to the pGL3-basic control (set to 1). Error bars indicate one standard deviation. P values for the indicated comparisons are *P < 0.001, §P = 0.034 and §§P = 0.041.

6.3.4. The UGT1A3 Promoter 2 allele is under-represented in a colon cancer cohort and is associated with the W11R/V47A protein variant

When the genomic DNA donors were categorised according to UGT1A3 promoter genotype and colon cancer status, it was found that while the control population continued to obey Hardy-Weinberg equilibrium (df(10) $\chi^2 = 1.004$, P = 1.000), the cancer patient population deviated from the expected genotype frequencies (df(10) χ^2 = 20.73, P = 0.023). Analysis of the genotypes present in each sub-population revealed a significant reduction (P = 0.004) in the occurrence of the UGT1A3 Promoter 2 homozygous genotype in the cancer patients relative to the control population (Table 6.3). All other genotypes were equally distributed ($P \ge 0.298$). Since functional assays did not suggest any mechanism by which the Promoter 2/Promoter 2 UGT1A3 promoter genotype could be protective against colon cancer, it was postulated that it was, instead, behaving as a biomarker. Therefore, it was investigated whether the UGT1A3 Promoter 2 allele was associated with any known UGT1A3 protein variant. Six of the eight genomes that were homozygous for the UGT1A3 Promoter 2 allele were sequenced over the UGT1A3 exon 1 sequence using the method of Iwai et al. (2004). The UGT1A3 first exon was also sequenced from six individuals known to be homozygous for the published promoter sequence, for comparison. All genomes that were homozygous for the reference UGT1A3 promoter were also found to be homozygous for the UGT1A3 exon 1 sequence UGT1A3*1. However, all samples homozygous for the UGT1A3 Promoter 2 allele were found to be homozygous for the UGT1A3*2 allele, which encodes the W11R/V47A UGT1A3.2 protein variant reported by Iwai et al. (2004).

The finding that individuals homozygous for the *UGT1A3* Promoter 2 allele were under-represented among German colon cancer patients, similar to the previously

<i>UGT1A3</i> promoter genotype ^a	Control $(n = 51)$	Colon cancer $(n = 50)$
UGT1A3 Reference/Reference	15	14
UGT1A3 Reference/Promoter 2	19	21
UGT1A3 Reference/Promoter 3	1	2
UGT1A3 Reference/Promoter 4	2	1
UGT1A3 Reference/Promoter 5	2	2
UGT1A3 Promoter 2/Promoter 2	8	0*
UGT1A3 Promoter 2/Promoter 3	1	0
UGT1A3 Promoter 2/Promoter 4	1	2
UGT1A3 Promoter 2/Promoter 5	1	3
UGT1A3 Promoter 3/Promoter 3	0	1
UGT1A3 Promoter 3/Promoter 4	0	1
UGT1A3 Reference/Promoter A	0	1
UGT1A3 Promoter 2/Promoter B	0	1
UGT1A3 Reference/Promoter C	0	1
UGT1A3 Reference/Promoter D	1	0

Table 6.3:	Association of	of UGT1A3	promoter	genotype	with c	olon cancer.

^a Allelic sequences as defined in Table 6.2. * P = 0.004, compared with control by Pearson χ^2 .

reported reduction in the homozygous *UGT1A7*1* genotype among patients with colorectal cancer (Strassburg *et al.*, 2002b), was unexpected. Since all 6 individuals homozygous for the *UGT1A3* Promoter 2 allele tested were also homozygous for *UGT1A3*2*, it seems likely that the *UGT1A3* Promoter 2 sequence is part of the *UGT1A3*2* allele, or at least strongly linked to the W11R and V47A polymorphisms. Evidence supporting the latter hypothesis is found in the recent publication by Caillier *et al.* (2007), where 98.6% of the *UGT1A3* H2 promoter (Promoter 2) haplotypes also contained the W11R and V47A coding region polymorphisms. The remaining two H2 promoter alleles only occurred once each in 498 chromosomes

and, therefore, would be considered mutations by the criteria used in this study and are indeed designated "hypothetical alleles" by Caillier and colleagues. Of the provably genuine *UGT1A3* alleles in the Québecian study containing the H2 promoter, 92.9% would result in the UGT1A3.2 (W11R/V47A) protein and the remainder would produce UGT1A3.6 (W11R/V47A/M270V).

The in vivo functional consequences of possessing one or more copies of the UGT1A3*2 allele remain unknown. Iwai and colleagues found that the W11R/V47A UGT1A3.2 protein variant had 369% of UGT1A3.1 activity towards oestrone (Iwai et al., 2004). On the basis of these results, it was hypothesised that UGT1A3.2 may be protective for diseases that have serum oestrone levels as a risk factor, such as osteoporosis and colon carcinoma (Iwai et al., 2004). However, the only subsequent study (Caillier et al., 2007) to investigate the enzymatic activity of UGT1A3.2 towards oestrone found that this variant had only 61% of UGT1A3.1 activity towards this substrate (difference not significant). Furthermore, it was found that UGT1A3.6 was an extremely low activity enzyme, with only 0.1% of UGT1A3.1 oestroneglucuronidating activity (Caillier et al., 2007). As discussed by Callier et al. (2007), there are a number of possible reasons for the observed differences between the two studies, including dissimilarities in the expression, assay and analytical methods used, and it is likely that the second study is the more accurate assessment of UGT1A3 activity. However, further investigations are necessary to definitively characterise the relationship between polymorphisms in UGT1A3 and its ability to glucuronidate oestrone, and ideally, between the presence of UGT1A3 variants and oestrone levels in humans in vivo.

If it transpires that differences in oestrone glucuronidation between UGT1A3 variants is not a likely risk factor for colon cancer, another possibility is that

UGT1A3.2 has increased activity towards its known carcinogenic substrates (such as primary aromatic amines and benzo[a]pyrene metabolites) (Mojarrabi *et al.*, 1996; Green *et al.*, 1998a) or is able to metabolise a carcinogen(s) that is not a substrate of the wild-type protein, and is consequently better able to protect the colonic mucosa from chemical damage. It is already becoming evident that different UGT1A3 protein variants are affected differently in their ability to metabolise various UGT1A3 substrates. For instance, UGT1A3.4, which has an unchanged or lowered capacity to glucuronidate oestrone (Iwai *et al.*, 2004; Caillier *et al.*, 2007), has a greatly increased ability to glucuronidate the flavonoids quercetin, luteolin and kaempferol relative to UGT1A3.1, with a changed preference in regioselectivity (Chen *et al.*, 2006b). Thus, it is clear that it will be necessary to better characterise common variants of all UGT enzymes for their preferred substrates before we can fully understand how genetic variation in these proteins is likely to affect human health.

One final possible explanation for the observed absence of Promoter 2 homozygotes among colon cancer patients worth discussing is that, although this association is statistically significant, that it is the result of a statistical type I error. As the presented study is limited in scope, only encompassing 101 individuals, it is possible that the association seen is a product of chance, and that inclusion of more individuals will cause the association to lose significance. Similar difficulties have been reported for other studies, including one that attempted to identify an association between amino acid variants of HNF1 α and pancreatic beta-cell function. In this instance, a study of 74 individuals returned a statistically significant association of the HNF1 α I27L variant with poor beta-cell function in an oral glucose tolerance test, but a second, larger study of 230 individuals failed to replicate these findings (Urhammer *et al.*, 1998b). It would certainly be wise to further investigate the potentially important association of *UGT1A3* gene variants with colon cancer risk, and determine whether the findings of this study hold true in a larger cohort.

6.4 General discussion and summary

6.4.1. Achievement of aims

This study confirmed the presence of 5 known *UGT1A3* promoter polymorphisms in a German population and detected a further two. These polymorphisms were able to be allocated to 5 distinct alleles, fulfilling the first stated aim of this work. The remaining two aims, to determine whether these polymorphisms or variations in HNF1 α affected *UGT1A3* promoter activity were also carried out successfully, with the results indicating that neither are likely to be major contributors to interindividual variation in UGT1A3 expression *in vivo*.

6.4.2. Future directions

There are a number of improvements and extensions that could be made to this study to further investigate the impact of genetic variation on the expression of UGT1A3. These include:

- a) Sequencing longer segments of the *UGT1A3* proximal promoter in order to identify nucleotide polymorphisms further upstream and investigate their relationship with promoter function.
- b) Testing a larger colon cancer case-control cohort for evidence that the *UGT1A3* Promoter 2 allele is associated with a protective effect against the development of colon cancer.

- c) Further investigating the relationship between HNF1 α and UGT1A3 transcription; in particular, titration of the HNF1 α variants against the UGT1A3 promoter to determine whether any HNF1 α variants are less able to activate UGT1A3 transcription when present at low concentration. Also, variation in HNF1 α levels may be more important in determining UGT1A3 expression than the presence of HNF1 α variants. Therefore, it would be of interest to investigate whether HNF1 α mRNA/protein levels correlate with UGT1A3 levels, in a similar manner as was found for UGT2B7 (Toide *et al.*, 2002).
- d) Identifying additional regulators of the UGT1A3 promoter and investigating whether variation in their activity or expression level affects UGT1A3 expression. Candidates include PXR (Rae *et al.*, 2001; Gardner-Stephen *et al.*, 2004) and AhR (Chen *et al.*, 2005a).
- e) Investigating genetic variation in other *UGT* genes, particularly the closely related gene *UGT1A4*. As there is some overlap in UGT1A3 and UGT1A4 substrates (see Chapter 3, section 3.1.3), it may be that the combination of *UGT1A3* and *UGT1A4* alleles inherited is more informative than either gene considered in isolation. Such a hypothesis would be reasonable, as certain combinations of *UGT1A1* and *UGT1A9* alleles have already been proposed to be important in predicting patient response and the likelihood of suffering toxicity when treated with irinotecan (Innocenti *et al.*, 2005; Girard *et al.*, 2006).

6.4.3. Relevance to pharmacogenetics

Although it has been known for some time that UGT1A3 is expressed in a polymorphic manner along the gastrointestinal tract (Strassburg et al., 2000), and with a significant degree of interindividual variation in the liver (Congiu et al., 2002), it has only more recently become clear that different human populations have very different frequencies of the various UGT1A3 variants. The second most common UGT1A3 variant, UGT1A3.2, occurs with a frequency of 35.9% in Caucasians from Québec (Caillier et al., 2007), but only 12.5% in people of Japanese extraction (Iwai et al., 2004) and 14% in a Chinese Han population (Chen et al., 2006b). The data presented in this chapter indicates that it is likely that the UGT1A3.2 protein variant is also approximately twice as common in German Caucasians as in Japanese/Chinese people; a result supported by two earlier studies of the UGT1A3 V47A polymorphism (which coincides with UGT1A3.2 at a frequency of 93.2%, at least in Caucasians (Caillier et al., 2007)). The first showed that in German people, alanine occurs in UGT1A3 amino acid position 47 at a frequency of 35% (Ehmer et al., 2004); the second showed that UGT1A3 47A occurs at a frequency of 37.9% in Caucasians and 16.3% in Asians (Thomas et al., 2006). Therefore, it would be prudent to not only continue investigating the determinants of interindividual variation in UGT1A3 expression, but also to comprehensively characterise the catalytic activity of UGT1A3.2 and other UGT1A3 variants. It is clear that not all UGT1A3 variants are equally able to glucuronidate the same substrates, and that there may be substances that are better substrates for UGT1A3.2 or the other variants, than they are for the widely tested UGT1A3.1.

Whether interindividual variation in UGT1A3 expression has a significant impact on human health and will therefore be useful as a pharmacogenetic target remains to be seen. In the past, UGT1A3 has not been regarded highly as an important contributor to human glucuronidation. This opinion was largely based on the observation that for many of the originally recognised UGT1A3 substrates, other UGT enzymes appeared to be more relevant *in vivo* (Tukey and Strassburg, 2000). However, UGT1A3 was recently reported to be important in the metabolism of 26,26,26,27,27,27,27-F₆-1 α ,23*S*,25-trihydroxyvitamin D₃ (Kasai *et al.*, 2005), the anti-oestrogenic drug fulvestrant (Chouinard *et al.*, 2006) and the bile acids CDCA and lithocholic acid (Verreault *et al.*, 2006). As such, it is probable that other unique or major substrates exist. Now, with the discovery that there is at least one highly prevalent UGT1A3 protein variant other than UGT1A3.1 present in humans, particularly in Caucasians, the possibility that UGT1A3 is an important polymorphic contributor to human glucuronidation is an interesting prospect. In particular, further work should be done to investigate the possible link between UGT1A3 and the risk of colon cancer.

Although no mechanisms were identified that can adequately explain the absence of UGT1A3 expression in the colon of some individuals (Strassburg *et al.*, 2000) or the hepatic interindividual variation seen in humans (Congiu *et al.*, 2002), this study adds to our knowledge of genetic variation within the *UGT1A* locus. If the glucuronidative capacity of an individual is eventually going to become a useful predictor/tool in personalised medicine, it is likely that it will be necessary to consider genetic variation within the *UGT1A* locus, and possibly all *UGT* genes, as a single entity (although the importance of particular variants may also need to be weighted towards the most relevant enzymes depending on the disease being studied). It is already known that linkage between distant *UGT1A* genes occurs in the human genome (Innocenti *et al.*, 2005; Thomas *et al.*, 2006). In addition, UGT enzymes can affect the activity of each other when present in the same cell (Fujiwara

et al., 2007a; Fujiwara *et al.*, 2007b); therefore, the combination of enzymes present in a tissue is likely to be more important than the sum of the activities of each. Furthermore, the *UGT1A* locus is subject to regulation through alternative splicing of the common exons that results in the production of inactive, truncated enzyme (Levesque *et al.*, 2007b). Thus, it seems likely that some polymorphisms of *UGT* genes not directly involved in the glucuronidation of a particular substrate will nonetheless be found to be important in pharmacogenetic predictions for that substrate. As such, the study presented in this chapter is one small but important step towards understanding the *UGT1A* locus, and thus, the greater goal of achieving effective personalised medical treatment.

6.4.4. Summary

Single nucleotide polymorphisms in *UGT* regulatory regions have previously been correlated with altered expression of UGT1A1 and UGT1A9 (Girard *et al.*, 2004; Girard *et al.*, 2005). Therefore, I chose to investigate whether SNPs in the *UGT1A3* promoter could also account for the variability in hepatic mRNA levels observed by Congiu *et al.* (2002) and Mojarrabi *et al.* (1996), and/or the polymorphic intestinal expression reported by Strassburg *et al.* (2000). Sequencing of 101 unrelated individuals of German descent revealed seven SNPs that occurred with sufficient frequency to be considered true polymorphisms. It was also determined that these polymorphisms could be accounted for by five alleles; which were subsequently cloned and tested for altered promoter activity. However, it was found that the basal activities of each promoter allele were comparable to the *UGT1A3* reference sequence in both Caco-2 and HepG2 cells, and that all promoters were similarly responsive to HNF1 α in HEK293T cells. Likewise, testing of the functional HNF1 α 127L, A98V and S487N variant proteins against the *UGT1A3* promoter only revealed

variations in activity of up to 25%. These results imply that genetic polymorphisms in the *UGT1A3* promoter and the transcription factor HNF1 α only contribute a small proportion of the observed variance *in vivo*.

CHAPTER SEVEN GENERAL SUMMARY AND CONCLUSIONS

7.1 Towards disease prevention and designer therapies

As discussed in Chapter 1, there are many incentives for thoroughly understanding the biology of human UGTs, including a better appreciation of their role in maintaining human health, and likely improvements in pharmaceutical drug design and usage. Indeed, clinically relevant outcomes have already been achieved from research into the UGT1A1 gene. The usefulness of such information will further increase once we can consider haplotype structures across both the UGT1 and UGT2 gene loci, as well as UGT variation in combination with other biotransformation enzyme variants. As it becomes realistic to consider whole metabolic networks rather than individual enzymes, it is reasonable to expect that it will be possible to make health care improvements such as safely developing and prescribing therapeutics that would ordinarily be discarded after causing severe adverse reactions in a small minority of patients, as high-risk genotypes could be identified and excluded from treatment (Thomas et al., 2006). For other drugs, where the ratio of two UGT forms, or of UGT to alternative metabolic enzyme, determines the efficacy or toxicity of a drug, an intimate knowledge of regulatory mechanisms may allow the development of strategies to temporally alter the relative expression of the relevant enzymes, achieving a better outcome from the original therapeutic. Variation in UGT activity and expression in humans is prevalent, and importantly, potentially manipulable; many UGT genes are now known to be targets of nuclear receptors whose activity can be altered by xenobiotic exposure. Thus, it is clear that research into the activity

and regulation of human biotransformation enzymes is an important investment in future medical practice.

7.2 Summary of the research findings presented in this thesis

The overall aim of this thesis, to substantially expand the knowledge of *UGT* regulation in humans, was achieved through four independent pieces of work, summarised as follows.

7.2.1. Chapter 3: *In vitro* characterisation of the *UGT1A3*, *UGT1A4* and *UGT1A5* proximal promoters

The UGT1A3, UGT1A4 and UGT1A5 genes are highly related, sharing greater than 85% nucleotide sequence identity in their 1 kb proximal promoters. Yet, they vary considerably in their expression patterns; to the extent that while UGT1A3 and UGT1A4 are considered key hepatic enzymes, UGT1A5 expression in humans is currently thought to be negligible. Furthermore, it is clear that the UGT1A3 and UGT1A4 genes are independently regulated, despite their extensive similarities. To explore the mechanisms responsible for these observations, the UGT1A3, UGT1A4 and UGT1A5 promoters were cloned and analysed by deletion, mutation and HNF1 α/β -over-expression experiments. The ensuing work established that putative HNF1-binding sites present in all three promoters are functional *in vitro*, but that while HNF1 factors are critical for UGT1A3 and UGT1A4 promoter activity, they are also insufficient to drive high levels of transcription. Two additional elements required for the maximal activity of the UGT1A3 promoter in liver-derived cells were also identified, at least one of which appears to be shared by the UGT1A4 promoter, but is only active in the context of UGT1A3. The discussion of this work highlights the likely functional relevance of these findings with respect to the

independent regulation of these genes, the lack of UGT1A5 expression in humans, and the implications for further pharmacogenomics research into these three genes.

7.2.2. Chapter 4: HNF1 transcription factors are essential for the UGT1A9 promoter response to HNF4α

Of the closely related UGT1A7, UGT1A8, UGT1A9 and UGT1A10 genes, UGT1A9 is the only member expressed in the liver. A study published during the course of this PhD candidature showed that, of this gene cluster, HNF4 α regulated only the UGT1A9 gene in a positive manner (Barbier *et al.*, 2005). The work presented in this chapter extends and refines these observations: identifying a major element through which HNF4 α interacts with the UGT1A9 promoter; showing that the HNF4 α response of the UGT1A9 promoter is completely dependent on the presence of HNF1 factors; and establishing that there are at least three major functional differences between the UGT1A8 and UGT1A9 promoters that allow HNF1 and HNF4 α to cooperatively regulate only the latter in hepatocyte-derived cells. The discussion of this work explores the differences between the regulation of UGT1A9 and other human genes by HNF1 and HNF4 transcription factors, and how variability in the expression and activity of such transcription factors could contribute to the variability of UGT1A9 expression in humans.

7.2.3. Chapter 5: Regulation of endogenous UGT expression in HepG2 cells by liver-enriched transcription factors

To overcome some of the potential drawbacks of studying gene regulation using transiently transfected promoters, and to identify new transcriptional regulators of the human hepatic *UGT* genes, liver-enriched transcription factors were over-expressed in HepG2 cells and the endogenous UGT mRNA levels subsequently measured. This experiment was performed both in the absence and presence of a

chromatin-relaxing agent, TSA. A number of interesting interactions between the chosen transcription factors and endogenous UGT transcription were observed, including previously unreported interactions of HNF4 α with *UGT1A1* and *UGT1A6*, HNF6 with *UGT1A4* and *UGT2B11*, FoxA1 and FoxA3 with *UGT2B11*, *UGT2B15* and *UGT2B28* and C/EBP α with *UGT2B17*. In addition, although UGT1A7 is not hepatically expressed, HNF1 α , HNF1 β , HNF4 α and C/EBP α were identified as potential regulators of the *UGT1A7* gene.

Another important set of observations presented and discussed in Chapter 5 were of the differences in the way that HNF1 α and HNF4 α interact with each of their target genes. In particular, HNF1 α has been proposed as an important regulator of all human hepatic UGT genes on the basis of DNA sequence and in vitro promoterreporter data, yet there is currently little information about the relative importance of this transcription factor in the expression of each gene in vivo. The second purpose of this study was, therefore, to investigate the hypothesis that HNF1 α is not equally important in the transcriptional hierarchy of each UGT gene. In support of this hypothesis, the hepatic UGT genes could be separated into four distinct groups based on their responses to HNF1 α and TSA. Four UGT genes showed no direct evidence of regulation by HNF1a (UGT1A1, UGT1A3, UGT2B7 and UGT2B10), two responded to the over-expression of HNF1 α alone (UGT1A6 and UGT1A9) and six responded to HNF1 α over-expression in the presence of TSA (UGT1A4, UGT2B4, UGT2B11, UGT2B15, UGT2B17 and UGT2B28). Of the four genes that did not respond to HNF1a over-expression in this system, two (UGT1A1 and UGT1A3) were increased in the presence of TSA despite a concomitant loss of HNF1 α expression, and two (UGT2B7 and UGT2B10) were decreased after TSA treatment.

Thus, this chapter extends our knowledge of the transcription factors likely to be important in the regulation of human *UGTs*, and shows that even when genes share conserved transcription factor binding sites, the mechanisms involved in their regulation may be significantly divergent. Knowing new potential transcriptional regulators for the human *UGTs* opens up fresh avenues for research into the determinants of interindividual variation in UGT expression.

7.2.4. Chapter 6: Genetic determinants of human UGT1A3 expression

UGT1A3 is expressed in a polymorphic manner along the gastrointestinal tract, and with a significant degree of interindividual variation in the liver. However, the cause of these variations remains unknown. The study presented in this chapter examined the genetic variation present in the *UGT1A3* promoter of people of German descent, finding seven polymorphisms that segregated into five alleles. These alleles were tested for promoter function *in vitro*, and were found to have similar activities to the reference sequence (within 1.4-fold activity of the reference allele). Similarly, none of the identified polymorphisms severely affected the ability of the *UGT1A3* promoter to respond to HNF1a over-expression *in vitro*. However, when the *UGT1A3* genotypes were grouped according to whether the donor was a colon cancer patient or matched control, it was found that there was a significant difference in the genotype distribution. Possible mechanisms for this association were discussed and further work was recommended to investigate the possible link between *UGT1A3* genotype and the risk of colon cancer.

Since HNF1 α was identified as an important regulator of *UGT1A3* promoter activity *in vitro*, the studies presented in Chapter 6 were also extended to investigate the effect of several common *HNF1* α coding polymorphisms on *UGT1A3* promoter

function. It was found that the chosen HNF1 α variants were still highly active towards the *UGT1A3* promoter in transfections.

Overall, the results presented in Chapter 6 imply that genetic polymorphisms in the UGT1A3 promoter and the transcription factor $HNF1\alpha$ are likely to contribute only a small proportion of the observed variance in UGT1A3 expression *in vivo*. Further research into the regulation of the UGT1A3 promoter will be required to discover why UGT1A3 expression is so inconsistent between individuals.

7.3 Relevant work published over the duration of this PhD candidature

Apart from the work presented in this thesis, there have been other significant advancements made into the research of human *UGT* regulation since the commencement of this PhD candidature in 2004, the majority of which has been in the understanding of the inducible expression of UGTs. However, a new focus of identifying functional polymorphisms in *UGT* gene promoters has also recently emerged.

Recent advancements regarding the inducible regulation of human *UGTs* include recognition that: a) multiple NRREs of the *UGT1A1* PBREM are responsible for the activation of the *UGT1A1* gene by flavonoids, with the greatest contribution from the PBREM XRE (Sugatani *et al.*, 2004); b) the *UGT2B7* promoter possesses a NRRE that allows negative regulation by FXR (Lu *et al.*, 2005); c) oxidants such as *tert*butylhydroquinone can increase UGT1A1 transcription through binding of Nrf2 to an anti-oxidant response element that flanks the *UGT1A1* PBREM XRE (Yueh and Tukey, 2007); d) liganded PPAR α also activates the *UGT1A1* promoter through a NRRE located within the PBREM (Senekeo-Effenberger *et al.*, 2007); e) bile acids can up-regulate the *UGT1A3* promoter through a NRRE that binds LXRa (Verreault *et al.*, 2006); f) the *UGT2B15* gene is unique among the *UGT2B* genes (although *UGT2B28* was not tested) in being oestrogen-responsive (Harrington *et al.*, 2006); and g) the *UGT1A1* promoter responds to GR ligand exposure *in vitro*. Two alternative mechanistic models have been proposed for the induction of UGT1A1 expression by glucocorticoids. Firstly, the research group lead by Dr. Mizutani found that induction of UGT1A1 transcription by dexamethasone or cortisol is reliant on GR and the *UGT1A1* HNF1 α -binding site, but is independent of the PBREM and appears to be mediated indirectly (Kanou *et al.*, 2004; Usui *et al.*, 2006a; Usui *et al.*, 2006b; Kuno *et al.*, 2007). Conversely, Sugatani and colleagues found that the *UGT1A1* PBREM contains functionally important glucocorticoid response elements and that liganded GR enhances the PXR- and CAR-mediated transactivation of the *UGT1A1* promoter through the co-activator GRIP-1 (Sugatani *et al.*, 2005a). Further work will be needed to resolve these apparently conflicting results.

In addition to the above research, the construction of a transgenic mouse model (Tg-*UGT1*) that carries a large portion of the human *UGT1A* locus has allowed further research into the induction of *UGT1A* genes by xenobiotics and hormones. AhR or PXR ligands were able to induce the transcription of all human UGT1A mRNAs in the gastrointestinal tract, and all but UGT1A5, UGT1A7 and UGT1A8 in the liver of Tg-*UGT1* mice. Furthermore, these effects were enhanced in the presence of glucocorticoid (Chen *et al.*, 2005a). The *in vitro* responses of the human *UGT1A1* gene to oxidative stress and a PPAR α agonist, and of *UGT1A3* to bile acids, were also confirmed in Tg-*UGT1* mice (Verreault *et al.*, 2006; Senekeo-Effenberger *et al.*, 2007; Yueh and Tukey, 2007). Recent research addressing the basic regulation of human UGT genes or the effect of promoter polymorphisms on UGT promoter function is still limited, but has shown that: a) UGT1A9 is an HNF4 α target gene (Barbier *et al.*, 2005); b) the UGT1A9 promoter contains polymorphisms at positions -275, -331/-440, -665 and -2152 that can be correlated with UGT1A9 expression levels in the liver (although the mechanisms responsible are yet to be elucidated) (Girard *et al.*, 2004); c) Cdx2 and HNF1 α co-operatively regulate the UGT2B7 promoter (Gregory *et al.*, 2006); d) UGT1A3 promoter variants have significantly lower transcriptional activity than the accepted reference promoter sequence (Caillier *et al.*, 2007) (contrary to the work shown in this thesis); and e) a UGT2B7 promoter polymorphism at nucleotide position -840 affects morphine glucuronidation in sickle cell disease patients (Darbari *et al.*, 2007). Finally, and somewhat controversially, UGT1A8 and UGT1A10 transcripts have recently been reported to be present in primary human hepatocytes (Li *et al.*, 2007), despite the widely held view that these UGT forms are strictly extrahepatic.

The work presented in this thesis complements the current bias that exists in human *UGT* research towards exploring inducible regulation. This is an important contribution, as a thorough understanding of inducible regulation will also ultimately require an understanding of the underlying constitutive control of *UGT* expression.

7.4 Modelling human UGT gene regulation

This thesis demonstrates the value of using different experimental models to investigate human *UGT* regulation. Unfortunately, there is no naturally occurring, convenient model of human UGT expression. Standard laboratory animals, such as rats and mice, do have similar *UGT* loci to humans (Mackenzie *et al.*, 2005b), but
many of the human UGT genes exhibit a higher similarity to each other than to any UGT genes in rodents, so meaningful orthologous relationships cannot be reliably determined. Maturation of glucuronidation during development is also very different between rats and humans (Ring et al., 1999) and there are significant gender-related differences in UGT expression in mice not seen in humans (Buckley and Klaassen, 2007). Even the humanised Tg-UGTI mouse, which is being used very successfully to provide much information about inducible expression of human UGT1A forms, is likely to be subject to significant limitations due to differences that exist between mice and humans in regulatory protein networks and in the activation of nuclear receptors by ligands. On the other hand, human cell lines usually exhibit very different gene expression patterns to the tissues from which they were derived, and their metabolism of xenobiotics can be very different from primary tissues (Smith et al., 2005b; Bonzo et al., 2007). Yet, primary cells, such as hepatocytes, are difficult to procure, don't transfect well, and undefined genetic variation or history of xenobiotic exposure between donors can confound experiments (McCarver and Hines, 2002). DNA introduced into primary or immortalised cells for transient reporter assays is not moderated by its normal chromosomal context, with unknown consequences. Finally, there are obvious ethical limitations to the extent of the data that can be collected using human subjects. However, as demonstrated by this thesis and the recent literature, significant progress can be made when multiple, complementary models are used to explore human UGT regulation.

7.5 Final remarks

The design of this thesis was influenced by two important concepts. Firstly, it has been estimated that, although nonsynonymous changes in gene coding regions are the most commonly studied genetic alterations, they are outnumbered by functional *cis*-acting regulatory polymorphisms that remain largely uncharacterised (Johnson *et al.*, 2005). Secondly, polymorphisms that alter the expression or activity of *trans*-acting factors have been predicted to be dominant determinants of gene expression patterns (Morley *et al.*, 2004). Thus, the intent and achievement of the work presented in this thesis was to provide significant new insight into the regulatory control of human *UGT* genes, and to identify mechanisms that can be further explored as potential contributors to interindividual variation in UGT expression.

APPENDIX ONE pGL3-BASIC REPORTER VECTOR MAP



APPENDIX TWO pGL3+ REPORTER VECTOR MAP



BIBLIOGRAPHY

Abernethy, D.R., Divoll, M., Greenblatt, D.J. and Ameer, B. (1982) Obesity, sex, and acetaminophen disposition. *Clin. Pharmacol. Ther.* **31**: 783-790.

Abernethy, D.R., Greenblatt, D.J., Divoll, M. and Shader, R.I. (1983) Enhanced glucuronide conjugation of drugs in obesity: studies of lorazepam, oxazepam, and acetaminophen. *J. Lab. Clin. Med.* **101**: 873-880.

Acuna, G., Foernzler, D., Leong, D., Rabbia, M., Smit, R., Dorflinger, E., Gasser, R., Hoh, J., Ott, J., Borroni, E., To, Z., Thompson, A., Li, J., Hashimoto, L. and Lindpaintner, K. (2002) Pharmacogenetic analysis of adverse drug effect reveals genetic variant for susceptibility to liver toxicity. *Pharmacogenomics J.* **2**: 327-334.

Adegoke, O.J., Shu, X.O., Gao, Y., Cai, Q., Breyer, J., Smith, J. and Zheng, W. (2004) Genetic polymorphisms in uridine diphospho-glucuronosyltransferase 1A1 (UGT1A1) and risk of breast cancer. *Breast Cancer Res. Treat.* **85**: 239-245.

Aitken, A.E., Richardson, T.A. and Morgan, E.T. (2006) Regulation of drugmetabolizing enzymes and transporters in inflammation. *Ann. Rev. Pharmacol. Toxicol.* **46**: 123-149.

Akiyama, T.E. and Gonzalez, F.J. (2003) Regulation of P450 genes by liver-enriched transcription factors and nuclear receptors. *Biochim. Biophys. Acta* **1619**: 223-234.

Alam, T., An, M.R. and Papaconstantinou, J. (1992) Differential expression of three C/EBP isoforms in multiple tissues during the acute phase response. *J. Biol. Chem.* **267**: 5021-5024.

Albert, C., Vallee, M., Beaudry, G., Belanger, A. and Hum, D.W. (1999) The monkey and human uridine diphosphate-glucuronosyltransferase UGT1A9, expressed in steroid target tissues, are estrogen-conjugating enzymes. *Endocrinology* **140**: 3292-3302.

Anders, M.W. (2004) Glutathione-dependent bioactivation of haloalkanes and haloalkenes. *Drug Metab. Rev.* **36**: 583-594.

Anderson, G.D. (2005) Pregnancy-induced changes in pharmacokinetics: a mechanistic-based approach. *Clin. Pharmacokinet.* **44**: 989-1008.

Andersson, S., Davis, D.L., Dahlback, H., Jornvall, H. and Russell, D.W. (1989) Cloning, structure, and expression of the mitochondrial cytochrome P-450 sterol 26hydroxylase, a bile acid biosynthetic enzyme. *J. Biol. Chem.* **264**: 8222-8229.

Ando, Y., Saka, H., Ando, M., Sawa, T., Muro, K., Ueoka, H., Yokoyama, A., Saitoh, S., Shimokata, K. and Hasegawa, Y. (2000) Polymorphisms of UDP-glucuronosyltransferase gene and irinotecan toxicity: a pharmacogenetic analysis. *Cancer Res.* **60**: 6921-6926.

Aninat, C., Piton, A., Glaise, D., Le Charpentier, T., Langouet, S., Morel, F., Guguen-Guillouzo, C. and Guillouzo, A. (2006) Expression of cytochromes P450, conjugating enzymes and nuclear receptors in human hepatoma HepaRG cells. *Drug Metab. Dispos.* **34**: 75-83.

Antonson, P., Stellan, B., Yamanaka, R. and Xanthopoulos, K.G. (1996) A novel human CCAAT/enhancer binding protein gene, C/EBPepsilon, is expressed in cells of lymphoid and myeloid lineages and is localized on chromosome 14q11.2 close to the T-cell receptor alpha/delta locus. *Genomics* **35**: 30-38.

Antonson, P. and Xanthopoulos, K.G. (1995) Molecular cloning, sequence, and expression patterns of the human gene encoding CCAAT/enhancer binding protein alpha (C/EBPalpha). *Biochem. Biophys. Res. Commun.* **215**: 106-113.

Aono, S., Adachi, Y., Uyama, E., Yamada, Y., Keino, H., Nanno, T., Koiwai, O. and Sato, H. (1995) Analysis of genes for bilirubin UDP-glucuronosyltransferase in Gilbert's syndrome. *Lancet* **345**: 958-959.

Aouabdi, S., Gibson, G. and Plant, N. (2006) Transcriptional regulation of the *PXR* gene: identification and characterization of a functional peroxisome proliferator-activated receptor alpha binding site within the proximal promoter of *PXR*. *Drug Metab*. *Dispos*. **34**: 138-144.

Araki, J., Kobayashi, Y., Iwasa, M., Urawa, N., Gabazza, E.C., Taguchi, O., Kaito, M. and Adachi, Y. (2005) Polymorphism of *UDP-glucuronosyltransferase 1A7* gene: a possible new risk factor for lung cancer. *Eur. J. Cancer* **41**: 2360-2365.

Archer, T.K., Lefebvre, P., Wolford, R.G. and Hager, G.L. (1992) Transcription factor loading on the MMTV promoter: a bimodal mechanism for promoter activation. *Science* **255**: 1573-1576.

Aronow, B.J., Ebert, C.A., Valerius, M.T., Potter, S.S., Wiginton, D.A., Witte, D.P. and Hutton, J.J. (1995) Dissecting a locus control region: facilitation of enhancer function by extended enhancer-flanking sequences. *Mol. Cell. Biol.* **15**: 1123-1135.

Artursson, P. and Karlsson, J. (1991) Correlation between oral drug absorption in humans and apparent drug permeability coefficients in human intestinal epithelial (Caco-2) cells. *Biochem. Biophys. Res. Commun.* **175**: 880-885.

Assenat, E., Gerbal-Chaloin, S., Larrey, D., Saric, J., Fabre, J.M., Maurel, P., Vilarem, M.J. and Pascussi, J.M. (2004) Interleukin 1beta inhibits CAR-induced expression of hepatic genes involved in drug and bilirubin clearance. *Hepatology* **40**: 951-960.

Australian Institute of Health and Welfare (2005) Health system expenditure on disease and injury in Australia, 2000-01. 2nd ed. Health and Welfare Expenditure Series 21. Canberra.

Auyeung, D.J., Kessler, F.K. and Ritter, J.K. (2001) An alternative promoter contributes to tissue- and inducer-specific expression of the rat UDP-glucuronosyltransferase 1A6 gene. *Toxicol. Appl. Pharmacol.* **174**: 60-68.

Auyeung, D.J., Kessler, F.K. and Ritter, J.K. (2003) Differential regulation of alternate UDP-glucuronosyltransferase 1A6 gene promoters by hepatic nuclear factor-1. *Toxicol. Appl. Pharmacol.* **191**: 156-166.

Babajko, S. and Groyer, A. (1993) Interplay of the liver-enriched trans-acting factors, DBP and HNF1, in the transactivation of human IGFBP-1 promoter. *Biochem. Biophys. Res. Commun.* **196**: 480-486.

Babaya, N., Ikegami, H., Fujisawa, T., Nojima, K., Itoi-Babaya, M., Inoue, K., Nakura, J., Abe, M., Yamamoto, M., Jin, J.J., Wu, Z., Miki, T., Fukuda, M. and

Ogihara, T. (2003) Association of I27L polymorphism of hepatocyte nuclear factorlalpha gene with high-density lipoprotein cholesterol level. *J. Clin. Endocrinol. Metab.* **88**: 2548-2551.

Bach, I., Mattei, M.G., Cereghini, S. and Yaniv, M. (1991) Two members of an HNF1 homeoprotein family are expressed in human liver. *Nucleic Acids Res.* **19**: 3553-3559.

Bach, I. and Yaniv, M. (1993) More potent transcriptional activators or a transdominant inhibitor of the HNF1 homeoprotein family are generated by alternative RNA processing. *EMBO J.* **12**: 4229-4242.

Bailey, M.J. and Dickinson, R.G. (2003) Acyl glucuronide reactivity in perspective: biological consequences. *Chem. Biol. Interact.* **145**: 117-137.

Bailly, A., Torres-Padilla, M.E., Tinel, A.P. and Weiss, M.C. (2001) An enhancer element 6 kb upstream of the mouse HNF4alpha1 promoter is activated by glucocorticoids and liver-enriched transcription factors. *Nucleic Acids Res.* **29**: 3495-3505.

Baraka, O.Z., Truman, C.A., Ford, J.M. and Roberts, C.J. (1990) The effect of propranolol on paracetamol metabolism in man. *Br. J. Clin. Pharmacol.* **29**: 261-264.

Barbacci, E., Chalkiadaki, A., Masdeu, C., Haumaitre, C., Lokmane, L., Loirat, C., Cloarec, S., Talianidis, I., Bellanne-Chantelot, C. and Cereghini, S. (2004) *HNF1beta/TCF2* mutations impair transactivation potential through altered coregulator recruitment. *Hum. Mol. Genet.* **13**: 3139-3149.

Barbier, O., Duran-Sandoval, D., Pineda-Torra, I., Kosykh, V., Fruchart, J.C. and Staels, B. (2003a) Peroxisome proliferator-activated receptor alpha induces hepatic expression of the human bile acid glucuronidating UDP-glucuronosyltransferase 2B4 enzyme. *J. Biol. Chem.* **278**: 32852-32860.

Barbier, O., Girard, H., Inoue, Y., Duez, H., Villeneuve, L., Kamiya, A., Fruchart, J.C., Guillemette, C., Gonzalez, F.J. and Staels, B. (2005) Hepatic expression of the UGT1A9 gene is governed by HNF4alpha. *Mol. Pharmacol.* **67**: 241-249.

Barbier, O., Torra, I.P., Sirvent, A., Claudel, T., Blanquart, C., Duran-Sandoval, D., Kuipers, F., Kosykh, V., Fruchart, J.C. and Staels, B. (2003b) FXR induces the UGT2B4 enzyme in hepatocytes: a potential mechanism of negative feedback control of FXR activity. *Gastroenterology* **124**: 1926-1940.

Barbier, O., Turgeon, D., Girard, C., Green, M.D., Tephly, T.R., Hum, D.W. and Belanger, A. (2000) 3'-azido-3'-deoxythimidine (AZT) is glucuronidated by human UDP-glucuronosyltransferase 2B7 (UGT2B7). *Drug Metab. Dispos.* **28**: 497-502.

Barbier, O., Villeneuve, L., Bocher, V., Fontaine, C., Torra, I.P., Duhem, C., Kosykh, V., Fruchart, J.C., Guillemette, C. and Staels, B. (2003c) The UDP-glucuronosyltransferase 1A9 enzyme is a peroxisome proliferator-activated receptor alpha and gamma target gene. *J. Biol. Chem.* **278**: 13975-13983.

Bartoov-Shifman, R., Hertz, R., Wang, H., Wollheim, C.B., Bar-Tana, J. and Walker, M.D. (2002) Activation of the insulin gene promoter through a direct effect of hepatocyte nuclear factor 4alpha. *J. Biol. Chem.* **277**: 25914-25919.

Barua, A.B. and Sidell, N. (2004) Retinoyl beta-glucuronide: a biologically active interesting retinoid. *J. Nutr.* **134**: 286S-289S.

Bas, A., Forsberg, G., Hammarstrom, S. and Hammarstrom, M.L. (2004) Utility of the housekeeping genes 18S rRNA, beta-actin and glyceraldehyde-3-phosphatedehydrogenase for normalization in real-time quantitative reverse transcriptasepolymerase chain reaction analysis of gene expression in human T lymphocytes. *Scand. J. Immunol.* **59**: 566-573.

Baumhueter, S., Mendel, D.B., Conley, P.B., Kuo, C.J., Turk, C., Graves, M.K., Edwards, C.A., Courtois, G. and Crabtree, G.R. (1990) HNF-1 shares three sequence motifs with the POU domain proteins and is identical to LF-B1 and APF. *Genes Dev.* **4**: 372-379.

Beaudry, J.B., Pierreux, C.E., Hayhurst, G.P., Plumb-Rudewiez, N., Weiss, M.C., Rousseau, G.G. and Lemaigre, F.P. (2006) Threshold levels of hepatocyte nuclear factor 6 (HNF-6) acting in synergy with HNF-4 and PGC-1alpha are required for time-specific gene expression during liver development. *Mol. Cell. Biol.* **26**: 6037-6046.

Beaulieu, M., Levesque, E., Hum, D.W. and Belanger, A. (1996) Isolation and characterization of a novel cDNA encoding a human UDP-glucuronosyltransferase active on C19 steroids. *J. Biol. Chem.* **271**: 22855-22862.

Belanger, A., Pelletier, G., Labrie, F., Barbier, O. and Chouinard, S. (2003) Inactivation of androgens by UDP-glucuronosyltransferase enzymes in humans. *Trends Endocrinol. Metab.* **14**: 473-479.

Benedetti, M.S., Whomsley, R., Baltes, E. and Tonner, F. (2005) Alteration of thyroid hormone homeostasis by antiepileptic drugs in humans: involvement of glucuronosyltransferase induction. *Eur. J. Clin. Pharmacol.* **61**: 863-872.

Berger, S.A., Cole, T.J., Schmid, W. and Schutz, G. (1996) Molecular genetic analysis of glucocorticoid and mineralocorticoid signaling in development and physiological processes. *Steroids* **61**: 236-239.

Bernard, O. and Guillemette, C. (2004) The main role of UGT1A9 in the hepatic metabolism of mycophenolic acid and the effects of naturally occurring variants. *Drug Metab. Dispos.* **32**: 775-778.

Bernard, P., Goudonnet, H., Artur, Y., Desvergne, B. and Wahli, W. (1999) Activation of the mouse TATA-less and human TATA-containing UDP-glucuronosyltransferase 1A1 promoters by hepatocyte nuclear factor 1. *Mol. Pharmacol.* **56**: 526-536.

Bernardini, S., Bellincampi, L., Ballerini, S., Federici, G., Iori, R., Trequattrini, A., Ciappi, F., Baldinetti, F., Bossu, P., Caltagirone, C. and Spalletta, G. (2005) Glutathione S-transferase P1 *C allelic variant increases susceptibility for late-onset Alzheimer disease: association study and relationship with apolipoprotein E epsilon4 allele. *Clin. Chem.* **51**: 944-951.

Bertilsson, G., Heidrich, J., Svensson, K., Asman, M., Jendeberg, L., Sydow-Backman, M., Ohlsson, R., Postlind, H., Blomquist, P. and Berkenstam, A. (1998) Identification of a human nuclear receptor defines a new signaling pathway for *CYP3A* induction. *Proc. Natl. Acad. Sci. U. S. A.* **95**: 12208-12213.

Besnard, V., Wert, S.E., Hull, W.M. and Whitsett, J.A. (2004) Immunohistochemical localization of Foxa1 and Foxa2 in mouse embryos and adult tissues. *Gene Expr. Patterns* **5**: 193-208.

Bhalla, S., Ozalp, C., Fang, S., Xiang, L. and Kemper, J.K. (2004) Ligand-activated pregnane X receptor interferes with HNF-4 signaling by targeting a common coactivator PGC-1alpha. Functional implications in hepatic cholesterol and glucose metabolism. *J. Biol. Chem.* **279**: 45139-45147.

Bigler, J., Whitton, J., Lampe, J.W., Fosdick, L., Bostick, R.M. and Potter, J.D. (2001) *CYP2C9* and *UGT1A6* genotypes modulate the protective effect of aspirin on colon adenoma risk. *Cancer Res.* **61**: 3566-3569.

Birkenmeier, E.H., Gwynn, B., Howard, S., Jerry, J., Gordon, J.I., Landschulz, W.H. and McKnight, S.L. (1989) Tissue-specific expression, developmental regulation, and genetic mapping of the gene encoding CCAAT/enhancer binding protein. *Genes Dev.* **3**: 1146-1156.

Bjornstrom, L. and Sjoberg, M. (2002) Mutations in the estrogen receptor DNAbinding domain discriminate between the classical mechanism of action and crosstalk with Stat5b and activating protein 1 (AP-1). *J. Biol. Chem.* **277**: 48479-48483.

Blackwood, E.M. and Kadonaga, J.T. (1998) Going the distance: a current view of enhancer action. *Science* **281**: 60-63.

Bock, K.W., Gschaidmeier, H., Heel, H., Lehmkoster, T., Munzel, P.A., Raschko, F. and Bock-Hennig, B. (1998) AH receptor-controlled transcriptional regulation and function of rat and human UDP-glucuronosyltransferase isoforms. *Adv. Enzyme Regul.* **38**: 207-222.

Bock, K.W. and Kohle, C. (2005) UDP-glucuronosyltransferase 1A6: structural, functional, and regulatory aspects. *Methods Enzymol.* **400**: 57-75.

Bock, K.W., Schrenk, D., Forster, A., Griese, E.U., Morike, K., Brockmeier, D. and Eichelbaum, M. (1994) The influence of environmental and genetic factors on CYP2D6, CYP1A2 and UDP-glucuronosyltransferases in man using sparteine, caffeine, and paracetamol as probes. *Pharmacogenetics* **4**: 209-218.

Bock, K.W., Wiltfang, J., Blume, R., Ullrich, D. and Bircher, J. (1987) Paracetamol as a test drug to determine glucuronide formation in man. Effects of inducers and of smoking. *Eur. J. Clin. Pharmacol.* **31**: 677-683.

Boj, S.F., Parrizas, M., Maestro, M.A. and Ferrer, J. (2001) A transcription factor regulatory circuit in differentiated pancreatic cells. *Proc. Natl. Acad. Sci. U. S. A.* **98**: 14481-14486.

Bombail, V., Taylor, K., Gibson, G.G. and Plant, N. (2004) Role of Sp1, C/EBPalpha, HNF3, and PXR in the basal- and xenobiotic-mediated regulation of the *CYP3A4* gene. *Drug Metab. Dispos.* **32**: 525-535.

Bonzo, J.A., Belanger, A. and Tukey, R.H. (2007) The role of chrysin and the ah receptor in induction of the human *UGT1A1* gene *in vitro* and in transgenic *UGT1* mice. *Hepatology* **45**: 349-360.

Bort, R., Gomez-Lechon, M.J., Castell, J.V. and Jover, R. (2004) Role of hepatocyte nuclear factor 3gamma in the expression of human CYP2C genes. *Arch. Biochem. Biophys.* **426**: 63-72.

Bosma, P.J. (2003) Inherited disorders of bilirubin metabolism. J. Hepatol. 38: 107-117.

Bosma, P.J., Chowdhury, J.R., Bakker, C., Gantla, S., de Boer, A., Oostra, B.A., Lindhout, D., Tytgat, G.N., Jansen, P.L. and Oude Elferink, R.P. (1995) The genetic basis of the reduced expression of bilirubin UDP-glucuronosyltransferase 1 in Gilbert's syndrome. *N. Engl. J. Med.* **333**: 1171-1175.

Bosma, P.J., Seppen, J., Goldhoorn, B., Bakker, C., Oude Elferink, R.P., Chowdhury, J.R., Chowdhury, N.R. and Jansen, P.L. (1994) Bilirubin UDP-glucuronosyltransferase 1 is the only relevant bilirubin glucuronidating isoform in man. *J. Biol. Chem.* **269**: 17960-17964.

Bouwmeester, N.J., Anderson, B.J., Tibboel, D. and Holford, N.H. (2004) Developmental pharmacokinetics of morphine and its metabolites in neonates, infants and young children. *Br. J. Anaesth.* **92**: 208-217.

Breyer-Pfaff, U., Mey, U., Green, M.D. and Tephly, T.R. (2000) Comparative *N*-glucuronidation kinetics of ketotifen and amitriptyline by expressed human UDP-glucuronosyltransferases and liver microsomes. *Drug Metab. Dispos.* **28**: 869-872.

Briancon, N., Bailly, A., Clotman, F., Jacquemin, P., Lemaigre, F.P. and Weiss, M.C. (2004) Expression of the alpha7 isoform of hepatocyte nuclear factor (HNF) 4 is activated by HNF6/OC-2 and HNF1 and repressed by HNF4alpha1 in the liver. *J. Biol. Chem.* **279**: 33398-33408.

Brooks, A.R., Nagy, B.P., Taylor, S., Simonet, W.S., Taylor, J.M. and Levy-Wilson, B. (1994) Sequences containing the second-intron enhancer are essential for transcription of the human apolipoprotein B gene in the livers of transgenic mice. *Mol. Cell. Biol.* **14**: 2243-2256.

Buck, M., Turler, H. and Chojkier, M. (1994) LAP (NF-IL-6), a tissue-specific transcriptional activator, is an inhibitor of hepatoma cell proliferation. *EMBO J.* **13**: 851-860.

Buckland, P.R. (2006) The importance and identification of regulatory polymorphisms and their mechanisms of action. *Biochim. Biophys. Acta* **1762**: 17-28.

Buckley, D.B. and Klaassen, C.D. (2007) Tissue- and gender-specific mRNA expression of UDP-glucuronosyltransferases (UGTs) in mice. *Drug Metab. Dispos.* **35**: 121-127.

Burger, D.M., Meenhorst, P.L., Koks, C.H. and Beijnen, J.H. (1993) Pharmacokinetic interaction between rifampin and zidovudine. *Antimicrob. Agents Chemother.* **37**: 1426-1431.

Bustin, S.A. (2000) Absolute quantification of mRNA using real-time reverse transcription polymerase chain reaction assays. *J. Mol. Endocrinol.* **25**: 169-193.

Bustin, S.A. (2002) Quantification of mRNA using real-time reverse transcription PCR (RT-PCR): trends and problems. *J. Mol. Endocrinol.* **29**: 23-39.

Butler, L.M., Duguay, Y., Millikan, R.C., Sinha, R., Gagne, J.F., Sandler, R.S. and Guillemette, C. (2005) Joint effects between UDP-glucuronosyltransferase 1A7 genotype and dietary carcinogen exposure on risk of colon cancer. *Cancer Epidemiol. Biomarkers Prev.* 14: 1626-1632.

Caillier, B., Lepine, J., Tojcic, J., Menard, V., Perusse, L., Belanger, A., Barbier, O. and Guillemette, C. (2007) A pharmacogenomics study of the human estrogen glucuronosyltransferase UGT1A3. *Pharmacogenet. Genomics* **17**: 481-495.

Campos, S.P. and Baumann, H. (1992) Insulin is a prominent modulator of the cytokine-stimulated expression of acute-phase plasma protein genes. *Mol. Cell. Biol.* **12**: 1789-1797.

Cao, Z., Umek, R.M. and McKnight, S.L. (1991) Regulated expression of three C/EBP isoforms during adipose conversion of 3T3-L1 cells. *Genes Dev.* **5**: 1538-1552.

Caraco, Y., Sheller, J. and Wood, A.J. (1997) Pharmacogenetic determinants of codeine induction by rifampin: the impact on codeine's respiratory, psychomotor and miotic effects. *J. Pharmacol. Exp. Ther.* **281**: 330-336.

Carlini, L.E., Meropol, N.J., Bever, J., Andria, M.L., Hill, T., Gold, P., Rogatko, A., Wang, H. and Blanchard, R.L. (2005) UGT1A7 and UGT1A9 polymorphisms predict response and toxicity in colorectal cancer patients treated with capecitabine/irinotecan. *Clin. Cancer Res.* **11**: 1226-1236.

Carter, M.E., Gulick, T., Raisher, B.D., Caira, T., Ladias, J.A., Moore, D.D. and Kelly, D.P. (1993) Hepatocyte nuclear factor-4 activates medium chain acyl-CoA dehydrogenase gene transcription by interacting with a complex regulatory element. *J. Biol. Chem.* **268**: 13805-13810.

Cartharius, K., Frech, K., Grote, K., Klocke, B., Haltmeier, M., Klingenhoff, A., Frisch, M., Bayerlein, M. and Werner, T. (2005) MatInspector and beyond: promoter analysis based on transcription factor binding sites. *Bioinformatics* **21**: 2933-2942.

Caslini, C., Capo-chichi, C.D., Roland, I.H., Nicolas, E., Yeung, A.T. and Xu, X.X. (2006) Histone modifications silence the GATA transcription factor genes in ovarian cancer. *Oncogene* **25**: 5446-5461.

Catlow, K., Ashurst, H.L., Varro, A. and Dimaline, R. (2007) Identification of a gastrin response element in the vesicular monoamine transporter type 2 promoter and requirement of 20S proteasome subunits for transcriptional activity. *J. Biol. Chem.* **282**: 17069-17077.

Cereghini, S., Ott, M.O., Power, S. and Maury, M. (1992) Expression patterns of vHNF1 and HNF1 homeoproteins in early postimplantation embryos suggest distinct and sequential developmental roles. *Development* **116**: 783-797.

Cervoni, N. and Szyf, M. (2001) Demethylase activity is directed by histone acetylation. J. Biol. Chem. 276: 40778-40787.

Cha, J.Y., Kim, H., Kim, K.S., Hur, M.W. and Ahn, Y. (2000) Identification of transacting factors responsible for the tissue-specific expression of human glucose transporter type 2 isoform gene. Cooperative role of hepatocyte nuclear factors 1alpha and 3beta. *J. Biol. Chem.* **275**: 18358-18365.

Chan, A.T., Tranah, G.J., Giovannucci, E.L., Hunter, D.J. and Fuchs, C.S. (2005) Genetic variants in the UGT1A6 enzyme, aspirin use, and the risk of colorectal adenoma. *J. Natl. Cancer Inst.* **97**: 457-460.

Chang, C.J., Shen, B.J. and Lee, S.C. (1995) Autoregulated induction of the acutephase response transcription factor gene, *agp/ebp. DNA Cell Biol.* **14**: 529-537.

Chartier, F.L., Bossu, J.P., Laudet, V., Fruchart, J.C. and Laine, B. (1994) Cloning and sequencing of cDNAs encoding the human hepatocyte nuclear factor 4 indicate the presence of two isoforms in human liver. *Gene* **147**: 269-272.

Chaudhary, I.P., Tuntaterdtum, S., McNamara, P.J., Robertson, L.W. and Blouin, R.A. (1993) Effect of genetic obesity and phenobarbital treatment on the hepatic conjugation pathways. *J. Pharmacol. Exp. Ther.* **265**: 1333-1338.

Chaya, D., Hayamizu, T., Bustin, M. and Zaret, K.S. (2001) Transcription factor FoxA (HNF3) on a nucleosome at an enhancer complex in liver chromatin. *J. Biol. Chem.* **276**: 44385-44389.

Chelly, J., Concordet, J.P., Kaplan, J.C. and Kahn, A. (1989) Illegitimate transcription: transcription of any gene in any cell type. *Proc. Natl. Acad. Sci. U. S. A.* **86**: 2617-2621.

Chelly, J., Hugnot, J.P., Concordet, J.P., Kaplan, J.C. and Kahn, A. (1991) Illegitimate (or ectopic) transcription proceeds through the usual promoters. *Biochem. Biophys. Res. Commun.* **178**: 553-557.

Chen, J., Cooper, A.D. and Levy-Wilson, B. (1999) Hepatocyte nuclear factor 1 binds to and transactivates the human but not the rat CYP7A1 promoter. *Biochem. Biophys. Res. Commun.* **260**: 829-834.

Chen, K., Jin, M., Zhu, Y., Jiang, Q., Yu, W., Ma, X. and Yao, K. (2006a) Genetic polymorphisms of the uridine diphosphate glucuronosyltransferase 1A7 and colorectal cancer risk in relation to cigarette smoking and alcohol drinking in a Chinese population. *J. Gastroenterol. Hepatol.* **21**: 1036-1041.

Chen, S., Beaton, D., Nguyen, N., Senekeo-Effenberger, K., Brace-Sinnokrak, E., Argikar, U., Remmel, R.P., Trottier, J., Barbier, O., Ritter, J.K. and Tukey, R.H. (2005a) Tissue-specific, inducible, and hormonal control of the human UDP-glucuronosyltransferase-1 (*UGT1*) locus. *J. Biol. Chem.* **280**: 37547-37557.

Chen, W.S., Manova, K., Weinstein, D.C., Duncan, S.A., Plump, A.S., Prezioso, V.R., Bachvarova, R.F. and Darnell, J.E. (1994) Disruption of the HNF-4 gene, expressed in visceral endoderm, leads to cell death in embryonic ectoderm and impaired gastrulation of mouse embryos. *Genes Dev.* **8**: 2466-2477.

Chen, Y., Chen, S., Li, X., Wang, X. and Zeng, S. (2006b) Genetic variants of human UGT1A3: functional characterization and frequency distribution in a Chinese Han population. *Drug Metab. Dispos.* **34**: 1462-1467.

Chen, Y., Ferguson, S.S., Negishi, M. and Goldstein, J.A. (2004) Induction of human *CYP2C9* by rifampicin, hyperforin, and phenobarbital is mediated by the pregnane X receptor. *J. Pharmacol. Exp. Ther.* **308**: 495-501.

Chen, Y., Kissling, G., Negishi, M. and Goldstein, J.A. (2005b) The nuclear receptors constitutive androstane receptor and pregnane X receptor cross-talk with hepatic nuclear factor 4alpha to synergistically activate the human *CYP2C9* promoter. *J. Pharmacol. Exp. Ther.* **314**: 1125-1133.

Cheng, Z., Radominska-Pandya, A. and Tephly, T.R. (1998a) Cloning and expression of human UDP-glucuronosyltransferase (UGT) 1A8. *Arch. Biochem. Biophys.* **356**: 301-305.

Cheng, Z., Rios, G.R., King, C.D., Coffman, B.L., Green, M.D., Mojarrabi, B., Mackenzie, P.I. and Tephly, T.R. (1998b) Glucuronidation of catechol estrogens by expressed human UDP-glucuronosyltransferases (UGTs) 1A1, 1A3, and 2B7. *Toxicol. Sci.* **45**: 52-57.

Chiba, T., Yokosuka, O., Arai, M., Tada, M., Fukai, K., Imazeki, F., Kato, M., Seki, N. and Saisho, H. (2004) Identification of genes up-regulated by histone deacetylase inhibition with cDNA microarray and exploration of epigenetic alterations on hepatoma cells. *J. Hepatol.* **41**: 436-445.

Chiu, K.C., Chuang, L.M., Chu, A. and Wang, M. (2003) Transcription factor 1 and beta-cell function in glucose-tolerant subjects. *Diabet. Med.* **20**: 225-230.

Chiu, K.C., Chuang, L.M., Ryu, J.M., Tsai, G.P. and Saad, M.F. (2000) The I27L amino acid polymorphism of hepatic nuclear factor-1alpha is associated with insulin resistance. *J. Clin. Endocrinol. Metab.* **85**: 2178-2183.

Chou, H.C., Lang, N.P. and Kadlubar, F.F. (1995) Metabolic activation of *N*-hydroxy arylamines and *N*-hydroxy heterocyclic amines by human sulfotransferase(s). *Cancer Res.* **55**: 525-529.

Chouinard, S., Tessier, M., Vernouillet, G., Gauthier, S., Labrie, F., Barbier, O. and Belanger, A. (2006) Inactivation of the pure antiestrogen fulvestrant and other synthetic estrogen molecules by UDP-glucuronosyltransferase 1A enzymes expressed in breast tissue. *Mol. Pharmacol.* **69**: 908-920.

Chowdhury, S., Gotoh, T., Mori, M. and Takiguchi, M. (1996) CCAAT/enhancerbinding protein beta (C/EBPbeta) binds and activates while hepatocyte nuclear factor-4 (HNF-4) does not bind but represses the liver-type arginase promoter. *Eur. J. Biochem.* **236**: 500-509.

Chung, I. and Bresnick, E. (1997) Identification of positive and negative regulatory elements of the human cytochrome P4501A2 (*CYP1A2*) gene. *Arch. Biochem. Biophys.* **338**: 220-226.

Chung, J.Y., Cho, J.Y., Yu, K.S., Kim, J.R., Jung, H.R., Lim, K.S., Jang, I.J. and Shin, S.G. (2005) Effect of the *UGT2B15* genotype on the pharmacokinetics, pharmacodynamics, and drug interactions of intravenous lorazepam in healthy volunteers. *Clin. Pharmacol. Ther*.77: 486-494.

Cirillo, L.A., Lin, F.R., Cuesta, I., Friedman, D., Jarnik, M. and Zaret, K.S. (2002) Opening of compacted chromatin by early developmental transcription factors HNF3 (FoxA) and GATA-4. *Mol. Cell* **9**: 279-289.

Cirillo, L.A. and Zaret, K.S. (1999) An early developmental transcription factor complex that is more stable on nucleosome core particles than on free DNA. *Mol. Cell* **4**: 961-969.

Claeyssens, S., Gangneux, C., Brasse-Lagnel, C., Ruminy, P., Aki, T., Lavoinne, A. and Salier, J.P. (2003) Amino acid control of the human glyceraldehyde 3-phosphate dehydrogenase gene transcription in hepatocyte. *Am. J. Physiol. Gastrointest. Liver Physiol.* **285**: G840-849.

Clark, H.F., Gurney, A.L., Abaya, E., Baker, K., Baldwin, D., Brush, J., Chen, J., Chow, B., Chui, C., Crowley, C., Currell, B., Deuel, B., Dowd, P., Eaton, D., Foster, J., Grimaldi, C., Gu, Q., Hass, P.E., Heldens, S., Huang, A., Kim, H.S., Klimowski, L., Jin, Y., Johnson, S., Lee, J., Lewis, L., Liao, D., Mark, M., Robbie, E., Sanchez, C., Schoenfeld, J., Seshagiri, S., Simmons, L., Singh, J., Smith, V., Stinson, J., Vagts, A., Vandlen, R., Watanabe, C., Wieand, D., Woods, K., Xie, M.H., Yansura, D., Yi, S., Yu, G., Yuan, J., Zhang, M., Zhang, Z., Goddard, A., Wood, W.I., Godowski, P. and Gray, A. (2003) The secreted protein discovery initiative (SPDI), a large-scale effort to identify novel human secreted and transmembrane proteins: a bioinformatics assessment. *Genome Res.* **13**: 2265-2270.

Coffman, B.L., King, C.D., Rios, G.R. and Tephly, T.R. (1998) The glucuronidation of opioids, other xenobiotics, and androgens by human UGT2B7Y(268) and UGT2B7H(268). *Drug Metab. Dispos.* **26**: 73-77.

Collier, A.C., Ganley, N.A., Tingle, M.D., Blumenstein, M., Marvin, K.W., Paxton, J.W., Mitchell, M.D. and Keelan, J.A. (2002a) UDP-glucuronosyltransferase activity, expression and cellular localization in human placenta at term. *Biochem. Pharmacol.* **63**: 409-419.

Collier, A.C., Keelan, J.A., Van Zijl, P.E., Paxton, J.W., Mitchell, M.D. and Tingle, M.D. (2004) Human placental glucuronidation and transport of 3'azido-3'deoxythymidine and uridine diphosphate glucuronic acid. *Drug Metab. Dispos.* **32**: 813-820.

Collier, A.C., Tingle, M.D., Paxton, J.W., Mitchell, M.D. and Keelan, J.A. (2002b) Metabolizing enzyme localization and activities in the first trimester human placenta: the effect of maternal and gestational age, smoking and alcohol consumption. *Hum. Reprod.* **17**: 2564-2572.

Combates, N.J., Rzepka, R.W., Chen, Y.N. and Cohen, D. (1994) NF-IL6, a member of the C/EBP family of transcription factors, binds and *trans*-activates the human MDR1 gene promoter. *J. Biol. Chem.* **269**: 29715-29719.

Congiu, M., Mashford, M.L., Slavin, J.L. and Desmond, P.V. (2002) UDP glucuronosyltransferase mRNA levels in human liver disease. *Drug Metab. Dispos.* **30**: 129-134.

Cooper, S.J., Trinklein, N.D., Anton, E.D., Nguyen, L. and Myers, R.M. (2006) Comprehensive analysis of transcriptional promoter structure and function in 1% of the human genome. *Genome Res.* **16**: 1-10.

Costa, R.H., Kalinichenko, V.V., Holterman, A.X. and Wang, X. (2003) Transcription factors in liver development, differentiation, and regeneration. *Hepatology* **38**: 1331-1347.

Coughtrie, M.W., Burchell, B., Leakey, J.E. and Hume, R. (1988) The inadequacy of perinatal glucuronidation: immunoblot analysis of the developmental expression of individual UDP-glucuronosyltransferase isoenzymes in rat and human liver microsomes. *Mol. Pharmacol.* **34**: 729-735.

Court, M.H., Duan, S.X., von Moltke, L.L., Greenblatt, D.J., Patten, C.J., Miners, J.O. and Mackenzie, P.I. (2001) Interindividual variability in acetaminophen glucuronidation by human liver microsomes: identification of relevant acetaminophen UDP-glucuronosyltransferase isoforms. *J. Pharmacol. Exp. Ther.* **299**: 998-1006.

Court, M.H., Hao, Q., Krishnaswamy, S., Bekaii-Saab, T., Al-Rohaimi, A., von Moltke, L.L. and Greenblatt, D.J. (2004) UDP-glucuronosyltransferase (UGT) 2B15 pharmacogenetics: UGT2B15 D85Y genotype and gender are major determinants of oxazepam glucuronidation by human liver. *J. Pharmacol. Exp. Ther.* **310**: 656-665.

Crowe, A.J., Sang, L., Li, K.K., Lee, K.C., Spear, B.T. and Barton, M.C. (1999) Hepatocyte nuclear factor 3 relieves chromatin-mediated repression of the alpha-fetoprotein gene. *J. Biol. Chem.* **274**: 25113-25120.

Cunningham, J.M., Hebbring, S.J., McDonnell, S.K., Cicek, M.S., Christensen, G.B., Wang, L., Jacobsen, S.J., Cerhan, J.R., Blute, M.L., Schaid, D.J. and Thibodeau, S.N. (2007) Evaluation of genetic variations in the androgen and estrogen metabolic pathways as risk factors for sporadic and familial prostate cancer. *Cancer Epidemiol. Biomarkers Prev.* **16**: 969-978.

Dallas, P.B., Yaciuk, P. and Moran, E. (1997) Characterization of monoclonal antibodies raised against p300: both p300 and CBP are present in intracellular TBP complexes. *J. Virol.* **71**: 1726-1731.

Darbari, D.S., van Schaik, R.H., Capparelli, E.V., Rana, S., McCarter, R. and van den Anker, J. (2007) UGT2B7 promoter variant -840G>A contributes to the variability in hepatic clearance of morphine in patients with sickle cell disease. *Am. J. Hematol.* In press.

de la Chapelle, A. (2004) Genetic predisposition to colorectal cancer. *Nat. Rev. Cancer* **4**: 769-780.

de Laat, W. and Grosveld, F. (2003) Spatial organization of gene expression: the active chromatin hub. *Chromosome Res.* **11**: 447-459.

de Lucas, S., Lopez-Alcorocho, J.M., Bartolome, J. and Carreno, V. (2004) Nitric oxide and TGF-beta1 inhibit HNF-4alpha function in HEPG2 cells. *Biochem. Biophys. Res. Commun.* **321**: 688-694.

De Simone, V., De Magistris, L., Lazzaro, D., Gerstner, J., Monaci, P., Nicosia, A. and Cortese, R. (1991) LFB3, a heterodimer-forming homeoprotein of the LFB1 family, is expressed in specialized epithelia. *EMBO J.* **10**: 1435-1443.

de Wildt, S.N., Kearns, G.L., Leeder, J.S. and van den Anker, J.N. (1999) Glucuronidation in humans. Pharmacogenetic and developmental aspects. *Clin. Pharmacokinet.* **36**: 439-452.

Dekker, J. (2006) The three 'C' s of chromosome conformation capture: controls, controls. *Nat. Methods* **3**: 17-21.

Delesque-Touchard, N., Park, S.H. and Waxman, D.J. (2000) Synergistic action of hepatocyte nuclear factors 3 and 6 on *CYP2C12* gene expression and suppression by growth hormone-activated STAT5b. Proposed model for female specific expression of *CYP2C12* in adult rat liver. *J. Biol. Chem.* **275**: 34173-34182.

Dellinger, R.W., Fang, J.L., Chen, G., Weinberg, R. and Lazarus, P. (2006) Importance of UDP-glucuronosyltransferase 1A10 (UGT1A10) in the detoxification of polycyclic aromatic hydrocarbons: decreased glucuronidative activity of the UGT1A10139Lys isoform. *Drug Metab. Dispos.* **34**: 943-949.

Desai, A.A., Innocenti, F. and Ratain, M.J. (2003) UGT pharmacogenomics: implications for cancer risk and cancer therapeutics. *Pharmacogenetics* **13**: 517-523.

Descombes, P. and Schibler, U. (1991) A liver-enriched transcriptional activator protein, LAP, and a transcriptional inhibitory protein, LIP, are translated from the same mRNA. *Cell* **67**: 569-579.

Dhe-Paganon, S., Duda, K., Iwamoto, M., Chi, Y.I. and Shoelson, S.E. (2002) Crystal structure of the HNF4alpha ligand binding domain in complex with endogenous fatty acid ligand. *J. Biol. Chem.* **277**: 37973-37976.

Ding, X., Lichti, K., Kim, I., Gonzalez, F.J. and Staudinger, J.L. (2006) Regulation of constitutive androstane receptor and its target genes by fasting, cAMP, hepatocyte nuclear factor alpha, and the coactivator peroxisome proliferator-activated receptor gamma coactivator-lalpha. *J. Biol. Chem.* **281**: 26540-26551.

Di-Poi, N., Desvergne, B., Michalik, L. and Wahli, W. (2005) Transcriptional repression of peroxisome proliferator-activated receptor beta/delta in murine keratinocytes by CCAAT/enhancer-binding proteins. *J. Biol. Chem.* **280**: 38700-38710.

Divine, J.K., McCaul, S.P. and Simon, T.C. (2003) HNF-1alpha and endodermal transcription factors cooperatively activate Fabpl: MODY3 mutations abrogate cooperativity. *Am. J. Physiol. Gastrointest. Liver Physiol.* **285**: G62-72.

Dohda, T., Kaneoka, H., Inayoshi, Y., Kamihira, M., Miyake, K. and Iijima, S. (2004) Transcriptional coactivators CBP and p300 cooperatively enhance HNF-1alpha-mediated expression of the albumin gene in hepatocytes. *J. Biochem.* **136**: 313-319.

Donato, M.F., Arosio, E., Del Ninno, E., Ronchi, G., Lampertico, P., Morabito, A., Balestrieri, M.R. and Colombo, M. (2001) High rates of hepatocellular carcinoma in cirrhotic patients with high liver cell proliferative activity. *Hepatology* **34**: 523-528.

Drewes, T., Senkel, S., Holewa, B. and Ryffel, G.U. (1996) Human hepatocyte nuclear factor 4 isoforms are encoded by distinct and differentially expressed genes. *Mol. Cell. Biol.* **16**: 925-931.

Dring, M.M., Goulding, C.A., Trimble, V.I., Keegan, D., Ryan, A.W., Brophy, K.M., Smyth, C.M., Keeling, P.W., O'Donoghue, D., O'Sullivan, M., O'Morain, C., Mahmud, N., Wikstrom, A.C., Kelleher, D. and McManus, R. (2006) The pregnane X receptor locus is associated with susceptibility to inflammatory bowel disease. *Gastroenterology* **130**: 341-348.

Duda, K., Chi, Y.I. and Shoelson, S.E. (2004) Structural basis for HNF-4alpha activation by ligand and coactivator binding. *J. Biol. Chem.* **279**: 23311-23316.

Duguay, Y., Baar, C., Skorpen, F. and Guillemette, C. (2004a) A novel functional polymorphism in the uridine diphosphate-glucuronosyltransferase 2B7 promoter with significant impact on promoter activity. *Clin. Pharmacol. Ther.***75**: 223-233.

Duguay, Y., McGrath, M., Lepine, J., Gagne, J.F., Hankinson, S.E., Colditz, G.A., Hunter, D.J., Plante, M., Tetu, B., Belanger, A., Guillemette, C. and De Vivo, I. (2004b) The functional *UGT1A1* promoter polymorphism decreases endometrial cancer risk. *Cancer Res.* **64**: 1202-1207.

Duncan, S.A., Manova, K., Chen, W.S., Hoodless, P., Weinstein, D.C., Bachvarova, R.F. and Darnell, J.E. (1994) Expression of transcription factor HNF-4 in the extraembryonic endoderm, gut, and nephrogenic tissue of the developing mouse embryo: HNF-4 is a marker for primary endoderm in the implanting blastocyst. *Proc. Natl. Acad. Sci. U. S. A.* **91**: 7598-7602.

Duong, D.T., Waltner-Law, M.E., Sears, R., Sealy, L. and Granner, D.K. (2002) Insulin inhibits hepatocellular glucose production by utilizing liver-enriched transcriptional inhibitory protein to disrupt the association of CREB-binding protein and RNA polymerase II with the phosphoenolpyruvate carboxykinase gene promoter. J. Biol. Chem. **277**: 32234-32242.

Duvic, M., Talpur, R., Ni, X., Zhang, C., Hazarika, P., Kelly, C., Chiao, J.H., Reilly, J.F., Ricker, J.L., Richon, V.M. and Frankel, S.R. (2007) Phase 2 trial of oral vorinostat (suberoylanilide hydroxamic acid, SAHA) for refractory cutaneous T-cell lymphoma (CTCL). *Blood* **109**: 31-39.

Ebert, U., Thong, N.Q., Oertel, R. and Kirch, W. (2000) Effects of rifampicin and cimetidine on pharmacokinetics and pharmacodynamics of lamotrigine in healthy subjects. *Eur. J. Clin. Pharmacol.* **56**: 299-304.

Ebner, T. and Burchell, B. (1993) Substrate specificities of two stably expressed human liver UDP-glucuronosyltransferases of the UGT1 gene family. *Drug Metab. Dispos.* **21**: 50-55.

Eeckhoute, J., Formstecher, P. and Laine, B. (2004) Hepatocyte nuclear factor 4alpha enhances the hepatocyte nuclear factor 1alpha-mediated activation of transcription. *Nucleic Acids Res.* **32**: 2586-2593.

Ehmer, U., Vogel, A., Schutte, J.K., Krone, B., Manns, M.P. and Strassburg, C.P. (2004) Variation of hepatic glucuronidation: novel functional polymorphisms of the UDP-glucuronosyltransferase UGT1A4. *Hepatology* **39**: 970-977.

Ek, J., Hansen, S.P., Lajer, M., Nicot, C., Boesgaard, T.W., Pruhova, S., Johansen, A., Albrechtsen, A., Yderstraede, K., Lauenborg, J., Parrizas, M., Boj, S.F., Jorgensen, T., Borch-Johnsen, K., Damm, P., Ferrer, J., Lebl, J., Pedersen, O. and Hansen, T. (2006) A novel -192c/g mutation in the proximal P2 promoter of the hepatocyte nuclear factor-4 alpha gene (*HNF4A*) associates with late-onset diabetes. *Diabetes* **55**: 1869-1873.

Elahi, A., Bendaly, J., Zheng, Z., Muscat, J.E., Richie, J.P., Schantz, S.P. and Lazarus, P. (2003) Detection of *UGT1A10* polymorphisms and their association with orolaryngeal carcinoma risk. *Cancer* **98**: 872-880.

Ellis, E., Wagner, M., Lammert, F., Nemeth, A., Gumhold, J., Strassburg, C.P., Kylander, C., Katsika, D., Trauner, M., Einarsson, C. and Marschall, H.U. (2006) Successful treatment of severe unconjugated hyperbilirubinemia via induction of UGT1A1 by rifampicin. *J. Hepatol.* **44**: 243-245.

Ellis, J., Tan-Un, K.C., Harper, A., Michalovich, D., Yannoutsos, N., Philipsen, S. and Grosveld, F. (1996) A dominant chromatin-opening activity in 5' hypersensitive site 3 of the human beta-globin locus control region. *EMBO J.* **15**: 562-568.

El-Omar, E.M., Rabkin, C.S., Gammon, M.D., Vaughan, T.L., Risch, H.A., Schoenberg, J.B., Stanford, J.L., Mayne, S.T., Goedert, J., Blot, W.J., Fraumeni, J.F. and Chow, W.H. (2003) Increased risk of noncardia gastric cancer associated with proinflammatory cytokine gene polymorphisms. *Gastroenterology* **124**: 1193-1201.

Evans, W.E. and McLeod, H.L. (2003) Pharmacogenomics - drug disposition, drug targets, and side effects. *N. Engl. J. Med.* **348**: 538-549.

Evans, W.E. and Relling, M.V. (1999) Pharmacogenomics: translating functional genomics into rational therapeutics. *Science* **286**: 487-491.

Falasca, L., Marcellini, P., Ara, C., Rufo, A. and Devirgiliis, L.C. (1999) Growth inhibition and induction of specific hepatic phenotype expression by retinoic acid in HEPG2 cells. *Anticancer Res.* **19**: 3283-3292.

Fang, J.L., Beland, F.A., Doerge, D.R., Wiener, D., Guillemette, C., Marques, M.M. and Lazarus, P. (2002) Characterization of benzo(*a*)pyrene-*trans*-7,8-dihydrodiol glucuronidation by human tissue microsomes and overexpressed UDP-glucuronosyltransferase enzymes. *Cancer Res.* **62**: 1978-1986.

Ferrini, J.B., Rodrigues, E., Dulic, V., Pichard-Garcia, L., Fabr, J.M., Blanc, P. and Maurel, P. (2001) Expression and DNA-binding activity of C/EBPalpha and C/EBPbeta in human liver and differentiated primary hepatocytes. *J. Hepatol.* **35**: 170-177.

Findlay, K.A., Kaptein, E., Visser, T.J. and Burchell, B. (2000) Characterization of the uridine diphosphate-glucuronosyltransferase-catalyzing thyroid hormone glucuronidation in man. *J. Clin. Endocrinol. Metab.* **85**: 2879-2883.

Finel, M., Li, X., Gardner-Stephen, D., Bratton, S., Mackenzie, P.I. and Radominska-Pandya, A. (2005) Human UDP-glucuronosyltransferase 1A5: identification, expression, and activity. *J. Pharmacol. Exp. Ther.* **315**: 1143-1149.

Food and Drug Administration (2005) Camptosar final label [on-line]. Available from: http:// www.fda.gov/medwatch/SAFETY/2005/Jun_PI/Camptosar_PI.pdf. [Accessed 10 September 2007].

Frain, M., Hardon, E., Ciliberto, G. and Sala-Trepat, J.M. (1990) Binding of a liver-specific factor to the human albumin gene promoter and enhancer. *Mol. Cell. Biol.* **10**: 991-999.

Fraser, J.D., Martinez, V., Straney, R. and Briggs, M.R. (1998) DNA binding and transcription activation specificity of hepatocyte nuclear factor 4. *Nucleic Acids Res.* **26**: 2702-2707.

Friedman, J.R. and Kaestner, K.H. (2006) The Foxa family of transcription factors in development and metabolism. *Cell Mol. Life Sci.* **63**: 2317-2328.

Fujiwara, R., Nakajima, M., Yamanaka, H., Katoh, M. and Yokoi, T. (2007a) Interactions between human UGT1A1, UGT1A4, and UGT1A6 affect their enzymatic activities. *Drug Metab. Dispos.* **35**: 1781-1787.

Fujiwara, R., Nakajima, M., Yamanaka, H., Nakamura, A., Katoh, M., Ikushiro, S., Sakaki, T. and Yokoi, T. (2007b) Effects of coexpression of UGT1A9 on enzymatic activities of human UGT1A isoforms. *Drug Metab. Dispos.* **35**: 747-757.

Furchert, S.E., Lanvers-Kaminsky, C., Juurgens, H., Jung, M., Loidl, A. and Fruhwald, M.C. (2007) Inhibitors of histone deacetylases as potential therapeutic tools for high-risk embryonal tumors of the nervous system of childhood. *Int. J. Cancer* **120**: 1787-1794.

Furuta, H., Iwasaki, N., Oda, N., Hinokio, Y., Horikawa, Y., Yamagata, K., Yano, N., Sugahiro, J., Ogata, M., Ohgawara, H., Omori, Y., Iwamoto, Y. and Bell, G.I. (1997) Organization and partial sequence of the hepatocyte nuclear factor-4

alpha/MODY1 gene and identification of a missense mutation, R127W, in a Japanese family with MODY. *Diabetes* **46**: 1652-1657.

Gagne, J.F., Montminy, V., Belanger, P., Journault, K., Gaucher, G. and Guillemette, C. (2002) Common human *UGT1A* polymorphisms and the altered metabolism of irinotecan active metabolite 7-ethyl-10-hydroxycamptothecin (SN-38). *Mol. Pharmacol.* **62**: 608-617.

Gagnon, J.F., Bernard, O., Villeneuve, L., Tetu, B. and Guillemette, C. (2006) Irinotecan inactivation is modulated by epigenetic silencing of *UGT1A1* in colon cancer. *Clin. Cancer Res.* **12**: 1850-1858.

Galijatovic, A., Walle, U.K. and Walle, T. (2000) Induction of UDPglucuronosyltransferase by the flavonoids chrysin and quercetin in Caco-2 cells. *Pharm. Res.* **17**: 21-26.

Gall, W.E., Zawada, G., Mojarrabi, B., Tephly, T.R., Green, M.D., Coffman, B.L., Mackenzie, P.I. and Radominska-Pandya, A. (1999) Differential glucuronidation of bile acids, androgens and estrogens by human UGT1A3 and 2B7. *J. Steroid Biochem. Mol. Biol.* **70**: 101-108.

Gallagher, C.J., Muscat, J.E., Hicks, A.N., Zheng, Y., Dyer, A.M., Chase, G.A., Richie, J. and Lazarus, P. (2007) The UDP-glucuronosyltransferase 2B17 gene deletion polymorphism: sex-specific association with urinary 4-(methylnitrosamino)-1-(3-pyridyl)-1-butanol glucuronidation phenotype and risk for lung cancer. *Cancer Epidemiol. Biomarkers Prev.* **16**: 823-828.

Gallicano, K.D., Sahai, J., Shukla, V.K., Seguin, I., Pakuts, A., Kwok, D., Foster, B.C. and Cameron, D.W. (1999) Induction of zidovudine glucuronidation and amination pathways by rifampicin in HIV-infected patients. *Br. J. Clin. Pharmacol.* **48**: 168-179.

Galson, D.L., Tsuchiya, T., Tendler, D.S., Huang, L.E., Ren, Y., Ogura, T. and Bunn, H.F. (1995) The orphan receptor hepatic nuclear factor 4 functions as a transcriptional activator for tissue-specific and hypoxia-specific erythropoietin gene expression and is antagonized by EAR3/COUP-TF1. *Mol. Cell. Biol.* **15**: 2135-2144.

Gao, N., Zhang, J., Rao, M.A., Case, T.C., Mirosevich, J., Wang, Y., Jin, R., Gupta, A., Rennie, P.S. and Matusik, R.J. (2003) The role of hepatocyte nuclear factor-3alpha (Forkhead Box A1) and androgen receptor in transcriptional regulation of prostatic genes. *Mol. Endocrinol.* **17**: 1484-1507.

Garcia-Carbonero, R. and Supko, J.G. (2002) Current perspectives on the clinical experience, pharmacology, and continued development of the camptothecins. *Clin. Cancer Res.* **8**: 641-661.

Gardner-Stephen, D., Heydel, J.M., Goyal, A., Lu, Y., Xie, W., Lindblom, T., Mackenzie, P. and Radominska-Pandya, A. (2004) Human PXR variants and their differential effects on the regulation of human UDP-glucuronosyltransferase gene expression. *Drug Metab. Dispos.* **32**: 340-347.

Gardner-Stephen, D.A. and Mackenzie, P.I. (2005) Identification and characterization of functional Hepatocyte Nuclear Factor 1 binding sites in UDP-glucuronosyltransferase genes. *Methods Enzymol.* **400**: 22-46.

Gardner-Stephen, D.A. and Mackenzie, P.I. (2007a) Hepatocyte nuclear factor1 transcription factors are essential for the UDP-glucuronosyltransferase 1A9 promoter response to hepatocyte nuclear factor 4alpha. *Pharmacogenet. Genomics* **17**: 25-36.

Gardner-Stephen, D.A. and Mackenzie, P.I. (2007b) Isolation of the UDP-glucuronosyltransferase 1A3 and 1A4 proximal promoters and characterization of their dependence on the transcription factor hepatocyte nuclear factor 1alpha. *Drug Metab. Dispos.* **35**: 116-120.

Gautier-Stein, A., Zitoun, C., Lalli, E., Mithieux, G. and Rajas, F. (2006) Transcriptional regulation of the glucose-6-phosphatase gene by cAMP/vasoactive intestinal peptide in the intestine. Role of HNF4alpha, CREM, HNF1alpha, and C/EBPalpha. *J. Biol. Chem.* **281**: 31268-31278.

Geier, A., Dietrich, C.G., Voigt, S., Kim, S.K., Gerloff, T., Kullak-Ublick, G.A., Lorenzen, J., Matern, S. and Gartung, C. (2003) Effects of proinflammatory cytokines on rat organic anion transporters during toxic liver injury and cholestasis. *Hepatology* **38**: 345-354.

Ghosal, A., Hapangama, N., Yuan, Y., Achanfuo-Yeboah, J., Iannucci, R., Chowdhury, S., Alton, K., Patrick, J.E. and Zbaida, S. (2004a) Identification of human UDP-glucuronosyltransferase enzyme(s) responsible for the glucuronidation of ezetimibe (Zetia). *Drug Metab. Dispos.* **32**: 314-320.

Ghosal, A., Hapangama, N., Yuan, Y., Achanfuo-Yeboah, J., Iannucci, R., Chowdhury, S., Alton, K., Patrick, J.E. and Zbaida, S. (2004b) Identification of human UDP-glucuronosyltransferase enzyme(s) responsible for the glucuronidation of posaconazole (Noxafil). *Drug Metab. Dispos.* **32**: 267-271.

Ghosh, S.S., Sappal, B.S., Kalpana, G.V., Lee, S.W., Chowdhury, J.R. and Chowdhury, N.R. (2001) Homodimerization of human bilirubin-uridinediphosphoglucuronate glucuronosyltransferase-1 (UGT1A1) and its functional implications. *J. Biol. Chem.* **276**: 42108-42115.

Gibson, G.G., Phillips, A., Aouabdi, S., Plant, K. and Plant, N. (2006) Transcriptional regulation of the human pregnane-X receptor. *Drug Metab. Rev.* **38**: 31-49.

Ginsberg, G., Hattis, D., Sonawane, B., Russ, A., Banati, P., Kozlak, M., Smolenski, S. and Goble, R. (2002) Evaluation of child/adult pharmacokinetic differences from a database derived from the therapeutic drug literature. *Toxicol. Sci.* **66**: 185-200.

Girard, H., Court, M.H., Bernard, O., Fortier, L.C., Villeneuve, L., Hao, Q., Greenblatt, D.J., von Moltke, L.L., Perussed, L. and Guillemette, C. (2004) Identification of common polymorphisms in the promoter of the UGT1A9 gene: evidence that UGT1A9 protein and activity levels are strongly genetically controlled in the liver. *Pharmacogenetics* 14: 501-515.

Girard, H., Thibaudeau, J., Court, M.H., Fortier, L.C., Villeneuve, L., Caron, P., Hao, Q., von Moltke, L.L., Greenblatt, D.J. and Guillemette, C. (2005) UGT1A1 polymorphisms are important determinants of dietary carcinogen detoxification in the liver. *Hepatology* **42**: 448-457.

Girard, H., Villeneuve, L., Court, M.H., Fortier, L.C., Caron, P., Hao, Q., von Moltke, L.L., Greenblatt, D.J. and Guillemette, C. (2006) The novel UGT1A9 intronic I399 polymorphism appears as a predictor of 7-ethyl-10-

hydroxycamptothecin glucuronidation levels in the liver. *Drug Metab. Dispos.* 34: 1220-1228.

Giuliani, L., Ciotti, M., Stoppacciaro, A., Pasquini, A., Silvestri, I., De Matteis, A., Frati, L. and Agliano, A.M. (2005) UDP-glucuronosyltransferases 1A expression in human urinary bladder and colon cancer by immunohistochemistry. *Oncol. Rep.* **13**: 185-191.

Gombart, A.F., Hofmann, W.K., Kawano, S., Takeuchi, S., Krug, U., Kwok, S.H., Larsen, R.J., Asou, H., Miller, C.W., Hoelzer, D. and Koeffler, H.P. (2002) Mutations in the gene encoding the transcription factor CCAAT/enhancer binding protein alpha in myelodysplastic syndromes and acute myeloid leukemias. *Blood* **99**: 1332-1340.

Gong, Q.H., Cho, J.W., Huang, T., Potter, C., Gholami, N., Basu, N.K., Kubota, S., Carvalho, S., Pennington, M.W., Owens, I.S. and Popescu, N.C. (2001) Thirteen UDP-glucuronosyltransferase genes are encoded at the human *UGT1* gene complex locus. *Pharmacogenetics* **11**: 357-368.

Gragnoli, C., Lindner, T., Cockburn, B.N., Kaisaki, P.J., Gragnoli, F., Marozzi, G. and Bell, G.I. (1997) Maturity-onset diabetes of the young due to a mutation in the hepatocyte nuclear factor-4alpha binding site in the promoter of the hepatocyte nuclear factor-1alpha gene. *Diabetes* **46**: 1648-1651.

Grancharov, K., Engelberg, H., Naydenova, Z., Muller, G., Rettenmeier, A.W. and Golovinsky, E. (2001) Inhibition of UDP-glucuronosyltransferases in rat liver microsomes by natural mutagens and carcinogens. *Arch. Toxicol.* **75**: 609-612.

Grech-Belanger, O., Gilbert, M., Turgeon, J. and LeBlanc, P.P. (1985) Effect of cigarette smoking on mexiletine kinetics. *Clin. Pharmacol. Ther.***37**: 638-643.

Green, M.D., Bishop, W.P. and Tephly, T.R. (1995) Expressed human UGT1.4 protein catalyzes the formation of quaternary ammonium-linked glucuronides. *Drug Metab. Dispos.* **23**: 299-302.

Green, M.D., King, C.D., Mojarrabi, B., Mackenzie, P.I. and Tephly, T.R. (1998a) Glucuronidation of amines and other xenobiotics catalyzed by expressed human UDP-glucuronosyltransferase 1A3. *Drug Metab. Dispos.* **26**: 507-512.

Green, M.D., Oturu, E.M. and Tephly, T.R. (1994) Stable expression of a human liver UDP-glucuronosyltransferase (UGT2B15) with activity toward steroid and xenobiotic substrates. *Drug Metab. Dispos.* **22**: 799-805.

Green, M.D. and Tephly, T.R. (1996) Glucuronidation of amines and hydroxylated xenobiotics and endobiotics catalyzed by expressed human UGT1.4 protein. *Drug Metab. Dispos.* **24**: 356-363.

Green, M.D. and Tephly, T.R. (1998) Glucuronidation of amine substrates by purified and expressed UDP-glucuronosyltransferase proteins. *Drug Metab. Dispos.* **26**: 860-867.

Green, V.J., Kokkotou, E. and Ladias, J.A. (1998b) Critical structural elements and multitarget protein interactions of the transcriptional activator AF-1 of hepatocyte nuclear factor 4. *J. Biol. Chem.* **273**: 29950-29957.

Greenblatt, D.J., Divoll, M., Harmatz, J.S. and Shader, R.I. (1980) Oxazepam kinetics: effects of age and sex. *J. Pharmacol. Exp. Ther.* **215**: 86-91.

Gregori, C., Kahn, A. and Pichard, A.L. (1993) Competition between transcription factors HNF1 and HNF3, and alternative cell-specific activation by DBP and C/EBP contribute to the regulation of the liver-specific aldolase B promoter. *Nucleic Acids Res.* **21**: 897-903.

Gregori, C., Kahn, A. and Pichard, A.L. (1994) Activity of the rat liver-specific aldolase B promoter is restrained by HNF3. *Nucleic Acids Res.* **22**: 1242-1246.

Gregori, C., Porteu, A., Lopez, S., Kahn, A. and Pichard, A.L. (1998) Characterization of the aldolase B intronic enhancer. *J. Biol. Chem.* **273**: 25237-25243.

Gregory, P.A., Gardner-Stephen, D.A., Lewinsky, R.H., Duncliffe, K.N. and Mackenzie, P.I. (2003) Cloning and characterization of the human UDP-glucuronosyltransferase 1A8, 1A9, and 1A10 gene promoters. Differential regulation through an initiator-like region. *J. Biol. Chem.* **278**: 36107-36114.

Gregory, P.A., Gardner-Stephen, D.A., Rogers, A., Michael, M.Z. and Mackenzie, P.I. (2006) The caudal-related homeodomain protein Cdx2 and hepatocyte nuclear factor 1alpha cooperatively regulate the UDP-glucuronosyltransferase 2B7 gene promoter. *Pharmacogenet. Genomics* **16**: 527-536.

Gregory, P.A., Hansen, A.J. and Mackenzie, P.I. (2000) Tissue specific differences in the regulation of the UDP glucuronosyltransferase 2B17 gene promoter. *Pharmacogenetics* **10**: 809-820.

Gregory, P.A., Lewinsky, R.H., Gardner-Stephen, D.A. and Mackenzie, P.I. (2004) Coordinate regulation of the human UDP-glucuronosyltransferase 1A8, 1A9, and 1A10 genes by hepatocyte nuclear factor 1alpha and the caudal-related homeodomain protein 2. *Mol. Pharmacol.* **65**: 953-963.

Gregory, P.A. and Mackenzie, P.I. (2002) The homeodomain Pbx2-Prep1 complex modulates hepatocyte nuclear factor 1alpha-mediated activation of the UDP-glucuronosyltransferase 2B17 gene. *Mol. Pharmacol.* **62**: 154-161.

Grubb, N.G., Rudy, D.W., Brater, D.C. and Hall, S.D. (1999) Stereoselective pharmacokinetics of ketoprofen and ketoprofen glucuronide in end-stage renal disease: evidence for a 'futile cycle' of elimination. *Br. J. Clin. Pharmacol.* **48**: 494-500.

Grunstein, M. (1997) Histone acetylation in chromatin structure and transcription. *Nature* **389**: 349-352.

Gsur, A., Preyer, M., Haidinger, G., Schatzl, G., Madersbacher, S., Marberger, M., Vutuc, C. and Micksche, M. (2002) A polymorphism in the *UDP-glucuronosyltransferase 2B15* gene (D85Y) is not associated with prostate cancer risk. *Cancer Epidemiol. Biomarkers Prev.* **11**: 497-498.

Guengerich, F.P. (2000) Metabolism of chemical carcinogenes. *Carcinogenesis* **21**: 345-351.

Guillemette, C., De Vivo, I., Hankinson, S.E., Haiman, C.A., Spiegelman, D., Housman, D.E. and Hunter, D.J. (2001) Association of genetic polymorphisms in UGT1A1 with breast cancer and plasma hormone levels. *Cancer Epidemiol. Biomarkers Prev.* **10**: 711-714.

Guillemette, C., Millikan, R.C., Newman, B. and Housman, D.E. (2000) Genetic polymorphisms in uridine diphospho-glucuronosyltransferase 1A1 and association with breast cancer among African Americans. *Cancer Res.* **60**: 950-956.

Gunaratna, C. (2000) Drug metabolism and pharmacokinetics in drug discovery: a primer for bioanalytical chemists, Part 1. *Curr. Separations* **19**: 17-23.

Haberkorn, V., Heydel, J.M., Mounie, J., Artur, Y. and Goudonnet, H. (2002) Vitamin A modulates the effects of thyroid hormone on UDP-glucuronosyltransferase expression and activity in rat liver. *Mol. Cell. Endocrinol.* **190**: 167-175.

Haga, S.B., Khoury, M.J. and Burke, W. (2003) Genomic profiling to promote a healthy lifestyle: not ready for prime time. *Nat. Genet.* **34**: 347-350.

Hajdinjak, T. and Zagradisnik, B. (2004) Prostate cancer and polymorphism D85Y in gene for dihydrotestosterone degrading enzyme UGT2B15: frequency of DD homozygotes increases with Gleason Score. *Prostate* **59**: 436-439.

Hall, R.K., Sladek, F.M. and Granner, D.K. (1995) The orphan receptors COUP-TF and HNF-4 serve as accessory factors required for induction of phosphoenolpyruvate carboxykinase gene transcription by glucocorticoids. *Proc. Natl. Acad. Sci. U. S. A.* **92**: 412-416.

Hanioka, N., Ozawa, S., Jinno, H., Ando, M., Saito, Y. and Sawada, J. (2001) Human liver UDP-glucuronosyltransferase isoforms involved in the glucuronidation of 7-ethyl-10-hydroxycamptothecin. *Xenobiotica* **31**: 687-699.

Hansen, A.J., Lee, Y.H., Gonzalez, F.J. and Mackenzie, P.I. (1997) HNF1alpha activates the rat UDP glucuronosyltransferase UGT2B1 gene promoter. *DNA Cell Biol.* **16**: 207-214.

Hansen, A.J., Lee, Y.H., Sterneck, E., Gonzalez, F.J. and Mackenzie, P.I. (1998) C/EBPalpha is a regulator of the UDP glucuronosyltransferase *UGT2B1* gene. *Mol. Pharmacol.* **53**: 1027-1033.

Hara, K., Horikoshi, M., Kitazato, H., Ito, C., Noda, M., Ohashi, J., Froguel, P., Tokunaga, K., Tobe, K., Nagai, R. and Kadowaki, T. (2006) Hepatocyte nuclear factor-4alpha P2 promoter haplotypes are associated with type 2 diabetes in the Japanese population. *Diabetes* **55**: 1260-1264.

Harp, J.B., Franklin, D., Vanderpuije, A.A. and Gimble, J.M. (2001) Differential expression of signal transducers and activators of transcription during human adipogenesis. *Biochem. Biophys. Res. Commun.* **281**: 907-912.

Harrington, W.R., Sengupta, S. and Katzenellenbogen, B.S. (2006) Estrogen regulation of the glucuronidation enzyme UGT2B15 in estrogen receptor-positive breast cancer cells. *Endocrinology* **147**: 3843-3850.

Harrow, F., Amuta, J.U., Hutchinson, S.R., Akwaa, F. and Ortiz, B.D. (2004) Factors binding a non-classical Cis-element prevent heterochromatin effects on locus control region activity. *J. Biol. Chem.* **279**: 17842-17849.

Hata, S., Inoue, T., Kosuga, K., Nakashima, T., Tsukamoto, T. and Osumi, T. (1995) Identification of two splice isoforms of mRNA for mouse hepatocyte nuclear factor 4 (HNF-4). *Biochim. Biophys. Acta* **1260**: 55-61. Hata, S., Tsukamoto, T. and Osumi, T. (1992) A novel isoform of rat hepatocyte nuclear factor 4 (HNF-4). *Biochim. Biophys. Acta* **1131**: 211-213.

Hatzis, P. and Talianidis, I. (2001) Regulatory mechanisms controlling human hepatocyte nuclear factor 4alpha gene expression. *Mol. Cell. Biol.* **21**: 7320-7330.

Hatzis, P. and Talianidis, I. (2002) Dynamics of enhancer-promoter communication during differentiation-induced gene activation. *Mol. Cell* **10**: 1467-1477.

Haumaitre, C., Fabre, M., Cormier, S., Baumann, C., Delezoide, A.L. and Cereghini, S. (2006) Severe pancreas hypoplasia and multicystic renal dysplasia in two human fetuses carrying novel *HNF1beta/MODY5* mutations. *Hum. Mol. Genet.* **15**: 2363-2375.

Hayhurst, G.P., Lee, Y.H., Lambert, G., Ward, J.M. and Gonzalez, F.J. (2001) Hepatocyte nuclear factor 4alpha (nuclear receptor 2A1) is essential for maintenance of hepatic gene expression and lipid homeostasis. *Mol. Cell. Biol.* **21**: 1393-1403.

Hazai, E., Gagne, P.V. and Kupfer, D. (2004) Glucuronidation of the oxidative cytochrome P450-mediated phenolic metabolites of the endocrine disruptor pesticide: methoxychlor by human hepatic UDP-glucuronosyl transferases. *Drug Metab. Dispos.* **32**: 742-751.

Hecht, S.S. (2003) Tobacco carcinogens, their biomarkers and tobacco-induced cancer. *Nat. Rev. Cancer* **3**: 733-744.

Heijmans, B.T., Westendorp, R.G., Droog, S., Kluft, C., Knook, D.L. and Slagboom, P.E. (2002) Association of the tumour necrosis factor alpha -308G/A polymorphism with the risk of diabetes in an elderly population-based cohort. *Genes Immun.* **3**: 225-228.

Henkens, T., Papeleu, P., Elaut, G., Vinken, M., Rogiers, V. and Vanhaecke, T. (2007) Trichostatin A, a critical factor in maintaining the functional differentiation of primary cultured rat hepatocytes. *Toxicol. Appl. Pharmacol.* **218**: 64-71.

Herr, W., Sturm, R.A., Clerc, R.G., Corcoran, L.M., Baltimore, D., Sharp, P.A., Ingraham, H.A., Rosenfeld, M.G., Finney, M. and Ruvkun, G. (1988) The POU domain: a large conserved region in the mammalian *pit-1*, *oct-1*, *oct-2*, and *Caenorhabditis elegans unc-86* gene products. *Genes Dev.* **2**: 1513-1516.

Hertz, R., Kalderon, B., Byk, T., Berman, I., Za'tara, G., Mayer, R. and Bar-Tana, J. (2005) Thioesterase activity and acyl-CoA/fatty acid cross-talk of hepatocyte nuclear factor-4alpha. *J. Biol. Chem.* **280**: 24451-24461.

Hertz, R., Magenheim, J., Berman, I. and Bar-Tana, J. (1998) Fatty acyl-CoA thioesters are ligands of hepatic nuclear factor-4alpha. *Nature* **392**: 512-516.

Hertz, R., Sheena, V., Kalderon, B., Berman, I. and Bar-Tana, J. (2001) Suppression of hepatocyte nuclear factor-4alpha by acyl-CoA thioesters of hypolipidemic peroxisome proliferators. *Biochem. Pharmacol.* **61**: 1057-1062.

Hess-Stumpp, H. (2005) Histone deacetylase inhibitors and cancer: from cell biology to the clinic. *Eur. J. Cell Biol.* **84**: 109-121.

Hiemisch, H., Schutz, G. and Kaestner, K.H. (1997) Transcriptional regulation in endoderm development: characterization of an enhancer controlling *Hnf3g* expression by transgenesis and targeted mutagenesis. *EMBO J.* **16**: 3995-4006.

Hiesberger, T., Shao, X., Gourley, E., Reimann, A., Pontoglio, M. and Igarashi, P. (2005) Role of the hepatocyte nuclear factor-1beta (HNF-1beta) C-terminal domain in *Pkhd1* (ARPKD) gene transcription and renal cystogenesis. *J. Biol. Chem.* **280**: 10578-10586.

Hirata-Koizumi, M., Saito, M., Miyake, S. and Hasegawa, R. (2007) Adverse events caused by drug interactions involving glucuronoconjugates of zidovudine, valproic acid and lamotrigine, and analysis of how such potential events are discussed in package inserts of Japan, UK and USA. *J. Clin. Pharm. Ther.* **32**: 177-185.

Hirota, K., Aoyama, H. and Fukamizu, A. (2005) Mutation analysis of HNF-4 binding sites in the human glucose-6-phosphatase promoter. *Int. J. Mol. Med.* **15**: 487-490.

Hoffmann, U. and Kroemer, H.K. (2004) The ABC transporters MDR1 and MRP2: multiple functions in disposition of xenobiotics and drug resistance. *Drug Metab. Rev.* **36**: 669-701.

Holewa, B., Zapp, D., Drewes, T., Senkel, S. and Ryffel, G.U. (1997) *HNF4* beta, a new gene of the *HNF4* family with distinct activation and expression profiles in oogenesis and embryogenesis of *Xenopus laevis*. *Mol. Cell. Biol.* **17**: 687-694.

Holmkvist, J., Cervin, C., Lyssenko, V., Winckler, W., Anevski, D., Cilio, C., Almgren, P., Berglund, G., Nilsson, P., Tuomi, T., Lindgren, C.M., Altshuler, D. and Groop, L. (2006) Common variants in *HNF-1* alpha and risk of type 2 diabetes. *Diabetologia* **49**: 2882-2891.

Honda, H., Pazin, M.J., Ji, H., Wernyj, R.P. and Morin, P.J. (2006) Crucial roles of Sp1 and epigenetic modifications in the regulation of the *CLDN4* promoter in ovarian cancer cells. *J. Biol. Chem.* **281**: 21433-21444.

Hoyumpa, A.M., Patwardhan, R., Maples, M., Desmond, P.V., Johnson, R.F., Sinclair, A.P. and Schenker, S. (1981) Effect of short-term ethanol administration on lorazepam clearance. *Hepatology* **1**: 47-53.

Hsieh, T.Y., Shiu, T.Y., Huang, S.M., Lin, H.H., Lee, T.C., Chen, P.J., Chu, H.C., Chang, W.K., Jeng, K.S., Lai, M.M. and Chao, Y.C. (2007) Molecular pathogenesis of Gilbert's syndrome: decreased TATA-binding protein binding affinity of UGT1A1 gene promoter. *Pharmacogenet. Genomics* **17**: 229-236.

Hsu, W., Kerppola, T.K., Chen, P.L., Curran, T. and Chen-Kiang, S. (1994) Fos and Jun repress transcription activation by NF-IL6 through association at the basic zipper region. *Mol. Cell. Biol.* **14**: 268-276.

Hu, C. and Perlmutter, D.H. (1999) Regulation of alpha1-antitrypsin gene expression in human intestinal epithelial cell line caco-2 by HNF-1alpha and HNF-4. *Am. J. Physiol.* **276**: G1181-1194.

Hu, Z. and Wells, P.G. (2004) Human interindividual variation in lymphocyte UDPglucuronosyltransferases as a determinant of *in vitro* benzo[a]pyrene covalent binding and cytotoxicity. *Toxicol. Sci.* **78**: 32-40.

Huang, Y.C. and Hung, W.C. (2006) 1,25-dihydroxyvitamin D3 transcriptionally represses p45Skp2 expression via the Sp1 sites in human prostate cancer cells. *J. Cell. Physiol.* **209**: 363-369.

Hubner, R.A., Muir, K.R., Liu, J.F., Logan, R.F., Grainge, M., Armitage, N., Shepherd, V., Popat, S., Houlston, R.S. and United Kingdom Colorectal Adenoma Prevention, C. (2006) Genetic variants of *UGT1A6* influence risk of colorectal adenoma recurrence. *Clin. Cancer Res.* **12**: 6585-6589.

Huggett, J., Dheda, K., Bustin, S. and Zumla, A. (2005) Real-time RT-PCR normalisation; strategies and considerations. *Genes Immun.* **6**: 279-284.

Hume, R., Coughtrie, M.W. and Burchell, B. (1995) Differential localisation of UDP-glucuronosyltransferase in kidney during human embryonic and fetal development. *Arch. Toxicol.* **69**: 242-247.

Hung, H.L. and High, K.A. (1996) Liver-enriched transcription factor HNF-4 and ubiquitous factor NF-Y are critical for expression of blood coagulation factor X. J. *Biol. Chem.* **271**: 2323-2331.

Huo, D., Kim, H.J., Adebamowo, C.A., Ogundiran, T.O., Akang, E.E., Campbell, O., Adenipekun, A., Niu, Q., Sveen, L., Fackenthal, J.D., Fackenthal, D.L., Das, S., Cox, N., Rienzo, A.D. and Olopade, O.I. (2007) Genetic polymorphisms in uridine diphospho-glucuronosyltransferase 1A1 and breast cancer risk in Africans. *Breast Cancer Res. Treat.* In press.

Huo, X. and Zhang, J. (2005) Important roles of reversible acetylation in the function of hematopoietic transcription factors. *J. Cell. Mol. Med.* **9**: 103-112.

Hustert, E., Zibat, A., Presecan-Siedel, E., Eiselt, R., Mueller, R., Fuss, C., Brehm, I., Brinkmann, U., Eichelbaum, M., Wojnowski, L. and Burk, O. (2001) Natural protein variants of pregnane X receptor with altered transactivation activity toward *CYP3A4*. *Drug Metab. Dispos.* **29**: 1454-1459.

Ikushiro, S., Emi, Y. and Iyanagi, T. (1997) Protein-protein interactions between UDP-glucuronosyltransferase isozymes in rat hepatic microsomes. *Biochemistry* **36**: 7154-7161.

Ikushiro, S., Emi, Y., Kato, Y., Yamada, S. and Sakaki, T. (2006) Monospecific antipeptide antibodies against human hepatic UDP-glucuronosyltransferase 1A subfamily (UGT1A) isoforms. *Drug Metab. Pharmacokinet.* **21**: 70-74.

Innocenti, F., Liu, W., Chen, P., Desai, A.A., Das, S. and Ratain, M.J. (2005) Haplotypes of variants in the UDP-glucuronosyltransferase *1A9* and *1A1* genes. *Pharmacogenet. Genomics* **15**: 295-301.

Innocenti, F., Undevia, S.D., Iyer, L., Chen, P.X., Das, S., Kocherginsky, M., Karrison, T., Janisch, L., Ramirez, J., Rudin, C.M., Vokes, E.E. and Ratain, M.J. (2004) Genetic variants in the *UDP-glucuronosyltransferase 1A1* gene predict the risk of severe neutropenia of irinotecan. *J. Clin. Oncol.* **22**: 1382-1388.

International HapMap Consortium (2003) The International HapMap Project. *Nature* **426**: 789-796.

International Human Genome Sequencing Consortium (2001) Initial sequencing and analysis of the human genome. *Nature* **409**: 860-921.

Ishii, Y., Hansen, A.J. and Mackenzie, P.I. (2000) Octamer transcription factor-1 enhances hepatic nuclear factor-1alpha-mediated activation of the human UDP glucuronosyltransferase 2B7 promoter. *Mol. Pharmacol.* **57**: 940-947.

Ishii, Y., Miyoshi, A., Maji, D., Yamada, H. and Oguri, K. (2004) Simultaneous expression of guinea pig UDP-glucuronosyltransferase 2B21 (UGT2B21) and 2B22 in COS-7 cells enhances UGT2B21-catalyzed chloramphenicol glucuronidation. *Drug Metab. Dispos.* **32**: 1057-1060.

Ishii, Y., Miyoshi, A., Watanabe, R., Tsuruda, K., Tsuda, M., Yamaguchi-Nagamatsu, Y., Yoshisue, K., Tanaka, M., Maji, D., Ohgiya, S. and Oguri, K. (2001) Simultaneous expression of guinea pig UDP-glucuronosyltransferase 2B21 and 2B22 in COS-7 cells enhances UDP-glucuronosyltransferase 2B21-catalyzed morphine-6-glucuronide formation. *Mol. Pharmacol.* **60**: 1040-1048.

Ishiyama, T., Kano, J., Minami, Y., Iijima, T., Morishita, Y. and Noguchi, M. (2003) Expression of HNFs and C/EBPalpha is correlated with immunocytochemical differentiation of cell lines derived from human hepatocellular carcinomas, hepatoblastomas and immortalized hepatocytes. *Cancer Sci.* **94**: 757-763.

Issa, A.M. (2000) Ethical considerations in clinical pharmacogenomics research. *Trends Pharmacol. Sci.* **21**: 247-249.

Itoh, M., Nakajima, M., Higashi, E., Yoshida, R., Nagata, K., Yamazoe, Y. and Yokoi, T. (2006) Induction of human CYP2A6 is mediated by the pregnane X receptor with peroxisome proliferator-activated receptor-gamma coactivator 1alpha. *J. Pharmacol. Exp. Ther.* **319**: 693-702.

Iwai, M., Maruo, Y., Ito, M., Yamamoto, K., Sato, H. and Takeuchi, Y. (2004) Six novel UDP-glucuronosyltransferase (UGT1A3) polymorphisms with varying activity. *J. Hum. Genet.* **49**: 123-128.

Iyanagi, T., Haniu, M., Sogawa, K., Fujii-Kuriyama, Y., Watanabe, S., Shively, J.E. and Anan, K.F. (1986) Cloning and characterization of cDNA encoding 3-methylcholanthrene inducible rat mRNA for UDP-glucuronosyltransferase. *J. Biol. Chem.* **261**: 15607-15614.

Jacquemin, P., Durviaux, S.M., Jensen, J., Godfraind, C., Gradwohl, G., Guillemot, F., Madsen, O.D., Carmeliet, P., Dewerchin, M., Collen, D., Rousseau, G.G. and Lemaigre, F.P. (2000) Transcription factor hepatocyte nuclear factor 6 regulates pancreatic endocrine cell differentiation and controls expression of the proendocrine gene ngn3. *Mol. Cell. Biol.* **20**: 4445-4454.

Jacquemin, P., Lannoy, V.J., Rousseau, G.G. and Lemaigre, F.P. (1999) OC-2, a novel mammalian member of the ONECUT class of homeodomain transcription factors whose function in liver partially overlaps with that of hepatocyte nuclear factor-6. *J. Biol. Chem.* **274**: 2665-2671.

Jakobsson, J., Ekstrom, L., Inotsume, N., Garle, M., Lorentzon, M., Ohlsson, C., Roh, H.K., Carlstrom, K. and Rane, A. (2006) Large differences in testosterone excretion in Korean and Swedish men are strongly associated with a UDP-glucuronosyl transferase 2B17 polymorphism. *J. Clin. Endocrinol. Metab.* **91**: 687-693.

Jansen, P.L. (1999) Diagnosis and management of Crigler-Najjar syndrome. *Eur. J. Pediatr.* **158**: S89-94.

Jedlitschky, G., Cassidy, A.J., Sales, M., Pratt, N. and Burchell, B. (1999) Cloning and characterization of a novel human olfactory UDP glycosyltransferase. *Biochem. J.* **340**: 837-843.

Jiang, G., Nepomuceno, L., Hopkins, K. and Sladek, F.M. (1995) Exclusive homodimerization of the orphan receptor hepatocyte nuclear factor 4 defines a new subclass of nuclear receptors. *Mol. Cell. Biol.* **15**: 5131-5143.

Jin, C., Miners, J.O., Lillywhite, K.J. and Mackenzie, P.I. (1993) Complementary deoxyribonucleic acid cloning and expression of a human liver uridine diphosphate-glucuronosyltransferase glucuronidating carboxylic acid-containing drugs. *J. Pharmacol. Exp. Ther.* **264**: 475-479.

Jinno, H., Saeki, M., Tanaka-Kagawa, T., Hanioka, N., Saito, Y., Ozawa, S., Ando, M., Shirao, K., Minami, H., Ohtsu, A., Yoshida, T., Saijo, N. and Sawada, J. (2003) Functional characterization of wild-type and variant (T202I and M59I) human UDP-glucuronosyltransferase 1A10. *Drug Metab. Dispos.* **31**: 528-532.

Johnson, A.D., Wang, D. and Sadee, W. (2005) Polymorphisms affecting gene regulation and mRNA processing: broad implications for pharmacogenetics. *Pharmacol. Ther.* **106**: 19-38.

Johnson, D.R., Li, C.W., Chen, L.Y., Ghosh, J.C. and Chen, J.D. (2006) Regulation and binding of pregnane X receptor by nuclear receptor corepressor silencing mediator of retinoid and thyroid hormone receptors (SMRT). *Mol. Pharmacol.* **69**: 99-108.

Josephy, P.D., Guengerich, F.P. and Miners, J.O. (2005) "Phase I" and "Phase II" drug metabolism: terminology that we should phase out? *Drug Metab. Rev.* **37**: 575-580.

Jover, R., Bort, R., Gomez-Lechon, M.J. and Castell, J.V. (1998) Re-expression of C/EBPalpha induces CYP2B6, CYP2C9 and CYP2D6 genes in HepG2 cells. *FEBS Lett.* **431**: 227-230.

Jover, R., Bort, R., Gomez-Lechon, M.J. and Castell, J.V. (2001) Cytochrome P450 regulation by hepatocyte nuclear factor 4 in human hepatocytes: a study using adenovirus-mediated antisense targeting. *Hepatology* **33**: 668-675.

Jover, R., Bort, R., Gomez-Lechon, M.J. and Castell, J.V. (2002) Down-regulation of human CYP3A4 by the inflammatory signal interleukin-6: molecular mechanism and transcription factors involved. *FASEB J.* **16**: 1799-1801.

Jung, D., Elferink, M.G., Stellaard, F. and Groothuis, G.M. (2007) Analysis of bile acid-induced regulation of FXR target genes in human liver slices. *Liver Int.* **27**: 137-144.

Jung, D. and Kullak-Ublick, G.A. (2003) Hepatocyte nuclear factor 1alpha: a key mediator of the effect of bile acids on gene expression. *Hepatology* **37**: 622-631.

Kaestner, K.H., Hiemisch, H., Luckow, B. and Schutz, G. (1994) The HNF-3 gene family of transcription factors in mice: gene structure, cDNA sequence, and mRNA distribution. *Genomics* **20**: 377-385.

Kaestner, K.H., Hiemisch, H. and Schutz, G. (1998) Targeted disruption of the gene encoding hepatocyte nuclear factor 3gamma results in reduced transcription of hepatocyte-specific genes. *Mol. Cell. Biol.* **18**: 4245-4251.

Kaestner, K.H., Katz, J., Liu, Y., Drucker, D.J. and Schutz, G. (1999) Inactivation of the winged helix transcription factor *HNF3* alpha affects glucose homeostasis and islet glucagon gene expression in vivo. *Genes Dev.* **13**: 495-504.

Kaivosaari, S., Toivonen, P., Hesse, L.M., Koskinen, M., Court, M.H. and Finel, M. (2007) Nicotine glucuronidation and the human UDP-glucuronosyltransferase UGT2B10. *Mol. Pharmacol.* **72**: 761-768.

Kaji, H. and Kume, T. (2005) Characterization of afloqualone *N*-glucuronidation: species differences and identification of human UDP-glucuronosyltransferase isoform(s). *Drug Metab. Dispos.* **33**: 60-67.

Kaku, T., Ogura, K., Nishiyama, T., Ohnuma, T., Muro, K. and Hiratsuka, A. (2004) Quaternary ammonium-linked glucuronidation of tamoxifen by human liver microsomes and UDP-glucuronosyltransferase 1A4. *Biochem. Pharmacol.* **67**: 2093-2102.

Kalderon, B., Sheena, V., Shachrur, S., Hertz, R. and Bar-Tana, J. (2002) Modulation by nutrients and drugs of liver acyl-CoAs analyzed by mass spectrometry. *J. Lipid Res.* **43**: 1125-1132.

Kamiya, A., Inoue, Y. and Gonzalez, F.J. (2003) Role of the hepatocyte nuclear factor 4alpha in control of the pregnane X receptor during fetal liver development. *Hepatology* **37**: 1375-1384.

Kanou, M., Usui, T., Ueyama, H., Sato, H., Ohkubo, I. and Mizutani, T. (2004) Stimulation of transcriptional expression of human UDP-glucuronosyltransferase 1A1 by dexamethasone. *Mol. Biol. Rep.* **31**: 151-158.

Kari, F.W., Kauffman, F.C. and Thurman, R.G. (1984) Characterization of mutagenic glucuronide formation from benzo(*a*)pyrene in the nonrecirculating perfused rat liver. *Cancer Res.* 44: 5073-5078.

Karvonen, U., Janne, O.A. and Palvimo, J.J. (2006) Androgen receptor regulates nuclear trafficking and nuclear domain residency of corepressor HDAC7 in a ligand-dependent fashion. *Exp. Cell Res.* **312**: 3165-3183.

Kasai, N., Sakaki, T., Shinkyo, R., Ikushiro, S., Iyanagi, T., Kamao, M., Okano, T., Ohta, M. and Inouye, K. (2004) Sequential metabolism of 2,3,7-trichlorodibenzo-*P*-dioxin (2,3,7-triCDD) by cytochrome P450 and UDP-glucuronosyltransferase in human liver microsomes. *Drug Metab. Dispos.* **32**: 870-875.

Kasai, N., Sakaki, T., Shinkyo, R., Ikushiro, S., Iyanagi, T., Ohta, M. and Inouye, K. (2005) Metabolism of 26,26,26,27,27,27-F6-1 alpha,23S,25-trihydroxyvitamin D3 by human UDP-glucuronosyltransferase 1A3. *Drug Metab. Dispos.* **33**: 102-107.

Kast, H.R., Goodwin, B., Tarr, P.T., Jones, S.A., Anisfeld, A.M., Stoltz, C.M., Tontonoz, P., Kliewer, S., Willson, T.M. and Edwards, P.A. (2002) Regulation of multidrug resistance-associated protein 2 (ABCC2) by the nuclear receptors pregnane X receptor, farnesoid X-activated receptor, and constitutive androstane receptor. *J. Biol. Chem.* **277**: 2908-2915.

Kawade, N. and Onishi, S. (1981) The prenatal and postnatal development of UDPglucuronyltransferase activity towards bilirubin and the effect of premature birth on this activity in the human liver. *Biochem. J.* **196**: 257-260.

Kelly, W.K., O'Connor, O.A., Krug, L.M., Chiao, J.H., Heaney, M., Curley, T., MacGregore-Cortelli, B., Tong, W., Secrist, J.P., Schwartz, L., Richardson, S., Chu, E., Olgac, S., Marks, P.A., Scher, H. and Richon, V.M. (2005) Phase I study of an

oral histone deacetylase inhibitor, suberoylanilide hydroxamic acid, in patients with advanced cancer. J. Clin. Oncol. 23: 3923-3931.

Kiang, T.K., Ensom, M.H. and Chang, T.K. (2005) UDP-glucuronosyltransferases and clinical drug-drug interactions. *Pharmacol. Ther.* **106**: 97-132.

Kijima, M., Yoshida, M., Sugita, K., Horinouchi, S. and Beppu, T. (1993) Trapoxin, an antitumor cyclic tetrapeptide, is an irreversible inhibitor of mammalian histone deacetylase. *J. Biol. Chem.* **268**: 22429-22435.

Kikuchi, R., Kusuhara, H., Hattori, N., Shiota, K., Kim, I., Gonzalez, F.J. and Sugiyama, Y. (2006) Regulation of the expression of human organic anion transporter 3 by hepatocyte nuclear factor 1alpha/beta and DNA methylation. *Mol. Pharmacol.* **70**: 887-896.

Kim, J.Y., Kim, H.J., Kim, K.T., Park, Y.Y., Seong, H.A., Park, K.C., Lee, I.K., Ha, H., Shong, M., Park, S.C. and Choi, H.S. (2004) Orphan Nuclear Receptor Small Heterodimer Partner Represses Hepatocyte Nuclear Factor 3/Foxa Transactivation via Inhibition of Its DNA Binding. *Mol. Endocrinol.* **18**: 2880-2894.

King, C.D., Rios, G.R., Green, M.D. and Tephly, T.R. (2000) UDP-glucuronosyltransferases. *Curr. Drug Metab.* 1: 143-161.

Kinoshita, S., Akira, S. and Kishimoto, T. (1992) A member of the C/EBP family, NF-IL6beta, forms a heterodimer and transcriptionally synergizes with NF-IL6. *Proc. Natl. Acad. Sci. U. S. A.* **89**: 1473-1476.

Kitabayashi, I., Kawakami, Z., Chiu, R., Ozawa, K., Matsuoka, T., Toyoshima, S., Umesono, K., Evans, R.M., Gachelin, G. and Yokoyama, K. (1992) Transcriptional regulation of the *c-jun* gene by retinoic acid and E1A during differentiation of F9 cells. *EMBO J.* **11**: 167-175.

Kitagawa, C., Ando, M., Ando, Y., Sekido, Y., Wakai, K., Imaizumi, K., Shimokata, K. and Hasegawa, Y. (2005) Genetic polymorphism in the phenobarbital-responsive enhancer module of the *UDP-glucuronosyltransferase 1A1* gene and irinotecan toxicity. *Pharmacogenet. Genomics* **15**: 35-41.

Klapper, M., Bohme, M., Nitz, I. and Doring, F. (2007) The human intestinal fatty acid binding protein (hFABP2) gene is regulated by HNF-4alpha. *Biochem. Biophys. Res. Commun.* **356**: 147-152.

Kleinjan, D.A. and van Heyningen, V. (2005) Long-range control of gene expression: emerging mechanisms and disruption in disease. *Am. J. Hum. Genet.* **76**: 8-32.

Kliewer, S.A., Goodwin, B. and Willson, T.M. (2002) The nuclear pregnane X receptor: a key regulator of xenobiotic metabolism. *Endocr. Rev.* **23**: 687-702.

Kliewer, S.A., Moore, J.T., Wade, L., Staudinger, J.L., Watson, M.A., Jones, S.A., McKee, D.D., Oliver, B.B., Willson, T.M., Zetterstrom, R.H., Perlmann, T. and Lehmann, J.M. (1998) An orphan nuclear receptor activated by pregnanes defines a novel steroid signaling pathway. *Cell* **92**: 73-82.

Knowles, B.B., Howe, C.C. and Aden, D.P. (1980) Human hepatocellular carcinoma cell lines secrete the major plasma proteins and hepatitis B surface antigen. *Science* **209**: 497-499.

Kohler, J.J., Metallo, S.J., Schneider, T.L. and Schepartz, A. (1999) DNA specificity enhanced by sequential binding of protein monomers. *Proc. Natl. Acad. Sci. U. S. A.* **96**: 11735-11739.

Koiwai, O., Aono, S., Adachi, Y., Kamisako, T., Yasui, Y., Nishizawa, M. and Sato, H. (1996) Crigler-Najjar syndrome type II is inherited both as a dominant and as a recessive trait. *Hum. Mol. Genet.* **5**: 645-647.

Kootstra-Ros, J.E., Van Weelden, M.J., Hinrichs, J.W., De Smet, P.A. and van der Weide, J. (2006) Therapeutic drug monitoring of antidepressants and Cytochrome P450 genotyping in general practice. *J. Clin. Pharmacol.* **46**: 1320-1327.

Krishnaswamy, S., Duan, S.X., Von Moltke, L.L., Greenblatt, D.J. and Court, M.H. (2003) Validation of serotonin (5-hydroxtryptamine) as an in vitro substrate probe for human UDP-glucuronosyltransferase (UGT) 1A6. *Drug Metab. Dispos.* **31**: 133-139.

Krishnaswamy, S., Hao, Q., Al-Rohaimi, A., Hesse, L.M., von Moltke, L.L., Greenblatt, D.J. and Court, M.H. (2005a) UDP glucuronosyltransferase (UGT) 1A6 pharmacogenetics: I. Identification of polymorphisms in the 5'-regulatory and exon 1 regions, and association with human liver UGT1A6 gene expression and glucuronidation. *J. Pharmacol. Exp. Ther.* **313**: 1331-1339.

Krishnaswamy, S., Hao, Q., Al-Rohaimi, A., Hesse, L.M., von Moltke, L.L., Greenblatt, D.J. and Court, M.H. (2005b) UDP glucuronosyltransferase (UGT) 1A6 pharmacogenetics: II. Functional impact of the three most common nonsynonymous UGT1A6 polymorphisms (S7A, T181A, and R184S). *J. Pharmacol. Exp. Ther.* **313**: 1340-1346.

Kritis, A.A., Argyrokastritis, A., Moschonas, N.K., Power, S., Katrakili, N., Zannis, V.I., Cereghini, S. and Talianidis, I. (1996) Isolation and characterization of a third isoform of human hepatocyte nuclear factor 4. *Gene* **173**: 275-280.

Kritis, A.A., Ktistaki, E., Barda, D., Zannis, V.I. and Talianidis, I. (1993) An indirect negative autoregulatory mechanism involved in hepatocyte nuclear factor-1 gene expression. *Nucleic Acids Res.* **21**: 5882-5889.

Ktistaki, E., Ktistakis, N.T., Papadogeorgaki, E. and Talianidis, I. (1995) Recruitment of hepatocyte nuclear factor 4 into specific intranuclear compartments depends on tyrosine phosphorylation that affects its DNA-binding and transactivation potential. *Proc. Natl. Acad. Sci. U. S. A.* **92**: 9876-9880.

Ktistaki, E. and Talianidis, I. (1997) Modulation of hepatic gene expression by hepatocyte nuclear factor 1. *Science* **277**: 109-112.

Kuehl, G.E. and Murphy, S.E. (2003) *N*-glucuronidation of nicotine and cotinine by human liver microsomes and heterologously expressed UDP-glucuronosyltransferases. *Drug Metab. Dispos.* **31**: 1361-1368.

Kuno, T., Togawa, H. and Mizutani, T. (2007) Induction of human UGT1A1 by a complex of dexamethasone-GR dependent on proximal site and independent of PBREM. *Mol. Biol. Rep.* In press.

Kuo, C.J., Conley, P.B., Chen, L., Sladek, F.M., Darnell, J.E. and Crabtree, G.R. (1992) A transcriptional hierarchy involved in mammalian cell-type specific specification. *Nature* **355**: 457-461.

Kuo, C.J., Conley, P.B., Hsieh, C.L., Francke, U. and Crabtree, G.R. (1990) Molecular cloning, functional expression, and chromosomal localization of mouse hepatocyte nuclear factor 1. *Proc. Natl. Acad. Sci. U. S. A.* **87**: 9838-9842.

Kurkela, M., Garcia-Horsman, J.A., Luukkanen, L., Morsky, S., Taskinen, J., Baumann, M., Kostiainen, R., Hirvonen, J. and Finel, M. (2003) Expression and characterization of recombinant human UDP-glucuronosyltransferases (UGTs). UGT1A9 is more resistant to detergent inhibition than other UGTs and was purified as an active dimeric enzyme. *J. Biol. Chem.* **278**: 3536-3544.

Kuypers, D.R., Naesens, M., Vermeire, S. and Vanrenterghem, Y. (2005) The impact of uridine diphosphate-glucuronosyltransferase 1A9 (*UGT1A9*) gene promoter region single-nucleotide polymorphisms *T-275A* and *C-2152T* on early mycophenolic acid dose-interval exposure in de novo renal allograft recipients. *Clin. Pharmacol. Ther.***78**: 351-361.

Kwon, H.S., Huang, B., Ho Jeoung, N., Wu, P., Steussy, C.N. and Harris, R.A. (2006) Retinoic acids and trichostatin A (TSA), a histone deacetylase inhibitor, induce human pyruvate dehydrogenase kinase 4 (PDK4) gene expression. *Biochim. Biophys. Acta* **1759**: 141-151.

Ladias, J.A., Hadzopoulou-Cladaras, M., Kardassis, D., Cardot, P., Cheng, J., Zannis, V. and Cladaras, C. (1992) Transcriptional regulation of human apolipoprotein genes ApoB, ApoCIII, and ApoAII by members of the steroid hormone receptor superfamily HNF-4, ARP-1, EAR-2, and EAR-3. *J. Biol. Chem.* **267**: 15849-15860.

Laganiere, J., Deblois, G., Lefebvre, C., Bataille, A.R., Robert, F. and Giguere, V. (2005) Location analysis of estrogen receptor alpha target promoters reveals that FOXA1 defines a domain of the estrogen response. *Proc. Natl. Acad. Sci. U. S. A.* **102**: 11651-11656.

Lahuna, O., Rastegar, M., Maiter, D., Thissen, J.P., Lemaigre, F.P. and Rousseau, G.G. (2000) Involvement of STAT5 (signal transducer and activator of transcription 5) and HNF-4 (hepatocyte nuclear factor 4) in the transcriptional control of the *hnf6* gene by growth hormone. *Mol. Endocrinol.* 14: 285-294.

Lai, K., Harnish, D.C. and Evans, M.J. (2003) Estrogen receptor alpha regulates expression of the orphan receptor small heterodimer partner. *J. Biol. Chem.* **278**: 36418-36429.

Landry, C., Clotman, F., Hioki, T., Oda, H., Picard, J.J., Lemaigre, F.P. and Rousseau, G.G. (1997) HNF-6 is expressed in endoderm derivatives and nervous system of the mouse embryo and participates to the cross-regulatory network of liver-enriched transcription factors. *Dev. Biol.* **192**: 247-257.

Landschulz, W.H., Johnson, P.F. and McKnight, S.L. (1989) The DNA binding domain of the rat liver nuclear protein C/EBP is bipartite. *Science* **243**: 1681-1688.

Lankisch, T.O., Vogel, A., Eilermann, S., Fiebeler, A., Krone, B., Barut, A., Manns, M.P. and Strassburg, C.P. (2005) Identification and characterization of a functional TATA box polymorphism of the *UDP glucuronosyltransferase 1A7* gene. *Mol. Pharmacol.* **67**: 1732-1739.

Lannoy, V.J., Burglin, T.R., Rousseau, G.G. and Lemaigre, F.P. (1998) Isoforms of hepatocyte nuclear factor-6 differ in DNA-binding properties, contain a bifunctional

homeodomain, and define the new ONECUT class of homeodomain proteins. *J. Biol. Chem.* **273**: 13552-13562.

Lannoy, V.J., Decaux, J.F., Pierreux, C.E., Lemaigre, F.P. and Rousseau, G.G. (2002) Liver glucokinase gene expression is controlled by the onecut transcription factor hepatocyte nuclear factor-6. *Diabetologia* **45**: 1136-1141.

Lannoy, V.J., Rodolosse, A., Pierreux, C.E., Rousseau, G.G. and Lemaigre, F.P. (2000) Transcriptional stimulation by hepatocyte nuclear factor-6. Target-specific recruitment of either CREB-binding protein (CBP) or p300/CBP-associated factor (p/CAF). *J. Biol. Chem.* **275**: 22098-22103.

Lausen, J., Thomas, H., Lemm, I., Bulman, M., Borgschulze, M., Lingott, A., Hattersley, A.T. and Ryffel, G.U. (2000) Naturally occurring mutations in the human HNF4alpha gene impair the function of the transcription factor to a varying degree. *Nucleic Acids Res.* **28**: 430-437.

Leakey, J.E., Hume, R. and Burchell, B. (1987) Development of multiple activities of UDP-glucuronyltransferase in human liver. *Biochem. J.* 243: 859-861.

Leclerc, I., Lenzner, C., Gourdon, L., Vaulont, S., Kahn, A. and Viollet, B. (2001) Hepatocyte nuclear factor-4alpha involved in type 1 maturity-onset diabetes of the young is a novel target of AMP-activated protein kinase. *Diabetes* **50**: 1515-1521.

Lee, Y.H., Sauer, B. and Gonzalez, F.J. (1998) Laron dwarfism and non-insulindependent diabetes mellitus in the *Hnf-1*alpha knockout mouse. *Mol. Cell. Biol.* 18: 3059-3068.

Lee, Y.H., Sauer, B., Johnson, P.F. and Gonzalez, F.J. (1997) Disruption of the *c/ebp* alpha gene in adult mouse liver. *Mol. Cell. Biol.* **17**: 6014-6022.

Lee, Y.K., Dell, H., Dowhan, D.H., Hadzopoulou-Cladaras, M. and Moore, D.D. (2000) The orphan nuclear receptor SHP inhibits hepatocyte nuclear factor 4 and retinoid X receptor transactivation: two mechanisms for repression. *Mol. Cell. Biol.* **20**: 187-195.

Lehman, D.M., Richardson, D.K., Jenkinson, C.P., Hunt, K.J., Dyer, T.D., Leach, R.J., Arya, R., Abboud, H.E., Blangero, J., Duggirala, R. and Stern, M.P. (2007) P2 Promoter Variants of the Hepatocyte Nuclear Factor 4alpha Gene Are Associated With Type 2 Diabetes in Mexican Americans. *Diabetes* **56**: 513-517.

Lehmann, J.M., McKee, D.D., Watson, M.A., Willson, T.M., Moore, J.T. and Kliewer, S.A. (1998) The human orphan nuclear receptor PXR is activated by compounds that regulate *CYP3A4* gene expression and cause drug interactions. *J. Clin. Invest.* **102**: 1016-1023.

Lekstrom-Himes, J. and Xanthopoulos, K.G. (1998) Biological role of the CCAAT/enhancer-binding protein family of transcription factors. *J. Biol. Chem.* **273**: 28545-28548.

Lemaigre, F.P., Durviaux, S.M., Truong, O., Lannoy, V.J., Hsuan, J.J. and Rousseau, G.G. (1996) Hepatocyte nuclear factor 6, a transcription factor that contains a novel type of homeodomain and a single *cut* domain. *Proc. Natl. Acad. Sci. U. S. A.* **93**: 9460-9464.

Lertora, J.J., Rege, A.B., Greenspan, D.L., Akula, S., George, W.J., Hyslop, N.E. and Agrawal, K.C. (1994) Pharmacokinetic interaction between zidovudine and valproic

acid in patients infected with human immunodeficiency virus. *Clin. Pharmacol. Ther*.**56**: 272-278.

Leslie, E.M., Deeley, R.G. and Cole, S.P. (2005) Multidrug resistance proteins: role of P-glycoprotein, MRP1, MRP2, and BCRP (ABCG2) in tissue defense. *Toxicol. Appl. Pharmacol.* **204**: 216-237.

Levesque, E., Beaulieu, M., Green, M.D., Tephly, T.R., Belanger, A. and Hum, D.W. (1997) Isolation and characterization of UGT2B15(Y85): a UDP-glucuronosyltransferase encoded by a polymorphic gene. *Pharmacogenetics* **7**: 317-325.

Levesque, E., Delage, R., Benoit-Biancamano, M.O., Caron, P., Bernard, O., Couture, F. and Guillemette, C. (2007a) The impact of UGT1A8, UGT1A9, and UGT2B7 genetic polymorphisms on the pharmacokinetic profile of mycophenolic acid after a single oral dose in healthy volunteers. *Clin. Pharmacol. Ther.***81**: 392-400.

Levesque, E., Girard, H., Journault, K., Lepine, J. and Guillemette, C. (2007b) Regulation of the UGT1A1 bilirubin-conjugating pathway: role of a new splicing event at the UGT1A locus. *Hepatology* **45**: 128-138.

Levesque, E., Turgeon, D., Carrier, J.S., Montminy, V., Beaulieu, M. and Belanger, A. (2001) Isolation and characterization of the UGT2B28 cDNA encoding a novel human steroid conjugating UDP-glucuronosyltransferase. *Biochemistry* **40**: 3869-3881.

Lewinsky, R.H., Smith, P.A. and Mackenzie, P.I. (2005) Glucuronidation of bioflavonoids by human UGT1A10: structure-function relationships. *Xenobiotica* **35**: 117-129.

Li, J., Ning, G. and Duncan, S.A. (2000) Mammalian hepatocyte differentiation requires the transcription factor HNF-4alpha. *Genes Dev.* **14**: 464-474.

Li, T. and Chiang, J.Y. (2006) Rifampicin induction of CYP3A4 requires pregnane X receptor cross talk with hepatocyte nuclear factor 4alpha and coactivators, and suppression of small heterodimer partner gene expression. *Drug Metab. Dispos.* **34**: 756-764.

Li, T., Jahan, A. and Chiang, J.Y. (2006) Bile acids and cytokines inhibit the human cholesterol 7alpha-hydroxylase gene via the JNK/c-Jun pathway in human liver cells. *Hepatology* **43**: 1202-1210.

Li, X., Bratton, S. and Radominska-Pandya, A. (2007) Human UGT1A8 and UGT1A10 mRNA are expressed in primary human hepatocytes. *Drug Metab. Pharmacokinet.* **22**: 152-161.

Lin, F.T., MacDougald, O.A., Diehl, A.M. and Lane, M.D. (1993) A 30-kDa alternative translation product of the CCAAT/enhancer binding protein alpha message: transcriptional activator lacking antimitotic activity. *Proc. Natl. Acad. Sci. U. S. A.* **90**: 9606-9610.

Lin, J.P., O'Donnell, C.J., Schwaiger, J.P., Cupples, L.A., Lingenhel, A., Hunt, S.C., Yang, S. and Kronenberg, F. (2006) Association between the UGT1A1*28 allele, bilirubin levels, and coronary heart disease in the Framingham Heart Study. *Circulation* **114**: 1476-1481.

Lindpaintner, K. (2002) Pharmacogenetics and the future of medical practice. *Br. J. Clin. Pharmacol.* **54**: 221-230.

Liu, R., McEachin, R.C. and States, D.J. (2003) Computationally identifying novel NF-kappa B-regulated immune genes in the human genome. *Genome Res.* **13**: 654-661.

Liu, S.Y. and Gonzalez, F.J. (1995) Role of the liver-enriched transcription factor HNF-1alpha in expression of the *CYP2E1* gene. *DNA Cell Biol.* **14**: 285-293.

Locker, J., Ghosh, D., Luc, P.V. and Zheng, J. (2002) Definition and prediction of the full range of transcription factor binding sites - the hepatocyte nuclear factor 1 dimeric site. *Nucleic Acids Res.* **30**: 3809-3817.

Love-Gregory, L.D., Wasson, J., Ma, J., Jin, C.H., Glaser, B., Suarez, B.K. and Permutt, M.A. (2004) A common polymorphism in the upstream promoter region of the hepatocyte nuclear factor-4alpha gene on chromosome 20q is associated with type 2 diabetes and appears to contribute to the evidence for linkage in an Ashkenazi Jewish population. *Diabetes* **53**: 1134-1140.

Lu, Y., Heydel, J.M., Li, X., Bratton, S., Lindblom, T. and Radominska-Pandya, A. (2005) Lithocholic acid decreases expression of UGT2B7 in Caco-2 cells: a potential role for a negative farnesoid X receptor response element. *Drug Metab. Dispos.* **33**: 937-946.

Lucas, B., Grigo, K., Erdmann, S., Lausen, J., Klein-Hitpass, L. and Ryffel, G.U. (2005) HNF4alpha reduces proliferation of kidney cells and affects genes deregulated in renal cell carcinoma. *Oncogene* **24**: 6418-6431.

Luedde, T., Duderstadt, M., Streetz, K.L., Tacke, F., Kubicka, S., Manns, M.P. and Trautwein, C. (2004) C/EBP beta isoforms LIP and LAP modulate progression of the cell cycle in the regenerating mouse liver. *Hepatology* **40**: 356-365.

Luukkanen, L., Taskinen, J., Kurkela, M., Kostiainen, R., Hirvonen, J. and Finel, M. (2005) Kinetic characterization of the 1A subfamily of recombinant human UDP-glucuronosyltransferases. *Drug Metab. Dispos.* **33**: 1017-1026.

Mackenzie, P.I. (1986) Rat liver UDP-glucuronosyltransferase. Sequence and expression of a cDNA encoding a phenobarbital-inducible form. *J. Biol. Chem.* **261**: 6119-6125.

Mackenzie, P.I., Gregory, P.A., Lewinsky, R.H., Yasmin, S.N., Height, T., McKinnon, R.A. and Gardner-Stephen, D.A. (2005a) Polymorphic variations in the expression of the chemical detoxifying UDP glucuronosyltransferases. *Toxicol. Appl. Pharmacol.* **207**: S77-S83.

Mackenzie, P.I., Owens, I.S., Burchell, B., Bock, K.W., Bairoch, A., Belanger, A., Fournel-Gigleux, S., Green, M., Hum, D.W., Iyanagi, T., Lancet, D., Louisot, P., Magdalou, J., Chowdhury, J.R., Ritter, J.K., Schachter, H., Tephly, T.R., Tipton, K.F. and Nebert, D.W. (1997) The UDP glycosyltransferase gene superfamily: recommended nomenclature update based on evolutionary divergence. *Pharmacogenetics* **7**: 255-269.

Mackenzie, P.I., Walter Bock, K., Burchell, B., Guillemette, C., Ikushiro, S., Iyanagi, T., Miners, J.O., Owens, I.S. and Nebert, D.W. (2005b) Nomenclature update for the

mammalian UDP glycosyltransferase (UGT) gene superfamily. *Pharmacogenet*. *Genomics* **15**: 677-685.

MacLeod, S.L., Nowell, S., Plaxco, J. and Lang, N.P. (2000) An allele-specific polymerase chain reaction method for the determination of the D85Y polymorphism in the human UDP-glucuronosyltransferase 2B15 gene in a case-control study of prostate cancer. *Ann. Surg. Oncol.* **7**: 777-782.

Maeda, Y., Seidel, S.D., Wei, G., Liu, X. and Sladek, F.M. (2002) Repression of hepatocyte nuclear factor 4alpha tumor suppressor p53: involvement of the ligand-binding domain and histone deacetylase activity. *Mol. Endocrinol.* **16**: 402-410.

Magenheim, J., Hertz, R., Berman, I., Nousbeck, J. and Bar-Tana, J. (2005) Negative autoregulation of HNF-4alpha gene expression by HNF-4alpha1. *Biochem. J.* **388**: 325-332.

Maglich, J.M., Stoltz, C.M., Goodwin, B., Hawkins-Brown, D., Moore, J.T. and Kliewer, S.A. (2002) Nuclear pregnane x receptor and constitutive androstane receptor regulate overlapping but distinct sets of genes involved in xenobiotic detoxification. *Mol. Pharmacol.* **62**: 638-646.

Maitland, M.L., Vasisht, K. and Ratain, M.J. (2006) *TPMT*, *UGT1A1* and *DPYD*: genotyping to ensure safer cancer therapy? *Trends Pharmacol. Sci.* **27**: 432-437.

Malfatti, M.A. and Felton, J.S. (2001) *N*-glucuronidation of 2-amino-1-methyl-6-phenylimidazo[4,5-*b*]pyridine (PhIP) and *N*-hydroxy-PhIP by specific human UDP-glucuronosyltransferases. *Carcinogenesis* **22**: 1087-1093.

Mansur, N.R., Meyer-Siegler, K., Wurzer, J.C. and Sirover, M.A. (1993) Cell cycle regulation of the glyceraldehyde-3-phosphate dehydrogenase/uracil DNA glycosylase gene in normal human cells. *Nucleic Acids Res.* **21**: 993-998.

Marks, P.A., Richon, V.M. and Rifkind, R.A. (2000) Histone deacetylase inhibitors: inducers of differentiation or apoptosis of transformed cells. *J. Natl. Cancer Inst.* **92**: 1210-1216.

Martinez-Jimenez, C.P., Castell, J.V., Gomez-Lechon, M.J. and Jover, R. (2006a) Transcriptional activation of *CYP2C9*, *CYP1A1*, and *CYP1A2* by hepatocyte nuclear factor 4alpha requires coactivators peroxisomal proliferator activated receptor-gamma coactivator 1alpha and steroid receptor coactivator 1. *Mol. Pharmacol.* **70**: 1681-1692.

Martinez-Jimenez, C.P., Gomez-Lechon, M.J., Castell, J.V. and Jover, R. (2005) Transcriptional regulation of the human hepatic CYP3A4: identification of a new distal enhancer region responsive to CCAAT/enhancer-binding protein beta isoforms (liver activating protein and liver inhibitory protein). *Mol. Pharmacol.* **67**: 2088-2101.

Martinez-Jimenez, C.P., Gomez-Lechon, M.J., Castell, J.V. and Jover, R. (2006b) Underexpressed coactivators PGC1alpha and SRC1 impair hepatocyte nuclear factor 4 alpha function and promote dedifferentiation in human hepatoma cells. *J. Biol. Chem.* **281**: 29840-29849.

Mazure, N.M., Nguyen, T.L. and Danan, J.L. (2001) Severe hypoxia specifically downregulates hepatocyte nuclear factor-4 gene expression in HepG2 human hepatoma cells. *Tumour Biol.* **22**: 310-317.
McCarver, D.G. and Hines, R.N. (2002) The ontogeny of human drug-metabolizing enzymes: phase II conjugation enzymes and regulatory mechanisms. *J. Pharmacol. Exp. Ther.* **300**: 361-366.

McFie, P.J., Wang, G.L., Timchenko, N.A., Wilson, H.L., Hu, X. and Roesler, W.J. (2006) Identification of a co-repressor that inhibits the transcriptional and growtharrest activities of CCAAT/enhancer-binding protein alpha. *J. Biol. Chem.* **281**: 18069-18080.

McGurk, K.A., Brierley, C.H. and Burchell, B. (1998) Drug glucuronidation by human renal UDP-glucuronosyltransferases. *Biochem. Pharmacol.* **55**: 1005-1012.

McPherson, C.E., Shim, E.Y., Friedman, D.S. and Zaret, K.S. (1993) An active tissue-specific enhancer and bound transcription factors existing in a precisely positioned nucleosomal array. *Cell* **75**: 387-398.

Meech, R. and Mackenzie, P.I. (1997a) Structure and function of uridine diphosphate glucuronosyltransferases. *Clin. Exp. Pharamcol. Physiol.* **24**: 907-915.

Meech, R. and Mackenzie, P.I. (1997b) UDP-glucuronosyltransferase, the role of the amino terminus in dimerization. *J. Biol. Chem.* **272**: 26913-26917.

Mendel, D.B., Hansen, L.P., Graves, M.K., Conley, P.B. and Crabtree, G.R. (1991a) HNF-1alpha and HNF-1beta (vHNF-1) share dimerization and homeo domains, but not activation domains, and form heterodimers in vitro. *Genes Dev.* **5**: 1042-1056.

Mendel, D.B., Khavari, P.A., Conley, P.B., Graves, M.K., Hansen, L.P., Admon, A. and Crabtree, G.R. (1991b) Characterization of a cofactor that regulates dimerization of a mammalian homeodomain protein. *Science* **254**: 1762-1767.

Metz, R.P., Auyeung, D.J., Kessler, F.K. and Ritter, J.K. (2000) Involvement of hepatocyte nuclear factor 1 in the regulation of the UDP-glucuronosyltransferase 1A7 (UGT1A7) gene in rat hepatocytes. *Mol. Pharmacol.* **58**: 319-327.

Miao, J., Fang, S., Bae, Y. and Kemper, J.K. (2006) Functional inhibitory cross-talk between constitutive androstane receptor and hepatic nuclear factor-4 in hepatic lipid/glucose metabolism is mediated by competition for binding to the DR1 motif and to the common coactivators, GRIP-1 and PGC-1alpha. *J. Biol. Chem.* **281**: 14537-14546.

Mietus-Snyder, M., Sladek, F.M., Ginsburg, G.S., Kuo, C.F., Ladias, J.A., Darnell, J.E. and Karathanasis, S.K. (1992) Antagonism between apolipoprotein AI regulatory protein 1, Ear3/COUP-TF, and hepatocyte nuclear factor 4 modulates apolipoprotein CIII gene expression in liver and intestinal cells. *Mol. Cell. Biol.* **12**: 1708-1718.

Milutinovic, S., D'Alessio, A.C., Detich, N. and Szyf, M. (2007) Valproate induces widespread epigenetic reprogramming which involves demethylation of specific genes. *Carcinogenesis* **28**: 560-571.

Minami, H., Sai, K., Saeki, M., Saito, Y., Ozawa, S., Suzuki, K., Kaniwa, N., Sawada, J., Hamaguchi, T., Yamamoto, N., Shirao, K., Yamada, Y., Ohmatsu, H., Kubota, K., Yoshida, T., Ohtsu, A. and Saijo, N. (2007) Irinotecan pharmacokinetics/pharmacodynamics and UGT1A genetic polymorphisms in Japanese: roles of *UGT1A1*6* and **28*. *Pharmacogenet. Genomics* **17**: 497-504.

Minchin, R.F., Reeves, P.T., Teitel, C.H., McManus, M.E., Mojarrabi, B., Ilett, K.F. and Kadlubar, F.F. (1992) *N*-and *O*-acetylation of aromatic and heterocyclic amine carcinogens by human monomorphic and polymorphic acetyltransferases expressed in COS-1 cells. *Biochem. Biophys. Res. Commun.* **185**: 839-844.

Miners, J.O., Attwood, J. and Birkett, D.J. (1983) Influence of sex and oral contraceptive steroids on paracetamol metabolism. *Br. J. Clin. Pharmacol.* **16**: 503-509.

Miners, J.O., Attwood, J. and Birkett, D.J. (1984a) Determinants of acetaminophen metabolism: effect of inducers and inhibitors of drug metabolism on acetaminophen's metabolic pathways. *Clin. Pharmacol. Ther*.**35**: 480-486.

Miners, J.O. and Day, R.O. (2007) Relationships between the adverse effects of drugs and genetic polymorphism in genes encoding drug metabolizing enzymes. *Br. J. Clin. Pharmacol.* **63**: 380-381.

Miners, J.O. and Mackenzie, P.I. (1991) Drug glucuronidation in humans. *Pharmacol. Ther.* **51**: 347-369.

Miners, J.O., Robson, R.A. and Birkett, D.J. (1984b) Gender and oral contraceptive steroids as determinants of drug glucuronidation: effects on clofibric acid elimination. *Br. J. Clin. Pharmacol.* **18**: 240-243.

Miners, J.O., Robson, R.A. and Birkett, D.J. (1986) Paracetamol metabolism in pregnancy. *Br. J. Clin. Pharmacol.* 22: 359-362.

Mischoulon, D., Rana, B., Bucher, N.L. and Farmer, S.R. (1992) Growth-dependent inhibition of CCAAT enhancer-binding protein (C/EBPalpha) gene expression during hepatocyte proliferation in the regenerating liver and in culture. *Mol. Cell. Biol.* **12**: 2553-2560.

Mitchelmore, C., Troelsen, J.T., Spodsberg, N., Sjostrom, H. and Noren, O. (2000) Interaction between the homeodomain proteins Cdx2 and HNF1alpha mediates expression of the lactase-phlorizin hydrolase gene. *Biochem. J.* **346**: 529-535.

Miura, N. and Tanaka, K. (1993) Analysis of the rat hepatocyte nuclear factor (HNF) 1 gene promoter: synergistic activation by HNF4 and HNF1 proteins. *Nucleic Acids Res.* **21**: 3731-3736.

Miyagi, S.J. and Collier, A.C. (2007) Pediatric development of glucuronidation: the ontogeny of hepatic UGT1A4. *Drug Metab. Dispos.* **35**: 1587-1592.

Mogal, A. and Abdulkadir, S.A. (2006) Effects of Histone Deacetylase Inhibitor (HDACi); Trichostatin-A (TSA) on the expression of housekeeping genes. *Mol. Cell. Probes* **20**: 81-86.

Mojarrabi, B., Butler, R. and Mackenzie, P.I. (1996) cDNA cloning and characterization of the human UDP glucuronosyltransferase, UGT1A3. *Biochem. Biophys. Res. Commun.* **225**: 785-790.

Mojarrabi, B. and Mackenzie, P.I. (1998) Characterization of two UDP glucuronosyltransferases that are predominantly expressed in human colon. *Biochem. Biophys. Res. Commun.* **247**: 704-709.

Moore, L.B., Goodwin, B., Jones, S.A., Wisely, G.B., Serabjit-Singh, C.J., Willson, T.M., Collins, J.L. and Kliewer, S.A. (2000) St. John's wort induces hepatic drug

metabolism through activation of the pregnane X receptor. *Proc. Natl. Acad. Sci. U. S. A.* **97**: 7500-7502.

Morgan, D.J. (1997) Drug disposition in mother and foetus. *Clin. Exp. Pharamcol. Physiol.* **24**: 869-873.

Morissette, P., Albert, C., Busque, S., St-Louis, G. and Vinet, B. (2001) In vivo higher glucuronidation of mycophenolic acid in male than in female recipients of a cadaveric kidney allograft and under immunosuppressive therapy with mycophenolate mofetil. *Therapeutic Drug Monitoring* **23**: 520-525.

Morley, M., Molony, C.M., Weber, T.M., Devlin, J.L., Ewens, K.G., Spielman, R.S. and Cheung, V.G. (2004) Genetic analysis of genome-wide variation in human gene expression. *Nature* **430**: 743-747.

Mouchel, N., Henstra, S.A., McCarthy, V.A., Williams, S.H., Phylactides, M. and Harris, A. (2004) HNF1alpha is involved in tissue-specific regulation of *CFTR* gene expression. *Biochem. J.* **378**: 909-918.

Mucklow, J.C., Fraser, H.S., Bulpitt, C.J., Kahn, C., Mould, G. and Dollery, C.T. (1980) Environmental factors affecting paracetamol metabolism in London factory and office workers. *Br. J. Clin. Pharmacol.* **10**: 67-74.

Mueller, C.R., Maire, P. and Schibler, U. (1990) DBP, a liver-enriched transcriptional activator, is expressed late in ontogeny and its tissue specificity is determined posttranscriptionally. *Cell* **61**: 279-291.

Munzel, P.A., Bookjans, G., Mehner, G., Lehmkoster, T. and Bock, K.W. (1996) Tissue-specific 2,3,7,8-tetrachlorodibenzo-*p*-dioxin-inducible expression of human UDP-glucuronosyltransferase UGT1A6. *Arch. Biochem. Biophys.* **335**: 205-210.

Munzel, P.A., Lehmkoster, T., Bruck, M., Ritter, J.K. and Bock, K.W. (1998) Aryl hydrocarbon receptor-inducible or constitutive expression of human UDP glucuronosyltransferase UGT1A6. *Arch. Biochem. Biophys.* **350**: 72-78.

Munzel, P.A., Schmohl, S., Buckler, F., Jaehrling, J., Raschko, F.T., Kohle, C. and Bock, K.W. (2003) Contribution of the Ah receptor to the phenolic antioxidantmediated expression of human and rat UDP-glucuronosyltransferase UGT1A6 in Caco-2 and rat hepatoma 5L cells. *Biochem. Pharmacol.* **66**: 841-847.

Munzel, P.A., Schmohl, S., Heel, H., Kalberer, K., Bock-Hennig, B.S. and Bock, K.W. (1999) Induction of human UDP glucuronosyltransferases (UGT1A6, UGT1A9, and UGT2B7) by *t*-butylhydroquinone and 2,3,7,8-tetrachlorodibenzo-*p*-dioxin in Caco-2 cells. *Drug Metab. Dispos.* **27**: 569-573.

Murao, K., Bassyouni, H., Taylor, A.H., Wanke, I.E. and Wong, N.C. (1997) Hepatocyte nuclear factor 4 inhibits the activity of site A from the rat apolipoprotein AI gene. *Biochemistry* **36**: 301-306.

Nacer-Cherif, H., Bois-Joyeux, B., Rousseau, G.G., Lemaigre, F.P. and Danan, J.L. (2003) Hepatocyte nuclear factor-6 stimulates transcription of the alpha-fetoprotein gene and synergizes with the retinoic-acid-receptor-related orphan receptor alpha-4. *Biochem. J.* **369**: 583-591.

Naganuma, M., Saruwatari, A., Okamura, S. and Tamura, H. (2006) Turmeric and curcumin modulate the conjugation of 1-naphthol in Caco-2 cells. *Biol. Pharm. Bull.* **29**: 1476-1479.

Naiki, T., Nagaki, M., Shidoji, Y., Kojima, H. and Moriwaki, H. (2004) Functional activity of human hepatoma cells transfected with adenovirus-mediated hepatocyte nuclear factor (HNF)-4 gene. *Cell Transplant.* **13**: 393-403.

Nakajima, M., Tanaka, E., Kobayashi, T., Ohashi, N., Kume, T. and Yokoi, T. (2002) Imipramine *N*-glucuronidation in human liver microsomes: biphasic kinetics and characterization of UDP-glucuronosyltransferase isoforms. *Drug Metab. Dispos.* **30**: 636-642.

Nakajima, M. and Yokoi, T. (2005) Interindividual variability in nicotine metabolism: *C*-oxidation and glucuronidation. *Drug Metab. Pharmacokinet.* **20**: 227-235.

Nakhei, H., Lingott, A., Lemm, I. and Ryffel, G.U. (1998) An alternative splice variant of the tissue specific transcription factor HNF4alpha predominates in undifferentiated murine cell types. *Nucleic Acids Res.* **26**: 497-504.

Nakshatri, H. and Chambon, P. (1994) The directly repeated RG(G/T)TCA motifs of the rat and mouse cellular retinol-binding protein II genes are promiscuous binding sites for RAR, RXR, HNF-4, and ARP-1 homo- and heterodimers. *J. Biol. Chem.* **269**: 890-902.

Navalon-Garcia, K., Mendoza-Alcantar, L., Diaz-Vargas, M.E., Martinez-Godinez, M.A., Reyna-Garfias, H., Aguilar-Salinas, C.A., Riba, L., Canizales-Quinteros, S., Villarreal-Molina, T., Gonzalez-Chavez, A., Argueta-Villamar, V., Tusie-Luna, M.T. and Miliar-Garcia, A. (2006) HNF-1alpha G574S is a functional variant with decreased transactivation activity. *Diabet. Med.* **23**: 1295-1300.

Navas, M.A., Munoz-Elias, E.J., Kim, J., Shih, D. and Stoffel, M. (1999) Functional characterization of the MODY1 gene mutations HNF4(R127W), HNF4(V255M), and HNF4(E276Q). *Diabetes* **48**: 1459-1465.

Nebert, D.W. (1994) Drug-metabolizing enzymes in ligand-modulated transcription. *Biochem. Pharmacol.* **47**: 25-37.

Ng, H.H. and Bird, A. (2000) Histone deacetylases: silencers for hire. *Trends Biochem. Sci.* 25: 121-126.

Nishiyama, C., Hi, R., Osada, S. and Osumi, T. (1998) Functional interactions between nuclear receptors recognizing a common sequence element, the direct repeat motif spaced by one nucleotide (DR-1). *J. Biochem.* **123**: 1174-1179.

Nishiyama, T., Ogura, K., Nakano, H., Ohnuma, T., Kaku, T., Hiratsuka, A., Muro, K. and Watabe, T. (2002) Reverse geometrical selectivity in glucuronidation and sulfation of *cis-* and *trans-*4-hydroxytamoxifens by human liver UDP-glucuronosyltransferases and sulfotransferases. *Biochem. Pharmacol.* **63**: 1817-1830.

Nitsch, D., Boshart, M. and Schutz, G. (1993) Activation of the tyrosine aminotransferase gene is dependent on synergy between liver-specific and hormone-responsive elements. *Proc. Natl. Acad. Sci. U. S. A.* **90**: 5479-5483.

Nowell, S.A., Massengill, J.S., Williams, S., Radominska-Pandya, A., Tephly, T.R., Cheng, Z., Strassburg, C.P., Tukey, R.H., MacLeod, S.L., Lang, N.P. and Kadlubar, F.F. (1999) Glucuronidation of 2-hydroxyamino-1-methyl-6-phenylimidazo[4,5b]pyridine by human microsomal UDP-glucuronosyltransferases: identification of specific UGT1A family isoforms involved. *Carcinogenesis* **20**: 1107-1114. Oberkofler, H., Schraml, E., Krempler, F. and Patsch, W. (2004) Restoration of sterol-regulatory-element-binding protein-1c gene expression in HepG2 cells by peroxisome-proliferator-activated receptor-gamma co-activator-1alpha. *Biochem. J.* **381**: 357-363.

Ockenga, J., Vogel, A., Teich, N., Keim, V., Manns, M.P. and Strassburg, C.P. (2003) UDP glucuronosyltransferase (UGT1A7) gene polymorphisms increase the risk of chronic pancreatitis and pancreatic cancer. *Gastroenterology* **124**: 1802-1808.

Ocker, M., Alajati, A., Ganslmayer, M., Zopf, S., Luders, M., Neureiter, D., Hahn, E.G., Schuppan, D. and Herold, C. (2005) The histone-deacetylase inhibitor SAHA potentiates proapoptotic effects of 5-fluorouracil and irinotecan in hepatoma cells. *J. Cancer Res. Clin. Oncol.* **131**: 385-394.

Odom, D.T., Zizlsperger, N., Gordon, D.B., Bell, G.W., Rinaldi, N.J., Murray, H.L., Volkert, T.L., Schreiber, J., Rolfe, P.A., Gifford, D.K., Fraenkel, E., Bell, G.I. and Young, R.A. (2004) Control of pancreas and liver gene expression by HNF transcription factors. *Science* **303**: 1378-1381.

Ogryzko, V.V., Schiltz, R.L., Russanova, V., Howard, B.H. and Nakatani, Y. (1996) The transcriptional coactivators p300 and CBP are histone acetyltransferases. *Cell* **87**: 953-959.

Oguri, T., Takahashi, T., Miyazaki, M., Isobe, T., Kohno, N., Mackenzie, P.I. and Fujiwara, Y. (2004) UGT1A10 is responsible for SN-38 glucuronidation and its expression in human lung cancers. *Anticancer Res.* **24**: 2893-2896.

Ohyama, K., Maki, S., Sato, K. and Kato, Y. (2004) *In vitro* metabolism of [14C]methoxychlor in rat, mouse, Japanese quail and rainbow trout in precision-cut liver slices. *Xenobiotica* **34**: 741-754.

Okugi, H., Nakazato, H., Matsui, H., Ohtake, N., Nakata, S. and Suzuki, K. (2006) Association of the polymorphisms of genes involved in androgen metabolism and signaling pathways with familial prostate cancer risk in a Japanese population. *Cancer Detect. Prev.* **30**: 262-268.

Onishi, S., Kawade, N., Itoh, S., Isobe, K. and Sugiyama, S. (1979) Postnatal development of uridine diphosphate glucuronyltransferase activity towards bilirubin and 2-aminophenol in human liver. *Biochem. J.* **184**: 705-707.

Ordovas, J.M. and Mooser, V. (2004) Nutrigenomics and nutrigenetics. *Curr. Opin. Lipidol.* **15**: 101-108.

Osada, S., Yamamoto, H., Nishihara, T. and Imagawa, M. (1996) DNA binding specificity of the CCAAT/enhancer-binding protein transcription factor family. *J. Biol. Chem.* **271**: 3891-3896.

Osborne, R., Thompson, P., Joel, S., Trew, D., Patel, N. and Slevin, M. (1992) The analgesic activity of morphine-6-glucuronide. *Br. J. Clin. Pharmacol.* **34**: 130-138.

Oscarson, M., Zanger, U.M., Rifki, O.F., Klein, K., Eichelbaum, M. and Meyer, U.A. (2006) Transcriptional profiling of genes induced in the livers of patients treated with carbamazepine. *Clin. Pharmacol. Ther*.**80**: 440-456.

Osoegawa, K., Mammoser, A.G., Wu, C., Frengen, E., Zeng, C., Catanese, J.J. and de Jong, P.J. (2001) A bacterial artificial chromosome library for sequencing the complete human genome. *Genome Res.* **11**: 483-496.

Ossipow, V., Descombes, P. and Schibler, U. (1993) CCAAT/enhancer-binding protein mRNA is translated into multiple proteins with different transcription activation potentials. *Proc. Natl. Acad. Sci. U. S. A.* **90**: 8219-8223.

Ourlin, J.C., Lasserre, F., Pineau, T., Fabre, J.M., Sa-Cunha, A., Maurel, P., Vilarem, M.J. and Pascussi, J.M. (2003) The small heterodimer partner interacts with the pregnane X receptor and represses its transcriptional activity. *Mol. Endocrinol.* **17**: 1693-1703.

Overbergh, L., Valckx, D., Waer, M. and Mathieu, C. (1999) Quantification of murine cytokine mRNAs using real time quantitative reverse transcriptase PCR. *Cytokine* **11**: 305-312.

Overdier, D.G., Porcella, A. and Costa, R.H. (1994) The DNA-binding specificity of the hepatocyte nuclear factor 3/forkhead domain is influenced by amino-acid residues adjacent to the recognition helix. *Mol. Cell. Biol.* **14**: 2755-2766.

Ozeki, T., Takahashi, Y., Kume, T., Nakayama, K., Yokoi, T., Nunoya, K., Hara, A. and Kamataki, T. (2001) Co-operative regulation of the transcription of human dihydrodiol dehydrogenase (*DD*)4/aldo-keto reductase (*AKR*)1C4 gene by hepatocyte nuclear factor (HNF)-4alpha/gamma and HNF-1alpha. *Biochem. J.* **355**: 537-544.

Ozeki, T., Takahashi, Y., Nakayama, K., Funayama, M., Nagashima, K., Kodama, T. and Kamataki, T. (2003) Hepatocyte nuclear factor-4 alpha/gamma and hepatocyte nuclear factor-1 alpha as causal factors of interindividual difference in the expression of human dihydrodiol dehydrogenase 4 mRNA in human livers. *Pharmacogenetics* **13**: 49-53.

Pacifici, G.M., Sawe, J., Kager, L. and Rane, A. (1982) Morphine glucuronidation in human fetal and adult liver. *Eur. J. Clin. Pharmacol.* **22**: 553-558.

Page, R.D. (1996) TreeView: an application to display phylogenetic trees on personal computers. *Comput. Appl. Biosci.* **12**: 357-358.

Pani, L., Quian, X.B., Clevidence, D. and Costa, R.H. (1992) The restricted promoter activity of the liver transcription factor hepatocyte nuclear factor 3beta involves a cell-specific factor and positive autoactivation. *Mol. Cell. Biol.* **12**: 552-562.

Pantuck, E.J., Pantuck, C.B., Anderson, K.E., Wattenberg, L.W., Conney, A.H. and Kappas, A. (1984) Effect of brussels sprouts and cabbage on drug conjugation. *Clin. Pharmacol. Ther.***35**: 161-169.

Pantuck, E.J., Pantuck, C.B., Kappas, A., Conney, A.H. and Anderson, K.E. (1991) Effects of protein and carbohydrate content of diet on drug conjugation. *Clin. Pharmacol. Ther*.**50**: 254-258.

Park, I.N., Cho, I.J. and Kim, S.G. (2004a) Ceramide negatively regulates glutathione S-transferase gene transactivation via repression of hepatic nuclear factor-1 that is degraded by the ubiquitin proteasome system. *Mol. Pharmacol.* **65**: 1475-1484.

Park, I.N., Cho, I.J. and Kim, S.G. (2004b) Ceramide, an apoptotic rheostat, inhibits CCAAT/enhancer binding protein-beta and NF-E2-related factor-2 activation: the role in glutathione S-transferase A2 gene repression. *Drug Metab. Dispos.* **32**: 893-897.

Park, J., Chen, L., Ratnashinge, L., Sellers, T.A., Tanner, J.P., Lee, J.H., Dossett, N., Lang, N., Kadlubar, F.F., Ambrosone, C.B., Zachariah, B., Heysek, R.V., Patterson, S. and Pow-Sang, J. (2006) Deletion polymorphism of UDP-glucuronosyltransferase 2B17 and risk of prostate cancer in African American and Caucasian men. *Cancer Epidemiol. Biomarkers Prev.* **15**: 1473-1478.

Park, J., Chen, L., Shade, K., Lazarus, P., Seigne, J., Patterson, S., Helal, M. and Pow-Sang, J. (2004c) Asp85tyr polymorphism in the udp-glucuronosyltransferase (UGT) 2B15 gene and the risk of prostate cancer. *J. Urol.* **171**: 2484-2488.

Parrizas, M., Maestro, M.A., Boj, S.F., Paniagua, A., Casamitjana, R., Gomis, R., Rivera, F. and Ferrer, J. (2001) Hepatic nuclear factor 1-alpha directs nucleosomal hyperacetylation to its tissue-specific transcriptional targets. *Mol. Cell. Biol.* **21**: 3234-3243.

Pascussi, J.M., Jounaidi, Y., Drocourt, L., Domergue, J., Balabaud, C., Maurel, P. and Vilarem, M.J. (1999) Evidence for the presence of a functional pregnane X receptor response element in the CYP3A7 promoter gene. *Biochem. Biophys. Res. Commun.* **260**: 377-381.

Patrinos, G.P., de Krom, M., de Boer, E., Langeveld, A., Imam, A.M., Strouboulis, J., de Laat, W. and Grosveld, F.G. (2004) Multiple interactions between regulatory regions are required to stabilize an active chromatin hub. *Genes Dev.* **18**: 1495-1509.

Paul, N.W. and Roses, A.D. (2003) Pharmacogenetics and pharmacogenomics: recent developments, their clinical relevance and some ethical, social, and legal implications. *J. Mol. Med.* **81**: 135-140.

Pennell, P.B., Newport, D.J., Stowe, Z.N., Helmers, S.L., Montgomery, J.Q. and Henry, T.R. (2004) The impact of pregnancy and childbirth on the metabolism of lamotrigine. *Neurology* **62**: 292-295.

Perera, F.P. (2000) Molecular epidemiology: on the path to prevention? J. Natl. Cancer Inst. 92: 602-612.

Perera, F.P. and Weinstein, I.B. (2000) Molecular epidemiology: recent advances and future directions. *Carcinogenesis* **21**: 517-524.

Peterson, R.S., Clevidence, D.E., Ye, H. and Costa, R.H. (1997) Hepatocyte nuclear factor-3alpha promoter regulation involves recognition by cell-specific factors, thyroid transcription factor-1, and autoactivation. *Cell Growth Differ.* **8**: 69-82.

Peterson, S., Bigler, J., Horner, N.K., Potter, J.D. and Lampe, J.W. (2005) *Cruciferae* interact with the *UGT1A1*28* polymorphism to determine serum bilirubin levels in humans. *J. Nutr.* **135**: 1051-1055.

Petrescu, A.D., Hertz, R., Bar-Tana, J., Schroeder, F. and Kier, A.B. (2002) Ligand specificity and conformational dependence of the hepatic nuclear factor-4alpha (HNF-4alpha). *J. Biol. Chem.* **277**: 23988-23999.

Petrescu, A.D., Payne, H.R., Boedecker, A., Chao, H., Hertz, R., Bar-Tana, J., Schroeder, F. and Kier, A.B. (2003) Physical and functional interaction of Acyl-CoA-binding protein with hepatocyte nuclear factor-4 alpha. *J. Biol. Chem.* **278**: 51813-51824.

Pharoah, P.D., Dunning, A.M., Ponder, B.A. and Easton, D.F. (2004) Association studies for finding cancer-susceptibility genetic variants. *Nat. Rev. Cancer* **4**: 850-860.

Phiel, C.J., Zhang, F., Huang, E.Y., Guenther, M.G., Lazar, M.A. and Klein, P.S. (2001) Histone deacetylase is a direct target of valproic acid, a potent anticonvulsant, mood stabilizer, and teratogen. *J. Biol. Chem.* **276**: 36734-36741.

Picard, N., Ratanasavanh, D., Premaud, A., Le Meur, Y. and Marquet, P. (2005) Identification of the UDP-glucuronosyltransferase isoforms involved in mycophenolic acid phase II metabolism. *Drug Metab. Dispos.* **33**: 139-146.

Piepoli, A., Gentile, A., Valvano, M.R., Barana, D., Oliani, C., Cotugno, R., Quitadamo, M., Andriulli, A. and Perri, F. (2006) Lack of association between *UGT1A7*, *UGT1A9*, *ARP*, *SPINK1* and *CFTR* gene polymorphisms and pancreatic cancer in Italian patients. *World J. Gastroenterol.* **12**: 6343-6348.

Pineda-Torra, I., Jamshidi, Y., Flavell, D.M., Fruchart, J.C. and Staels, B. (2002) Characterization of the human PPARalpha promoter: identification of a functional nuclear receptor response element. *Mol. Endocrinol.* **16**: 1013-1028.

Pitarque, M., Rodriguez-Antona, C., Oscarson, M. and Ingelman-Sundberg, M. (2005) Transcriptional regulation of the human *CYP2A6* gene. *J. Pharmacol. Exp. Ther.* **313**: 814-822.

Plengvidhya, N., Antonellis, A., Wogan, L.T., Poleev, A., Borgschulze, M., Warram, J.H., Ryffel, G.U., Krolewski, A.S. and Doria, A. (1999) Hepatocyte nuclear factor-4gamma. cDNA sequence, gene organization, and mutation screening in early-onset autosomal-dominant type 2 diabetes. *Diabetes* **48**: 2099-2102.

Poll, A.V., Pierreux, C.E., Lokmane, L., Haumaitre, C., Achouri, Y., Jacquemin, P., Rousseau, G.G., Cereghini, S. and Lemaigre, F.P. (2006) A vHNF1/TCF2-HNF6 cascade regulates the transcription factor network that controls generation of pancreatic precursor cells. *Diabetes* **55**: 61-69.

Pollak, E.S., Hung, H.L., Godin, W., Overton, G.C. and High, K.A. (1996) Functional characterization of the human factor VII 5'-flanking region. *J. Biol. Chem.* **271**: 1738-1747.

Pontoglio, M., Barra, J., Hadchouel, M., Doyen, A., Kress, C., Bach, J.P., Babinet, C. and Yaniv, M. (1996) Hepatocyte nuclear factor 1 inactivation results in hepatic dysfunction, phenylketonuria, and renal Fanconi syndrome. *Cell* **84**: 575-585.

Pontoglio, M., Faust, D.M., Doyen, A., Yaniv, M. and Weiss, M.C. (1997) Hepatocyte nuclear factor 1alpha gene inactivation impairs chromatin remodeling and demethylation of the phenylalanine hydroxylase gene. *Mol. Cell. Biol.* **17**: 4948-4956.

Popowski, K., Eloranta, J.J., Saborowski, M., Fried, M., Meier, P.J. and Kullak-Ublick, G.A. (2005) The human organic anion transporter 2 gene is transactivated by hepatocyte nuclear factor-4 alpha and suppressed by bile acids. *Mol. Pharmacol.* **67**: 1629-1638.

Puigserver, P. and Spiegelman, B.M. (2003) Peroxisome proliferator-activated receptor-gamma coactivator 1 alpha (PGC-1 alpha): transcriptional coactivator and metabolic regulator. *Endocr. Rev.* **24**: 78-90.

Qadri, I., Iwahashi, M., Kullak-Ublick, G.A. and Simon, F.R. (2006) Hepatocyte nuclear factor (HNF) 1 and HNF4 mediate hepatic multidrug resistance protein 2 upregulation during hepatitis C virus gene expression. *Mol. Pharmacol.* **70**: 627-636.

Qi, H. and Ratnam, M. (2006) Synergistic induction of folate receptor beta by all*trans* retinoic acid and histone deacetylase inhibitors in acute myelogenous leukemia cells: mechanism and utility in enhancing selective growth inhibition by antifolates. *Cancer Res.* **66**: 5875-5882.

Qian, X., Samadani, U., Porcella, A. and Costa, R.H. (1995) Decreased expression of hepatocyte nuclear factor 3 alpha during the acute-phase response influences transthyretin gene transcription. *Mol. Cell. Biol.* **15**: 1364-1376.

Radominska-Pandya, A., Czernik, P.J., Little, J.M., Battaglia, E. and Mackenzie, P.I. (1999) Structural and functional studies of UDP-glucuronosyltransferases. *Drug Metab. Rev.* **31**: 817-899.

Rae, J.M., Johnson, M.D., Lippman, M.E. and Flockhart, D.A. (2001) Rifampin is a selective, pleiotropic inducer of drug metabolism genes in human hepatocytes: studies with cDNA and oligonucleotide expression arrays. *J. Pharmacol. Exp. Ther.* **299**: 849-857.

Ramirez, J., Iyer, L., Journault, K., Belanger, P., Innocenti, F., Ratain, M.J. and Guillemette, C. (2002) *In vitro* characterization of hepatic flavopiridol metabolism using human liver microsomes and recombinant UGT enzymes. *Pharm. Res.* **19**: 588-594.

Ramji, D.P. and Foka, P. (2002) CCAAT/enhancer-binding proteins: structure, function and regulation. *Biochem. J.* **365**: 561-575.

Rastegar, M., Rousseau, G.G. and Lemaigre, F.P. (2000) CCAAT/enhancer-binding protein-alpha is a component of the growth hormone-regulated network of liver transcription factors. *Endocrinology* **141**: 1686-1692.

Rausa, F., Samadani, U., Ye, H., Lim, L., Fletcher, C.F., Jenkins, N.A., Copeland, N.G. and Costa, R.H. (1997) The cut-homeodomain transcriptional activator HNF-6 is coexpressed with its target gene HNF-3 beta in the developing murine liver and pancreas. *Dev. Biol.* **192**: 228-246.

Rausa, F.M., Hughes, D.E. and Costa, R.H. (2004) Stability of the hepatocyte nuclear factor 6 transcription factor requires acetylation by the CREB-binding protein coactivator. *J. Biol. Chem.* **279**: 43070-43076.

Rausa, F.M., Tan, Y. and Costa, R.H. (2003) Association between hepatocyte nuclear factor 6 (HNF-6) and FoxA2 DNA binding domains stimulates FoxA2 transcriptional activity but inhibits HNF-6 DNA binding. *Mol. Cell. Biol.* **23**: 437-449.

Raymond, E., Fabbro, M., Boige, V., Rixe, O., Frenay, M., Vassal, G., Faivre, S., Sicard, E., Germa, C., Rodier, J.M., Vernillet, L. and Armand, J.P. (2003) Multicentre phase II study and pharmacokinetic analysis of irinotecan in chemotherapy-naive patients with glioblastoma. *Ann. Oncol.* **14**: 603-614.

Reijnen, M.J., Sladek, F.M., Bertina, R.M. and Reitsma, P.H. (1992) Disruption of a binding site for hepatocyte nuclear factor 4 results in hemophilia B Leyden. *Proc. Natl. Acad. Sci. U. S. A.* **89**: 6300-6303.

Ren, Q., Murphy, S.E., Zheng, Z. and Lazarus, P. (2000) *O*-Glucuronidation of the lung carcinogen 4-(methylnitrosamino)-1- (3-pyridyl)-1-butanol (NNAL) by human UDP-glucuronosyltransferases 2B7 and 1A9. *Drug Metab. Dispos.* **28**: 1352-1360.

Rey-Campos, J., Chouard, T., Yaniv, M. and Cereghini, S. (1991) vHNF1 is a homeoprotein that activates transcription and forms heterodimers with HNF1. *EMBO J.* **10**: 1445-1457.

Rhee, K.H., Stier, G., Becker, P.B., Suck, D. and Sandaltzopoulos, R. (1997) The bifunctional protein DCoH modulates interactions of the homeodomain transcription factor HNF1 with nucleic acids. *J. Mol. Biol.* **265**: 20-29.

Richardson, T.A., Sherman, M., Kalman, D. and Morgan, E.T. (2006) Expression of UDP-glucuronosyltransferase isoform mRNAs during inflammation and infection in mouse liver and kidney. *Drug Metab. Dispos.* **34**: 351-353.

Ring, J.A., Ghabrial, H., Ching, M.S., Smallwood, R.A. and Morgan, D.J. (1999) Fetal hepatic drug elimination. *Pharmacol. Ther.* **84**: 429-445.

Ringeisen, F., Rey-Campos, J. and Yaniv, M. (1993) The transactivation potential of variant hepatocyte nuclear factor 1 is modified by alternative splicing. *J. Biol. Chem.* **268**: 25706-25711.

Ritter, J.K. (2000) Roles of glucuronidation and UDP-glucuronosyltransferases in xenobiotic bioactivation reactions. *Chem. Biol. Interact.* **129**: 171-193.

Ritter, J.K., Chen, F., Sheen, Y.Y., Lubet, R.A. and Owens, I.S. (1992a) Two human liver cDNAs encode UDP-glucuronosyltransferases with 2 log differences in activity toward parallel substrates including hyodeoxycholic acid and certain estrogen derivatives. *Biochemistry* **31**: 3409-3414.

Ritter, J.K., Chen, F., Sheen, Y.Y., Tran, H.M., Kimura, S., Yeatman, M.T. and Owens, I.S. (1992b) A novel complex locus *UGT1* encodes human bilirubin, phenol, and other UDP-glucuronosyltransferase isozymes with identical carboxyl termini. *J. Biol. Chem.* **267**: 3257-3261.

Ritter, J.K., Kessler, F.K., Thompson, M.T., Grove, A.D., Auyeung, D.J. and Fisher, R.A. (1999) Expression and inducibility of the human bilirubin UDP-glucuronosyltransferase UGT1A1 in liver and cultured primary hepatocytes: evidence for both genetic and environmental influences. *Hepatology* **30**: 476-484.

Robertson, A.F. (2003) Reflections on errors in neonatology: II. The "Heroic" years, 1950 to 1970. *J. Perinatol.* **23**: 154-161.

Rodriguez, J.C., Ortiz, J.A., Hegardt, F.G. and Haro, D. (1998) The hepatocyte nuclear factor 4 (HNF-4) represses the mitochondrial HMG-CoA synthase gene. *Biochem. Biophys. Res. Commun.* **242**: 692-696.

Rodriguez-Antona, C., Bort, R., Jover, R., Tindberg, N., Ingelman-Sundberg, M., Gomez-Lechon, M.J. and Castell, J.V. (2003) Transcriptional regulation of human CYP3A4 basal expression by CCAAT enhancer-binding protein alpha and hepatocyte nuclear factor-3 gamma. *Mol. Pharmacol.* **63**: 1180-1189.

Rodriguez-Antona, C., Donato, M.T., Boobis, A., Edwards, R.J., Watts, P.S., Castell, J.V. and Gomez-Lechon, M.J. (2002) Cytochrome P450 expression in human hepatocytes and hepatoma cell lines: molecular mechanisms that determine lower expression in cultured cells. *Xenobiotica* **32**: 505-520.

Roe, A.L., Poloyac, S.M., Howard, G., Shedlofsky, S.I. and Blouin, R.A. (2001) The effect of endotoxin on hepatocyte nuclear factor 1 nuclear protein binding: potential implications on CYP2E1 expression in the rat. *J. Pharm. Pharmacol.* **53**: 1365-1371.

Rollini, P., Xu, L. and Fournier, R.E. (1999) Partial activation of gene activity and chromatin remodeling of the human 14q32.1 serpin gene cluster by HNF-1alpha and HNF-4 in fibroblast microcell hybrids. *Somat. Cell Mol. Genet.* **25**: 207-221.

Roman, C., Platero, J.S., Shuman, J. and Calame, K. (1990) Ig/EBP-1: a ubiquitously expressed immunoglobulin enhancer binding protein that is similar to C/EBP and heterodimerizes with C/EBP. *Genes Dev.* **4**: 1404-1415.

Ron, D. and Habener, J.F. (1992) CHOP, a novel developmentally regulated nuclear protein that dimerizes with transcription factors C/EBP and LAP and functions as a dominant-negative inhibitor of gene transcription. *Genes Dev.* **6**: 439-453.

Ross, J.S. and Ginsburg, G.S. (2002) Integration of molecular diagnostics with therapeutics: implications for drug discovery and patient care. *Expert Rev. Mol. Diagn.* **2**: 531-541.

Ross, J.S., Schenkein, D.P., Pietrusko, R., Rolfe, M., Linette, G.P., Stec, J., Stagliano, N.E., Ginsburg, G.S., Symmans, W.F., Pusztai, L. and Hortobagyi, G.N. (2004) Targeted therapies for cancer 2004. *Am. J. Clin. Pathol.* **122**: 598-609.

Rouet, P., Raguenez, G., Tronche, F., Mfou'ou, V. and Salier, J.P. (1995) Hierarchy and positive/negative interplays of the hepatocyte nuclear factors HNF-1, -3 and -4 in the liver-specific enhancer for the human alpha-1-microglobulin/bikunin precursor. *Nucleic Acids Res.* **23**: 395-404.

Rowland, A., Elliot, D.J., Williams, J.A., Mackenzie, P.I., Dickinson, R.G. and Miners, J.O. (2006) In vitro characterization of lamotrigine *N*2-glucuronidation and the lamotrigine-valproic acid interaction. *Drug Metab. Dispos.* **34**: 1055-1062.

Rowley, C.W., Staloch, L.J., Divine, J.K., McCaul, S.P. and Simon, T.C. (2006) Mechanisms of mutual functional interactions between HNF-4alpha and HNF-1alpha revealed by mutations that cause maturity onset diabetes of the young. *Am. J. Physiol. Gastrointest. Liver Physiol.* **290**: G466-475.

Rubins, N.E., Friedman, J.R., Le, P.P., Zhang, L., Brestelli, J. and Kaestner, K.H. (2005) Transcriptional networks in the liver: hepatocyte nuclear factor 6 function is largely independent of Foxa2. *Mol. Cell. Biol.* **25**: 7069-7077.

Ruse, M.D., Privalsky, M.L. and Sladek, F.M. (2002) Competitive cofactor recruitment by orphan receptor hepatocyte nuclear factor 4alpha1: modulation by the F domain. *Mol. Cell. Biol.* **22**: 1626-1638.

Ryffel, G.U. (2001) Mutations in the human genes encoding the transcription factors of the hepatocyte nuclear factor (HNF)1 and HNF4 families: functional and pathological consequences. *J. Mol. Endocrinol.* **27**: 11-29.

Sabers, A., Ohman, I., Christensen, J. and Tomson, T. (2003) Oral contraceptives reduce lamotrigine plasma levels. *Neurology* **61**: 570-571.

Sabolovic, N., Heydel, J.M., Li, X., Little, J.M., Humbert, A.C., Radominska-Pandya, A. and Magdalou, J. (2004) Carboxyl nonsteroidal anti-inflammatory drugs are efficiently glucuronidated by microsomes of the human gastrointestinal tract. *Biochim. Biophys. Acta* **1675**: 120-129. Saborowski, M., Kullak-Ublick, G.A. and Eloranta, J.J. (2006) The human organic cation transporter-1 gene is transactivated by hepatocyte nuclear factor-4alpha. *J. Pharmacol. Exp. Ther.* **317**: 778-785.

Samadani, U. and Costa, R.H. (1996) The transcriptional activator hepatocyte nuclear factor 6 regulates liver gene expression. *Mol. Cell. Biol.* **16**: 6273-6284.

Samadani, U., Porcella, A., Pani, L., Johnson, P.F., Burch, J.B., Pine, R. and Costa, R.H. (1995) Cytokine regulation of the liver transcription factor hepatocyte nuclear factor-3beta is mediated by the C/EBP family and interferon regulatory factor 1. *Cell Growth Differ.* **6**: 879-890.

Sambrook, J. and Russell, D.W. (2001) Molecular cloning: A laboratory handbook. 3rd ed. Cold Spring Harbor Press, New York.

Samokyszyn, V.M., Gall, W.E., Zawada, G., Freyaldenhoven, M.A., Chen, G., Mackenzie, P.I., Tephly, T.R. and Radominska-Pandya, A. (2000) 4-hydroxyretinoic acid, a novel substrate for human liver microsomal UDP-glucuronosyltransferase(s) and recombinant UGT2B7. *J. Biol. Chem.* **275**: 6908-6914.

Sanchez Pozzi, E.J., Crocenzi, F.A., Pellegrino, J.M., Catania, V.A., Luquita, M.G., Roma, M.G., Rodriguez Garay, E.A. and Mottino, A.D. (2003) Ursodeoxycholate reduces ethinylestradiol glucuronidation in the rat: role in prevention of estrogen-induced cholestasis. *J. Pharmacol. Exp. Ther.* **306**: 279-286.

Sanchez, R.I., Mesia-Vela, S. and Kauffman, F.C. (2001) Challenges of cancer drug design: a drug metabolism perspective. *Curr. Cancer Drug Targets* **1**: 1-32.

Sandanaraj, E., Jada, S.R., Shu, X., Lim, R., Lee, S.C., Zhou, Q., Zhou, S., Goh, B.C. and Chowbay, B. (2007) Influence of UGT1A9 intronic I399C>T polymorphism on SN-38 glucuronidation in Asian cancer patients. *Pharmacogenomics J*.

Sawe, J., Kager, L., Svensson Eng, J.O. and Rane, A. (1985) Oral morphine in cancer patients: *in vivo* kinetics and *in vitro* hepatic glucuronidation. *Br. J. Clin. Pharmacol.* **19**: 495-501.

Scavone, J.M., Greenblatt, D.J., LeDuc, B.W., Blyden, G.T., Luna, B.G. and Harmatz, J.S. (1990) Differential effect of cigarette smoking on antipyrine oxidation versus acetaminophen conjugation. *Pharmacology* **40**: 77-84.

Schaeffer, E., Guillou, F., Part, D. and Zakin, M.M. (1993) A different combination of transcription factors modulates the expression of the human transferrin promoter in liver and Sertoli cells. *J. Biol. Chem.* **268**: 23399-23408.

Schmittgen, T.D. and Zakrajsek, B.A. (2000) Effect of experimental treatment on housekeeping gene expression: validation by real-time, quantitative RT-PCR. *J. Biochem. Biophys. Methods* **46**: 69-81.

Schreiber, E., Matthias, P., Muller, M.M. and Schaffner, W. (1989) Rapid detection of octamer binding proteins with 'mini-extracts', prepared from a small number of cells. *Nucleic Acids Res.* **17**: 6419.

Schrem, H., Klempnauer, J. and Borlak, J. (2002) Liver-enriched transcription factors in liver function and development. Part I: the hepatocyte nuclear factor network and liver-specific gene expression. *Pharmacol. Rev.* **54**: 129-158.

Schrem, H., Klempnauer, J. and Borlak, J. (2004) Liver-enriched transcription factors in liver function and development. Part II: the C/EBPs and D site-binding protein in cell cycle control, carcinogenesis, circadian gene regulation, liver regeneration, apoptosis, and liver-specific gene regulation. *Pharmacol. Rev.* **56**: 291-330.

Schroeder, F., Huang, H., Hostetler, H.A., Petrescu, A.D., Hertz, R., Bar-Tana, J. and Kier, A.B. (2005) Stability of fatty acyl-coenzyme A thioester ligands of hepatocyte nuclear factor-4alpha and peroxisome proliferator-activated receptor-alpha. *Lipids* **40**: 559-568.

Schwabe, J.W., Chapman, L., Finch, J.T. and Rhodes, D. (1993) The crystal structure of the estrogen receptor DNA-binding domain bound to DNA: how receptors discriminate between their response elements. *Cell* **75**: 567-578.

Seideman, P., Ericsson, O., Groningsson, K. and von Bahr, C. (1981) Effect of pentobarbital on the formation of diastereomeric oxazepam glucuronides in man: analysis by high performance liquid chromatography. *Acta Pharmacol. Toxicol.* **49**: 200-204.

Selvey, S., Thompson, E.W., Matthaei, K., Lea, R.A., Irving, M.G. and Griffiths, L.R. (2001) Beta-actin - an unsuitable internal control for RT-PCR. *Mol. Cell. Probes* **15**: 307-311.

Senekeo-Effenberger, K., Chen, S., Brace-Sinnokrak, E., Bonzo, J.A., Yueh, M.F., Argikar, U., Kaeding, J., Trottier, J., Remmel, R.P., Ritter, J.K., Barbier, O. and Tukey, R.H. (2007) Expression of the human *UGT1* locus in transgenic mice by 4-chloro-6-(2,3-xylidino)-2-pyrimidinylthioacetic acid (WY-14643) and implications on drug metabolism through peroxisome proliferator-activated receptor alpha activation. *Drug Metab. Dispos.* **35**: 419-427.

Senkel, S., Lucas, B., Klein-Hitpass, L. and Ryffel, G.U. (2005) Identification of target genes of the transcription factor HNF1beta and HNF1alpha in a human embryonic kidney cell line. *Biochim. Biophys. Acta* **1731**: 179-190.

Seol, W., Choi, H.S. and Moore, D.D. (1996) An orphan nuclear hormone receptor that lacks a DNA binding domain and heterodimerizes with other receptors. *Science* **272**: 1336-1339.

Sharp, S., Mak, L.Y., Smith, D.J. and Coughtrie, M.W. (1992) Inhibition of human and rabbit liver steroid and xenobiotic UDP-glucuronosyltransferases by tertiary amine drugs - implications for adverse drug reactions. *Xenobiotica* **22**: 13-25.

Shatalova, E.G., Walther, S.E., Favorova, O.O., Rebbeck, T.R. and Blanchard, R.L. (2005) Genetic polymorphisms in human SULT1A1 and UGT1A1 genes associate with breast tumor characteristics: a case-series study. *Breast Cancer Res.* 7: R909-921.

Shedlofsky, S.I., Israel, B.C., McClain, C.J., Hill, D.B. and Blouin, R.A. (1994) Endotoxin administration to humans inhibits hepatic cytochrome P450-mediated drug metabolism. *J. Clin. Invest.* **94**: 2209-2214.

Shih, D.Q., Bussen, M., Sehayek, E., Ananthanarayanan, M., Shneider, B.L., Suchy, F.J., Shefer, S., Bollileni, J.S., Gonzalez, F.J., Breslow, J.L. and Stoffel, M. (2001) Hepatocyte nuclear factor-1alpha is an essential regulator of bile acid and plasma cholesterol metabolism. *Nat. Genet.* **27**: 375-382.

Shim, E.Y., Woodcock, C. and Zaret, K.S. (1998) Nucleosome positioning by the winged helix transcription factor HNF3. *Genes Dev.* **12**: 5-10.

Shimamoto, Y., Ishida, J., Yamagata, K., Saito, T., Kato, H., Matsuoka, T., Hirota, K., Daitoku, H., Nangaku, M., Fujii, H., Takeda, J. and Fukamizu, A. (2004) Inhibitory effect of the small heterodimer partner on hepatocyte nuclear factor-4 mediates bile acid-induced repression of the human angiotensinogen gene. *J. Biol. Chem.* **279**: 7770-7776.

Shimoi, K., Saka, N., Nozawa, R., Sato, M., Amano, I., Nakayama, T. and Kinae, N. (2001) Deglucuronidation of a flavonoid, luteolin monoglucuronide, during inflammation. *Drug Metab. Dispos.* **29**: 1521-1524.

Shuman, J.D., Cheong, J. and Coligan, J.E. (1997) ATF-2 and C/EBPalpha can form a heterodimeric DNA binding complex in vitro. Functional implications for transcriptional regulation. *J. Biol. Chem.* **272**: 12793-12800.

Silander, K., Mohlke, K.L., Scott, L.J., Peck, E.C., Hollstein, P., Skol, A.D., Jackson, A.U., Deloukas, P., Hunt, S., Stavrides, G., Chines, P.S., Erdos, M.R., Narisu, N., Conneely, K.N., Li, C., Fingerlin, T.E., Dhanjal, S.K., Valle, T.T., Bergman, R.N., Tuomilehto, J., Watanabe, R.M., Boehnke, M. and Collins, F.S. (2004) Genetic variation near the hepatocyte nuclear factor-4alpha gene predicts susceptibility to type 2 diabetes. *Diabetes* **53**: 1141-1149.

Sladek, F.M., Ruse, M.D., Nepomuceno, L., Huang, S.M. and Stallcup, M.R. (1999) Modulation of transcriptional activation and coactivator interaction by a splicing variation in the F domain of nuclear receptor hepatocyte nuclear factor 4alpha1. *Mol. Cell. Biol.* **19**: 6509-6522.

Sladek, F.M., Zhong, W.M., Lai, E. and Darnell, J.E. (1990) Liver-enriched transcription factor HNF-4 is a novel member of the steroid hormone receptor superfamily. *Genes Dev.* 4: 2353-2365.

Smale, S.T. and Kadonaga, J.T. (2003) The RNA polymerase II core promoter. *Ann. Rev. Biochem.* **72**: 449-479.

Smith, C.L. and Hager, G.L. (1997) Transcriptional regulation of mammalian genes *in vivo*. A tale of two templates. *J. Biol. Chem.* **272**: 27493-27496.

Smith, C.M., Faucette, S.R., Wang, H. and LeCluyse, E.L. (2005a) Modulation of UDP-glucuronosyltransferase 1A1 in primary human hepatocytes by prototypical inducers. *J. Biochem. Mol. Toxicol.* **19**: 96-108.

Smith, C.M., Graham, R.A., Krol, W.L., Silver, I.S., Negishi, M., Wang, H. and Lecluyse, E.L. (2005b) Differential UGT1A1 induction by chrysin in primary human hepatocytes and HepG2 Cells. *J. Pharmacol. Exp. Ther.* **315**: 1256-1264.

Soars, M.G., Petullo, D.M., Eckstein, J.A., Kasper, S.C. and Wrighton, S.A. (2004) An assessment of udp-glucuronosyltransferase induction using primary human hepatocytes. *Drug Metab. Dispos.* **32**: 140-148.

Song, K.H., Li, T. and Chiang, J.Y. (2006) A Prospero-related homeodomain protein is a novel co-regulator of hepatocyte nuclear factor 4alpha that regulates the cholesterol 7alpha-hydroxylase gene. *J. Biol. Chem.* **281**: 10081-10088.

Song, Y.H., Ray, K., Liebhaber, S.A. and Cooke, N.E. (1998) Vitamin D-binding protein gene transcription is regulated by the relative abundance of hepatocyte nuclear factors 1alpha and 1beta. *J. Biol. Chem.* **273**: 28408-28418.

Sonne, J., Boesgaard, S., Poulsen, H.E., Loft, S., Hansen, J.M., Dossing, M. and Andreasen, F. (1990) Pharmacokinetics and pharmacodynamics of oxazepam and metabolism of paracetamol in severe hypothyroidism. *Br. J. Clin. Pharmacol.* **30**: 737-742.

Soutoglou, E., Katrakili, N. and Talianidis, I. (2000a) Acetylation regulates transcription factor activity at multiple levels. *Mol. Cell* **5**: 745-751.

Soutoglou, E., Papafotiou, G., Katrakili, N. and Talianidis, I. (2000b) Transcriptional activation by hepatocyte nuclear factor-1 requires synergism between multiple coactivator proteins. *J. Biol. Chem.* **275**: 12515-12520.

Soutoglou, E., Viollet, B., Vaxillaire, M., Yaniv, M., Pontoglio, M. and Talianidis, I. (2001) Transcription factor-dependent regulation of CBP and P/CAF histone acetyltransferase activity. *EMBO J.* **20**: 1984-1992.

Spencer, T.E., Jenster, G., Burcin, M.M., Allis, C.D., Zhou, J., Mizzen, C.A., McKenna, N.J., Onate, S.A., Tsai, S.Y., Tsai, M.J. and O'Malley, B.W. (1997) Steroid receptor coactivator-1 is a histone acetyltransferase. *Nature* **389**: 194-198.

Sperker, B., Backman, J.T. and Kroemer, H.K. (1997) The role of beta-glucuronidase in drug disposition and drug targeting in humans. *Clin. Pharmacokinet.* **33**: 18-31.

Staudinger, J.L., Goodwin, B., Jones, S.A., Hawkins-Brown, D., MacKenzie, K.I., LaTour, A., Liu, Y., Klaassen, C.D., Brown, K.K., Reinhard, J., Willson, T.M., Koller, B.H. and Kliewer, S.A. (2001) The nuclear receptor PXR is a lithocholic acid sensor that protects against liver toxicity. *Proc. Natl. Acad. Sci. U. S. A.* **98**: 3369-3374.

Stegmann, A., Hansen, M., Wang, Y., Larsen, J.B., Lund, L.R., Ritie, L., Nicholson, J.K., Quistorff, B., Simon-Assmann, P., Troelsen, J.T. and Olsen, J. (2006) Metabolome, transcriptome, and bioinformatic cis-element analyses point to HNF-4 as a central regulator of gene expression during enterocyte differentiation. *Physiol. Genomics* **27**: 141-155.

Strandmann, E.P., Senkel, S. and Ryffel, G.U. (1998) The bifunctional protein DCoH/PCD, a transcription factor with a cytoplasmic enzymatic activity, is a maternal factor in the rat egg and expressed tissue specifically during embryogenesis. *Int. J. Dev. Biol.* **42**: 53-59.

Strassburg, C.P., Kneip, S., Topp, J., Obermayer-Straub, P., Barut, A., Tukey, R.H. and Manns, M.P. (2000) Polymorphic gene regulation and interindividual variation of UDP-glucuronosyltransferase activity in human small intestine. *J. Biol. Chem.* **275**: 36164-36171.

Strassburg, C.P., Manns, M.P. and Tukey, R.H. (1997a) Differential down-regulation of the *UDP-glucuronosyltransferase 1A* locus is an early event in human liver and biliary cancer. *Cancer Res.* **57**: 2979-2985.

Strassburg, C.P., Manns, M.P. and Tukey, R.H. (1998a) Expression of the UDP-glucuronosyltransferase 1A locus in human colon. Identification and characterization of the novel extrahepatic UGT1A8. *J. Biol. Chem.* **273**: 8719-8726.

Strassburg, C.P., Nguyen, N., Manns, M.P. and Tukey, R.H. (1998b) Polymorphic expression of the UDP-glucuronosyltransferase UGT1A gene locus in human gastric epithelium. *Mol. Pharmacol.* **54**: 647-654.

Strassburg, C.P., Oldhafer, K., Manns, M.P. and Tukey, R.H. (1997b) Differential expression of the *UGT1A* locus in human liver, biliary, and gastric tissue: identification of *UGT1A7* and *UGT1A10* transcripts in extrahepatic tissue. *Mol. Pharmacol.* **52**: 212-220.

Strassburg, C.P., Strassburg, A., Kneip, S., Barut, A., Tukey, R.H., Rodeck, B. and Manns, M.P. (2002a) Developmental aspects of human hepatic drug glucuronidation in young children and adults. *Gut* **50**: 259-265.

Strassburg, C.P., Strassburg, A., Nguyen, N., Li, Q., Manns, M.P. and Tukey, R.H. (1999) Regulation and function of family 1 and family 2 UDP-glucuronosyltransferase genes (*UGT1A*, *UGT2B*) in human oesophagus. *Biochem. J.* **338**: 489-498.

Strassburg, C.P., Vogel, A., Kneip, S., Tukey, R.H. and Manns, M.P. (2002b) Polymorphisms of the human UDP-glucuronosyltransferase (UGT) 1A7 gene in colorectal cancer. *Gut* **50**: 851-856.

Strasser, S.I., Mashford, M.L. and Desmond, P.V. (1998) Regulation of uridine diphosphate glucuronosyltransferase during the acute-phase response. *J. Gastroenterol. Hepatol.* **13**: 88-94.

Su, J.S., Tsai, T.F., Chang, H.M., Chao, K.M., Su, T.S. and Tsai, S.F. (2006) Distant HNF1 site as a master control for the human class I alcohol dehydrogenase gene expression. *J. Biol. Chem.* **281**: 19809-19821.

Suaud, L., Hemimou, Y., Formstecher, P. and Laine, B. (1999) Functional study of the E276Q mutant hepatocyte nuclear factor-4alpha found in type 1 maturity-onset diabetes of the young. Impaired synergy with chicken ovalbumin upstream promoter transcription factor II on the hepatocyte nuclear factor-1 promoter. *Diabetes* **48**: 1162-1167.

Sugatani, J., Kojima, H., Ueda, A., Kakizaki, S., Yoshinari, K., Gong, Q.H., Owens, I.S., Negishi, M. and Sueyoshi, T. (2001) The phenobarbital response enhancer module in the human bilirubin UDP-glucuronosyltransferase *UGT1A1* gene and regulation by the nuclear receptor CAR. *Hepatology* **33**: 1232-1238.

Sugatani, J., Nishitani, S., Yamakawa, K., Yoshinari, K., Sueyoshi, T., Negishi, M. and Miwa, M. (2005a) Transcriptional regulation of human *UGT1A1* gene expression: activated glucocorticoid receptor enhances constitutive androstane receptor/pregnane X receptor-mediated UDP-glucuronosyltransferase 1A1 regulation with glucocorticoid receptor-interacting protein 1. *Mol. Pharmacol.* **67**: 845-855.

Sugatani, J., Sueyoshi, T., Negishi, M. and Miwa, M. (2005b) Regulation of the human *UGT1A1* gene by nuclear receptors constitutive active/androstane receptor, pregnane X receptor, and glucocorticoid receptor. *Methods Enzymol.* **400**: 92-104.

Sugatani, J., Yamakawa, K., Tonda, E., Nishitani, S., Yoshinari, K., Degawa, M., Abe, I., Noguchi, H. and Miwa, M. (2004) The induction of human UDP-glucuronosyltransferase 1A1 mediated through a distal enhancer module by flavonoids and xenobiotics. *Biochem. Pharmacol.* **67**: 989-1000.

Sugatani, J., Yamakawa, K., Yoshinari, K., Machida, T., Takagi, H., Mori, M., Kakizaki, S., Sueyoshi, T., Negishi, M. and Miwa, M. (2002) Identification of a defect in the UGT1A1 gene promoter and its association with hyperbilirubinemia. *Biochem. Biophys. Res. Commun.* **292**: 492-497.

Sund, N.J., Ang, S.L., Sackett, S.D., Shen, W., Daigle, N., Magnuson, M.A. and Kaestner, K.H. (2000) Hepatocyte nuclear factor 3beta (Foxa2) is dispensable for maintaining the differentiated state of the adult hepatocyte. *Mol. Cell. Biol.* **20**: 5175-5183.

Sutomo, R., Laosombat, V., Sadewa, A.H., Yokoyama, N., Nakamura, H., Matsuo, M. and Nishio, H. (2002) Novel missense mutation of the UGT1A1 gene in Thai siblings with Gilbert's syndrome. *Pediatr. Int.* **44**: 427-432.

Synold, T.W., Dussault, I. and Forman, B.M. (2001) The orphan nuclear receptor SXR coordinately regulates drug metabolism and efflux. *Nat. Med.* **7**: 584-590.

Tanaka, T., Akira, S., Yoshida, K., Umemoto, M., Yoneda, Y., Shirafuji, N., Fujiwara, H., Suematsu, S., Yoshida, N. and Kishimoto, T. (1995) Targeted disruption of the NF-IL6 gene discloses its essential role in bacteria killing and tumor cytotoxicity by macrophages. *Cell* **80**: 353-361.

Tang, K.S., Chiu, H.F., Chen, H.H., Eng, H.L., Tsai, C.J., Teng, H.C. and Huang, C.S. (2005) Link between colorectal cancer and polymorphisms in the uridinediphosphoglucuronosyltransferase *1A7* and *1A1* genes. *World J. Gastroenterol.* **11**: 3250-3254.

Tarumi, T., Kravtsov, D.V., Zhao, M., Williams, S.M. and Gailani, D. (2002) Cloning and characterization of the human factor XI gene promoter. Transcription factor hepatocyte nuclear factor 4alpha (HNF-4alpha) is required for hepatocytespecific expression of factor XI. *J. Biol. Chem.* **277**: 18510-18516.

Taura, K., Naito, E., Ishii, Y., Mori, M.A., Oguri, K. and Yamada, H. (2004) Cytochrome P450 1A1 (CYP1A1) inhibitor alpha-naphthoflavone interferes with UDP-glucuronosyltransferase (UGT) activity in intact but not in permeabilized hepatic microsomes from 3-methylcholanthrene-treated rats: possible involvement of UGT-P450 interactions. *Biol. Pharm. Bull.* **27**: 56-60.

Taura, K.I., Yamada, H., Hagino, Y., Ishii, Y., Mori, M.A. and Oguri, K. (2000) Interaction between cytochrome P450 and other drug-metabolizing enzymes: evidence for an association of CYP1A1 with microsomal epoxide hydrolase and UDP-glucuronosyltransferase. *Biochem. Biophys. Res. Commun.* **273**: 1048-1052.

Tchernof, A., Levesque, E., Beaulieu, M., Couture, P., Despres, J.P., Hum, D.W. and Belanger, A. (1999) Expression of the androgen metabolizing enzyme UGT2B15 in adipose tissue and relative expression measurement using a competitive RT-PCR method. *Clin. Endocrinol.* **50**: 637-642.

Terakura, S., Murata, M., Nishida, T., Emi, N., Akatsuka, Y., Riddell, S.R., Morishima, Y., Kodera, Y. and Naoe, T. (2005) A UGT2B17-positive donor is a risk factor for higher transplant-related mortality and lower survival after bone marrow transplantation. *Br. J. Haematol.* **129**: 221-228.

Thomas, S.S., Li, S.S., Lampe, J.W., Potter, J.D. and Bigler, J. (2006) Genetic variability, haplotypes, and htSNPs for exons 1 at the human *UGT1A* locus. *Hum. Mut.* **27**: 717.

Thomassin, H., Hamel, D., Bernier, D., Guertin, M. and Belanger, L. (1992) Molecular cloning of two C/EBP-related proteins that bind to the promoter and the enhancer of the alpha 1-fetoprotein gene. Further analysis of C/EBPbeta and C/EBPgamma. *Nucleic Acids Res.* **20**: 3091-3098.

Thompson, J.D., Gibson, T.J., Plewniak, F., Jeanmougin, F. and Higgins, D.G. (1997) The CLUSTAL_X windows interface: flexible strategies for multiple sequence alignment aided by quality analysis tools. *Nucleic Acids Res.* **25**: 4876-4882.

Timchenko, N., Wilson, D.R., Taylor, L.R., Abdelsayed, S., Wilde, M., Sawadogo, M. and Darlington, G.J. (1995) Autoregulation of the human C/EBPalpha gene by stimulation of upstream stimulatory factor binding. *Mol. Cell. Biol.* **15**: 1192-1202.

Timchenko, N.A., Wang, G.L. and Timchenko, L.T. (2005) RNA CUG-binding protein 1 increases translation of 20-kDa isoform of CCAAT/enhancer-binding protein beta by interacting with the alpha and beta subunits of eukaryotic initiation translation factor 2. *J. Biol. Chem.* **280**: 20549-20557.

Timchenko, N.A., Welm, A.L., Lu, X. and Timchenko, L.T. (1999) CUG repeat binding protein (CUGBP1) interacts with the 5' region of C/EBPbeta mRNA and regulates translation of C/EBPbeta isoforms. *Nucleic Acids Res.* **27**: 4517-4525.

Timmers, C.M., Dekker, M., Buijsman, R.C., van der Marel, G.A., Ethell, B., Anderson, G., Burchell, B., Mulder, G.J. and van Boom, J.H. (1997) Synthesis and inhibitory effect of a trisubstrate transition state analogue for UDP glucuronosyltransferases. *Bioorg. Med. Chem. Lett.* **7**: 1501-1506.

Toide, K., Takahashi, Y., Yamazaki, H., Terauchi, Y., Fujii, T., Parkinson, A. and Kamataki, T. (2002) Hepatocyte nuclear factor-lalpha is a causal factor responsible for interindividual differences in the expression of UDP-glucuronosyltransferase 2B7 mRNA in human livers. *Drug Metab. Dispos.* **30**: 613-615.

Tomizawa, M., Watanabe, K., Saisho, H., Nakagawara, A. and Tagawa, M. (2003) Down-regulated expression of the CCAAT/enhancer binding protein alpha and beta genes in human hepatocellular carcinoma: a possible prognostic marker. *Anticancer Res.* **23**: 351-354.

Torres-Padilla, M.E., Fougere-Deschatrette, C. and Weiss, M.C. (2001) Expression of HNF4alpha isoforms in mouse liver development is regulated by sequential promoter usage and constitutive 3' end splicing. *Mech. Dev.* **109**: 183-193.

Torres-Padilla, M.E., Sladek, F.M. and Weiss, M.C. (2002) Developmentally regulated N-terminal variants of the nuclear receptor hepatocyte nuclear factor 4alpha mediate multiple interactions through coactivator and corepressor-histone deacetylase complexes. *J. Biol. Chem.* **277**: 44677-44687.

Torres-Padilla, M.E. and Weiss, M.C. (2003) Effects of interactions of hepatocyte nuclear factor 4alpha isoforms with coactivators and corepressors are promoter-specific. *FEBS Lett.* **539**: 19-23.

Tougou, K., Gotou, H., Ohno, Y. and Nakamura, A. (2004) Stereoselective glucuronidation and hydroxylation of etodolac by UGT1A9 and CYP2C9 in man. *Xenobiotica* **34**: 449-461.

Tronche, F., Ringeisen, F., Blumenfeld, M., Yaniv, M. and Pontoglio, M. (1997) Analysis of the distribution of binding sites for a tissue-specific transcription factor in the vertebrate genome. *J. Mol. Biol.* **266**: 231-245.

Tronche, F. and Yaniv, M. (1992) HNF1, a homeoprotein member of the hepatic transcription regulatory network. *Bioessays* 14: 579-587.

Trottier, J., Milkiewicz, P., Kaeding, J., Verreault, M. and Barbier, O. (2006a) Coordinate regulation of hepatic bile acid oxidation and conjugation by nuclear receptors. *Mol. Pharm.* **3**: 212-222.

Trottier, J., Verreault, M., Grepper, S., Monte, D., Belanger, J., Kaeding, J., Caron, P., Inaba, T.T. and Barbier, O. (2006b) Human UDP-glucuronosyltransferase (UGT)1A3 enzyme conjugates chenodeoxycholic acid in the liver. *Hepatology* **44**: 1158-1170.

Tseng, C.S., Tang, K.S., Lo, H.W., Ker, C.G., Teng, H.C. and Huang, C.S. (2005) UDP-glucuronosyltransferase 1A7 genetic polymorphisms are associated with hepatocellular carcinoma risk and onset age. *Am. J. Gastroenterol.* **100**: 1758-1763.

Tsezou, A., Tzetis, M., Giannatou, E., Gennatas, C., Pampanos, A., Kanavakis, E. and Kitsiou-Tzeli, S. (2007) Genetic polymorphisms in the UGT1A1 gene and breast cancer risk in Greek women. *Genet. Test.* **11**: 303-306.

Tukey, R.H. and Strassburg, C.P. (2000) Human UDP-glucuronosyltransferases: metabolism, expression, and disease. *Ann. Rev. Pharmacol. Toxicol.* **40**: 581-616.

Tukey, R.H. and Strassburg, C.P. (2001) Genetic multiplicity of the human UDP-glucuronosyltransferases and regulation in the gastrointestinal tract. *Mol. Pharmacol.* **59**: 405-414.

Turgeon, D., Carrier, J.S., Chouinard, S. and Belanger, A. (2003a) Glucuronidation activity of the UGT2B17 enzyme toward xenobiotics. *Drug Metab. Dispos.* **31**: 670-676.

Turgeon, D., Carrier, J.S., Levesque, E., Beatty, B.G., Belanger, A. and Hum, D.W. (2000) Isolation and characterization of the human *UGT2B15* gene, localized within a cluster of *UGT2B* genes and pseudogenes on chromosome 4. *J. Mol. Biol.* **295**: 489-504.

Turgeon, D., Carrier, J.S., Levesque, E., Hum, D.W. and Belanger, A. (2001) Relative enzymatic activity, protein stability, and tissue distribution of human steroid-metabolizing UGT2B subfamily members. *Endocrinology* **142**: 778-787.

Turgeon, D., Chouinard, S., Belanger, P., Picard, S., Labbe, J.F., Borgeat, P. and Belanger, A. (2003b) Glucuronidation of arachidonic and linoleic acid metabolites by human UDP-glucuronosyltransferases. *J. Lipid Res.* **44**: 1182-1191.

UGT Nomenclature Committee (2005) UGT Alleles Nomenclature Home Page [online]. Available from: <u>http://www.pharmacogenomics.pha.ulaval.ca/sgc/ugt_alleles/</u> [Accessed on 23 July 2007; 12 September 2007].

Umesono, K., Murakami, K.K., Thompson, C.C. and Evans, R.M. (1991) Direct repeats as selective response elements for the thyroid hormone, retinoic acid, and vitamin D3 receptors. *Cell* **65**: 1255-1266.

Uno, Y., Sakamoto, Y., Yoshida, K., Hasegawa, T., Hasegawa, Y., Koshino, T. and Inoue, I. (2003) Characterization of six base pair deletion in the putative HNF1binding site of human *PXR* promoter. *J. Hum. Genet.* **48**: 594-597.

Urhammer, S.A., Fridberg, M., Hansen, T., Rasmussen, S.K., Moller, A.M., Clausen, J.O. and Pedersen, O. (1997) A prevalent amino acid polymorphism at codon 98 in the hepatocyte nuclear factor-1alpha gene is associated with reduced serum C-peptide and insulin responses to an oral glucose challenge. *Diabetes* **46**: 912-916.

Urhammer, S.A., Hansen, T., Ekstrom, C.T., Eiberg, H. and Pedersen, O. (1998a) The Ala/Val98 polymorphism of the hepatocyte nuclear factor-1alpha gene contributes to the interindividual variation in serum C-peptide response during an oral glucose tolerance test: evidence from studies of 231 glucose-tolerant first degree relatives of type 2 diabetic probands. *J. Clin. Endocrinol. Metab.* **83**: 4506-4509.

Urhammer, S.A., Moller, A.M., Nyholm, B., Ekstrom, C.T., Eiberg, H., Clausen, J.O., Hansen, T., Pedersen, O. and Schmitz, O. (1998b) The effect of two frequent amino acid variants of the hepatocyte nuclear factor-1alpha gene on estimates of the pancreatic beta-cell function in Caucasian glucose-tolerant first-degree relatives of type 2 diabetic patients. *J. Clin. Endocrinol. Metab.* **83**: 3992-3995.

Usui, T., Kuno, T. and Mizutani, T. (2006a) Induction of human UDP-glucuronosyltransferase 1A1 by cortisol-GR. *Mol. Biol. Rep.* **33**: 91-96.

Usui, T., Kuno, T., Ueyama, H., Ohkubo, I. and Mizutani, T. (2006b) Proximal HNF1 element is essential for the induction of human UDP-glucuronosyltransferase 1A1 by glucocorticoid receptor. *Biochem. Pharmacol.* **71**: 693-701.

Vaisse, C., Kim, J., Espinosa, R., Le Beau, M.M. and Stoffel, M. (1997) Pancreatic islet expression studies and polymorphic DNA markers in the genes encoding hepatocyte nuclear factor-3alpha, -3beta, -3gamma, -4gamma, and -6. *Diabetes* **46**: 1364-1367.

Vallee, M., Albert, C., Beaudry, G., Hum, D.W. and Belanger, A. (2001) Isolation and characterization of the monkey UDP-glucuronosyltransferase cDNA clone monUGT1A01 active on bilirubin and estrogens. *J. Steroid Biochem. Mol. Biol.* **77**: 239-249.

Vallet, V., Antoine, B., Chafey, P., Vandewalle, A. and Kahn, A. (1995) Overproduction of a truncated hepatocyte nuclear factor 3 protein inhibits expression of liver-specific genes in hepatoma cells. *Mol. Cell. Biol.* **15**: 5453-5460.

van der Logt, E.M., Bergevoet, S.M., Roelofs, H.M., van Hooijdonk, Z., te Morsche, R.H., Wobbes, T., de Kok, J.B., Nagengast, F.M. and Peters, W.H. (2004) Genetic polymorphisms in UDP-glucuronosyltransferases and glutathione S-transferases and colorectal cancer risk. *Carcinogenesis* **25**: 2407-2415.

van der Logt, E.M., Roelofs, H.M., Nagengast, F.M. and Peters, W.H. (2003) Induction of rat hepatic and intestinal UDP-glucuronosyltransferases by naturally occurring dietary anticarcinogens. *Carcinogenesis* **24**: 1651-1656.

van Dijk, M.A., Rodenburg, R.J., Holthuizen, P. and Sussenbach, J.S. (1992) The liver-specific promoter of the human insulin-like growth factor II gene is activated by CCAAT/enhancer binding protein (C/EBP). *Nucleic Acids Res.* **20**: 3099-3104.

Van Eerdewegh, P., Little, R.D., Dupuis, J., Del Mastro, R.G., Falls, K., Simon, J., Torrey, D., Pandit, S., McKenny, J., Braunschweiger, K., Walsh, A., Liu, Z., Hayward, B., Folz, C., Manning, S.P., Bawa, A., Saracino, L., Thackston, M., Benchekroun, Y., Capparell, N., Wang, M., Adair, R., Feng, Y., Dubois, J., FitzGerald, M.G., Huang, H., Gibson, R., Allen, K.M., Pedan, A., Danzig, M.R., Umland, S.P., Egan, R.W., Cuss, F.M., Rorke, S., Clough, J.B., Holloway, J.W., Holgate, S.T. and Keith, T.P. (2002) Association of the *ADAM33* gene with asthma and bronchial hyperresponsiveness. *Nature* **418**: 426-430.

van Oijen, M.G.H., Laheij, R.J.F. and Peters, W.H.M. (2007) Authors' response: Relationships between the adverse effects of aspirin and polymorphisms in genes encoding aspirin metabolizing enzymes. *Br. J. Clin. Pharmacol.* **63**: 381-382.

van Ommen, G.J. (2002) The Human Genome Project and the future of diagnostics, treatment and prevention. *J. Inherit. Metab. Dis.* **25**: 183-188.

van Ooij, C., Snyder, R.C., Paeper, B.W. and Duester, G. (1992) Temporal expression of the human alcohol dehydrogenase gene family during liver development correlates with differential promoter activation by hepatocyte nuclear factor 1, CCAAT/enhancer-binding protein alpha, liver activator protein, and D-element-binding protein. *Mol. Cell. Biol.* **12**: 3023-3031.

Verlaan, M., Drenth, J.P., Truninger, K., Koudova, M., Schulz, H.U., Bargetzi, M., Kunzli, B., Friess, H., Cerny, M., Kage, A., Landt, O., te Morsche, R.H., Rosendahl, J., Luck, W., Nickel, R., Halangk, J., Becker, M., Macek, M., Jansen, J.B. and Witt, H. (2005) Polymorphisms of UDP-glucuronosyltransferase 1A7 are not involved in pancreatic diseases. *J. Med. Genet.* **42**: e62.

Verreault, M., Senekeo-Effenberger, K., Trottier, J., Bonzo, J.A., Belanger, J., Kaeding, J., Staels, B., Caron, P., Tukey, R.H. and Barbier, O. (2006) The liver X-receptor alpha controls hepatic expression of the human bile acid-glucuronidating UGT1A3 enzyme in human cells and transgenic mice. *Hepatology* **44**: 368-378.

Villeneuve, L., Girard, H., Fortier, L.C., Gagne, J.F. and Guillemette, C. (2003) Novel functional polymorphisms in the UGT1A7 and UGT1A9 glucuronidating enzymes in Caucasian and African-American subjects and their impact on the metabolism of 7-ethyl-10-hydroxycamptothecin and flavopiridol anticancer drugs. *J. Pharmacol. Exp. Ther.* **307**: 117-128.

Viollet, B., Kahn, A. and Raymondjean, M. (1997) Protein kinase A-dependent phosphorylation modulates DNA-binding activity of hepatocyte nuclear factor 4. *Mol. Cell. Biol.* **17**: 4208-4219.

Vogel, A., Kneip, S., Barut, A., Ehmer, U., Tukey, R.H., Manns, M.P. and Strassburg, C.P. (2001) Genetic link of hepatocellular carcinoma with polymorphisms of the *UDP-glucuronosyltransferase UGT1A7* gene. *Gastroenterology* **121**: 1136-1144.

Vogel, A., Ockenga, J., Ehmer, U., Barut, A., Kramer, F.J., Tukey, R.H., Manns, M.P. and Strassburg, C.P. (2002) Polymorphisms of the carcinogen detoxifying UDP-glucuronosyltransferase UGT1A7 in proximal digestive tract cancer. *Z. Gastroenterol.* **40**: 497-502.

Vorachek, W.R., Steppan, C.M., Lima, M., Black, H., Bhattacharya, R., Wen, P., Kajiyama, Y. and Locker, J. (2000) Distant enhancers stimulate the albumin promoter through complex proximal binding sites. *J. Biol. Chem.* **275**: 29031-29041.

Vyhlidal, C.A., Rogan, P.K. and Leeder, J.S. (2004) Development and refinement of pregnane X receptor (PXR) DNA binding site model using information theory. Insights into PXR-mediated gene regulation. *J. Biol. Chem.* **279**: 46779-46786.

Wahlstrom, A., Lenhammar, L., Ask, B. and Rane, A. (1994) Tricyclic antidepressants inhibit opioid receptor binding in human brain and hepatic morphine glucuronidation. *Pharmacol. Toxicol.* **75**: 23-27.

Walle, T., Walle, U.K., Cowart, T.D. and Conradi, E.C. (1989) Pathway-selective sex differences in the metabolic clearance of propranolol in human subjects. *Clin. Pharmacol. Ther.***46**: 257-263.

Walle, T., Walle, U.K., Cowart, T.D., Conradi, E.C. and Gaffney, T.E. (1987) Selective induction of propranolol metabolism by smoking: additional effects on renal clearance of metabolites. *J. Pharmacol. Exp. Ther.* **241**: 928-933.

Walle, U.K. and Walle, T. (1999) Transport of the cooked-food mutagen 2-amino-1methyl-6-phenylimidazo-[4,5-*b*]pyridine (PhIP) across the human intestinal Caco-2 cell monolayer: role of efflux pumps. *Carcinogenesis* **20**: 2153-2157.

Walle, U.K. and Walle, T. (2002) Induction of human UDP-glucuronosyltransferase UGT1A1 by flavonoids - structural requirements. *Drug Metab. Dispos.* **30**: 564-569.

Wang, H., Maechler, P., Antinozzi, P.A., Hagenfeldt, K.A. and Wollheim, C.B. (2000) Hepatocyte nuclear factor 4alpha regulates the expression of pancreatic beta - cell genes implicated in glucose metabolism and nutrient-induced insulin secretion. *J. Biol. Chem.* **275**: 35953-35959.

Wang, J.C., Stafford, J.M. and Granner, D.K. (1998) SRC-1 and GRIP1 coactivate transcription with hepatocyte nuclear factor 4. *J. Biol. Chem.* **273**: 30847-30850.

Wang, L., Coffinier, C., Thomas, M.K., Gresh, L., Eddu, G., Manor, T., Levitsky, L.L., Yaniv, M. and Rhoads, D.B. (2004a) Selective deletion of the *Hnflbeta* (MODY5) gene in beta-cells leads to altered gene expression and defective insulin release. *Endocrinology* **145**: 3941-3949.

Wang, Y., Kato, N., Hoshida, Y., Otsuka, M., Taniguchi, H., Moriyama, M., Shiina, S., Kawabe, T., Ito, Y.M. and Omata, M. (2004b) UDP-glucuronosyltransferase 1A7 genetic polymorphisms are associated with hepatocellular carcinoma in japanese patients with hepatitis C virus infection. *Clin. Cancer Res.* **10**: 2441-2446.

Wang, Y., Moser, A.H., Shigenaga, J.K., Grunfeld, C. and Feingold, K.R. (2005) Downregulation of liver X receptor-alpha in mouse kidney and HK-2 proximal tubular cells by LPS and cytokines. *J. Lipid Res.* **46**: 2377-2387.

Weedon, M.N., Owen, K.R., Shields, B., Hitman, G., Walker, M., McCarthy, M.I., Love-Gregory, L.D., Permutt, M.A., Hattersley, A.T. and Frayling, T.M. (2004) Common variants of the hepatocyte nuclear factor-4alpha P2 promoter are associated with type 2 diabetes in the U.K. population. *Diabetes* **53**: 3002-3006.

Weinshilboum, R. (2003) Inheritance and drug response. N. Engl. J. Med. 348: 529-537.

Weintraub, D., Buchsbaum, R., Resor, S.R. and Hirsch, L.J. (2005) Effect of antiepileptic drug comedication on lamotrigine clearance. *Arch. Neurol.* **62**: 1432-1436.

Weissglas-Volkov, D., Huertas-Vazquez, A., Suviolahti, E., Lee, J., Plaisier, C., Canizales-Quinteros, S., Tusie-Luna, T., Aguilar-Salinas, C., Taskinen, M.R. and Pajukanta, P. (2006) Common hepatic nuclear factor-4alpha variants are associated with high serum lipid levels and the metabolic syndrome. *Diabetes* **55**: 1970-1977.

Wells, P.G., Mackenzie, P.I., Chowdhury, J.R., Guillemette, C., Gregory, P.A., Ishii, Y., Hansen, A.J., Kessler, F.K., Kim, P.M., Chowdhury, N.R. and Ritter, J.K. (2004) Glucuronidation and the UDP-glucuronosyltransferases in health and disease. *Drug Metab. Dispos.* **32**: 281-290.

Welm, A.L., Mackey, S.L., Timchenko, L.T., Darlington, G.J. and Timchenko, N.A. (2000) Translational induction of liver-enriched transcriptional inhibitory protein during acute phase response leads to repression of CCAAT/enhancer binding protein alpha mRNA. *J. Biol. Chem.* **275**: 27406-27413.

Welm, A.L., Timchenko, N.A. and Darlington, G.J. (1999) C/EBPalpha regulates generation of C/EBPbeta isoforms through activation of specific proteolytic cleavage. *Mol. Cell. Biol.* **19**: 1695-1704.

West, A.G. and Fraser, P. (2005) Remote control of gene transcription. *Hum. Mol. Genet.* **14**: R101-111.

Westmacott, A., Burke, Z.D., Oliver, G., Slack, J.M. and Tosh, D. (2006) C/EBPalpha and C/EBPbeta are markers of early liver development. *Int. J. Dev. Biol.* **50**: 653-657.

Whitcomb, D.C. and Block, G.D. (1994) Association of acetaminophen hepatotoxicity with fasting and ethanol use. *JAMA* **272**: 1845-1850.

Wiener, D., Doerge, D.R., Fang, J.L., Upadhyaya, P. and Lazarus, P. (2004) Characterization of *N*-glucuronidation of the lung carcinogen 4-(methylnitrosamino)-1-(3-pyridyl)-1-butanol (NNAL) in human liver: importance of UDPglucuronosyltransferase 1A4. *Drug Metab. Dispos.* **32**: 72-79.

Williams, S.C., Cantwell, C.A. and Johnson, P.F. (1991) A family of C/EBP-related proteins capable of forming covalently linked leucine zipper dimers in vitro. *Genes Dev.* **5**: 1553-1567.

Wilson, W., 3rd, Pardo-Manuel de Villena, F., Lyn-Cook, B.D., Chatterjee, P.K., Bell, T.A., Detwiler, D.A., Gilmore, R.C., Valladeras, I.C., Wright, C.C., Threadgill, D.W. and Grant, D.J. (2004) Characterization of a common deletion polymorphism of the *UGT2B17* gene linked to *UGT2B15*. *Genomics* **84**: 707-714.

Wisely, G.B., Miller, A.B., Davis, R.G., Thornquest, A.D., Johnson, R., Spitzer, T., Sefler, A., Shearer, B., Moore, J.T., Willson, T.M. and Williams, S.P. (2002) Hepatocyte nuclear factor 4 is a transcription factor that constitutively binds fatty acids. *Structure* **10**: 1225-1234.

Wogan, G.N., Hecht, S.S., Felton, J.S., Conney, A.H. and Loeb, L.A. (2004) Environmental and chemical carcinogenesis. *Semin. Cancer Biol.* **14**: 473-486.

Wu, K.J., Wilson, D.R., Shih, C. and Darlington, G.J. (1994) The transcription factor HNF1 acts with C/EBPalpha to synergistically activate the human albumin promoter through a novel domain. *J. Biol. Chem.* **269**: 1177-1182.

Wynalda, M.A., Wynalda, K.M., Amore, B.M., Fagerness, P.E. and Wienkers, L.C. (2003) Characterization of bropirimine *O*-glucuronidation in human liver microsomes. *Xenobiotica* **33**: 999-1011.

Xie, W., Yeuh, M.F., Radominska-Pandya, A., Saini, S.P., Negishi, Y., Bottroff, B.S., Cabrera, G.Y., Tukey, R.H. and Evans, R.M. (2003) Control of steroid, heme, and carcinogen metabolism by nuclear pregnane X receptor and constitutive androstane receptor. *Proc. Natl. Acad. Sci. U. S. A.* **100**: 4150-4155.

Xiong, W., Hsieh, C.C., Kurtz, A.J., Rabek, J.P. and Papaconstantinou, J. (2001) Regulation of CCAAT/enhancer-binding protein-beta isoform synthesis by alternative translational initiation at multiple AUG start sites. *Nucleic Acids Res.* **29**: 3087-3098.

Xu, C., Li, C.Y. and Kong, A.N. (2005) Induction of phase I, II and III drug metabolism/transport by xenobiotics. *Arch. Pharm. Res.* **28**: 249-268.

Xu, L., Hui, L., Wang, S., Gong, J., Jin, Y., Wang, Y., Ji, Y., Wu, X., Han, Z. and Hu, G. (2001) Expression profiling suggested a regulatory role of liver-enriched transcription factors in human hepatocellular carcinoma. *Cancer Res.* **61**: 3176-3181.

Yamada, T., Tobita, K., Osada, S., Nishihara, T. and Imagawa, M. (1997) CCAAT/enhancer-binding protein delta gene expression is mediated by APRF/STAT3. *J. Biochem.* **121**: 731-738.

Yamagata, K., Yang, Q., Yamamoto, K., Iwahashi, H., Miyagawa, J., Okita, K., Yoshiuchi, I., Miyazaki, J., Noguchi, T., Nakajima, H., Namba, M., Hanafusa, T. and Matsuzawa, Y. (1998) Mutation P291fsinsC in the transcription factor hepatocyte nuclear factor-1alpha is dominant negative. *Diabetes* **47**: 1231-1235.

Yamanaka, H., Nakajima, M., Katoh, M., Hara, Y., Tachibana, O., Yamashita, J., McLeod, H.L. and Yokoi, T. (2004) A novel polymorphism in the promoter region of human UGT1A9 gene (*UGT1A9*22*) and its effects on the transcriptional activity. *Pharmacogenetics* 14: 329-332.

Yamashita, Y., Shimada, M., Harimoto, N., Rikimaru, T., Shirabe, K., Tanaka, S. and Sugimachi, K. (2003) Histone deacetylase inhibitor trichostatin A induces cell-cycle arrest/apoptosis and hepatocyte differentiation in human hepatoma cells. *Int. J. Cancer* **103**: 572-576.

Yang, Q., Yamagata, K., Yamamoto, K., Cao, Y., Miyagawa, J., Fukamizu, A., Hanafusa, T. and Matsuzawa, Y. (2000) R127W-HNF-4alpha is a loss of function mutation but not a rare polymorphism and causes Type II diabetes in a Japanese family with MODY1. *Diabetologia* **43**: 520-524.

Yang, X.J., Ogryzko, V.V., Nishikawa, J., Howard, B.H. and Nakatani, Y. (1996) A p300/CBP-associated factor that competes with the adenoviral oncoprotein E1A. *Nature* **382**: 319-324.

Yoon, J.C., Puigserver, P., Chen, G., Donovan, J., Wu, Z., Rhee, J., Adelmant, G., Stafford, J., Kahn, C.R., Granner, D.K., Newgard, C.B. and Spiegelman, B.M. (2001)

Control of hepatic gluconeogenesis through the transcriptional coactivator PGC-1. *Nature* **413**: 131-138.

Yoshida, E., Aratani, S., Itou, H., Miyagishi, M., Takiguchi, M., Osumu, T., Murakami, K. and Fukamizu, A. (1997) Functional association between CBP and HNF4 in *trans*-activation. *Biochem. Biophys. Res. Commun.* **241**: 664-669.

Yoshida, M., Kijima, M., Akita, M. and Beppu, T. (1990) Potent and specific inhibition of mammalian histone deacetylase both *in vivo* and *in vitro* by trichostatin A. J. Biol. Chem. **265**: 17174-17179.

Yoshida, Y., Hughes, D.E., Rausa, F.M., Kim, I.M., Tan, Y., Darlington, G.J. and Costa, R.H. (2006) C/EBPalpha and HNF6 protein complex formation stimulates HNF6-dependent transcription by CBP coactivator recruitment in HepG2 cells. *Hepatology* **43**: 276-286.

Yue, Q., von Bahr, C., Odar-Cederlof, I. and Sawe, J. (1990) Glucuronidation of codeine and morphine in human liver and kidney microsomes: effect of inhibitors. *Pharmacol. Toxicol.* **66**: 221-226.

Yueh, M.F., Huang, Y.H., Hiller, A., Chen, S., Nguyen, N. and Tukey, R.H. (2003) Involvement of the xenobiotic response element (XRE) in Ah receptor-mediated induction of human UDP-glucuronosyltransferase *1A1. J. Biol. Chem.* **278**: 15001-15006.

Yueh, M.F. and Tukey, R.H. (2007) Nrf2-Keap1 signaling pathway regulates human UGT1A1 expression *in vitro* and in transgenic *UGT1* mice. *J. Biol. Chem.* **282**: 8749-8758.

Yuen, A.W., Land, G., Weatherley, B.C. and Peck, A.W. (1992) Sodium valproate acutely inhibits lamotrigine metabolism. *Br. J. Clin. Pharmacol.* **33**: 511-513.

Zee, R.Y., Cook, N.R., Cheng, S., Reynolds, R., Erlich, H.A., Lindpaintner, K. and Ridker, P.M. (2004) Polymorphism in the P-selectin and interleukin-4 genes as determinants of stroke: a population-based, prospective genetic analysis. *Hum. Mol. Genet.* **13**: 389-396.

Zhang, J. and Byrne, C.D. (1997) A novel highly reproducible quantitative competitve RT PCR system. *J. Mol. Biol.* **274**: 338-352.

Zhang, J. and Byrne, C.D. (1999) Differential priming of RNA templates during cDNA synthesis markedly affects both accuracy and reproducibility of quantitative competitive reverse-transcriptase PCR. *Biochem. J.* **337**: 231-241.

Zhang, J., Kuehl, P., Green, E.D., W., T.J., Watkins, P.B., Daly, A., Hall, S.D., Maurel, P., Relling, M., Brimer, C., Yasuda, K., Wrighton, S.A., Hancock, M., Kim, R.B., Strom, S., Thummel, K., Russell, C.G., Hudson, J.R., Schuetz, E.G. and Boguski, M.S. (2001) The human pregnane X receptor: genomic structure and identification and functional characterization of natural allelic variants. *Pharmacogenetics* **11**: 555-572.

Zhang, L., Rubins, N.E., Ahima, R.S., Greenbaum, L.E. and Kaestner, K.H. (2005) Foxa2 integrates the transcriptional response of the hepatocyte to fasting. *Cell Metab.* **2**: 141-148.

Zhang, M. and Chiang, J.Y. (2001) Transcriptional regulation of the human sterol 12alpha-hydroxylase gene (*CYP8B1*). Roles of heaptocyte nuclear factor 4alpha in mediating bile acid repression. *J. Biol. Chem.* **276**: 41690-41699.

Zheng, Z., Fang, J.L. and Lazarus, P. (2002) Glucuronidation: an important mechanism for detoxification of benzo[a]pyrene metabolites in aerodigestive tract tissues. *Drug Metab. Dispos.* **30**: 397-403.

Zheng, Z., Park, J.Y., Guillemette, C., Schantz, S.P. and Lazarus, P. (2001) Tobacco carcinogen-detoxifying enzyme UGT1A7 and its association with orolaryngeal cancer risk. *J. Natl. Cancer Inst.* **93**: 1411-1418.

Zhong, W., Mirkovitch, J. and Darnell, J.E. (1994) Tissue-specific regulation of mouse hepatocyte nuclear factor 4 expression. *Mol. Cell. Biol.* **14**: 7276-7284.

Zhong, W., Sladek, F.M. and Darnell, J.E. (1993) The expression pattern of a *Drosophila* homolog to the mouse transcription factor HNF-4 suggests a determinative role in gut formation. *EMBO J.* **12**: 537-544.

Zhuang, X., Bulter, B., Hawkins, E., Paguio, A., Orr, L., Wood, M. and Wood, K. (2001) A new age of enlightenment: new synthetic renilla gene and assay system increase expression, reliability and sensitivity. *Promega Notes* **79**: 6-11.

Zimmerman, H.J. and Maddrey, W.C. (1995) Acetaminophen (paracetamol) hepatotoxicity with regular intake of alcohol: analysis of instances of therapeutic misadventure. *Hepatology* **22**: 767-773.